Implementing school self-evaluation:

The experience of one school in making the process meaningful.

by

Vincent Thorpe M.Ed (SL), MA.Comp.Mus, H.Dip.Prim.Ed, H.Dip.Mus.Tech, P.Grad.Dip.Ed.St(SEN), B.Mus.Ed, Dip.Mus.Ed, T.Dip.ICT

A Dissertation

Presented in Fulfilment of the Requirements

For the Award of

Professional Doctorate in Education

Ed.D.

Department of Education

NUI Maynooth

February 2022

Head of Department: Prof. Aislinn O'Donnell

Supervisor: Dr. Anthony Malone

Table of Contents

Table of contents	2
List of figures	7
List of tables	8
Abbreviations	10
Declaration	11
Dedication	12
Acknowledgments	13
Abstract	14
Chapter 1 – Introduction	
1.1 Introduction	15
1.2 Research question / aims	16
1.3 Theoretical framework	17
1.4 Overview of the research process	18
1.5 Contributions to knowledge	19
1.6 Key terms and concepts	19
1.7 Overview of this thesis	21
Chapter 2 – Literature Review	
2.1 Introduction	24
2.2 Quality and education	24
2.3 School self-evaluation	28
2.4 Implementing school self-evaluation	30
2.4.1 Quality criteria and local knowledge	31
2.4.2 Capability and capacity	32
2.5 Leading school self-evaluation	36
2.6 Conclusion	38

Chapter 3 – Theoretical Framework

3.1 Introduction	41
3.2 Complexity theory	41
3.2.1 The diversity and dynamic nature of agents	43
3.2.2 Nested systems, connectedness and patterns of interaction	45
3.2.3 Information flow and feedback loops	47
3.2.4 Change, adaptability, co-evolution, self-organisation and emergence	49
3.3 The potential limitations of complexity theory	50
3.4 Conclusion	52
Chapter 4 – Research Methodology	
4.1 Introduction	54
4.2 Research aims	54
4.3 Research setting	54
4.4 Research participants	55
4.5 Research paradigm	55
4.5.1 Pragmatic paradigm	56
4.6 Action research	57
4.6.1 Action research and school self-evaluation	57
4.6.2 Action research and school leadership	62
4.6.3 Action research in this project	63
4.7 Data collection and analysis	69
4.7.1 Data collection	70
4.7.2 Analysis of Data	78
4.8 Reliability and Validity	92
4.9 Ethical considerations	94

Chapter 5 – First Phase of Reconnaissance

5.1 Introduction	97
5.2 Reconnaissance	98
5.3 Findings and discussion	101
5.3.1 Staff involvement in whole school decision-making	101
5.3.2 Resourcing school self-evaluation	103
5.3.3 Leadership and staff motivation	106
5.4 Further analysis through the lens of complexity theory	107
5.4.1 The diversity and dynamic nature of agents	107
5.4.2 Nested systems, connectedness and patterns of interaction	108
5.4.3 Information flow and feedback loops	108
5.4.4 Change, adaptability, co-evolution, self-organisation, and emergence	109
5.5 The next step	109
5.5.1 Prioritising our ideas	114
Chapter 6 – Action Research Cycle 1	
6.1 Introduction	118
6.2 Planning	118
6.3 Implementation	125
6.3.1 Preparatory step 1: Create a school self-evaluation timeline	125
6.3.2 Preparatory step 2: Create a school self-evaluation information board	127
6.3.3 Preparatory step 3: Create school self-evaluation resource trays	128
6.3.4 Preparatory step 4: Show and talk staff members through	
the Department of Education school self-evaluation guidelines	
and quality framework	129
6.3.5 Action step 1: Initiate a school self-evaluation cycle	131
6.3.5.1 Implementing the action	131

6.3.5.2 Monitoring implementation of the action and its effects	138
6.4 Conclusion	143
Chapter 7 – Second Phase of Reconnaissance	
7.1 Introduction	144
7.2 Reconnaissance	144
7.3 Findings and discussion	145
7.3.1 Attitudes to school self-evaluation	146
7.3.2 Leadership and staff motivation	158
7.4 Further analysis through the lens of complexity theory	160
7.4.1 The diversity and dynamic nature of agents	160
7.4.2 Nested systems, connectedness and patterns of interaction	161
7.4.3 Information flow and feedback loops	162
7.4.4 Change, adaptability, co-evolution, self-organisation, and emergence	163
7.5 Summary of findings	164
7.6 The next step	166
Chapter 8 – Action Research Cycle 2	
8.1 Introduction	170
8.2 Planning	170
8.3 Implementation	173
8.3.1 Action step 1: Create an agreed definition of school self-evaluation	173
8.3.1.1 Implementing the action	173
8.3.1.2 Monitoring implementation of the action and its effects	181
8.3.2 Action step 2: List and organise into a graph all the general activities	
undertaken in our school for accountability and improvement	182
8.3.2.1 Implementing the action	182

8.3.2.2 Monitoring implementation of the action and its effects	189
8.4 Findings and discussion	190
8.5 Further analysis through the lens of complexity theory	193
8.5.1 The diversity and dynamic nature of agents	193
8.5.2 Nested systems, connectedness and patterns of interaction	194
8.5.3 Information flow and feedback loops	194
8.5.4 Change, adaptability, co-evolution, self-organisation, and emergence	195
8.6 Summary of findings	195
Chapter 9 – Conclusion and Recommendations	
9.1 Introduction	198
9.2 Summary of findings	198
9.3 The next step	203
9.4 Limitations	204
9.5 Recommendations for future work	206
9.6 Final thoughts	208
Bibliography	210
Appendices	258

List of figures

Figure 3.1	My school as a complex adaptive system	
Figure 4.1	Elliot's model of action research	66
Figure 4.2	My research design	67 – 68
Figure 4.3	The process of SSE is easy to understand – bar chart results	80
Figure 6.1	The school self-evaluation cycle	125
Figure 6.2	School self-evaluation information board	127
Figure 6.3	School self-evaluation resource trays	128
Figure 8.1	Four-quadrant evaluation diagram	183
Figure 8.2	Diagram of all the activities undertaken in our school for	188
	accountability and improvement	

List of tables

Table 2.1	Evaluation activities and the capability required	34
Table 2.2	Evaluation activities and the capability, capacity required	35
Table 4.1	Sample of participants for each data gathering process within the	
	study	69 – 70
Table 4.2	Reflective journal template	77
Table 4.3	What are the positive aspects of SSE as a government policy?	
	– sample answers	80
Table 4.4	Survey Part 1:Likert-scale and yes/no questions – sample answers	
	Survey Part 2: open-ended questions – sample answers	81
Table 4.5	Focus group interviews – sample answers	82
Table 4.6	Coding data extracts process sample	83 - 84
Table 4.7	Codes and descriptions	85 - 86
Table 4.8	Coding and theme process	86 - 88
Table 4.9	Developing themes process	89 – 91
Table 5.1	SWOT analysis on the process of SSE in our school	98 – 99
Table 5.2	Brainwriting 6-3-5 'Idea creation' strategy	110
Table 5.3	Idea prioritization strategy	114
Table 5.4	Idea prioritization grid – How can we improve the process of	
	SSE in our school?	115
Table 6.1	Action research cycle 1: Action plan	121 – 123
Table 6.2	School self-evaluation timeline	126
Table 6.3	School self-evaluation 'area of focus' survey results	132
Table 6.4	Useful documents regarding SSE according to staff members	
	 sample survey answers 	139 – 140
Table 6.5	The purpose of SSE according to staff members	
	– sample survey results	140
Table 6.6	The positive aspects of SSE according to staff members	
	 sample survey answers 	140
Table 6.7	The negative aspects of SSE according to staff members	
	– sample survey answers	141
Table 6.8	Emotions felt by staff members when they think of SSE	
	– sample survey answers	141

Table 6.9	Staff member opinions, thoughts, ideas, observations and	
	comments in relation to SSE – sample survey answers	142
Table 8.1	Action research cycle 2: Action plan	172 - 173
Table 8.2	Definitions of school self-evaluation thematic analysis	175 – 177
Table 8.3	An agreed definition of what school self-evaluation means to us	
	for our school	181
Table 8.4	Evaluation activities undertaken in our school and why,	
	according to our school staff	184 – 186

Abbreviations

School Self-Evaluation
National Council for Curriculum and Assessment
Department of Education and Skills
Organisation for Economic Co-operation and Development
Office for Standards in Education, Children's Services and Skills
Continuous Professional Development
Evaluation Support and Research Unit
Looking At Our School
Programme for International Student Assessment
National Parents Council
Professional Development Service for Teachers
Education Review Office
United Nations Educational, Scientific and Cultural Organization
Participatory Action Research
Action Research
Community-Based Participatory Research
Strengths, Weakness, Opportunities, Threats
Special Education Teacher
Special Needs Assistant
Irish National Teachers Organisation
Irish Primary Principals Network
Collaborative Action Research Network
Network for Educational Action Research in Ireland

Declaration

I hereby certify that this material, which I now submit for assessment on the programme of study leading to the award of Doctorate in Education is entirely my own work, and that I have exercised reasonable care to ensure that the work is original, and does not to the best of my knowledge breach any law of copyright, and has not been taken from the work of others save and to the extent that such work has been cited and acknowledged within the text of my work.

Signed: _____ (Vincent Thorpe)

Student No: 60129999

Date: 14th February 2022

Dedication

I dedicate this thesis to my wife Sonia and my two sons Harry and Seán. You are and always will be, the most special and best of friends I will ever be lucky enough to have in this whole wide world. Your firm faith in me has sustained me through this project and your unwavering support and love made completing this thesis possible. *Míle buíochas. Tá grá agam daoibh.*

Acknowledgements

I wish to acknowledge the help received from a number of people in the preparation and presentation of this thesis.

First, I would like to offer my sincere thanks to Dr. Anthony Malone, my thesis supervisor, whose wisdom and guidance challenged me to consider the topic of school evaluation from a broad and wide-ranging perspective. His leadership directed me to read widely, to explore new pathways and possibilities and to reflect deeply. His knowledge and insights were particularly useful at crucial junctures on the journey, particularly in discussing the research question and in drawing the final conclusions. His expertise and encouraging feedback were invaluable in enabling me to bring this thesis to completion.

Heartfelt appreciation goes to my work colleagues and all those who participated in this research. Thank you to the teachers and special needs assistants who met me willingly and were very generous in providing their time and their ideas as this project progressed. Their willingness to take time from their very busy schedules, to listen to and develop an understanding of the research framework and to discuss the relevant topics with honesty and openness enabled me to conduct the research successfully. The sharing of their experience, insights and reflections greatly enriched my knowledge and helped me to gain a deeper appreciation of the value and potential in the work that I do. Thank you most sincerely.

Many thanks to the coordinators of the professional Doctorate in Education programme in NUI Maynooth, in particular Dr. Rose Dolan who worked tirelessly to help me learn and grow, and to all staff associated with the course for their support and assistance.

Particular thanks to my EdD colleagues whose ongoing contact, discussion, encouragement, good humour and sharing of wisdom ensured that the study and thesis writing did not become an isolating activity, but a beautiful journey of learning and lifelong friendship.

A very special word of thanks to Dr. Majella Dempsey who patiently read all sections of the thesis, made very useful suggestions and ensured the final product reached the desired standard.

Finally, I offer deep appreciation to my family, my hometown of Newbridge and the people within it and God. Anything I achieve, or the person I live to be, is due to the faith, education, sacrifice, kindness, love and support you have given and continue to give me freely each and every day. I am deeply grateful and profoundly humbled. *Dia Beannaigh sibh go léir agus go raibh maith agaibh*.

Abstract

Implementing school self-evaluation: The experience of one school in making the process meaningful.

Vincent Thorpe

This study aimed to improve the implementation of school self-evaluation in our school and how I, as principal, in collaboration with staff, could create the optimal conditions necessary. The study investigated school self-evaluation in practice. From this perspective it describes the characteristics which promote good quality implementation. The research used an action research design, putting school staff at the centre of all efforts to improve the school selfevaluation process.

I worked through two cycles of action research, based on Elliot's (1991) action research for educational change model using a range of predominantly qualitative tools including surveys, interviews, a reflective journal, field notes and interaction with a critical friend. The data was analysed using thematic analysis techniques. The constructs of complexity theory and school improvement theory were used as the basis for coding the data and examining the generated themes for key findings.

From a school evaluation perspective, the work provides the reader with a deeper understanding of school self-evaluation in practice. The study also examines the impact of values and reflective practice on a school principal's ability to lead the school self-evaluation process successfully.

The study concluded that when school leaders resource school self-evaluation, in terms of professional and personal supports, the process can be implemented successfully. The study showed a positive change in staff members' attitudes, commitment and capacity to implement school self-evaluation when the school leadership itself was reflexive and positive towards the process and provided opportunities for staff members to take ownership of the process in a spirit of collegiality and collaboration.

Chapter 1 – Introduction

1.1 Introduction

School self-evaluation became mandatory in Ireland in 2012. At that time, I was a teaching principal in a small rural country school. However, I felt school self-evaluation was just another job to be added on to what was becoming an already hard and growing workload. So, I despised it. In the two days 'Continuous Professional Development' (hereafter cited as CPD) I received from the 'Professional Development Service for Teachers' (hereafter cited as PDST) as leader of school self-evaluation in my school, I felt the process was never fully explained to me or maybe I never fully understood the reasoning behind it. In fact, my first CPD as school leader happened before I even got copies of the actual guidelines. Not a positive start.

I now work as an administrative principal in a medium-sized rural school of two-hundred and sixteen pupils, with a staff of ten teachers, two special needs assistants and four support staff. However, after nine years of school self-evaluation, I was still unsure of 'how' to do it, let alone 'lead' it. I put work into it, but it seems like a box ticking exercise and the creation of a paper trail in case *'someone from the department'* comes looking for it. I lack confidence in my own ability to lead evaluation effectively. Although trying my best, I feel ill-equipped, unskilled, un-trained, and apathetic for school self-evaluation due to a lack of results and very little progress, despite the work and effort involved. I do not enjoy the process. School leaders and school staff can organise and attend continuous professional development in school self-evaluation through the Professional Development Service for Teachers (PDST). Schools can also book advisory visits on the process from the Inspectorate. However, competing curricula and legislative demands have forced schools to prioritise and make productive choices (O'Day, 2002, Coolahan et al., 2017).

My feelings regarding school self-evaluation, affect my overt behaviour and this has knock-on and consequential effects on how my staff engage with school self-evaluation and any possible learning benefits it may have for our organisation. Staff members are involved in the process, but at a very low level, as I lead school self-evaluation with a quick fix approach, doing most of the work on it myself, so as not to burden my staff with the extra workload. This is limiting any possibility of collaborative benefit or meaningful organisational change, limiting the school's capacity to develop and learn. However, my practice as school principal, is guided by my values of authenticity and collaboration and my current approach to leading school selfevaluation negates those values. With a renewed focus on my leadership, I believe we as a school can do self-evaluation better. By leading the school to engage with the school self-evaluation process in more collaborative, dialogical and co-constructivist way, my hope is that the process can become more meaningful, with genuine school improvement possible.

This is important, as school self-evaluation has become a key mechanism, in many educational systems, for school improvement and accountability, and a mechanism to manage change in a school organisation (Hopkins, 2005; MacBeath, 2006; McNamara and O'Hara, 2008; Vanhoof et al., 2011; DES, 2012). Many education systems have applied newly developed evaluation methods at school level in the form of self-evaluation (Nevo, 2010), evident by the fact that it is now compulsory in two thirds of European education systems (European Commission/EACEA/Eurydice, 2015). School self-evaluation is a form of internal evaluation, counterbalancing a tendency in many educational systems to rely exclusively on external evaluations as a measure of educational quality (OECD, 2013).

According to McNamara et al. (2021a) however, there has been limited research to date exploring how school self-evaluation is being received in schools both as a policy and a concept and how successfully or otherwise it takes place. This study adds to the body of literature available on school self-evaluation by giving an account from the viewpoint of school staff in an Irish primary school as they implement the process. From this perspective it describes the characteristics which promote good quality implementation.

This research study investigated school self-evaluation in practice, examining its implementation by school staff in a primary school. The focus of the research was on how to improve the implementation process. A qualitative action research design was used based on a 'bottom-up' approach to change, putting school staff at the centre of all efforts to improve the implementation of school self-evaluation. The overall aim of this action research study therefore, was to improve the school self-evaluation process in my school and how I, as school leader, in collaboration with my school staff, could create the optimal conditions necessary for its successful implementation, whilst also addressing any barriers which may have existed.

1.2 Research question / aims

The research and development of new models of evaluation, such as school self-evaluation and their successful implementation to ensure quality in schools, is gaining increased interest at international level (Nevo, 2010; Chapman and Sammons, 2013; Hamzah, 2013; Ladden, 2015).

Whilst there is growing recognition of the important role school self-evaluation can play in improving schools and ensuring quality (MacBeath, 1999, 2006; Hopkins, 2005; McNamara and O'Hara, 2008; Vanhoof et al., 2011; DES, 2012; Hamzah, 2013), Meuret and Morlaix, (2003) suggest it is more praised by policymakers than schools themselves. The evidence that exists regarding school self-evaluation in Irish schools suggests similar findings (McNamara and O'Hara, 2012; O'Brien et al., 2015). However, as school self-evaluation is now on a formal footing in Ireland, research at implementation level is important to aid its success. This thesis seeks to augment the growing body of research relating to school self-evaluation in Ireland (Brown et al., 2020; McNamara and O'Hara, 2005; McNamara and O'Hara, 2006; McNamara and O'Hara, 2008; McNamara and O'Hara, 2012; O'Brien et al., 2017; Brown et al., 2018; O'Brien et al., 2019; McNamara et al., 2021a; McNamara et al. 2021b; Skerrit et al., 2021) especially within the context of an Irish primary school. The key question driving this qualitative study therefore is how we can improve the implementation of school self-evaluation in our school.

In order to address this question, I worked through two cycles of Elliot's (1991) action research model (described in detail in chapters five to eight).

1.3 Theoretical framework

In this section, I will briefly outline the origins and principles of complexity theory and school improvement theory and their application to school self-evaluation. I discuss briefly the potential and the limitations of both theories for this study and school leadership in general.

A review of the literature reveals an ongoing debate into the nature of evaluation and how it should be carried out (Ladden, 2015). It shows the implementation of school self-evaluation to be complex and sensitive due to a number of tensions and dilemmas which form the basis of four key debates surrounding the topic, namely (i) what is the purpose of school self-evaluation? (ii) should it be internally or externally driven? (iii) should it be seen from the 'bottom-up' or the ''top-down'? (iv) what is the appropriate blend of these elements? (Faubert, 2009; Chapman and Sammons, 2013). In my experience, Government pass education policies and expect school management to implement those policies with absolutely fidelity to the goals that were initially established, but the reality is that this rarely happens. The policies get adapted to local contexts and often times are not exactly what was intended (Howlett and Raynor, 2006; Terhart, 2013; Brown et al., 2020; Hudson et al., 2019; Skerritt et al. 2021). This research

project draws upon insights from complexity theory and school improvement theory as a conceptual framework to better understand why this happens and how a school leader could use that knowledge in a positive way to introduce, implement and sustain a process of school self-evaluation within their own school context.

Complexity theory challenges any analysis which sees organisations in linear terms as mechanistic and predictable, performing as they are programmed to do (Morrison, 2002; Mennin, 2010) based on simple rules of cause and effect. Complexity theorists believe that the human environment is non-linear, dynamic and complex, continuously changing in unpredictable ways. Similarly, school improvement theory too sees human environments as dynamic social organisations engaged in the implementation of change to improve teaching and learning. A school leader can use both theories to analyse staff members responses about their experiences with school self-evaluation and gain a better understanding of how, why and when school staff implement school self-evaluation. This gives school leaders more of an awareness of the optimal conditions and potential barriers to successfully implementing school self-evaluation in their own school.

1.4 Overview of the research process

The methodology adopted to complete this research was based on Elliot's (1991) action research for educational change model and was completed over a number of cycles to allow for implementation and evaluation at different stages of the study. Data was gathered using a range of predominantly qualitative tools including surveys, interviews, a reflective journal, field notes and interaction with a critical friend. The data was analysed using the six-phase thematic analysis technique proposed by Braun and Clarke (2019) which facilitated themes being generated from the data collected during each cycle of study.

In order to describe the complex systemic factors that work together to create the conditions that foster or inhibit any successful implementation of school self-evaluation, the constructs of complexity theory and school improvement theory were used as the basis for coding the data and examining the generated themes for key findings. These constructs, as well as action research and the method of thematic analysis used, are explained in detail in sections 2.2, 3.2, 4.6 and 4.7.2

1.5 Contributions to knowledge

This study will be of interest to researchers from the fields of school evaluation and school leadership. This work provides the reader with a deeper understanding of school self-evaluation in practice. The various cycles of study which are presented, address a gap in the literature by demonstrating and analysing the implementation of school self-evaluation in practice by school staff in an Irish primary school (McNamara et al., 2021a). The study expands our knowledge and understanding of the conditions which enable school staff to implement the process of school self-evaluation in a successful and meaningful way.

While the context of this study focused on one cohort of school staff implementing school selfevaluation in a specific school and the strategies employed to enable this, learning from this research may have implications for the use of such strategies in other areas of study. The successful implementation of school self-evaluation to ensure quality education in schools is of particular interest at both international and national level. As school self-evaluation is being developed and formally introduced around the world, including Ireland, it is pertinent to conduct this research at grass roots implementation level. What these research participants have to say can inform the future direction of school self-evaluation in Ireland.

This research has shown that the implementation of school self-evaluation at local level is a complex problem and one that deserves study. It also indicates that a complexity theory approach to analysing how, why and when school staff change their practices maybe a useful theoretical construct for understanding the myriad of complex factors involved and how they interact with each other. This study contributes to change by providing knowledge to school leaders that will lead to a better understanding of how to create the optimal conditions necessary to support the successful implementation of school self-evaluation in their school.

1.6 Key terms and concepts

Throughout this dissertation, the reader may encounter terms and concepts with which they are unfamiliar or which have a different meaning in the context of this study. For this reason, it is important at this stage of the study to define the key terms and concepts which will be used throughout. **School self-evaluation (SSE):** In the Irish education system, school self-evaluation is a collaborative, inclusive and reflective process of internal self-review. An evidence-based approach, it involves gathering information from a range of sources, and then making judgements.

Continuous Professional Development (CPD): Ongoing training in the content, pedagogy and skills necessary for school staff to stay current in their field (Jaquith et al., 2010).

Capacity: The ability of school staff to grow in leadership, content, knowledge and skills (Kaniuka, 2012).

Meaningful organisational change: The process of moving from the present or current operational phase of an organisation, such as a school, into more advanced functional operations, where the quality of the interactions and relationships among the people involved, not the people themselves, determine if the change is meaningful and a success (Fullan, 2001).

Agent of change: Anyone in their area of responsibility who is skilled in initiating, facilitating and implementing organisational change and enables others to deal with these change efforts (Gerwing, 2016).

School Improvement (SI): A process by which schools become more effective both in terms of academic outcomes as well as in developing the social and cultural wellbeing of the children and adults within the school. It describes conscious efforts both inside and outside the school to raise school achievements by modifying classroom practices and adapting management arrangements to improve both teaching and learning (Baines, 2019).

Quality education: An education that enables pupils to achieve their full potential and to participate fully as members of society, and contribute to Ireland's social, cultural and economic development (Government of Ireland: National Development Plan, 2018).

New Public Management (NPM): A term coined by Hood (1991) to describe the policy to modernise the public sector and render it more effective. The basic idea of new public management is that market-orientated management of the public sector will lead to greater cost-efficiency for governments. Ferlie et al. (1996) describe it as introducing the 'three Ms' of markets, managers and measurement into public services.

Neoliberalism: A political approach that favours free-market capitalism, deregulation and reductions in government spending (Stevenson, 2010).

Core values: For this study, core values were defined as personal ethics or ideals that guide a person when making decisions, building relationships and solving problems. Identifying the values that are meaningful in your life can help a person to develop and achieve both personal and professional goals.

Classroom practices: For this study, classroom practices were defined as any practices used by teachers that were related to curriculum, instructional methods and classroom management techniques because changes in these practices often have impact on pupil learning.

Learning community: A term used in education circles to mean a learning organisation (Williams et al., 2012).

Learning organisation: An organisation that explicitly seeks to enhance the capacity of its members to learn and change through their adherence to the disciplines of personal mastery, metal models, shared vision, team building and system thinking (Senge et al., 2012).

Professional Development: Ongoing training in the content, pedagogy and skills necessary for teachers to stay current in their field (Jaquith et al., 2010).

Professional Development Service for Teachers (PDST): The government funded body tasked with delivering professional development to teachers in Ireland.

Professional Learning Community (PLC): Teams of teachers working collaboratively to reflect upon and improve educational practices (Johnson et al., 2010).

Systems thinking: An approach to understanding individuals, groups and organisations through the concepts of interdependency, interaction, feedback, stability and change (Senge et al., 1999).

1.7 Overview of this thesis

Chapter 1 introduces the reader to the overall background and context for the study. It summarises the key motivations, objectives and research questions that drive the research. This chapter also explains the theoretical framework used and the methodological approach taken.

Chapter 2 provides the reader with a review of the key international literature, reports and other documentation related to school self-evaluation. It outlines the context and purpose of school self-evaluation, examining its accountability and improvement function. To do this it examines what is meant by the term 'quality' in education, highlighting the competition for

primacy over what counts as educational quality and how this should be measured. It explores the meaning of school self-evaluation and what's involved in the process. It then conducts a review of the literature relevant to the research question, relating to the introduction and implementation of school self-evaluation in the Irish context and what has been learned so far. The literature review concludes by examining the key role school leaders play in embracing school self-evaluation and building the evaluative capacity of their schools.

Chapter 3 outlines the theoretical aspects of the project. It begins by presenting the origins and principles of complexity theory and its application to human organisation systems. It then explores these principles when applied to school self-evaluation. The chapter concludes by examining both the potential and the limitations of complexity theory for this study and school leadership in general.

Chapter 4 describes the methodology used in the research. It describes the aims of the research, the research setting and the participants involved. The philosophical paradigms that underpin the research are explained. The rationale for selecting a pragmatic, mixed method approach to the research is presented. The chapter explains and justifies action research as the most appropriate design for this study. It describes the predominantly qualitative research methods employed throughout this study and the data analysis procedures used. The chapter concludes by addressing both the reliability and validity of the research and its ethical considerations.

Chapter 5 presents the first phase of reconnaissance. It describes the qualitative research methods employed to collect data and the data analysis procedures used to examine the process of school self-evaluation in our school. The findings are presented. They are then analysed, compared and contrasted in the light of findings from the literature review. The salient issues regarding the implementation of school self-evaluation in our school are outlined. Based on these findings the chapter concludes by presenting the solutions generated to address the identified issues.

Chapter 6 outlines cycle one of this research. This cycle involved the creation and implementation of action steps to improve the implementation of school self-evaluation in our school. Taking into account the cyclical nature of action research, the chapter traces the various stages of planning, implementing and monitoring. This cycle examined the creation of school self-evaluation resources by school staff, a discussion by staff regarding Department of Education SSE guidelines and the initiation of a school self-evaluation cycle, whilst also describing staff members' experiences of using SSE as a method for school improvement.

22

Chapter 7 presents the second phase of reconnaissance. It describes the qualitative research methods employed to collect data and the data analysis procedures used to examine the implementation of our action plan. The findings from a survey and two semi-structured interviews are presented. They are then analysed, compared and contrasted in the light of findings from both the literature review and the first phase of reconnaissance. The salient issues regarding the implementation of school self-evaluation and the lived experience of staff members tasked with implementing the process in our school are outlined. The chapter concludes by presenting the solutions and action strategies generated to address issues identified in the findings.

Chapter 8 details cycle two of the study. This cycle involved developing a cohesive whole school definition of school self-evaluation by consensus and a clear understanding of where it fits within our school evaluative practice. Taking into account the cyclical nature of action research, the chapter traces the various stages of planning, implementing and monitoring. This cycle examined the creation of an agreed definition of school self-evaluation and a diagram to list and explain all the activities undertaken in our school for accountability and improvement, whilst also describing staff members' attitude towards and knowledge of school self-evaluation as a tool for school improvement. The chapter concludes by presenting the solutions and action strategies generated to address issues identified in the findings.

Chapter 9 concludes the research. It details the key contributions to knowledge which enabled the successful implementation of school self-evaluation in our school, whilst also addressing the limitations of the study. The chapter draws together findings from all cycles of the study. From these findings, conclusions and recommendations arising from the research are drawn and suggestions for further research are made.

Chapter 2 – Literature Review

2.1 Introduction

In this chapter I introduce the reader to the themes of this research which are relevant to the implementation of school self-evaluation in my school. This review draws together published literature, reports and other documentation which provide a theoretical and practical background to the study. The themes discussed are: (1) Quality and education, which examines what is meant by the term 'quality' and its relationship to education. There is varied opinion on how quality in education should be measured and the unit of analysis to be used. I discuss how such diverse views bring tension and competition for primacy over what counts as educational quality even within each individual school. This had major implications for the implementation of school self-evaluation in my school. (2) School self-evaluation as a method to improve quality in education. In a complex setting such as a school, school self-evaluation can mean many different things to many different people. I discuss the meaning of school self-evaluation, what is involved in the process and its implementation in Ireland to date. (3) Leading school self-evaluation. I discuss the key role leadership plays in embracing school self-evaluation and building the evaluative capacity of their school.

2.2 Quality and education

Understanding what quality means in terms of education varies between countries. However, most tend to agree on three broad principles: the need for relevance, for equity of access and outcome, and for proper observance of individual rights (UNESCO, 2004). According to UNESCO (2004), education quality is based on five variables (i) Learner characteristics (ii) Context (iii) Enabling inputs (iv) Teaching and Learning and (v) Outcomes. They describe education as a production process, whereby *input*, by means of a *process*, results in *output*. However, all are influenced by context. The context gives input, provides resources for the process and sets requirements to the output. All are interconnected to each other (Scheerens, 1990).

Schools are coming under increased external pressure and heightened levels of scrutiny and regulation (Murphy, 2019) to improve quality outcomes by being (i) accountable for what they do and (ii) continuously striving for quality improvement (MacGilchrist, 2000; Simons, 2013; Klenowski and Woods, 2013; Day and Sammons, 2014). For educational quality and learning

outcomes to improve however, policy makers and educators need access to evidence-based analyses of the current situation, trends over time and information on the strengths and weaknesses of a system, and their causes. Using relevant indicators, a strong monitoring and evaluation system such as inspection or school self-evaluation, can provide that evidence (UNESCO, 2004).¹

How best to determine the quality of schools however, is contentious, as is the unit of analysis (Simons, 2013). Quality can be multifaceted, contested and never fully representable (Stake and Schwandt, 2006; Brown et al., 2016) and therefore difficult to define. Evaluating quality is not a neutral process (Klenowski and Woods, 2013). According to Stake and Schwandt (2006), how we assess quality depends on whether we see quality as something measured, or quality as something experienced. Tension and competition can arise for primacy over what counts as educational quality (Lai and Kushner, 2013). The politics, economics and prevailing logic of the time tend to dictate the area of focus (Simons, 2013). In Ireland for example, public service inspection policy and practice have aligned with the philosophy of new public management drawn from theories of neo-liberal economics. New public management takes the school as the unit of accountability and seeks to improve pupil learning by improving the functioning of the school organisation. It puts the focus on outputs over inputs, measured in terms of performance indicators, where schools are both accountable and responsible for the delivery and improvement of all its services (Lynch et al., 2015).

Most schools tend to operate within two governance philosophies, namely accountability and continuous improvement (Van Bruggen, 2010; Smith and Benavot, 2019). Within these two philosophies, there are different innovative ways of improving schools ranging from quality assurance, inspection and the implementation of school improvement/effectiveness initiatives such as school self-evaluation (MacBeath, 2006; Vanhoof et al., 2009; Demetriou and

¹ In Ireland, the Department of Education is tasked with regulating this process (Government of Ireland, 1998; Government of Ireland: National Development Plan, 2018). To do this it is given a large budget. In the year 2019, 10.763 billion euro in public revenue was allocated to education, representing the highest ever expenditure on education and skills (DES, 2018). It is reasonable that the public and its politically elected representatives want to know how that money is spent and whether it is producing quality outcomes (Demetriou and Kyriakides, 2012).

Kyriakides, 2012; Simons, 2013). According to Simons (2013) however, externally inspired innovations such as these can fail if they do not allow for the cultural and local differences which can exist in specific school contexts. Some believe that parents, local communities and schools are best placed to improve pupil learning, because context matters (O'Day, 2002; Fancy, 2007; Lai, 2013). From this debate on context emerged the school effectiveness and school improvement movement, both focused on the school as the major unit of change (Simons, 2013).

School effectiveness is strongly focused on pupil learning outcomes as a measurement of school quality. It examines the characteristics (factors) of classrooms, schools and systems associated with these outcomes. It does not however, examine the processes that maybe needed to change the situation in classes, schools or systems (Creemers and Kyriakides, 2010). The emergence of the school effectiveness movement has provided growing evidence that schools can, and do, make a difference to academic outcomes for pupils (Chapman and Sammons, 2013). School effectiveness researchers (Tymms, 1992; Teddlie and Reynolds, 2000; Hendriks et al., 2002; Kyriakides and Campbell, 2004; Creemers and Reezigt, 2005; Bollen et al., 2005) have tended to focus on exploring differences between more or less effective schools from a quantitative perspective i.e., quality can be measured (Chapman and Sammons, 2013). The research methodology has often involved quantitative measurement of a range of parameters associated with educational quality and performance such as academic achievement, teacher behaviour, curriculum, class grouping, school organisation and education policy in an attempt to (i) assess the size of school effects (Tymms, 1992) (ii) identify characteristics exhibited by more effective schools (Sammons et al., 1995) (iii) quantify school effects for different groups of pupils and (iv) explore the stability of school effects over time (Nuttall et al., 1989).

In contrast, the school improvement movement believes such a narrow focus on academic achievement only captures but a fragment of what counts in schooling, both in terms of processes and outcomes (Carney, 2003). For example, the role of ethos, community relations, values, beliefs, attitudes to school, flexibility, health, productivity and lifelong learning, to name but a few (Davies, 2001; Carney, 2003). School improvement is mainly concerned with the process of change in classes and to a larger extent in schools, as the measure of school quality. It does not look too much at the consequences for pupil outcomes. The main interest of this movement is in establishing a process of improvement in schools and to keep it going by emphasising the importance of the process, rather than being critical of its impact on learning outcomes (Creemers and Kyriakides, 2010). The school improvement movement has

tended to consider schools as social organisations engaged in the implementation of change to improve teaching and learning. In contrast to school effectiveness research, this has usually involved a 'bottom-up' rather than a 'top-down' approach to change. It locates power and control with those actually tasked with securing improvements (Chapman and Sammons, 2013). Improvement and the capacity to improve comes from within the school. To address achievement problems, it is considered important to allow contextually driven approaches that encourage flexibility and local ownership (Parr and Timperley, 2013). The methodologies relied upon to achieve these aims are largely qualitative i.e., quality is experienced (Chapman and Sammons, 2013). School improvement research and practice is concerned with making schools 'better' paces for all in the school community (Bollen et al., 2005). This approach draws upon the principals of inquiry, reflection and self-evaluation to initiate improvement (Hopkins et al., 1994).

Whilst school effectiveness and school improvement are two distinct movements within the field of educational research, both are influential in how quality outcomes are measured in many education systems. Various governments from around the world, including Ireland, have listened to the findings and arguments of academics and researchers from both movements, drawing on the research findings to develop a range of quality assurance policies to intervene in schools to raise educational standards (Chapman and Sammons, 2013). In both movements educational quality is taken to mean the delivery of effective learning which is evidence-based and can be measured. In addition, any issues which arise can be identified by those held accountable and solved, using proven strategies and methods to improve quality (Anderson and Boyle, 2020). School self-evaluation is one such method of improvement used in both. To get a balanced view of the quality of education provision therefore, educational systems in many countries, including Ireland, now conduct various evaluation type formats (Hamzah and Tahir, 2013). Typical components can include (i) pupil assessment (e.g., standardised assessments, summative assessment), (ii) teacher or school leader appraisal (e.g., self-assessment, informal/formal feedback on practice, performance management processes such as probation), (iii) school evaluation (e.g., external validation, school self-evaluation) and (iv) system evaluation (e.g., policy or programme evaluation) (OECD, 2013, pg.87).

School evaluation can be external or internal (MacBeath, 2006) and can play a very influential part in determining the 'quality' of education in schools. Both external and internal evaluation are interrelated and can form an important part of school improvement and reform policies (Jaffer, 2010). School self-evaluation serves as internal evaluation, whilst school inspection

functions as external evaluation. Internal evaluation is conducted by schools themselves (Hamzah and Tahir, 2013) and is often described as self-evaluation (O'Brien et al., 2019). External evaluation on the other hand, is usually done by an appointed inspection or supervision body (Janssens and Amelsvoort, 2008; Vanhoof and Petegem, 2010). It is possible for a primary school and an external quality assurance agency such as an inspectorate, to base its evaluations on a negotiated and collaborative approach to whole school evaluation (Simons, 2013). This would be a form of increased cooperation and partnership between schools and the inspectorate in a dual system of internal/external quality assurance (McNamara and O'Hara, 2012; Brown et al., 2016; McNamara et al., 2021b). Such an approach is perceived as having a developmental and improvement focus and a role in equipping schools to analyse performance and drive change (Brown et al., 2016; Coolahan et al., 2017). In the Irish education system for example, external and internal evaluation are both seen as complementary contributors to school improvement. It maintains that the most powerful agent of improved quality is a well-integrated system of evaluation that combines the external perspective of the inspectorate with the reflective and collective insights of school leaders, teachers, parents and pupils (DES, 2016).

2.3 School self-evaluation

School self-evaluation is an important concept related to school improvement plans and forms an integral part of the whole-school evaluation process (Van der Westhuizen, 2014). In many educational systems, school self-evaluation has become a key mechanism for school improvement and accountability and a necessary mechanism to manage changes in the school organisation (Hopkins, 2005; MacBeath, 2006; McNamara and O'Hara, 2008; Vanhoof et al., 2011; DES, 2012). Many education systems have applied newly developed evaluation methods at school level in the form of self-evaluation (Nevo, 2010), evident by the fact that it is now compulsory thirds of European education in two systems (European Commission/EACEA/Eurydice, 2015). The overarching goals for school self-evaluation are twofold according to Kyriakides and Campbell (2004). First, to improve the quality of the organisation and secondly, to improve teaching and learning. Improving the quality of the organisation includes social relations between members of the organisation, organisational climate and culture (Freiberg, 2005), the nature of decision making (Hoy et al., 2012) and the responsiveness of the school as an organisation to external and internal change forces (Fullan, 2016). Improving teaching and learning involves the concepts of teacher effectiveness (Muijs and Reynolds, 2002), school effectiveness (Teddlie and Reynolds, 2000) and decisions about how these concepts are measured within the school setting (Goldstein, 2011). It stresses the schools' own responsibility for quality (Hofman et al., 2010), creating space for the school to evaluate its policies and practices (Simons, 2013). It has gained a prominent position as an important leverage for quality assurance and school improvement (Faddar et al., 2018). Countries are developing this evaluative expertise and building evaluative capacity (O'Brien et al., 2015; Mcnamara et al., 2021a), resulting in a greater variety of evaluation activities and research data. Studies into the effects of self-evaluation have shown that it can improve school quality in terms of improved pupil achievement levels (Campbell and Levin, 2009; McNaughton et al., 2012). Not mentioned by these advocates however, is the broader emphasis on attitudes and practices which are also significant when evaluating school improvement measures (Vanhoof et al., 2009; O'Hara et al., 2016; O'Brien et al., 2019). Nonetheless, school self-evaluation is expanding in school systems and has gained a lot of supporters worldwide claiming it to be vital in ensuring quality education (Hamzah, 2013).

School self-evaluation is an internal process in large part initiated by a school itself (Vanhoof and Petegem, 2011) to ensure quality and to improve the teaching-learning process and school performance (Hofman et al., 2009). When a school community engages and participates with school self-evaluation, they get an opportunity to commit to speaking for themselves, to develop and change from within (MacBeath, 1999). The school self-evaluation process has been described as a process for helping schools improve, which should be shaped by themselves, and integrated into their routine management systems (Chapman, 2008). It aims to assess the functioning of the school in a systematic way, with a view to taking decisions or adopting initiatives within the framework of overall school development and policy (Vanhoof and Petegem, 2011). School self-evaluation works as a reflection process which will identify and analyse the school's strengths and weaknesses (Hamzah, 2013). It involves the systematic collection and analysis of data to form value judgements based on firm evidence (Neil and Johnston, 2005). In undertaking school self-evaluation, stakeholders can understand the current situation, including the strengths and area for development (weaknesses) and the opportunities and threats to their organisation. This enables them to determine the goals needed and develop the strategies for achieving those goals (Setlalentoa, 2014). From that analysis, school improvement strategies are constructed (Hamzah, 2013).

Barber (1997) argues that the essence of a successful organisation is the search for improvement and that effective self-evaluation is the key to it. He describes school self-

evaluation as restless in its quest for evidence in a schools' transparent sense of purpose, behaviour, relationships and classroom performance (cited in Demetris and Kyriakides, 2012). It is both a process (gathering data to make claims) and a product (the summary statement of what the evidence shows) to inform the goal setting, the implementation of strategies and the measuring of their success (Mutch, 2013).

2.4 Implementing school self-evaluation

In a complex adaptive system such as a school, school self-evaluation can mean many different things to many different people (Skerritt et al., 2021). According to Klenowski and Woods (2013), evaluation involves the processes of description, analysis and judgement of educational programmes, practices, institutions and policies for a range of purposes. Evaluation can serve the purpose of accountability (measurement of results or efficiency), development (to develop or improve) and knowledge (deeper understanding of practice or policy). Any evaluation will involve key questions about

(i) What is the purpose of the evaluation itself?

• Why is this evaluation being undertaken?

(ii) Who is sponsoring the evaluation?

• Who is conducting the evaluation?

(iii) Ethical issues

- Whose judgement counts here?
- What rights do those being evaluated have?
- What rights and responsibilities do the evaluators have?

(iv) Political stance

- Who is the audience?
- *How is the evaluation to be reported?*
- Who will take action as a result of the evaluation?

(v) Procedural choices

- How is the evaluation to be carried out?
- Whose criterion of value is being applied?

(Klenowski and Woods, 2013, pg.204).

Implementing school self-evaluation is therefore a complex activity for any school community (Skerritt et al., 2021) where issues can arise in terms of knowledge, skills, workload and leadership (McNamara and O'Hara, 2012; O'Brien et al., 2015; McNamara et al., 2021a). I will now discuss these potential problems in more detail and their relevance to this research.

2.4.1 Quality criteria and local knowledge

There are a myriad of models, frameworks, and definitions associated with the process of school self-evaluation, as well as a range of commonly used terms. Definitions and understanding of the process can vary from country to country and region to region (Chapman and Sammons, 2013). A review of the salient literature, reveals many definitions of school selfevaluation (Appendix 1). They contain many common aspects. Most of the definitions (Swaffield and MacBeath, 2005; MacBeath, 2005; Hopkins, 2005; Schildkamp, 2007; Vanhoof and Petegem, 2010) see school self-evaluation as an internal process which uses reflection and judgement to develop pupils, teachers and the school as a whole. In terms of new public management and its neo-liberal agenda, development here is taken to mean being accountable for increased efficiency and effectiveness in reaching set targets and delivering improved quality outcomes (Tolofari, 2005). What is also interesting is that reference to the word 'quality' is often absent. As mentioned earlier in the chapter, this could be evidence that quality can be multi-faceted, contested and never fully representable (Stake and Schwandt, 2006) and therefore difficult to define. Also of note is the lack of reference to the technical aspects of school self-evaluation i.e., the cyclical process of systematically collecting evidence, then planning and setting targets and implementing them. This is significant and an important issue in terms of school self-evaluation practice. School staff must be clear on the practicalities of school self-evaluation and what is expected of them (McNamara and O'Hara, 2005; Ryan et al., 2007; McNamara and O'Hara, 2012; O'Brien et al., 2015).

Stake (2004) argues that educational instruments and standards, used to measure or make judgements about performance, derive quality criteria from 'somewhere else' and import them into another educational setting 'out of context', thereby displacing 'local' judgements.

However, Brown (2013) suggests it is possible to use such a criterion-based approach, whilst at the same time taking into consideration more local, experiential and contextual criteria.

To improve the implementation of school self-evaluation in our school therefore, it was imperative that any engagement with the process was underpinned by an unambiguous definition of school self-evaluation. It needed to be meaningful, free of confusion, easy to work with and both agreeable and understood by all staff members. Published research shows how necessary such understandings are in helping school staff assume ownership of the process and engage more effectively with it (MacBeath, 1999; Vanhoof and Petegem, 2009; Vanhoof et al., 2009; O'Brien et al., 2015).

2.4.2 Capability and capacity

Schools are increasingly being asked to shoulder a greater proportion of the responsibility for developing and guaranteeing educational quality (Vanhoof et al., 2009; Faddar et al., 2017). To do this, school management and staff are being asked to make professional judgements regarding the operation of their school through school self-evaluation. This increased responsibility has implications for school leaders and staff members in terms of training and implementation. It requires schools to have the necessary capability and capacity to evaluate their practices, processes and outcomes (Vanhoof and Petegem, 2011; McNamara et al., 2021a).²

A review of the literature shows how external evaluations become effective and meaningful only when schools have well developed self-evaluation processes in place. In other words, to carry out a full and proper evaluation of a school, the school itself must have the capacity to tell its own story in a reliable and accurate way (MacBeath, 1999). Both external and internal evaluations are important, but neither can exist by itself (DES, 2012) As pointed to earlier, schools need a view from outside to protect themselves from self-delusions (MacBeath, 2006; Setalentoa, 2013). However, external evaluation on its own, has the potential to become a

² For the first time in our careers, my staff and I were being asked to use a new quality indicator framework to systematically gather and analyse various types of data, devise improvement plans and implement improvements on an annual basis, feeding the results into a school improvement plan, which we would then implement and monitor (Westraad, 2011; O'Brien et al. 2019).

meaningless 'once-off' event. A challenge therefore, is how to balance the strengths and weaknesses of both internal and external perspectives, within a school evaluation process (MacBeath, 2006; Vanhoof and Petegem, 2007; McNamara and O'Hara, 2008; Blok et al., 2008; Brown et al., 2016). According to Janssens and van Amelsvoort (2008), accountability demands imposed on school self-evaluation generate accountability-oriented self-evaluations, while improvement demands generate improvement-oriented self-evaluations. Respondents can behave in different ways depending on whether the school self-evaluation is performed in a context strongly characterised by accountability or in a context more orientated towards development (Van Petegem, 1998; Faddar et al., 2018). Mutch (2013) describes how school stakeholders in New Zealand and their inspectorate improved the success of school self-evaluation in their schools by coming together to build a shared conceptual framework of evaluation which portrayed an internal-external continuum, whilst also showing an accountability and improvement focus.

Gathering and interpreting data is a key part of school self-evaluation procedure and forms the basis for school development plans and policy decisions (Schildkamp et al., 2013; Faddar et al., 2018). In terms of school self-evaluation, data can be defined as information that is gathered and organised to represent some aspect of schools (Schildkamp et al., 2013). Data is the raw material of school self-evaluation. Therefore, an understanding of data in terms of definition, usage, problems, solutions and guidelines is essential when implementing a school self-evaluation process (Chapman and Sammons, 2013).

The literature points to many problems in the gathering and interpreting of data, pointing to different kinds of errors that can occur, resulting in distorted data (Groves et al., 2009; Faddar et al., 2017). For example, an interesting study by Faddar et al. (2017), argues that a survey demands a lot of cognitive effort from respondents, suggesting that problems in cognitive processes can lead to cognitively invalid results, particularly in the context of school self-evaluation surveys. All of this is concerning, particularly when schools rely on the conclusions drawn from such data as the source of information for policy decisions and actions, which in turn can have a large impact on school processes and their outcomes (Hofman et al., 2005).

With all these activities Timperley (2013) asks the question of whether it is more in the form of busy work or in the form of the kinds of deep change needed to address real priorities. She argues that schools use a lot of energy engaged in activities that involve reporting to a range of external agencies, all in an attempt to develop capability (pg.39). For example, she questions

the belief that teachers who continuously assess and report on their pupils in terms of an 'end product', will somehow develop a capability to make complex overall judgements about pupil achievement or initiate any real improvement in teaching and learning. Other examples of these busy reporting activities and their questionable outcomes are provided by Timperley (2013) in *Table 2.1*

Reporting activity	Capability
Pupil assessment	Teachers are being asked to make complex overall judgments about pupil achievement
Teacher appraisal	Principals and Droichead teams are being asked to improve teacher performance
School self-evaluation	Schools are being asked to have high quality self-evaluation processes in place ³

Table 2.1Evaluation activities and the capability required

Timperley (2013) contends that these reporting activities are actually taking away from the more educationally relevant but highly demanding task of using this information to improve

³Taking school self-evaluation in my own school for example, we have a school self-evaluation policy in place. Each year our school surveys different people about the education service we provide (e.g., parents, pupils, staff). Pupil work samples are examined as are their standardised test results. From an analysis of these various sources of data, the school produces a school improvement plan. For accountability purposes this plan is reported to the school board of management and placed on the school website for all parents, pupils and staff to read. A copy of the plan is also kept onsite at the school in a folder in a filing cabinet for inspection. The report is consulted a year later to assess if targets have been reached and progress made. This information is then reported in the next improvement plan. This process is repeated each year. For all intent and purposes, by engaging in these activities our school is meeting its mandated regulatory school self-evaluation requirements.

teaching and learning. Reporting accountabilities are mostly just that, providing an account. Timperley (2013) believes however, there is little accountability for making real improvement.

This view is echoed by Sjøbakken (2013) who states that "'quality' when self-defined is less a question of 'summing up' in terms of the quick achievement of instrumental goals and is more realistically understood as a complex process where different forms of self-evaluation are at play" (pg.184).

This action research project is a recognition of this complexity and a genuine attempt to make the school self-evaluation process more collaborative and meaningful for everyone in an attempt to improve our school. To do this, Timperley (2013) makes a very strong case to move away from this focus on capability and instead put more emphasis on a school's capacity. For example, in reference to the previous example of pupil assessment, Timperley (2013) believes this would involve teachers shifting from a role of judging and reporting on pupils in a summative way, to one of enabling pupils to develop their own capacity for self and peer assessment. This, she believes, leads to real change and improvement. Timperley (2013) provides other similar examples of this shift from capability to capacity and its consequences (including for school self-evaluation) in *Table 2.2*

Activity	Capability	Capacity
Pupil assessment	Teachers are being asked to	Pupils developing their own
	make complex overall	capacity for self and peer
	judgments about pupil	assessment
	achievement	
Teacher appraisal	Principals and Droichead	Continuous professional
	teams are being asked to	development and self-
	improve teacher	evaluation, rather than
	performance	compliance and
		accountability
School self-evaluation	Schools are being asked to	Continuous organisational
	have high quality self-	learning rather than just
	evaluation processes in place	reporting

Table 2.2Evaluation activities and the capability, capacity required

This capacity is evident in classrooms on a daily basis where teachers display a capacity to engage in evaluation and improvement. Simons (2013) observes that teachers on a daily basis theorise their own practice, are keen observers and questioners in their classrooms and skilfully interpret documents and curriculum materials. Whilst the context is different, she believes that the potential for using these skills at an institutional whole school level are there. Maybe all that is needed is some leadership?

To improve the implementation of school self-evaluation in our school therefore, it is imperative that I, as school leader, facilitate plenty of opportunities for staff members to realise this potential at whole school level. Every action must be encouraged to allow staff discover that school self-evaluation is basically the same in operation as the self-evaluation they do in their classrooms every day.

2.5 Leading school self-evaluation

Mulford (2003) asks the question, how can a school leader build the capacity of their school and add value, especially when faced with the busyness of new national policies and schoolbased interventions. Increased expectations of the education system in many countries, including greater accountability pressures and emphasis on raising standards has put growing demands on school leaders (Day and Sammons, 2014). Enacting reform, implementing school self-evaluation and embedding effective professional learning places leadership of learning and school improvement at the heart of modern school leadership (Mulford, 2003). Whilst school self-evaluation maybe a necessary ingredient to stimulate school improvement, and raise standards, Chapman and Sammons (2013) believe it is insufficient without the key change agent of school leadership. Bubb and Early (2008) highlight strong leadership as a 'consistent ingredient' to school improvement, while Roby (2011) discusses the positive effect of a leader's efforts towards a culture of continuous learning (cited in Karagiorgi et al., 2015). The school leader is vital, according to Vanhoof and Van Petegem (2012), in creating a culture of school self-evaluation with an emphasis on school improvement, where all teachers can communicate openly and honestly about their own experiences, opinions and expectations to generate success. This sentiment is shared by Karagiorgi et al. (2015) who state that the potential impact of a school leader on the development of a culture receptive to school selfevaluation should not be underestimated. How the process of school self-evaluation is led and managed is vital to its success in promoting school improvement (Chapman and Sammons, 2013). The literature shows leadership to be a consistent positive factor in self-evaluation
activities which have a positive impact on the quality of teaching and pupil outcomes (Bubb and Earley, 2008).

School principals have a key role to play in setting direction and creating a positive school culture including the proactive school mindset needed to foster improvement and promote success for schools (Day and Sammons, 2014). Improving school performance often comes without any guaranteed solution, ready-made blueprint or failsafe model of improving organisational performance, according to Jones and Harris (2014). However, the research evidence shows, that collective professional learning generates the shared leadership and the social capital needed for positive, and lasting organisational change and improved pupil achievement (Jones and Harris, 2014).

According to Day and Sammons (2014) however, school leaders improve teaching and learning indirectly and most powerfully through their influence on staff motivation, commitment and working conditions. They contend that this is a key task for leadership as staff performance is a function of staff members' motivations, commitments, capacities (skills and knowledge) and the culture and conditions in which they work. How school staff perceive the process of school self-evaluation and how they engage with it is therefore determined by a leader's actions (Chapman and Sammons, 2013). A principal who understands the context of their own school environment, is essential to the successful implementation of school self-evaluation. Miller (2018) argues that leadership is situated, and both enabled and constrained by context. Each school has unique characteristics which the school principal must take into consideration when fulfilling their role. Chapman (2008) points out that a principal not understanding context, makes it difficult to develop school self-evaluation that would make a difference to the work of those involved in educating children and their experiences and outcomes. School leaders, he contends, must have a detailed understanding of their context, if school self-evaluation is to fit with those involved in the process. This involves establishing a strong, professional collaborative community, built on trust, that is receptive to critical review and improvement.

In the past, the responsibility for educational quality seemed to rest with the school principal and the middle management team. Sometimes school self-evaluation was carried out with little or no consultation of the school community (MacBeath, 2005). However, the Department of Education (hereafter cited as DES) school self-evaluation guidelines (2016) actively seek a wider range of perspectives, encouraging the participation and input of all school community stakeholders (such as staff, parents, pupils, local community members), thereby offering more

detailed and complex insights into the depths of the organisation. All members of a school community may now be subjects of and participants in a school self-evaluation process. Leaders have come to understand how to do things 'with' rather than 'to' participants (Parr and Timperley, 2013). The research indicates the necessary move away from more traditional school relationships, and a hierarchal autocratic type of leadership, where knowledge is handed down to participants (Parr and Timperley, 2013). If evaluation processes are done in such an imposed and rigid way, a meaningful process of evaluation can become school self-inspection, with the school principal becoming the primary inspector. Chapman and Sammons (2013) argue therefore that school leaders need to maintain a healthy balance between robust structures and room for localised organic growth. One way to do this, according to Parr and Timperley (2013), is for a school leader to co-construct the knowledge with school self-evaluation participants. In such an approach, the leadership is more dispersed throughout the school, resulting in a more bottom-up process of self-evaluation owned by school staff, which the literature shows to contribute to high levels of achievement for the vast majority of pupils (Harris, 2002).

So, whilst school self-evaluation has an important role in supporting pupils learning, at the same time school self-evaluation also supports professional learning too. As teachers improve their own assessment and evaluation standards, they directly support pupils' learning too. This highlights the importance of understanding context and securing strong leadership to stimulate a 'bottom-up' approach to self-evaluation for school improvement, which is internally driven and owned by school staff (National Union of Teachers, 2005).

2.6 Conclusion

The meaning of 'quality' in terms of education varies between countries. What's deemed to be of 'quality' can often be multifaceted and contested. According to UNESCO (2004), education quality is based on the five variables of (i) learner characteristics (ii) context (iii) enabling inputs (iv) teaching and learning and (v) outcomes. In this approach, education is seen as a production process, whereby *input* by means of a *process*, results in *output*. However, all are influenced by context. The context not only gives input, but also provides resources for the process and sets requirements to the output. All are interconnected to each other (Scheerens, 1990). For educational quality and learning outcomes to improve therefore, policy makers and educators need access to evidence-based analyses of the entire process from beginning to end.

Using relevant indicators, a strong monitoring and evaluation system can provide that evidence (UNESCO, 2004).

In Ireland, DES policy and practice have aligned with the philosophy of new public management drawn from theories of neo-liberal economics. Schools are accountable and responsible for the delivery and improvement of all its services, measured in terms of performance indicators (Lynch et al., 2015). As a result, schools operate within two governance philosophies, namely accountability and continuous improvement (Van Bruggen, 2010; Smith and Benavot, 2019). In order to know how to improve and develop, a school must be able to evaluate where it is, what it needs to improve and use indicators to ascertain if it has achieved its aims.

There are different ways of improving schools stemming from quality assurance, inspection and the implementation of school improvement/effectiveness initiatives such as school selfevaluation (MacBeath, 2006; Vanhoof et al., 2009; Demetriou and Kyriakides, 2012; Simons, 2013). From this debate emerged the school effectiveness and school improvement movement. School effectiveness is strongly focused on pupil outcomes as a measurement of school quality, whilst school improvement is mainly concerned with the process of change in classes and to a larger extent in schools, as the measure of school quality. However, in both, a quality education is taken to mean the delivery of effective learning to learners which is evidence based and can be measured. Any issues which arise can be identified by those held accountable and solved using proven strategies and methods to improve quality (Anderson and Boyle, 2020). School self-evaluation is one such method of improvement used in both.

In an effort to get a balanced view of the quality of education provision, educational systems now conduct various evaluation type formats (Hamzah and Tahir, 2013). In the Irish education system external and internal evaluation are both seen as complementary contributors to school improvement. School self-evaluation has become central to school improvement plans and forms an integral part of the whole-school evaluation process (Van der Westhuizen, 2014). Studies into the effects of self-evaluation have shown that it can improve school quality in terms of improved pupil achievement levels (Campbell and Levin, 2009; McNaughton et al., 2012). As a result, school self-evaluation is expanding in school systems and has gained a lot of supporters worldwide claiming it to be vital in ensuring quality education (Hamzah, 2013). There are a myriad of models, frameworks and definitions associated with school self-

evaluation, as well as a range of commonly used terms. Definitions and understanding of the process also vary from country to country and region to region (Chapman and Sammons, 2013).

In 2003, the Irish inspectorate published "Looking At Our Schools" (LAOS) (DES, 2003), a framework for school inspection and self-evaluation to monitor and evaluate the work of schools (McNamara and O'Hara, 2012; Brady, 2016). It was a model of quality assurance that emphasised school development planning through internal school review and self-evaluation, with the support of external validation carried out by the inspectorate (DES, 2003). In 2012 the DES put school self-evaluation on a formal footing, making it a mandatory process for all primary schools (DES, 2012), and reaffirmed LAOS as the key framework for school self-evaluation.

Whilst school self-evaluation maybe a necessary ingredient to stimulate school improvement, and raise standards however, Chapman and Sammons (2013) believe it is insufficient without the key change agent of school leadership. How school staff perceive the process of school self-evaluation and how they engage with it is determined by a leader's actions (Chapman and Sammons, 2013). Strong leadership is vital in stimulating a 'bottom-up' approach to self-evaluation for school improvement. An approach which is internally driven and owned by staff members, helping them to improve their own assessment and evaluation standards and support learning.

Chapter 3 – Theoretical Framework

3.1 Introduction

In this section I will outline the theoretical aspects of the project. Governments pass education policies and expect school management to implement those policies with absolutely fidelity to the goals that were initially established, but the reality is that this rarely happens (Howlett and Raynor, 2006; Terhart, 2013; Hudson et al., 2019). The policies get adapted to local contexts and often times are not exactly what was intended. This research project draws upon insights from complexity theory as a framework to better understand why this happens and how a school leader could use that knowledge in a positive way to introduce, implement and sustain a process of school self-evaluation within their own school context. I will firstly discuss the origins and principles of complexity theory and their application to human organisation systems. I will then explore these principles in the implementation of school self-evaluation. To conclude, I discuss both the potential and the limitations of complexity theory for this study.

3.2 Complexity theory

Complexity theory first began and emerged from the work of natural science to develop models for understanding things like weather systems, which are highly complex systems. Complexity science is really the application of the language of mathematics to try model and explain how smaller systems can work together interdependently to create larger systems. Investigators from many different disciplines began to explore phenomena in similar ways and a new theory emerged known as 'complexity theory'. Complexity theory suggests that the universe is full of systems and that these systems are complex and constantly adapting to their environment (Morrison, 2002; Osberg, 2008; Fenwick, 2012). The most widely used definition of complexity is the one developed by the Santa Fe Institute in New Mexico. According to this definition, complexity refers to the integrated, rich and varied condition of the universe which we cannot comprehend in a usual mechanical way or in a linear fashion (Fidan and Balci, 2017). Marion (1999) provides this useful definition "A complex system is one whose component parts interact with sufficient intricacy that they cannot be predicted by standard linear equations; so many variables are at work in the system that its overall behaviour can only be understood as an emergent consequence of the holistic sum of all the myriad behaviours embedded within" (pg.27-28).

These same principles can be applied to the systems that we work with in the human world to form complex systems theory. This theory challenges any analysis which sees organisations in linear terms as mechanistic and predictable, performing as they are programmed to do (Morrison, 2002; Mennin, 2010) based on simple rules of cause and effect. In contrast, proponents of complex systems theory believe that the human environment is non-linear, dynamic and complex, continuously changing in unpredictable ways. This research draws upon insights from complexity theory as a framework that a school leader can use in a positive way to introduce, implement and sustain a process of school self-evaluation for school improvement within their own school context. I will now describe the main principles and characteristics of a complex system and how they are related to the aims of this study.

Complex systems are considered adaptive because they are constantly adapting to their environment. This adaptiveness helps define the complexity of the system. Complex adaptive systems theory is only one of a number of theories of complexity, but it is well regarded for unifying the notion of complexity (Chu et al., 2003). The most common definition of a complex adaptive system, is a dynamic network of agents acting in parallel, constantly reacting to what the other agents are doing, which in turn influences behaviour and the network as a whole (Holland, 1992, 1999, 2006; Alhadeff-Jones, 2008). Complex adaptive systems thinking provides a way to understand educational processes (Biesta and Osberg, 2010) and has become prominent in educational studies (Fenwick, 2012). The language of complexity makes it possible to see the interdependency between people within those systems, as their roles constantly adapt, self-organise, emerge and evolve overtime (Biesta and Osberg, 2010). This would be in terms of how they are connected to one another, how they negotiate varying pressures from outside forces (like the economy and state policy), as well as the internal pressures (from pupils and school staff). Fullan (2001) believes that all schools, if they are to survive, must understand complexity theory.

Figure 3.1 below, provides a visual representation of a complex adaptive system (Morrison, 2002, pg.10). A school can be described as a complex adaptive system as it is comprised of a population of diverse people (elements). These people are numerous, dynamic, autonomous and highly interactive, with the ability to learn and adapt. The school is characterised by their individual behaviour. They are interconnected through networks nested at various levels within the school system. Within this nested networked system, they interact and adapt to each other based on a combination of elements such as their knowledge, experience, feedback from their environment, local values and formal system rules. These elements can also change over time,

therefore leading to continuously changing interactions that are often novel and hard to predict. Complexity theory contends that from these interactions, people within a school system selforganise and emerge with structures and patterns, which then feed back into the school and further influence the interactions of the people within it (Morrison, 2002; Keshavarz et al., 2010).

My School as a complex adaptive system



Figure 3.1 My school as a complex adaptive system

Fenwick (2012), Keshavarz et al., (2010), Fidan & Balci (2017) and Morrison (2002), list and describe the key concepts of a complex adaptive system and how they are exhibited in schools. When these are applied to the implementation of school self-evaluation, they are useful in understanding the potential and limitations of the process and the implications for this study. I will now discuss these key concepts in more detail and their relevance to this research.

3.2.1 The diversity and dynamic nature of agents

Schools exhibit diversity in terms of size, resources, and context. Schools also have a diverse range of members. These include teachers, principals and support staff, as well as pupils, parents and families. Each person is living their own life journey, each with its own unique

history. Each person also possesses their own unique inherent qualities such as likes, dislikes, skills and opinion. Each individual follows their own agenda (goals) and selects their own way of achieving it (strategies) (Stacey, 2005). Complexity theory suggests that such diversity and dynamism among members enables the system to continually generate new possibilities and the resilience needed to sustain itself throughout both challenges and losses (Fenwick, 2012). I would argue that it also has implications for the implementation of school self-evaluation. The scope of self-evaluation can vary from individual teacher reflective practice to schoolwide analysis of data to inform planning and resource allocation (Dyson, 2018). Each member is going to have very diverse views on all aspects of school self-evaluation and its implementation in their school. This is problematic, as the model of school self-evaluation mandated by the DES is a one-size-fits all approach that seeks input from all school stakeholders as a collective process. The whole school in its entirety is seen as the unit of intervention (Simons, 2013; Chapman and Sammons, 2013). Of course, this is nothing new in Irish education policy, with most schools having policies, rules and procedures in place for a variety of things. However, this is at odds with complexity theory which would suggest that schools are a collection of individuals. Each member in a school community is the unit of intervention and action. Schools are fundamentally characterised by the behaviour of individuals such as teachers, principals, and pupils who base and adapt their everyday actions on their interpretation of such policies, rules and procedures. The success of the system is very dependent on the capacity of individuals, in a variety of settings with differing skills, knowledge and ability levels (Ryan and Timmer, 2013). On this basis, a big question for this study was, how could we reach beyond the collective level of a school, to mobilise the change needed at individual level, to implement school self-evaluation successfully? The variety of people involved in a school makes any schoolwork complex (Keshavarz et al. 2010; Fidan and Balci, 2017). The involvement of dynamic and diverse individuals such as teachers, parents, principals, special needs assistants, pupils, deputy principals, board of management members, patron body members, post holders, community members and special education teachers in decision making can add complex layers to the implementation of any school process, such as This may explain the gap between policy and practice which school self-evaluation. McNamara and O'Hara (2012) describe in relation to the implementation of school selfevaluation in Ireland. Whenever a particular set of policy goals are put into place by state agencies, often times the realisation of those policy goals gets lost in implementation challenges. This refers to the people that are actually responsible for enacting those policies, having different understandings of what they mean and how they fit within their particular role,

within the institutional context in which they work (McNamara and O'Hara, 2006; McNamara and O'Hara, 2012).

Complexity theory provides a helpful framework to describe, characterise and better understand the dynamics of this implementation gap, in the hope of avoiding it (Biesta and Osberg, 2010). Schools are dependent on external agencies, like the government, parent groups and management bodies for the provision of supports such as finance and human resources. However, they are also autonomous having the freedom to act within a relatively fixed set of possibilities. School decision making is influenced by school "culture", formal policies, expressed local community needs, the resources available and time. Policies get adapted to local contexts by the diverse people within them and often times they are not exactly what was intended (McNamara et al., 2002; Meuret and Morlaix, 2003; Howlett and Raynor, 2006; Fullan et al., 2006; Ngan et al., 2010; Terhart, 2013; McNamara and Nayir, 2015; Hudson et al., 2019). However, for policies to be implemented successfully acknowledging diversity alone is not sufficient. According to Fenwick (2012), there must also be interaction.

3.2.2 Nested systems, connectedness and patterns of interaction

Most social systems are nested within other systems, co-implicating and co-habiting, yet each retaining its own distinct identity, organising logic and emerging patterns. Within these nested networks, people are interconnected (Holland, 1998; Cilliers, 2001; Keshavarz, 2010; Fenwick, 2012). Schools for example have observable subsystems such as individuals, classes, year groups, and disciplines, all with complex relationship networks and different traits. They are also part of parallel systems such as families, community and friends. In addition, schools are part of larger supra systems. For example, the OECD, the Government of Ireland, the DES, Inspectoral regions and Patron management bodies, each with their own complex relationship networks and diverse traits. Within the structures of these nested systems, diverse interactions are observable between schools and within schools, as well as with families and the local community. For example, teachers interact with pupils, with other teachers, with special needs assistants, with school management and with parents, through interactive processes such as conversation and dialogue. These interactions range from being compulsory (e.g., parent teacher meetings or engagement with the Inspectorate) to voluntary (e.g., being part of a committee or joining a teaching union) and formal (e.g., being called for a job interview) to informal (e.g., having a quick chat at the school gate). As each person is diverse and dynamic and different from the other individuals they meet, information and resources are usually exchanged during these various types of interactions. This relationship between people, as they connect and relate to one another, is critical to a school's success, as the more frequent and powerful these interactions are, the more influence they are likely to have on the behaviour of school members (Holland, 1992, 1999; Keshavarz et al., 2010; Fenwick, 2012), for example learning new ideas, trying new tools, reaching consensus or making sense of new concepts or information. These processes provide an important link between the individual and the collective. Within masses of interaction, the system as a whole can become energised and be sensitive to fluctuation. This puts the system in a state of uncertainty, continually operating with unfolding configurations and a multiplicity of possibilities (Osberg, 2008). This has implications for school self-evaluation. For it to be implemented successfully as a whole school process, members must be allowed to connect with each other and build relationships. To connect with each other, individuals need the resources to do so, for example frequently scheduled meeting time, a meeting place and a meeting plan. To build powerful relationships, individuals need opportunities to share information and knowledge as a collective, with protocols for doing so, such as rules.

Complexity theory contends that in the right conditions, these interactions have the power to alter behaviour patterns at individual level and in turn collective behaviour patterns at wholeschool level (Holland, 1992, 1999; Keshavarz et al., 2010; Fenwick, 2012). This would suggest that for school self-evaluation to be implemented successfully it needs to begin with a 'bottomup' community approach. This however can be problematic. School self-evaluation, as envisaged by the DES is externally mandated (Brown et al., 2016), received by schools as a 'top-down' transmission of information through the education network. This serves to maintain an interaction pattern that fosters compliance and hierarchy (O'Day, 2002). This pattern of interaction caused me to see the school self-evaluation process as little more than a symbolic exercise in response to a formulaic requirement, with an emphasis on compliance, rather than a self-reflective and inclusive learning experience for school members. These inner beliefs affected my behaviour towards school self-evaluation and therefore my leadership practice, hence the significance of this action research study.

Inferences from complexity theory highlighted this as a problem for the school as an organisation and posed certain questions about my management and planning skills as a school leader (Fidan and Balci, 2017). I did not know exactly what to do, or the right process to follow, to allow me make the significant changes needed. Using complexity theory as a framework to understand my school as a complex adaptive system, helped me to adopt a more sophisticated

approach to account for the diverse, complex and context specific nature of my school (Keshavarz et al., 2010) and as such, the challenges and the methods by which an intervention could be structured, introduced and sustained (Cunningham, 2001). Complexity theory provided me with a powerful way to think about how to negotiate the day-to-day challenges involved in identifying and addressing my school's changing needs (Ladd, 2009). In an environment of unpredictability such as a school, complexity theory contends that leaders must try not to manage change, as I was doing, but instead support their school on its change journey, releasing school members to adapt as the school moves forward (Morrison, 2002).

3.2.3 Information flow and feedback loops

Most schools have well organised systems to support the flow of information relating to educational issues. Schools can send and receive information using a variety of communication channels such as text, email, letter, phone call or in-person. Through these channels of reciprocal communication, information can flow back and forth between various networked systems of diverse individual people such as, the DES, a board of management, a school principal, families, staff and pupils. Schools also receive information from many diverse individuals outside the realm of education such as the Health Service Executive, various state departments or local community groups. Schools in turn, can transfer this information to families, staff members or the DES. All schools have systems for feedback, including systems for both internal and external performance reporting. This feedback can be both formal and informal and is usually about educational issues and performance reporting. Schools utilise this feedback to reorganise their plans and practices. Examples can include pupil school reports, parent teacher meetings, assessment portfolios, evaluations, assessment for learning and staff meetings.

According to complexity theorists (Cilliers, 1998) however, this information is not a static entity that is disseminated and unchanged. Rather this information is dynamic and is constantly changing, being interpreted by each individual based on their prior experience and recombined with other information and knowledge they have passed on, through interaction with other people. This movement of information, through the interaction of people, lies at the heart of any learning and change process within a complex system, such as a school (Morrison, 2002; Stacey, 2005). When information is freely provided, and structures support use of that information, people engage more productively in their work (Bower, 2006).

However, O'Day (2002) points out that not all information leads to learning and change in a school and not all learning and change leads to improvement. She describes five conditions which must be in place for information to be useful in any learning or change process. The first condition sets out that in a process such as school self-evaluation, participants must be given access to information, be it from other school members or the school's environment. However, it must not be of such quantity that they become overloaded with it. O'Day (2002) argues that schools are constantly bombarded with information and by demands to do something about that information. She describes schools moving chaotically from one demand or source of information to another, with insufficient focus and time to digest it or learn from it. She describes some schools literally closing their door to new information as a coping strategy to allow them focus. Principals and teachers take protection by isolating themselves in the sanctity of their classrooms, a form of walled buffer from change. This prevents school members from encountering variation and the information it engenders, thus limiting any opportunity to learn and grow from one another and select more successful ways of doing things (Morrison, 2002). Working in isolation such as this, devoid of any meaningful communication or rich feedback, can make it difficult to know how you are doing and how your work fits within the larger school picture (Bower, 2006). This resonates with a study of health promotion programs in Australian primary school by Keshavarz et al. (2010). Using a complex adaptive system framework, it found that information quantity and quality is not always productive in its impact on a school system. They found that in general, schools receive too much formal and informal information. This they contend, causes schools to prioritise communication and information within their network. Information from the DES tended to have a much higher priority to be followed up, than communication from entities outside the education system, such as the local health authority. This information overload has implications for the school self-evaluation process externally mandated by the DES. In the event of information overload and the deployment of coping strategies, school members may resent having no choice but to prioritise such communication and information from the DES at the expense of more local issues. This could affect their motivation to engage with school self-evaluation.

The second condition outlined by O'Day (2002) states that school members must be given the opportunity to attend to information, if it is to prove useful. School members must be given the adequate time and resources needed to deal with information regarding school self-evaluation. Of course, attending to this information may not be enough. The third condition sets out that school members must also have the necessary knowledge and skillset to process this

information. This raises many important questions for any study on the implementation of school self-evaluation, such as, do individuals interpret school self-evaluation the same way? Is there a chance of misinterpretation therefore leading to inadequate information, ineffective learning, unreliable results and maladaptive declining outcomes?

Lastly and most crucially of all for O'Day (2002) however, are the final two conditions of motivation and resources. Both must be in place to act on information. These in particular are critical to the implementation of any school self-evaluation process, for example, are school members individually motivated and resourced enough to engage with and implement school self-evaluation in a meaningful way? (MacBeath, 1999; Schildkamp, 2007; Bubb and Earley, 2008; Vanhoof et al., 2009; O'Brien et al., 2015).

A breakdown in any of the conditions described above – *access*, *attention*, *knowledge*, *motivation* and *resources* – can disrupt the connection between information and change (O'Day, 2002, pg.7), thus limiting the potential of any school self-evaluation process.

3.2.4 Change, adaptability, co-evolution, self-organisation and emergence

Complex systems, such as a school, are part of a broader complex open education system and a wider complex community system. These systems are always in a state of dynamic interaction ad infinitum where change is constant in almost all aspects of school life, occurring with different frequency and intensity (Morrison, 2002; Osberg, 2008; Keshavarz et al., 2010; Fenwick, 2012). Changes in one part of the education system can lead to unpredictable outcomes, just as easy as predictable outcomes, in another part of the education system. Changes maybe planned, for example, a school setting their own internal educational planning priorities. Others however, may be introduced by an external source with no control or preplan, such as the government's mandated policy of school self-evaluation. When there is a change in conditions or new information, schools can choose to make adaptive changes and evolve to ensure a best fit. Through multiple interactions among diverse school members, a unity and coherence can emerge without any imposition or planning (Fenwick, 2012). School members as a diverse, but yet dependent collective, can randomly organise themselves and emerge with a best fit for their context, with the freedom to develop along undetermined alternative trajectories (Osberg, 2008). The informal human architecture of a school can be self-organising, as school leadership structures tend to be flat and flexible, thereby allowing the space needed for any necessary innovation.

These adaptions can be radically indeterminate and open-ended despite any previous incarnations of the school, with the potential to feedback into and influence the evolution and emergence of the system (Osberg, 2008), for example through evaluation studies, audits or practitioner action research. Schools can also undergo internal changes, such as a change in its membership. Here schools can respond and adapt by setting criteria and holding interviews to seek the best fit and most suitable candidate for their school. Schools who are compelled to try new and different things move away from the status quo (or what's called 'equilibrium') to a place of change (or what's called 'the edge of chaos'). It is here that new relationships and new structures are forged causing the school to evolve (Morrison, 2002). However, in the true spirit of complexity theory, the outcome of this evolution is hard to predict. For example, in the case of a newly appointed staff member one possible outcome could see the new member bringing change to school practices. This could happen through the selection, recombination and adaption of information derived from interaction with other school staff members and the school's environment. On the other hand, another possible outcome could see the new staff member gradually learn and become socialised into the strong and stable language, beliefs and routines that make up the dominant behaviour of the school, thereby keeping the school system stable. A schools' general behaviour therefore, is the result of the interplay of multiple factors, and is accordingly an emergent phenomenon that is not easily or fully predictable (Osberg, 2008). A schools' collective behaviour is dependent on the behaviour of its components, rules, interactions, information, values, context, time, and other systems' action, as well as the availability of resources.

3.3 The potential limitations of complexity theory

As outlined above, central concepts from complexity theory, when applied to an organisation such as a school, are particularly instructive in understanding the dilemmas of practice in implementing school self-evaluation. Governments cannot govern and implement policy by themselves, they need partners (Morçöl and Wackhaus, 2009). My school works in partnership with the DES to deliver quality education to young people. To do this, my school is connected to the government through a relatively stable and structured networked system of multiple, interdependent and self-organising representative groups and bodies, for example, the NCSE, NCCA, teaching council, inspectorate, NPC and teacher unions, to name but a few. Whilst my school may be an interdependent part of this structured network and dependent on the government, it has a high degree of autonomy. This has implications for the implementation of

government policy. Policies get adapted to local contexts and often times are not exactly what was intended (Hudson et al., 2019).

There are numerous organisational or social theories I could have used to explore this phenomenon (Haque, 2015; Jones and Bradbury, 2018). Actor network theory for example, is conceptually useful in helping to understand organisational reality. It too believes that systems are complex, composed of multiple interdependent members (actors) with the ability to selforganise (Morçöl and Wackhaus, 2009). It provides fundamentals for understanding uncertainties that are created by human relations and organisational arrangements, in an attempt to study them and bring about stability. It is an approach to understanding humans and their interactions with inanimate objects. For example, it would describe school self-evaluation as a process in which human 'actors' (e.g., teachers, pupils, care staff, school leaders) interact with each other and with aspects of their environment considered to have agentive roles (e.g., inanimate objects such as guidance documents, classrooms, online spaces), to form structured networks which embody change and improvement (Carroll, 2018). Actor network theory sees reality as the stability of these structures. In contrast, complexity theory sees reality as a dynamic and ever-changing series of events. It is more concerned with the dynamic interactions and processes between people within that structure (Morçöl and Wackhaus, 2009). Complexity theory suggests that improvement (adaption) is based on the information (feedback) that school members receive from one another as they interact in a process of school self-evaluation. The interpretation and dispersal of this information leads to the invention, selection and combination of strategies to produce improvement (O'Day, 2002). For that very reason, complexity theory was chosen as the theoretical framework for this study.

However, that is not to say that complexity theory is not without its limitations. Bush (2011) believes that on its own, it is not sufficiently comprehensive to explain behaviour and events in education. He references significant weaknesses in researching schools using a complexity theory approach. To begin, he believes it is difficult to reconcile such a perspective of reality with the customary structures and processes of schools. School members generally understand and accept the broad aims of education. Whilst school members may move in and out of decision-making situations, the broad aims and policy framework of a school remain intact, continually influencing the outcome of any discussions. Secondly, Bush (2011) argues that complexity theory can exaggerate the degree of unpredictability. Schools have many predictable features in terms of professional conduct, expected behaviour, standard rules and procedures. All school members are accountable to someone. Timetables regulate the location

and movement of all school members on a daily basis. There are also generally plans to guide classroom activities. Thirdly, complexity theory according to Bush (2011), seems irrelevant and less appropriate for schools during periods of stability. Levels of predictability can vary from school to school depending on their context.

In its defence however, Cunningham (2001) believes that complexity theory makes a valuable contribution to educational research, as conditions in schools may be too uncertain to allow an informed choice among alternatives. Its emphasis on the unpredictability of organisations is a significant counter to the view that problems can be solved through a rational process (Bush, 2011).

3.4 Conclusion

An ever changing and complex environment poses certain questions about the strategic management and strategic planning skills of school principals (Fidan & Balci, 2017). Complexity theory provides guidance and a framework for a school leader to think about their role in leading school self-evaluation in a complex system such as a school. Complexity theory provides a powerful way to think about the interaction needed between various components in a school system in order to implement school self-evaluation successfully (Morrison, 2002). I used the following key concepts from complexity theory as my lens in this study.

Each member of school staff is going to have their own unique interpretation of school selfevaluation and its implementation in our school. The process mandated by the DES is a onesize-fits all collective approach where the school as a whole is seen as the unit of intervention. In contrast however, complexity theory describes each individual member of staff as the unit of action. Careful consideration and acknowledgment were therefore given to the diversity between school members regarding the implementation of school self-evaluation in this study.

Within a nested system structure, such as a school, diverse interactions can occur between the diverse people within it. If these interactions are frequent and meaningful, they have the power and potential, according to complexity theorists, to influence individual and collective behaviour (Holland, 1992, 1999; Keshavarz et al., 2010; Fenwick, 2012) regarding the implementation of school self-evaluation. The movement of information through interaction such as this, lies at the heart of any learning and change process within a complex system such as a school and allows staff members to engage more productively with their work. In this study therefore, careful consideration was given to facilitate, in as much as possible, effective

interaction between school members, in an attempt to encourage the flow of information between them, regarding their experiences of implementing school self-evaluation.

From this process, school members as a diverse, but yet dependent collective can, according to complexity theorists, randomly self-organise, forging new relationships and new structures, and emerge with the best fit for their school context. Any approach to school self-evaluation is an emergent product of many factors over time. This study was very cognisant therefore of the complexity of our school, and its functioning as a complex adaptive system.

Chapter 4 – Research Methodology

4.1 Introduction

This action research study is focused on how to improve the implementation of school selfevaluation in my school. The purpose of this chapter is to outline the design of this research and the approach taken. I discuss the methodological choices made and their implications for the research methodology, research methods and data analysis techniques adopted. To begin I will discuss the aims of the research and provide details of the research questions, as well as a description of the setting and the participants involved. This is followed by a discussion of the key data collection and analysis techniques used. To conclude, ethical issues and how these were dealt with are then discussed.

4.2 Research aims

The overall aim of this action research study was to improve the school self-evaluation process in our school based on a '*bottom-up*' approach to change, putting school stakeholders at the centre of all efforts to improve the implementation of school self-evaluation in our school (Chapman and Sammons, 2013, pg.6). In our school self-evaluation process, we must learn to identify our weaknesses and embrace our strengths for the betterment of all. We need to learn how to gather information and feedback from all the stakeholders in our school community and negotiate realistic solutions and decisions through agreed consensus. This action research was an opportunity to engage in such activity and document our experiences. However, in such a 'bottom-up' approach, Smith (2012) cautions against the simple '*replication*' of imported school self-evaluation models (pg.150-152). Evidence suggests the need for approaches to school improvement that are more suitably adjusted to the setting (Harris and Chrispeels, 2009). This study therefore reflected on our school context and the appropriate blend of elements to optimise the implementation of school self-evaluation in our school community.

4.3 Research setting

The setting for this research project is a co-ed, vertical, single stream, rural school, under Catholic patronage. The school is made up of two hundred and sixteen pupils, four school support staff, eight classroom teachers, two special education teachers (one of whom is the deputy principal), one shared resource teacher, two special needs assistants and an administrative principal. I, the author of the research, am the administrative principal.

4.4 Research participants

The school is a close-knit community of eighteen colleagues (teachers, special needs assistants and school support staff) who work hard every day supporting each other, our pupils and their school community. Part of my daily lived experience, as school principal, involves working within that team. I see myself as part of that team. Having worked with these staff colleagues over the past nine years, I was very aware of how knowledgeable and talented they were. Their input to this research project was seen as positive, as they brought a wealth of in-depth experience, skills and insight. Such a mixture of staff offered a broad range of views and experiences. Each colleague was different in terms of their focus, their daily activities and their life experience. The entire fourteen members of the teaching and care needs staff gave consent to take part in this research. Out of these fourteen staff members, eight were classroom teachers, three were special education teachers (of which one is deputy principal), one was an administrative principal and two were special needs assistants. Twelve were female and two were male. Three of the teaching staff had over twenty years teaching experience, five teachers had ten years' experience or more, whilst the remaining four had less than 10 years' experience. The two special needs assistants also had less than 10 years' experience. For six members of the teaching staff, this was their first school, and for the two special needs assistants, this was also their first school. Such diversity and range of views helped to triangulate the data gathered, open up questions for further research and ensure the research was rooted in a community approach.

4.5 Research Paradigm

In this section I describe in general terms the assumptions, beliefs, norms and values of research paradigms and the implications of these on the positionings, values and judgements of researchers (Creswell, 1994). I then discuss and present the rationale for selecting the pragmatic paradigm applied in this action research study.

A research paradigm is a model of research that reflects a general agreement on the nature of the world and how to investigate it (Bassey, 1990). It is a set of beliefs and principles that shape how a researcher sees the world (MacKenzie and Knipe, 2006) and how they interpret and act within that world. It helps a researcher to consider the fundamental beliefs underpinning their research process and how these frame the ways in which they understand the world around them (Burton and Bartlett, 2009). Paradigms can clarify and organise a researcher's thinking

about their work (Cohen et al., 2018). According to Lincoln and Guba (1985), a paradigm comprises four elements, namely, ontology, epistemology, methodology and axiology. Hitchcock and Hughes (1995) suggest that ontological assumptions (assumptions about the nature of reality and the nature of things) give rise to epistemological assumptions (ways of researching and enquiring into the nature of reality and the nature of things); these, in turn, give rise to methodological considerations; and these, in turn, give rise to issues of instrumentation and data collection. Added to ontology and epistemology is axiology (the values and beliefs that we hold), which Hitchcock and Hughes (1995) describe as being concerned with understanding the world, by how we view our world(s), what we take understandings to be, what we see as the purposes of understanding and what is deemed valuable (cited in Cohen et al., 2018, pg.3). A researcher's work is upheld and guided by the ontological, epistemological, methodological and axiological perspectives of their chosen paradigm (Guba and Lincoln, 1994; Keeves, 1997; Benton and Craib, 2001; Burton and Bartlett, 2009; Cooksey and McDonald, 2011; Kivunja and Kuyini, 2017; Denzin and Lincoln, 2018).

I will now outline the research paradigm which underpinned the assumptions, beliefs, norms and values of this work, before addressing the research methodology used in this study.

4.5.1 Pragmatic paradigm

The aim of this study was to improve the implementation of school self-evaluation in our school. The problem as leader of the process was to stay true to my core values of authenticity and collaboration. It was important to me that any approach to school self-evaluation in our school adopted a practitioner-based and collaborative approach. Having considered interpretivism, criticality and constructivism, the position I eventually adopted was pragmatism because it acknowledges that reality, truth and objectivity can be viewed from multiple perspectives.

Pragmatists argue that it is not possible to access 'truth' in the real world by using the single scientific method as championed by the positivist paradigm. Equally however, they also argue that it is not possible to determine social reality constructed under an interpretivist paradigm (Patton, 2002; Alise and Teddlie, 2010; Biesta, 2010; Creswell, 2014; Mertens, 2014). Instead, pragmatism advocates the use of multiple methods as a pragmatic way to understand human behaviour. It therefore focuses on using a combination of research tools to gather both quantitative and qualitative data, but more importantly, data that is useful (Greenwood and Levin, 2006) to achieve what Dewey (1998, 1938) calls *'warranted assertability'* (pg.161).

That is, strong data which is meaningful to us and to those who share our context (Lofthouse et al., 2012). This evidence gives us the support and confidence we need to guide future practice (Hibberts and Johnson, 2012), choosing whether to continue with current practice, to scrap something that no longer works or to introduce something new (Dewey, 1922, 1938). According to Powell (2001) such an approach can help organisations such as schools to solve problems and implement education initiatives, such as school self-evaluation, in their own context. For that reason, the pragmatic paradigm was applied in this study. Research conducted within this paradigm advocates the use of multiple research methods according to need. One such methodology is action research, which through a focus on qualitative data and improvement in practice, has the potential to "professionalise the work of educators...by reducing the gap between theory and practice" (Ary et al., 2010, pg.516).

4.6 Action research

In this section I discuss the methodological commonalities between action research and school self-evaluation. I outline how action research has influenced the model of school self-evaluation in Ireland, a model of active internal review within local settings based on collaboration, inclusion and continuous reflection. I describe the action research method and its application to education. I then explain my decision not to use participatory action research (PAR), but instead an action research approach (AR) to study the implementation of school self-evaluation in my school, and the importance of interaction and collaboration in the process amongst school members.

4.6.1 Action research and school self-evaluation

It is often argued that the scope of school self-evaluation can be wide and embrace several activities, ranging from the process evaluation of classroom practices to organisational change (Blok et al., 2008). Policies designed to support improvement, such as school self-evaluation, can often be traced back to traditions associated with Kurt Lewin i.e., action research (Chapman and Sammons, 2013). To further conceptualise school self-evaluation Sjøbakken and Dobson (2013) draw parallels with action research as defined by Carr and Kemmis (1986) i.e., a form of self-reflective enquiry undertaken by participants in social situations, in order to improve the rationality and justice of their own practices, their understanding of these practices, and the situations in which the practices are carried out. The inspectorate in Ireland similarly draws on the principles of action research in its model of school self-evaluation. The inspectorate describes it as a collaborative, inclusive and reflective iterative process of internal review

which involves gathering evidence from a range of sources, analysing it, interpreting it and acting on it with a view to bringing about improvements in pupils' learning (DES, 2016). Sjøbakken et al. (2013) claim that both action research and school self-evaluation seek to move beyond a technical and practical interest. The goal instead they argue, is to increase the level of reflection by stakeholders and institute forms of change to school practice that are more inclusionary, democratic and emancipatory. Practitioners in this instance assume ownership of the improvement process through professional enquiry (Walters, 2014). Such a participatory approach is vital because action research, like school self-evaluation, is something that schools do to themselves, by themselves and for themselves (Swaffield and MacBeath, 2005). Similarly, school self-evaluation, like action research, is often viewed as a 'cyclical process' whereby a school itself, describes and assesses aspects of its own functioning in a systematic manner (Vanhhof & Van Petegem, 2012 pg.50). The similarities do not end there. In both school self-evaluation and action research, school teachers shift from a role of passiveness and compliance, to an active role, developing their own agenda, focusing on areas of relevance to their own context using self-developed evaluation and improvement strategies to initiate a process of change (MacBeath, 2008). Evaluations need to be custom fit to the situation, with the interventions being systematically and deliberately designed with partnerships at their core (McNaughton et al., 2013). According to McNaughton et al. (2013), effective research-school partnerships cannot be left to chance, but rather must be developed systematically from the start of any intervention. This is important, as initiatives such as action research and school self-evaluation can contribute in efforts to turn schools into effective learning environments (Karagiorgi et al., 2018).

Using an action research approach was therefore an obvious choice in this study of change due to its methodological parallels with school self-evaluation. This study of school self-evaluation in our school presented us with a prime opportunity to engage in a methodology and change process at a substantively deep level. One that was well grounded in both theory and practice and supported by our vision for change in our own school.

Action research has been defined as "a participatory democratic process concerned with developing practical knowing in the pursuit of worthwhile human purposes" (Reason & Bradbury, 2001, pg. 1). Within action-oriented research processes, Auemaneekul (2010) describes three broad types: Action Research (AR), Participatory Action Research (PAR) and Community Based Participation Research (CBPR) (pg.86). The methods on this continuum share many similarities, such as active participation, open-ended objectives and high levels of

commitment from the researcher and the participants to the research problem and active learning (Bell et al., 2004). They differ however, in their specific end goal and the methods used to get there. Each has a different purpose reflecting a different level of commitment and influence.

In AR, it is the researcher who initiates change based on a feeling that something needs to change to create a better situation. The researcher may act as an individual or with a team of colleagues as the facilitator. Although the researcher can be studying themselves in the context of working with an organisation, it can be a collaborative effort when the organisation is being supported by the action research process. In this scenario, the researcher and the group identify actions to take and jointly analyse the results. They then reflect on these actions and results and propose new courses of action. Here, the researcher and the clients are acting together to create or actualise satisfying results for change. But at all times it is the researcher who leads the process of identifying the problem, drawing facts and opinions from colleagues, and leading the group to identify any gaps in understanding.

In contrast however, PAR is initiated by the organisation itself. Here the research approach is jointly designed through discussions with professional researchers and active participation by some members of the organisation. Community-based participatory research (CBPR) is similar in approach to PAR but instead, it takes place in community settings and involves community members as full and equal partners in all phases of the research, from design to implementation. Unlike AR, participants carry out the data collection and analyse the results.

Educational Action Research (EAR) can be placed within this broad continuum of action research. It applies the concepts and practices of action research to the context of education (Rauch et al., 2019) in an attempt to improve educational outcomes for pupils, address educational inequality, and create more critical and innovative educational practice (Brydon-Miller et al., 2017). An action-oriented research process is a potent methodology for educational reform, because of its core principle of combining action with research to challenge the routines of the status quo. It gives those who carry it out, a means to develop agency to bring about changes which are locally appropriate, within the globalised world they live in (Somekh et al., 2009). It is applied research using a cyclical, action-reflection model, carried out by practitioners who have themselves identified a need for change or improvement in an organisation (Bell, 2005; Noffke & Somekh, 2009).

It can be carried out by individual teachers in their own classrooms, by groups of teachers working in collaboration or at school, community, regional or national level to address a common question or concern (Brydon-Miller et al., 2017). Examples include classroom-based action research, teacher as researcher and practitioner inquiry. Educational action researchers are aware of the many diverse conceptions of value which may exist within a school setting and the uncertainty this can bring to any possible action. Within a school setting therefore, they must engage in ongoing cycles of reflection on their actions and thoughts to inform further planning, future actions and understandings. Through this cyclical process, they contribute to the conceptual and practical knowledge of education (Rauch et al., 2019).

The fundamental aim of action research according to Elliot (1991) is to improve practice. It is a way for people to take action in their personal and social situations with a view to improving them (McNiff, 2017). The focus of action research is to empower people to change their social situation and to raise awareness on a particular issue (Van Der Voort and Wood, 2014). This resonates with Kemmis and McTaggert's (1988) definition of action research as a form of collective self-enquiry by participants in social situations in order to improve the rationality and justice of their own social and educational practices, as well as their understanding of these practices and the situations in which these practices are carried out (cited in Cohen et al., 2007, pg.298). Action research can therefore, according to McNiff (2017), be a powerfully liberating form of professional enquiry where practitioners investigate their practices through a process of action and reflection, finding ways to live more fully in the direction of their personal and social values.

Leitch and Day (2000) argue however, that there are three different modes of action research, which according to purpose, will have different kinds of benefits. The first, '*technical action research*', is related to solving issues as a means of increasing efficiency in delivering the curriculum. It is likely to be orientated towards the development of pedagogical strategies, skills and tactics (pg. 183). The second, '*practical action research*', aims to improve practice through the application of practical judgement and the accumulated personal wisdom of the teacher. It is as concerned with building problem-solving skills and self-evaluation capacities, as with the end product of enquiry (Leitch and Day, 2000, pg. 182). Elliot's (1991) action research cycle with its recurrent feature of '*reconnaissance*' to analyse and reflect on a situation, provides an approach for undertaking practical action research of this kind (pgs.73-75; Leitch and Day, 2000, pg.184). In keeping with this orientation towards practical problem-

solving, self-evaluation and improved practice, this study adopted Elliot's (1991) action research design.

The third mode '*emancipatory action research*', is concerned with understanding the broader social and policy effects on teaching and learning. Within emancipatory action research, Leitch and Day (2000) distinguish two separate camps. The first camp is focused upon individual practice, typified by the work of Whitehead and McNiff (2006). Here researchers systematically relate their work to their values and draw on these values as standards of judgement for their work (Karagiorgi et al., 2018). This requires engaging in a process of explaining your present practice in terms of an evaluation of your past, with a view to creating improvements in present and future contexts. Through such processes, researchers construct their own 'living educational theory' (Leitch and Day., 2000). This project drew upon the work of Whitehead and McNiff (2006) to explore and describe how my values of authenticity and collaboration guided my school leadership practice. This allowed me to reflect on my practice and see how my hierarchal leadership of an unauthentic school self-evaluation process, negated those values. By leading the school self-evaluation process to become more meaningful, therein achieving genuine school improvement.

The second camp is outwardly focused on the social or educational system, seeing action research as a critical activity not only to increase the level of reflection by stakeholders, but also to increase participation by bringing change to school practices that are more inclusionary, democratic and emancipatory (cited in Sjøbakken et al., 2013, pg.171). Anything else, they claim, is not true action research (cited in Karagiorgi et al. 2018, pg.240).

This focus on increased participation is endorsed by Klenowski and Woods (2013). They believe there is much to be gained by utilising a participatory and democratic action research methodology for school self-evaluation. An action research methodology which operates within a participatory enquiry paradigm with a democratic ethic they argue, attempts to represent a wide range of perspectives and involve more stakeholders in the school self-evaluation process, with increased opportunities for dialogue and deliberation. Upholding the principles of such a participatory and democratic approach to evaluation they believe, will make the practice of school self-evaluation for improvement more ethical and respectful. The involvement of pupils, teachers, parents and the wider school community is seen as a crucial factor in determining the success of participatory modes of evaluation (Hopkins, 2001).

However, regardless of whatever model of action research is used to underpin a school selfevaluation process, Vanhoof et al. (2011), argue that "the way in which school self-evaluation is conducted is not something which can be considered in isolation from the broader functioning of the school" (pg. 284). That is because context matters, where local variations of a national problem require local solutions created by local communities (Lai, 2013). It is important to allow contextually driven approaches to deliver improvement with flexibility and local ownership being encouraged (Parr and Timperley, 2013). As Bana (2010) suggests, participatory action research allows for mutual consensus at all levels of research and provides the opportunity to stakeholders "to initiate the process of improvement in their own context, by close examination of the effects of their own practices" (pg.218). Continuous improvement occurs when individuals within a working environment collaborate with one another to understand and explain action (Van Der Voort and Wood, 2014). The methods used should be interactive and iterative, requiring an engagement with critical questioning, reflection and an action learning process (Klenowski and Woods, 2013). Indeed, the contribution of action research to school improvement is well documented. Various studies point towards the employment of action research as a powerful tool for school improvement (Karagiorgi et al., 2018). However, to make this a reality, it is critical to have buy in to the process from school staff. Research shows both teachers and more importantly for this study, school leaders to be the key change agents for improvement (MacBeath, 1999; Hopkins et al., 2001; Leithwood et al., 2006; Chapman and Sammons, 2013; Fullan, 2015).

4.6.2 Action research and school leadership

In this section, I discuss the positive impact that engagement with action research can have on my role as a school leader. I describe the transformative impact that critical reflection and values-based action research can have on school leaders and their school. I then discuss briefly the work of Jack Whitehead (2006) and Jean McNiff (2017), two leading scholars in the field of values-based action research.

According to McNaughton et al. (2013), leadership plays a key role in overseeing the use of evidence to define problems, designing solutions and evaluating those solutions through successive iterations. Various guides (Elliot, 1991; Whitehead and McNiff, 2006; Mertler, 2013; McNiff, 2017) describe and discuss the knowledge and skills participants need to apply such a problem-solving action research approach within their own context (Karagiorgi et al., 2018). The strategies and techniques proposed for school leaders are not different from those

of teachers (Glanz, 2014). Various studies have tried to document the impact of a leader's engagement with action research on a personal level, as well as a school level (Karagiorgi et al., 2018). Karagiorgi et al. (2018) contend that a leader's involvement in action research impacts positively on their role and approach to improving instruction and learning. Other studies describe changes in beliefs and values for school leaders involved in action research and the positive impacts this can have for their schools. For example, Wood and Govender (2013) highlight the contribution of systematic engagement with critical reflection and action to the development of leaders' epistemological and ontological shifts and changes in their perceptions and leadership practices. Wood and Damons (2013) also describe the energising power of values-based action research in helping school principals to lead in transformative ways and the positive impact on their school environment (cited in Karagiorgi et al., 2018, pg.241).

The work of school principals, regardless of context, is informed and driven by their strong, clearly articulated moral and ethical values (Day and Sammons, 2014). I have been inspired by the writings of Jack Whitehead (2006) and Jean McNiff (2017) who provide numerous examples of education practitioners (e.g., Bosher, 2001; DeLong, 2002; McDonagh, 2007; Cahill, 2007; Sullivan, 2007) engaging in a process of personal learning in their day-to-day professional life using insights from the paradigm of action research. These practitioners share written accounts of their attempts to improve their practice. In the research process they begin with an exploration of their core values and question how these are being lived in their daily practice and what improvements could be made to realise them. This is in an attempt to improve the teaching and learning experiences of those they encounter. To do this, they develop their critical reflective judgement and learn how to integrate any new learning into their daily lived practice. This project drew upon the work of Whitehead and McNiff (2006) to explore my values of authenticity and collaboration and how they were being lived in my leadership of school self-evaluation.

4.6.3 Action research in this project

In this section I describe the action research approach used in this project with a brief description of each action research cycle. To conclude, I present a pictorial diagram of the research design.

This project adopted an AR approach to reflect on and examine in a systematic and careful way my own values towards the improvement of my leadership practice. At the same time, it focused on a collaborative intervention under my leadership, designed to meet the identified needs of staff members in their implementation of school self-evaluation. This was in keeping with the suggestion by Swaffield and MacBeath (2005) that school self-evaluation is something that schools do to themselves, by themselves and for themselves. Through the active involvement of staff members in the research process, the aim was to empower our school to implement school self-evaluation in a more effective, authentic and meaningful way. To do this, the project adopted a practitioner-based approach with the expectation that any improvement strategies were to be developed by the school members themselves (Reason and Bradbury, 2001; Bana 2010; Klenowski and Woods, 2013; Walters, 2014; Jacobs, 2016). A self-evaluating school describes and assesses aspects of its own functioning in a systematic manner, developing its own agenda, focusing on areas of relevance to its own context (Davies & Rudd, 2001). In this regard, the project assumed that the school self-evaluation mechanisms, as well as the improvement strategies, would be developed by the school itself.

The project was oriented towards school improvement (SI), rather than school effectiveness (SE). This approach was in line with evaluation schemes at national level. An emphasis on school improvement also safeguarded against instrumentalist, rationalistic and managerial approaches to educational change (De Crauwe & Naidoo, 2004; Mulford, 2005; Blok et al., 2008). It is often argued that the scope of SSE can be wide and embrace several activities, ranging from the evaluation of classroom practices (action-research), school-focused in-service education for teachers and continuous organisational change to more mechanistic modes of accountability such as cost-benefit analysis and performance-based education (De Crauwe & Naidoo, 2004; Blok et al., 2008). In consideration of the breadth of emphasis within school self-evaluation, the teaching and learning of handwriting was chosen as the area of focus in this action research project.

In keeping with the orientation towards school improvement, the project adopted an action research design. Following Elliot's (1991) model of action research (*Figure 4.1*) as a guide, each research cycle used varying methods of data collection and involved a number of staged activities as follows:

1. General idea

For Elliot, the general idea is essentially a statement which links an idea to action, informing the reader of the situation to be improved, highlighting the central question to be addressed in each cycle. The situation should involve something that the researcher can impact or effect and measure.

2. Reconnaissance

This involves outlining the situation and describing the nature of the issue that the researcher wants to change or improve upon. This includes linking what is already known and what has already been uncovered through research.

3. General planning

The general plan outlines what is going to be changed during this cycle of research in order to improve the situation, clearly stating what is being attempted.

4. Developing and implementing action steps

Here, the researcher outlines exactly what course of action is going to be taken and how it will be implemented in practice. This stage gives details about how the plans will be put into operation, noting any challenges that may be useful for planning future cycles of research.

5. Monitoring the implementation of action steps and their impact

During this stage, the researcher shifts from planning and implementation, to gathering and analysing data to understand the effects of actions implemented.

1	and the second
IDENTIFYING INITIAL IDEA	
INITACIO	
RECONNAISSAINCE (fact finding & analysis)	
[PBCI IIII O	
+	
GENERAL PLAN	
ACTION STEPS 1	
ACTION STEPS 2 ACTION STEPS 3	and the second sec
& ACTION STELE	WER AS THE STATE OF A STATE OF A STATE OF A
IMPLEMENT	
ACTION STE	PS 1
MONITOR IMPLEMENTA-	a state of the sta
TION & EFFECTS	
'RECONNAISSANCE'	REVISE GENERAL IDEA
lovelain any tailure to	
implement, and effects)	the second s
	AMENDED PLAN
	ACTION STEPS 1
	ACTION STEPS 2 IMPLEMENT NEXT
	ACTION STEPS 3
and the second state of the second state of the	ACHOICOU
U N	
MONITOR IMPLEMENTA	
TION & EFFECTS	
'RECONNAISSANCE'	REVISE GENERAL IDEA
(explain any failure to implement, and effects)	
implement, and effects)	AMENDED PLAN
	ACTION STEPS 1
	ACTION STEPS 2 IMPLEMENT NEXT
	ACTION STEPS 3
	ACTION STEPS 5
n	
MONITOR IMPLEMENTA-	
MONITOR IMPLEMENTA- TION & EFFECTS	
LINIT & EFFECTS	
'RECONNAISSANCE'	
(explain any failure to implement, and effects)	
president, und effects)	and the second second second second second

Figure 4.1 Elliot's (1991) model of action research (pg.71)

Elliot's (1991) model of action research provided opportunities for both myself and school staff to participate in reflective processes, decision-making and actions, and take ownership of the interventions for improved school self-evaluation in our school. Hence, the study involved much collaboration and joint effort between school staff members.

All staff were called together at a meeting to explain fully the purposes of the project and invited to take part. The first cycle of action research began in January 2020 and involved fourteen staff members. Our aim in this first phase of action research was to raise our awareness of school self-evaluation for school improvement and to help us acquire the knowledge, skills and will to conduct such an evaluation successfully. Our reflection during and after this first cycle of action research resulted in significant learning on our part that guided our next phase of action research. The second cycle began in September 2020 and involved fourteen staff members. In each cycle, staff were supported to engage with school self-evaluation, but in all cases, they chose the focus of the evaluation and decided on the data to be collected and how.

My Research Design

based on Elliot's (1991) model of action research







4.7 Data collection and analysis

In this section, I discuss the multiple sources of data in this action research study. I describe in detail the methods used in terms of data collection and analysis and at what stages the methods were used in the project. To help the reader *Table 4.1* below outlines the overall population for each data gathering process in terms of the number invited to participate, the criteria for inclusion, the level of response and the profile of those who participated.

Table 4.1	Sample of participants f	or each data oathering	process within the study
1 ubie 4.1	Sumple of puril ipunis j	or each adia guinering	process within the study

Sample of participants for each data gathering process within this study						
Overall population in this study for data gathering purposes:		14				
Data collection tool and [identificatio n Code]	Description	Number of participants invited to the process	Criteria for inclusion	Number of participants who attended the process	Profile of those that responded	
Survey 1 [N/A]	SWOT analysis	14	Member of school staff	14	12 Teachers and 2 SNAs	
Survey 2 [N/A]	A tool to generate ideas called 'Brain Writing 6-3-5'	14	Member of school staff	14	12 Teachers and 2 SNAs	
Survey 3 [S3 respondent 1 etc.]	25 Likert- scale and 3 yes/no questions	14	Member of school staff	12	10 Teachers and 2 SNAs	
Survey 4 [N/A]	Questionnair e to select their three preferred definitions of SSE from a list of nineteen	14	Member of school staff	14	12 Teachers and 2 SNAs	
Survey 5 [N/A]	Questionnair e to select one preferred definitions of SSE from a list of three	14	Member of school staff	14	12 Teachers and 2 SNAs	

Survey 6 [S6 respondent 1 etc.]	Survey of responses to draft a definition of SSE for our school	14	Member of school staff	7	7 Teachers
Survey 7 [S7 respondent 1 etc.]	Survey of responses to a final draft definition of SSE for our school	14	Member of school staff	9	9 Teachers
Survey 8 [S8 respondent 1 etc.]	List all the activities undertaken in our school for the purposes of accountabilit y and improvement	14	Member of school staff	9	9 Teachers
Survey 9 [S9 respondent 1 etc.]	Survey of responses to draft a graphic of all the evaluation activities undertaken in our school	14	Member of staff	4	4 Teachers
Focus group Interviews [interviewee A etc.]	Semi- structured interviews	14	Member of staff	4	4 Teachers

4.7.1 Data collection

Surveys

Surveys gather data at a particular point in time with the intention of describing the nature of existing conditions (Cohen et al. 2018). Check and Schutt (2012) define survey research as *"the collection of information from a sample of individuals through their responses to questions"* (pg.160). Surveys are often used in social research to describe and explore human behaviour (Singleton and Straits,2010). A survey probes for respondents' perceptions and is particularly suited to monitor any change in opinions and attitudes regarding school processes (Muijs,

2012). Their use is referenced in the literature on school effectiveness (Hendriks et al., 2002). It can range from asking a few targeted questions of individuals in a school to obtain information related to their behaviours and preferences, to larger studies such as public political opinion on a national scale. Survey research can use a variety of data collection methods, the most common being questionnaires and interviews. Survey research can use quantitative research strategies (e.g., using questionnaires with numerically rated items), qualitative research strategies (e.g., using open-ended questions), or both strategies (i.e., mixed methods) (Singleton and Straits, 2010). Dillman et al. (2014) advocate the use of multiple methods for survey research, if no one method is adequate enough to address the planned research aims. For example, a multiple methods survey research approach may begin with distributing a questionnaire and then following it up with interviews to clarify any unclear survey responses (Singleton and Straits, 2010). Using a variety of methods such as this in a study gives a more comprehensive picture than one which does not (De Vos et al., 2007; Babbie, 2012; Cohen et al., 2018). In this action research study, surveys and interviews were both used to collect enough information to allow me begin the next phases of action research in terms of interpretation and implementation. In this study, nine survey instruments were administered to staff members in both paper format and online.

The first survey was used in the initial reconnaissance phase to ascertain where we were as a school, regarding the school self-evaluation process. The method of survey used was a strengths, weakness, opportunities and threats (SWOT) analysis. This qualitative research tool is used to assess the internal and external factors that influence and impact an organisation. It gives a snap shot of the 'current reality'. It is a vital step in both strategic planning and organisational design or redesign. The objective is to look at what's working well and what should be changed or improved. The SWOT analysis was completed by staff members collectively with broad input and discussion. The outcomes from it informed our decisions about action planning and setting priorities (Hart, 2007). Participants in this survey were not coded to identify the source of the data.

The second survey 'Brain Writing 6-3-5' was used in the initial reconnaissance phase to identify new ideas or solutions to improve the school self-evaluation process in our school. Brain Writing 6-3-5 is a silent brainstorming process developed by King and Schlicksupp (1998) to create ideas. The goal is to generate as many creative ideas as possible. As an approach, Brain Writing 6-3-5 (Hart, 2007) effectively lessens power differentials to create a richer dialogic and inclusive community, ensuring that highly verbal people do not overwhelm

quieter members. It enables individuals to see what others have written, allowing creative ideas to be shared for mutual benefit, but in a silent way. The process is conducted in five minutes or less using worksheets which teachers work on individually and privately at first but then publicly share their thoughts. The question in this case was 'how can we improve the process of school self-evaluation in our school?'. Participants in this survey were not coded to identify the source of the data.

The third survey was administered to monitor the implementation and effects of action steps taken in action research cycle 1 (Appendix 2). The survey consisted of two sections. Section 'A' compromised twenty-five Likert-scale questions and three yes/no questions regarding staff members' attitudes towards school self-evaluation in terms of input, process, output, outcomes and unintended consequences. In this section we used questions asked by the 'Centre for Evaluation, Quality and Inspection' in their national survey of principals' attitudes towards school self-evaluation in 2014/15 (O'Hara et al., 2016). This structure allowed us to get a quick snapshot of the current reality regarding staff members' attitudes towards school selfevaluation. In addition, in section 'B', I asked staff members six open-ended questions about their understanding of school self-evaluation and its implementation. This was an attempt to ascertain staff member's lived experience of school self-evaluation as a government policy. These additional questions were influenced by my ongoing reading of the literature regarding school self-evaluation, especially the work of Vanhoof et al. (2009) who explore the influence of teacher attitudes on school self-evaluation implementation, Mutch (2013) who explores the strengths and weaknesses of internal and external perspectives within the school evaluation process and in an Irish context, the work of O'Brien et al. (2019), who explore Irish teacher attitudes to school self-evaluation and the gaps which can occur between school self-evaluation policy and its implementation at local level.

For audit trail purposes, participants responses to this survey were coded systematically with the identification code 'S3 respondent' and a unique number e.g., S3 respondent 1, S3 respondent 2 etc. This same coding process was carried out in all subsequent surveys. However, this unique number was not assigned to the same individual across the different data collection processes in this study.

The fourth survey was administered to staff members as part of an agreed action step in action research cycle 2 (Appendix 3). Nineteen definitions of school self-evaluation from the literature (Swaffield and MacBeath, 2005; MacBeath, 2005; Hopkins, 2005; Schildkamp, 2007; Vanhoof
and Petegem, 2010) were compiled into an online questionnaire. Staff members were requested to select from this list, the three definitions of school self-evaluation they most agreed with. The aim was to reach consensus as a staff on a definition of school self-evaluation which would be the most suitable for our school context. Participants responses in this survey were not coded to identify the source of the data.

The fifth survey was administered as part of an agreed action step in action research cycle 2 (Appendix 4). The top three definitions of school self-evaluation chosen by staff members were compiled into a multiple-choice questionnaire. Staff were asked to select the one definition of school self-evaluation they most agreed with. The aim was to reach consensus as a staff on a definition of school self-evaluation suitable for our school context. The participants' responses in this survey were not coded to identify the source of the data.

The sixth survey was administered to school staff as part of an agreed action step in action research cycle 2 (Appendix 5). A draft definition of school self-evaluation was distributed to all staff members. They were told that this definition came from a process of reducing nineteen definitions down to five definitions using their votes, with key words and phrases from the final five being interwoven into one definition. The survey asked staff members for their thoughts on the draft definition of school self-evaluation and if they agreed with it. The survey also asked staff members had they any comments, thoughts or ideas on the way we reached the definition. To respond to the draft definition using a mixture of multiple choice and open-ended questions. The aim was to reach consensus as a staff on a definition of school self-evaluation for our school context. For audit trail purposes, participants responses to this survey were coded systematically with the identification code 'S6 respondent' and a unique number e.g., S6 respondent 1, S6 respondent 2 etc.

The seventh survey was administered to school staff as part of an agreed action step in action research cycle 2 (Appendix 6). A final draft definition of school self-evaluation was distributed to all staff members. Staff were asked to respond to the draft definition using a mixture of multiple choice and open-ended questions. The aim was to reach consensus as a staff on a final definition of school self-evaluation for our school context. For audit trail purposes, participants responses to this survey were coded systematically with the identification code 'S7 respondent' and a unique number e.g., S7 respondent 1, S7 respondent 2 etc.

The eighth survey was administered as part of an agreed action step in action research cycle 2 (Appendix 7). Staff were requested to list all the activities that are undertaken in our school for

accountability and improvement, why are they being done and for whom? This was in an effort to see if staff members deemed them to be for either accountability or improvement purposes. The intention was to see was there a diversity of opinion in relation to school self-evaluation, as this could have implications for its implementation in our school. As an aid, they were given a prompt in the form of open-ended questions. If they could not think of anything to contribute, they had the option to write the word 'pass'. For audit trail purposes, participants responses to this survey were coded systematically with the identification code 'S8 respondent' and a unique number e.g., S8 respondent 1, S8 respondent 2 etc.

The ninth and final survey was administered as part of an agreed action step in action research cycle 2 (Appendix 8). Staff were asked to respond, through a mixture of multiple choice and open-ended questions, to the final draft of a graphic diagram which communicates and explains all the evaluation activities undertaken in our school. For audit trail purposes, participants responses to this survey were coded systematically with the identification code 'S9 respondent' and a unique number e.g., S9 respondent 1, S9 respondent 2 etc.

Interview

Conducting interviews is another approach to data collection used in survey research (Cohen et al., 2018). An interviewer can use probing comments to obtain more information about a question or topic and can request clarification of an unclear response (Singleton and Straits, 2009). My choice of interview as a research tool was motivated by my interest in understanding the lived experience of my staff and the meaning they made of that experience (Seidman, 2006).

I employed a focus group interview format with four staff members in two groups of two, to monitor the implementation and effects of action steps taken in the first cycle of action research and to clarify data collected from the second survey instrument administered in the reconnaissance phase of the second action research cycle. According to Vaughan et al. (1996), focus groups are helpful in the interpretation of findings and in the generation of ideas for follow up studies. Focus groups are a form of group interview in which reliance is placed on interaction within the group to yield a collective rather than an individual view (Morgan, 1988). Focus groups have the attraction of synergy, with people simultaneously stimulating discussion and working together on the issues in hand (Cohen et al., 2018). This fosters interactivity and dialogue among participants, allowing for an increased richness of response (Bloomberg and Volpe, 2008). It is from this interaction that the views of group members emerge, leading to data and outcomes (Smithson, 2000; Hydén and Bülow, 2003). Focus groups (Morgan, 1988;

Bailey, 1994; Robson, 2002; Krueger, 2009; Gibbs, 2012) are useful for gathering data on attitudes, values, perceptions. Focus groups are also useful to triangulate with other data collection methods such as questionnaires or observation, and in the interpretation of findings (Cohen et al. 2018; Vaughn et al., 1996).

To this end, semi-structured focus group interviews between myself and school staff were conducted. This was to clarify and extend findings yielded from survey data collected during the reconnaissance phase of action research cycle 2, when staff were asked about their understanding and attitude towards school self-evaluation and its implementation (Goddard and Melville, 2011; Babbie, 2012; Coleman, 2012; Almalki, 2016; Cohen et al., 2018). An interview schedule was prepared to ensure consistency across all focus groups (Breen, 2006). In an enthusiastic discussion, my critical friend and I agreed that the interview should probe more deeply into staff members' experiences of and relationship with school self-evaluation to date, particularly what they felt both enabled and disabled the implementation of school selfevaluation. To do this, we formulated and constructed a set of eleven interview questions (Appendix 9) to ensure consistency between groups, to enable more efficient analysis and to bring greater precision to the discussion (Krueger and Casey, 2015). Questions one to five were much the same as the first two questions asked in section 'B' of the survey. They were an attempt to clarify in greater depth and find reason for the attitude of staff members to school self-evaluation as expressed in the survey. Questions six to eleven were an attempt to extend the survey data gathered, regarding both the positive and negative aspects of school selfevaluation as a government policy.

The sequencing of the questions also received attention. We wanted each staff member to be able to give their general view of school self-evaluation, listen to other views and then build from there. The questions moved from their general opinion regarding school self-evaluation at the start to the specifics of their own practice at the end. Each interview lasted one hour. For audit trail purposes, participants responses to this survey were coded systematically with the identification code 'interviewee' and a unique letter e.g., interviewee A, interviewee B etc.

As part of the signing up process to take part in the research, staff members were offered the opportunity to opt in and opt out of being interviewed. At a meeting with school staff, I explained that the interview would be audio recorded. I also explained that anyone interviewed would be given a full transcript of the interview for their approval. From the full complement of staff, four agreed to be interviewed and audio recorded in two groups of two. One was a

special education teacher; one was a junior infant teacher and the other two were classroom teachers from 1st to 6th class. I gave the four participants the proposed set of interview questions. I also arranged to check in with them a couple of days before the interview to see if they had any queries or concerns about the questions to be asked. However, they did not have any concerns. As the facilitator, I led the discussion, steering the group as necessary, keeping them focused on the discussion. When the interviews were completed, I transcribed the two focus group interviews into written form, checking the transcripts back against the original audio recordings for accuracy. Transcripts were then returned to the participants for individual checking.

Reflective journal and field notes

A core criteria of practitioner research is critical self-reflection. Reflection is necessary in becoming a reflective practitioner and to enhance the validity and quality of the research (Somekh, 2006). To this end, I created and maintained a reflective journal with fieldnotes throughout the entire project. This was mainly employed to reflect on and document how my values, positionings, choices and research practices influenced and shaped this study and the final analysis of the data. The recorded accounts related to ethical issues, tensions, problems and dilemmas and lines of further inquiry. I also used it to record impressions of the school improvement plans, as well as reflections of the activities that were designed and implemented. The field notes initially took the form of jotted notes (i.e., short, temporary memory triggers written in situ) (Neuman 2006). However, these were then developed systematically, usually after school hours each day, into reflection notes of short analytical memos. To do this, I drew on suggestions from the literature (McNiff, 2017; Whitehead and McNiff, 2006; Schein; cited in Coghlan, 2019, pg.43-50) to create a reflective journal template (Appendix 10). It provided me with prompt questions which were simple and easy to use, as can be seen below in *Table 4.2*.

I found this reflective journal template extremely helpful in critically self-reflecting on my actions in a systematic way throughout the duration of this action research project. Some worked examples can be found at Appendix 11.

Author		Vinny's	Reflective Jo	ournal	
Whitehead	Action	Reflection		Significance	New Action (Leads you back to the
					start/Left side of the grid)
McNiff		Thoughts/Thinking		What I learned	
Schein (taken form Coghlan)		Observation (what did you observe? Can you describe it?)	Reaction (how did you react? What feelings were aroused in you?)	Judgement (what was your judgement about what happened? What thoughts or evaluations did the event trigger?)	Intervention (what did you do about it? How did you intervene? Remember: doing nothing or remaining silent is also an intervention)

Table 4.2Reflective journal template

Critical Friend

Critical friendship is a versatile form of support for school colleagues engaged in leadership activities. Critical friends are significant because leadership is clearly linked to improving schools (Day and Sammons, 2014), and critical friends have been shown to support the work of school leaders (Gurr and Huerta, 2013). Critical friends have been widely used in a school context to promote individual and professional learning and reflection on practice (Bambino, 2002; Butler, 2011). According to Senge (2006), critical friends are key agents for organisations that see themselves as learning organisations because of their ability to foster reflection and improvement. Swaffield (2004) argues that the role of critical friend as a facilitator of change has become a significant component in school improvement initiatives such as school self-evaluation. They are able to listen, step back from the process, and assist through providing another perspective (Kember et al., 1997). Costa and Kallick (1993) define a critical friend as "a trusted person who asks provocative questions, provides data to be examined through another lens, and offers critiques of a person's work as friend. A critical friend takes the time to fully understand the context of the work presented and the outcomes that the person or group is working toward. The friend is an advocate for the success of that work" (pg.49-51). The benefits of using a critical friend as a methodology in terms of scholarly

rigour and validity are well documented (McNiff, 2007; Lofthouse et al., 2012). As part of our normal leadership practice, my deputy principal and I would meet once every week for an hour to discuss school issues. During this scheduled time and all times during the project, my deputy principal fulfilled the role of a critical friend as described in MacBeath (1999, 2006), offering both critical and supportive feedback in relation to my own leadership journey and the approaches taken in this project. She listened to my ideas, critiqued them and assisted in refining them. Field notes were taken during these meetings in the form of jotted notes. Throughout the study, the data and findings were also subjected to rigorous critique from a community forum of critical friends, made up of all research participants. Throughout each action research cycle, I checked the data and findings with all participants by publishing reports at each stage of the process and meeting with them in the school as a validation group to seek their feedback to ensure that the findings and decision-making process were both rigorous and transparent. These meetings were held as a collective in a round table and open forum format. However, with the arrival of Covid-19 in March 2020 and the national emergency measures which followed, these meetings were moved online. Presentations were given via a mixture of Zoom and PowerPoint, with feedback facilitated via both Zoom and email. I will now discuss each methodology in terms of data collection and where relevant coding and analysis.

4.7.2 Analysis of data

As a result of staff members sharing their insight in eight surveys, two focus group interviews and field notes, key data was available to help us improve the implementation of school selfevaluation in our school. In order to ensure that the analysis of this data was systematic, a deliberate, sequential process was followed at all times (Krueger and Casey, 2015). One approach to analysing narrative data is to reduce the data by using coding and thematic analysis techniques (Floyd, 2012). I adopted a thematic analysis approach to the qualitative data, which was the substantial kind of data in this research. Thematic analysis is a qualitative analysis technique to identify and analyse themes linked to a data set. It organises and describes data in rich detail (Braun & Clarke, 2006). Thematic analysis goes beyond counting phrases or words in a text and explores explicit and implicit meanings within the data (Guest et al., 2012). Coding is the primary process for developing themes by identifying items of analytic interest in the data and tagging these with a coding label (Boyatzis, 1998). Themes are then built from the codes (Braun and Clarke, 2019). To establish the most important themes, noteworthy quotes and any unexpected findings in this research (Breen, 2006), the recently revised and amended six-phase guide to performing thematic analysis proposed by Braun and Clarke (2019), was used as the principal approach to all data analysis in this study. The six phases are (1) Data familiarisation (2) Data coding (3) Generating initial themes (4) Reviewing and developing themes (5) Refining, defining and naming themes (6) Writing the report. It provides a vocabulary and 'recipe' to undertake thematic analysis (Braun and Clarke, 2006, pg.78), where themes are conceptualised as patterns of shared meaning across data items, underpinned or united by a central concept, which are important to the understanding of phenomena and the research questions being explored (Braun and Clarke, 2019). In the final phase of writing the report, Braun and Clarke (2006) say we must identify the essence of what the recognised themes are about and determine what aspect of the data set each theme captures. They contend it is important to identify the 'story' that each theme tells in relation to each other, and how it fits into the overall story of the research question. I describe below my engagement with this six-phase process, giving a worked example of the coding process being applied to data collected as part of the second round of reconnaissance in this action research study. The data was collected from a survey compromised of twenty-five Likert-scale questions and three yes/no questions regarding staff members' attitudes towards school self-evaluation in terms of input, process, output, outcomes and unintended consequences. It was also collected from six open-ended questions about staff members' understanding of school self-evaluation and its implementation. In addition, it includes data collected from focus group interviews with four members of staff. For full worked examples of thematic analysis being applied to all the surveys and interviews conducted in this project please refer to Appendix 12.

Phase 1 – Becoming familiar with the data

The data gathered from the twenty-five Likert-scale questions and three yes/no questions was illustrated in numerical graph format (bar chart) for analysis (Dixon and Woolner, 2012) for example:



Figure 4.3 The process of SSE is easy to understand – bar chart results

I immersed myself in these numerical graphs to make myself familiar with the depth and breadth of their content. This involved reading the numerical graphs in an active way searching for meanings and patterns and making notes of same.

I then combined and organised, for analysis, all the answers from the open-ended questions into one master document showing each respondent's response, for example:

Question 31: What are the positive aspects	of SSE as a government policy?
S3 Respondent 1	Allows schools to make practical changes
S3 Respondent 2	Allows time for staff to reflect and be
	objective and professional as we review all aspects of subject/strand in question
S3 Respondent 4	As an SNA I don't really know the positive
	aspects of SSE, as my role and SSE have no
	relevance to each other
S3 Respondent 8	Improvement of learning methodologies and
	strategies. Identify areas for development.
	Taking ownership of curriculum
S3 Respondent 9	(no answer)
S3 Respondent 12	All schools in the country are expected to
-	partake. It allows schools the autonomy to
	choose areas of most concern to them and
	their organisation. It acknowledges the
	inbuilt desire of the vast majority of
	teachers to improve their schools teaching and learning.

Table 4.3What are the positive aspects of SSE as a government policy? – sample answers

I immersed myself in this data to become familiar with the depth and breadth of its content. This involved repeated reading of the data in an active way by making notes and searching for meanings and patterns.

Phase 2 – Data coding

This phase involved the production of initial codes from the data. This initial coding was undertaken by using gerunds and keeping the codes active and as close to the original statements as possible (Charmaz, 2006). To begin, I underlined what I felt was every key word and phrase in both the survey data and the interview transcripts. The table presented below is a worked example of the methodological process adopted.

Table 4.4	Survey Part 1:Likert	-scale and ves/no	questions – sampl	e answers
10000		<i>Sectice cline jcSirc</i>	questions semipt	e en 15 // e. 5

	Survey part 1 (Likert-scale and yes/no questions)		
Question 24	 Unintended consequences: SSE places a lot of <u>stress</u> on staff (<u>yes</u> <u>90%</u>, 10% no) 		
Question 25	 Unintended consequences: SSE increase staff morale (no 90%, 10% yes) 		
Question 26	• Unintended consequences: SSE <u>takes up a lot of time</u> (<u>yes 90%</u> , 10% indifferent)		
	Survey part 2 (open-ended questions)		
Question 29: W	hat do you think is the purpose of SSE?		
S3 Respondent	Give a sense of agency to schools to work on what actually needs		
1	<u>improvement</u>		
S3 Respondent 4	I think it's a <u>tick the box exercise</u> and I am not really sure to be honest		
Question 31: W	hat are the negative aspects of SSE as a government policy?		
S3 Respondent 2	Very time consuming; lots of <u>abstract</u> , repetitive 'splitting of hairs'; <u>quite meaningless</u> exercise at times		
S3 Respondent 3	Additional paperwork on already stretched staff		
S3 Respondent 6	Added workload and time required to do it		
S3 Respondent 7	<u>Time consuming</u> – takes a lot of meeting times. <u>Increased paperwork</u> equals <u>increased stress</u> which <u>affects morale</u> .		

I then combined every underlined word and phrase into colour coded groups. For example, any words or phrases associated with making improvements were coloured pink, any words or phrases associated with paperwork were colour coded dark-green and words or phrases associated with being time consuming were colour coded light-orange. Words or phrases associated with attitudes were colour coded dark-purple. Words or phrases associated with being accountable were colour coded yellow and words or phrases associated with staff feeling over-stretched were colour coded light-blue. For a full list of my colour codes please refer to Appendix 13. The table presented below is a worked example of the methodological process adopted (*Table 4.5*).

Table 4.5Focus group interviews – sample answers

Focus Group One Interview		
• attitudes are one of confusion to it		
• there's so much going on between new language curriculum already being 		
• <i>it's the principal or the whole staff, decide what the school needs to improve</i>		
• I do see the benefit of it having one specific thing that the school overall has a focus on,		
Focus Group Two Interview		
• positive without being enthusiastic		
• that I think it makes sense that schools can identify their own problems and you know go from there but		
• I don't enjoy the process I don't enjoy that kind of work in terms of policies, policies and paperwork		
 any good school with a positive atmosphere, this happens, you see we've a little weakness, it's mentioned at a staff meeting and we do something to change it,		
we were trained we were ready but you know the strike then it was called I felt like I'd emotionally invested in it and then it was gone, then it came back,		

The colour coded words and phrases were then combined into one master document for ease of thematic analysis. They were then each assigned an alphanumeric code and description title. This removed the need for colour coding. The table presented below is a worked example of the methodological process adopted (*Table 4.6*).

Table 4.6	Coding	data	extracts	nrocess	samnlø
<i>Tuble</i> 4.0	Couing	uuiu	exilacis	process	sampie

Actual data extract quotes	Code	Description title
Reason/Purpose of school self-evaluation		
Accountability	R1	Accountability
• Box ticking exercise		
• Tick the box exercise		
• Improvement	R 2	Improvement
• Improvement in teaching and learning		
• A need to improve a school for both teachers and learners		
• To improve		
• Improve		
• Identify areas that need improvement		
• Improve the school to enhance learning		
• Improve outcomes for the better of each pupil and the whole school		
• Improvement plans based on findings		
Positive aspects of school self-evaluation?		
Practical changes	P1	Practical
To reflect	P2	Reflect and discuss
• Reflection and discussion		
Negative aspects of school self-evaluation?		
• Very long term aims	N1	Too long term
• Staff turnover is high. Would prefer one item done and dusted each year		
1		

 Very time consuming Time consuming Time required to do it 	N3	No time
Other comments		
 Happy to be told what to do if it saves on time/discussion/meetings Would like a directive from Dept. 	OC 2	Tell me what to do
saying what to do/how and when in each subject		
• I find it quite tiresome in general	OC3	Tiresome process
Emotions		
Confused	E1	All references to
• Frustrated		emotions
• Calm		
Annoyed		
Leadership		
• <i>it's the principal or the whole staff, decide what the school needs to improve</i>	L1	All references specific to leadership

As can be seen from the example above (*Table 4.5*), identifying the codes and matching them with the actual data extract quote that demonstrate that code was an important part of phase two. Braun and Clarke (2006) contend that it is important in this phase to ensure that all our actual data extract quotes are collated together within each code for scrutiny and reference. For a full list of my alphanumeric data codes and their matching actual data extract quotes please refer to Appendix 14.

Phase 3 – Generating initial themes

This phase involved sorting the different codes (and the actual data extract quotes) into potential themes. To help analyse the codes and consider how they might combine to form overarching themes, Braun and Clarke (2006) recommend the use of a visual representation. I therefore listed all the codes with their title description into a table. The table presented below (*Table 4.7*) is a worked example of the methodological process adopted.

Code	Title description	Code	Title description
R1	Accountability	N1	Too long term
R2	Improvement	N2	No momentum/No energy in process
R3	Evaluate (weakness and good practice)	N3	No time
R4	Planning	N4	Meaningless
		N5	Overworked staff
P1	Practical	N6	More paperwork
P2	Promotes reflection and discussion	N7	More stress
P3	Promotes professionalism	N8	Staff overworked
P4	SNA	N9	Not realistic
P5	Whole school review and work on weaknesses	N10	Lowers morale
P6	Focus on better outcomes	N11	Takes from classroom teaching and learning time
P7	Focus on improvement		
P8	Autonomy	OC1	Participate yes!
P9	Engagement with teaching and learning	OC2	Tell me what to do
		OC3	Tiresome
E1	Emotions	OC4	Irrelevant
L1	Leadership	OC5	Teaching and learning planning more than policy please
		OC6	Involves everyone

Table 4.7Codes and descriptions

OC7	Once target area improves
OC8	Pressure
OC9	Proper sub cover
OC10	School self- evaluation has to be done not sure why
OC11	Partnership with inspectorate would be good

The next task involved thinking about the relationship between the codes. I used colour coding once again to link the codes which had something in common. I collated into their own sub-thematic box for ease of analysis and further refinement as follows:

Table 4.8	Coding and theme process		
Code		Title description	

Code	Title description	Sub themes
R1	Accountability	
P2	Promotes reflection and discussion	
Р3	Promotes professionalism	
P8	Autonomy	
Р9	Engagement with teaching and learning	
OC1	Participate yes!	
OC11	Partnership with inspectorate would be good	

R2	Improvement	
R3	Evaluate (weakness and good practice)	
R4	Planning	
P1	Practical	
Р5	Whole school review and work on weaknesses	
P6	Focus on better outcomes	
P7	Focus on improvement	
OC6	Involve everyone	
OC7	Once target area improves	
R1	Accountability	
N1	Too long term	
N4	Meaningless	
N9	Not realistic	
OC2	Tell me what to do	
OC4	Irrelevant	
OC10	School self- evaluation has to be done not sure why	
OC11	Partnership with inspectorate would be good	
P4	School self- evaluation is meaningless to	

N2	No momentum/no energy in the process
N3	No time
N5	Overworked staff
N6	More paperwork
N7	More stress
N8	Staff overworked
N10	Lowers morale
N11	Takes from classroom teaching
	and learning time
OC3	Tiresome
OC5	Teaching and
	learning planning more than policy
	please
OC8	Pressure
OC9	Proper sub cover
E1	
L1	Leadership can motivate the
	involvement of staff
	in school self- evaluation

The orange codes represented what school self-evaluation gives to a school, for example community involvement in a drive for improvement and better outcomes. The yellow codes were representative of the perceived reality of school self-evaluation on the ground by those implementing it. For example, staff said they had no time to do it and that it caused them additional workload and stress and lowered their morale. The purple codes represented what, according to the literature, school self-evaluation in theory gives to school staff, such as

autonomy and a focus on professionalism and accountability. The blue codes represented the attitude of staff members to school self-evaluation (finding it generally unrealistic and meaningless), whilst the green code represented the leadership of school self-evaluation and its role in motivating staff members to engage in the process.

Phase 4 – Reviewing and developing themes

This phase involved the refinement of potential themes. To do this I revisited the actual data extracts and the literature, to thus ensure, that the themes worked in relation to my data set and that I had accurate representations for the purposes of this study. This involved a process of mapping and redrafting, to develop sub-themes which represented some level of meaning within the data set. During this process a few potential themes did not have enough data to support them. Whilst some blended quite well together, others separated, with clearly identifiable distinctions between them. The outcome of this process was as follows:

Code	Title description	Sub themes
R1	Accountability	• Internal accountability
P2	Promotes reflection and discussion	
Р3	Promotes professionalism	
P8	Autonomy	
Р9	Engagement with teaching and learning	
OC1	Participate yes!	
OC11	Partnership with inspectorate would be good	

Table 4.9Developing themes process

R2	Improvement	• Evaluate for improvement
R3	Evaluate (weakness and good practice)	
R4	Planning	
P1	Practical	
OC7	Once target area improves	
Рб	Focus on better outcomes	
Р7	Focus on improvement	
OC6	Involve everyone	Whole school approach
Р5	Whole school review and work on weaknesses	
R1	Accountability	• External accountability
N1	Too long term	
N4	Meaningless	
N9	Not realistic	
OC2	Tell me what to do	
OC4	Irrelevant	
OC10	School self- evaluation has to be done not sure why	
OC11	Partnership with inspectorate would be good	
Р4	School self- evaluation is meaningless to SNAs	• SNA

N2	No momentum/no energy in the process	Workload
N3	No time	
N5	Overworked staff	
N6	More paperwork	
N8	Staff overworked	
N11	Takes from classroom teaching and learning time	
OC3	Tiresome	
OC5	Teaching and learning planning more than policy please	
OC9	Proper sub cover	
N7	More stress	Emotions
N10	Lowers morale	
OC8	Pressure	
E1	Emotions	
Ll	Leadership can motivate the involvement of staff in school self- evaluation	• Leadership and staff motivation

<u>Phase 5 – Refining, defining and naming themes</u>

In this phase Braun and Clarke (2006) say we must identify the essence of what recognised themes are about and determine what aspect of the data set each theme captures. They contend it is important to identify the 'story' that each theme tells in relation to each other and how it fits into the overall story of the research question. This required going back to the collated data

extracts for each theme, and organising them into a coherent and consistent narrative account. To do this, I engaged in a process of writing and redrafting to describe the scope and content of each theme in a couple of sentences. Using this approach in my analysis of the data, I generated two key themes to frame a thematic analysis of the data.

The first theme '*Attitudes to school self-evaluation*', is about the people tasked with implementing the school self-evaluation process to achieve the end goal of improved outcomes in teaching and learning. Whilst the second theme, '*Leadership and staff motivation*' describes the important role school leaders play in motivating and sustaining staff to engage meaningfully with the school self-evaluation process.

Phase 6 – Writing the report

This final phase involved relating back to my research question and the literature to produce a scholarly report of my findings. This report is presented in sections 5.3, 7.3 and 7.4.

4.8 Reliability and Validity

A pragmatic paradigm was applied in this study using an action research methodology. Action research within a school is, by its very nature, a collaborative endeavour. It's about working 'with' people and not 'on' people. According to Soobrayan (2003), the main research instrument in such an approach as action research, is the actual researchers themselves, concluding that the ethics, truth and politics of research is a deliberate exercise in making choices and taking responsibility. With this in mind, this study was undertaken from the standpoint of care for all participants (Noddings, 2003). The success of this project depended on others. It involved continuous consultation and effective communication with colleagues throughout, a critical community where feedback was both received and given. In such an approach, the researcher is not external to the process, but instead an 'insider' who reflects on and acknowledges their own values, beliefs and aspirations and builds these into their daily practice (McNiff and Whitehead, 2002). According to Nixon (2008), engaging in such reflection at work and undertaking insider led research can make significant contributions to work practices. With in-depth insider knowledge, a researcher can study a particular issue in a detailed and thorough way. However, when researchers are insiders, they draw upon the shared understandings and trust of their colleagues with whom they interact and work with as a community each day (Lave and Wenger, 1991). Whilst this shared information can enhance insider knowledge, it can have a bearing on how knowledge is understood and used in a

research project (Costley et al., 2010). When immersed in such work-based learning, it can be possible to fail to see the obvious. An insider researcher needs feedback on what they are doing. For that reason, the adoption of a reflective approach to insider research is crucial (Costley et al., 2010). The insider researcher must invite critique of both their own perspective and of everyone else, so that they can fully defend the validity of the interpretations they make from the data (Punch, 2005). This too requires having reliably collected evidence on which to base their findings (Burton and Bartlett, 2009).

In order to ensure such reliability and validity in this study, I outline in detail the methodological decisions made throughout the entire project and the rationale for those decisions. This includes comprehensive notes on the contextual background of the data and how that data was collected, managed, coded and analysed (Bloomberg and Volpe, 2008). A consistent approach was used in relation to the distribution of the surveys and the questions asked at each focus group interview, which were also audio recorded and transcribed. Where appropriate, responses were coded to identify the source of the data for audit trail purposes.

For analysis purposes, data was systematically coded and analysed using the recently revised and amended six-phase guide to performing thematic analysis proposed by Braun and Clarke (2019). Data triangulation and methodological triangulation were both used to compare data from different sources and validate any findings. The first involved comparing data from a variety of staff members with different roles within our school (Silverman, 2005; Woods, 2006). Specifically, these included classroom teachers, special education teachers, school principal, school deputy principal and special needs assistants. The second involved a comparison of data gathered from both quantitative and qualitative methods. Specifically, qualitative data from both semi-structured interviews and surveys were triangulated with qualitative/quantitative data from questionnaires. This allowed me to investigate the implementation of school self-evaluation in our school in a number of different ways and from a variety of different sources. Using this approach throughout the study, staff members and I were able to cross-check the data to determine the accuracy of the information (Miles and Hubermann, 2014). This added depth to the results and increased the validity of the study's findings, as is evident from both the positive and negative feedback received regarding the implementation of school self-evaluation in our school.

Care was also taken to limit the potential for bias due to the involvement of myself in the implementation of the project. To ensure validity, the data and findings were continually

subjected to rigorous scrutiny, discussion and critique by all staff members at all stages during the project (McNiff and Whitehead, 2002). During the study, four staff members went on leave and did not return by the end of the project. To fulfil my ethical obligations, these staff members were reminded of their right to withdraw from involvement in the project without any repercussions. These issues are dealt with in more detail in the section below. However, all four staff members asked to be kept informed of progress in the study and wished to maintain the right to respond with feedback. The same information sent to research participants therefore, continued to be sent to these four staff members, with channels of communication monitored and checked daily for any responses, concerns or issues raised to ensure the validity and reliability of the research.

Throughout the study I kept a reflective journal (Whitehead and McNiff, 2006; Schein, 2013; McNiff, 2017; Coughlan, 2019) and field notes, whilst my deputy principal also fulfilled the role of a critical friend as described in MacBeath (1999; 2006), offering both critical and supportive feedback in relation to my leadership journey and the approaches taken in the project.

4.9 Ethical considerations

This research was conducted in accordance with the British Educational Research Association's Revised Ethical Guidelines for Educational Research (BERA, 2018). Permission to conduct the research was sought and granted from the school's Board of Management. I undertook the dual role of school principal and active researcher. Participants may have felt obliged to be involved in the research as it was being carried out by the school principal. I fully acknowledged this power relationship and its implications. However, I was fully aware of the ethical guidelines for educational research and my responsibilities in this regard.

I was aware that such ethical implications needed further reflection and consideration. Brinkmann and Kvale (2005) advise that research ethics are just as important as methodology. In their view, the onus is on the insider researcher to be fully aware of their dual role and the power dynamics at play when making any request to participants. No action should be carried out by the insider researcher without due consideration of the ethical considerations. Costley et al. (2010) identify the insider researcher's position as both a researcher and as a practitioner, engaging with fellow colleagues as a major ethical challenge. It requires careful management of issues regarding power and vulnerability. A way to deal with this, according to Kvale and Brinkmann (2000), is to present the proposed method as transparently as possible, so that others can ascertain the potential power effects.

With this in mind, all staff were called together at a meeting to explain fully the purposes of the project and to emphasise to all participants, that I was working as a researcher and not as a principal teacher. It was outlined to staff that their involvement and shared ownership of the project would be most welcome as their wealth of knowledge, talent, skills, expertise and insight would be of great benefit to the project and the school self-evaluation process in our school. I was therefore, adopting a collaborative, dialogical and co-constructivist approach, where my fellow work colleagues were invited to be co-researchers and share in the ownership and direction setting of the thesis. This approach enabled all participants to be alert to any issues which may have arisen regarding the conduct and direction of the project. Whilst this invitation helped negate against any power dynamics that may have stemmed from my role as school principal, I remained both sensitive and alert to these throughout the research process. I told staff that at all times, I would treat each staff member fairly, sensitively, with dignity and an ethic of respect and freedom from prejudice.

Staff were offered every opportunity to discuss in private, any concerns they may have had about the project. A spirit of collaboration and partnership underpinned the research and respect for confidentiality. As a safeguard, a 'gate keeper' was put in place to allow participants bring attention to any concerns they may have during the project. The 'gate keeper' was the chairperson of the Board of Management. Regular reminders were given about the facility to communicate concerns or tensions to the 'gate keeper' by phone, email or in person. This contact email address and phone number was given to every participant. Participants were also supplied with my phone and email contact details.

Staff were informed, both verbally and in writing, that participation in the research was entirely voluntary and that they were under no obligation to take part in the research. Staff were informed of their right to withdraw from involvement in the project at any time without repercussions, whether at the beginning, middle or end, by informing the gate keeper or I in person or through phone or email. Regular reminders were also given about this option to withdraw from the research project at any time throughout the entire process. No staff withdrew from the research project.

All potential participants were supplied in advance with a written consent form explaining the nature and purpose of the research, their involvement in it, how the research will be used and

who it will be reported to (Appendix 15). All research participants were given guarantees of anonymity and that any information they provided would be treated in the strictest confidence (subject to the proviso that the researcher had an overriding legal obligation in relation to the disclosure of certain kinds of information, such as child welfare). Full notes would be kept relating to all such matters. No inducements were offered to encourage participation.

The project did not overburden participants. All work carried out by school staff in relation to this project was completed during working hours and the normal allocation of staff meeting hours. The research project did not take from the normal activity of a school day and pupils' right to an education. Under the terms of DES Circular 0008/2011, primary school teachers in Ireland are obliged to work an additional thirty-six hours to provide for certain essential activities without reducing class contact or tuition time. These additional hours were used throughout the project to carry out most of the planning and development work. When this was not possible, infant class teachers were used to provide substitute teaching cover for colleagues working on the project. In Ireland, infant class pupils finish one hour earlier than all other classes.

The research fully complied with all legal requirements in relation to the storage and use of personal data. All data obtained was processed fairly and kept only for the specific purpose of my research. All data given was used only for the purposes for which it was volunteered initially. All data collected was relevant to my research. All primary data will be held for a minimum of ten years. All data stored electronically was secured using passwords, encryption and backup. There was appropriate firewalls and anti-virus software in place. Any manual data was held securely in a locked filing cabinet. Any waste papers, printouts etc. were shredded and disposed of carefully. Data was organised in a manner that allowed for ready verification in either hard copy or electronically. All original data was authenticated. This was in accordance with Maynooth University Research Ethics Policy, Maynooth University Research Integrity Policy, BERA guidelines (2018) and the Data Protection Act (Government of Ireland, 2018) and any subsequent acts.

Chapter 5 – First Phase of Reconnaissance

5.1 Introduction

In Ireland, school self-evaluation operates within a national policy framework which provides a set of themes related to educational priorities, to help guide schools in their evaluation and planning. Schools are required to use this framework to plan and review outcomes (DES, 2016). A standardised, evidence-based school self-evaluation process normally begins by conducting a needs analysis which involves (i) examining pupil achievement information to uncover pupils' learning needs and (ii) identifying teaching and management support needs. Schools then must develop a programme of intervention, which may include professional development to improve aspects of teaching. Finally, schools are required to review their intervention annually, using the data collected and produce a written report of their findings (Lai, 2013). However, in my school I have led school self-evaluation with a quick-fix approach in an attempt to shield my staff from this burden. This approach was down to assumptions I made about my staff and indications they had given me suggesting an apathy for school selfevaluation. I assumed that staff members had a hard and busy workload and that this additional imposed task would only impact negatively on their morale. So, I despised it. This caused me to 'serve' my staff, resulting in an unhealthy practice of going it alone, limiting any possibility of collaborative benefit or meaningful organisational change.

However, leading school self-evaluation in this way negated my core values of authenticity and collaboration. This was problematic as evidence shows that the attitude of school leadership towards school self-evaluation is significant, especially in supporting and enhancing the staff motivation and commitment needed to implement the process successfully and foster improvement (Day and Sammons, 2014). In addition, as I engaged with my own thinking around school self-evaluation and read more about the process, I began to appreciate its potential in having a positive impact on teaching and learning practice. This caused me to engage in this research, to reflect on and give a renewed focus to my leadership practice, in the belief that I could lead the process better and improve its implementation for the betterment of our school. A change in my leadership approach was inevitable as my core values of authenticity and collaboration were not being met or lived as leader of my school. These values were important and really mattered to me. It was important therefore, that any approach I adopted to school self-evaluation was both meaningful and practitioner-based, adopting a collaborative, dialogical and co-constructivist approach.

5.2 Reconnaissance

The first cycle of the Action Research began in January 2020. School staff met to ascertain where we were as a school regarding the school self-evaluation process in our school. My aim was to create and maintain a positive, relaxed, friendly and informal space for the sharing of ideas by staff members as a group. One presumed benefit of using groups is that they are able to tackle a problem more effectively than would be possible had the problem been left to any single individual (Hart, 2007; National College, 2012; Lessard et al., 2016; Franco & Nielsen, 2018).

To get a snap shot of the 'current reality' regarding the implementation of school selfevaluation in our school, the facilitation tool I chose to use with staff was a SWOT analysis (*Table 5.1*). The objective was to ascertain what was working well and what could be changed or improved.⁴ The quadrant below sets out a detailed picture of the key points noted under each of the sub-headings:

The process of school self-evaluation in our school			
Strengths	Weaknesses		
 Staff relationships, motivation and interest Manageability of school size Frequent meetings and communication amongst staff Involving everyone Works well Specific Focus on learning and teaching Knowledgeable and professional staff 	 Interruptions with exceptional events Turnover of staff Perceived lack of confidence in staff members I'm not sure there are weaknesses, a great job is being done Time consuming Extra stress and the feeling of <i>"just another area of work that has to be done"</i> Lack of interest due to constant moving of goalposts regarding SSE 		

Table 5.1	SWOT	analysis	on the	process	of SSE in	our school
1 0000 5.1	51101	and yous	on me	process	0,552 11	our senoor

⁴ My deputy principal agreed to be the scribe. This I felt was important as it allowed me greater opportunities to actively listen and observe how ideas were being discussed and decisions made (Hart, 2007; National College, 2012; Franco & Nielsen, 2018).

Opportunities	Threats (obstacles)		
 Potential to organise more teams to help with each area of school planning Great opportunity to identify and agree on areas that need our focus and attention Works well Clearer layout from the DES Increased supports for schools Good opportunity to harness staff energy and interest if led well 	 Willingness of staff Availability of sub cover to do it Not having all staff on board Time Support from the DES Lack of interest and enthusiasm Feelings that teachers' voices won't be heard and respected 		

Staff members saw the relationships and the interactions between them regarding school selfevaluation as a strength. They reported these interactions to be frequent with everyone involved. In addition, staff members reported that the school self-evaluation process in our school was strengthened in a positive way by frequent communication to support the flow of information.

Staff members also saw their interest, knowledge, professionalism and motivation for school self-evaluation as key strengths for the process. It was not surprising therefore, that any change in staff was seen as a weakness for the school self-evaluation process. Conversation on this point revealed that any change in staff meant extra work for remaining school members to fill any gaps in the implementation of school self-evaluation, whilst also retraining and bringing up to speed new staff members with the process.

Staff members saw the focus on evaluating teaching and learning at whole school level as a strength, as highlighted below. Interruptions the work of school self-evaluation were reported by staff members to be a weakness to the process.

However, this commitment to the work of school self-evaluation was not held by everyone. Some staff members stated quite clearly in our discussion, that they did not have any interest or enthusiasm in the school self-evaluation process. During the discussion, one staff member found some support in a quite downbeat assessment that school self-evaluation was *"just another area of work that has to be done"*. These negative attitudes towards school selfevaluation and the possible unwillingness amongst some staff to engage with the process (other than in some form of compliance), were seen as a threat to its success by some staff members.

A few staff members also saw the school self-evaluation process as time consuming and an increase in their workload, causing them extra stress. They said these issues were both a weakness and a threat to the successful implementation of the process in our school. In contrast, some staff members said that they would welcome more opportunities to engage in teamwork as part of the school self-evaluation process, to agree on areas that needed attention. Probing questions revealed however, genuine concerns that their voices would not be heard and respected.

Another obstacle to the successful implementation of school self-evaluation in our school according to the data, was the lack of tangible support for school staff from the DES. Conversation on this point revealed a real desire amongst some staff members for increased supports and a cleaner layout from the DES on how to implement school self-evaluation, as evidenced below. Coupled with this, the data suggested there was a lack of confidence in doing school self-evaluation amongst some staff members.

To analyse the data, I used an inductive process of thematic analysis (Braun and Clarke, 2019) with identified themes being strongly linked to the data. For a full worked example of the thematic analysis applied to the survey, please refer to Appendix 16.

I generated three key themes to frame a thematic analysis of the data. The first, entitled '*Staff involvement in whole school decision-making*', was about the people who drive the implementation of school self-evaluation to achieve the end goal of improved outcomes in teaching and learning.

The second theme '*Resourcing school self-evaluation*', was about the daily experiences of people working in a school doing school self-evaluation and all that entails, physically, mentally and emotionally.

The potential outcomes of both of these themes are grounded in the effective enactment of the final and third theme, '*Leadership and staff motivation*'. School leadership plays an important role in motivating staff to engage meaningfully with the school self-evaluation process by facilitating the development and provision of any resources required and tapping into the energy and interest of staff members.

5.3 Findings and discussion

5.3.1 Staff involvement in whole school decision-making

From this reconnaissance data, school staff saw the relationships and the interactions between them as a strength and reported these interactions to be frequent. Follow up discussions revealed staff members were both motivated and interested in any activity, such as school selfevaluation, which improved teaching and learning in our school. I was not surprised therefore that probing questions during the discussion revealed a view amongst a few staff members that any change in staff was a loss to the school self-evaluation process in terms of knowledge and experience and extra work for remaining school members to fill any gaps in the implementation of school self-evaluation, whilst also retraining and bringing up to speed new staff members with the process.

However, the positivity by staff to issues regarding teaching and learning at a whole school level was most welcome as research shows teachers to be primarily interested in teaching and not so much in what happens at whole school level (Schildkamp et al., 2012). It was concerning therefore, to see that interruptions to the school self-evaluation process for exceptional events were viewed by some staff as a weakness in our school. The conversation on this point revealed that for them, it removed the focus from their recognised teaching and learning priorities. Exceptional events could include fundraising events, the arrival of new immediate mandated initiatives, a sudden change in staff or unannounced visitors to the school or a global pandemic to name but a few. Therefore, to ensure the success of school self-evaluation at whole school level in this project, it was important that participants had plenty of meaningful opportunities to interact and discuss teaching and learning matters at all levels within our school, with interruptions being kept to a minimum where possible.

According to my data and in follow up discussions, staff members reported that the flow of information between them during the school self-evaluation process, was supported by regular opportunities for communication. Empirical evidence tells us that policy implementation processes are more effective when they involve effective communication strategies (MacBeath, 1999). This implied that if school self-evaluation was to be implemented successfully in our school, the continuation of clear, open and honest communication was vital.

Staff members saw the knowledge and professionalism they brought to school self-evaluation as a strength and were happy that involvement in our school self-evaluation process involved everyone. The view that the school self-evaluation process in our school currently involved everyone surprised me. Self-reflection on my leadership of the process showed me to have implemented school self-evaluation in a structure of hierarchal compliance. In this approach, staff voted on an area of teaching and learning to improve, I then did the evaluative work and drew up the draft improvement plan for staff. The plan was then communicated to the wider school community and filed away for future review or inspection. Staff were obliged to implement the improvement plan. School staff had to engage with school self-evaluation in this way whether they liked it or not. However, this finding from the data suggested that some staff were more than happy to engage. My data seemed to suggest that any desire amongst staff members to use school self-evaluation for school improvement was questionable, other than in the area of simple compliance with rules. This resonated with similar research from the literature (McNamara and O'Hara, 2012). Vanhoof et al. (2009), believe schools experience self-evaluation as more of an obligation than anything else, the principal objective being compliance, rather than using school self-evaluation as a tool for improving the functioning of the school. This argument finds support in my findings that some staff members view school self-evaluation in a negative way as "just another area of work that had to be done". Hall and Noyes (2009) argue that being engaged in such an activity that is more about compliance than educational endeavour, makes teachers feel oppressed and professionally compromised. Other teachers see it as a threat to their autonomy in the classroom (Vanhoof et al., 2009). Sentiments such as these in any evaluation process can lead to inauthenticity and fabrication (Webb, 2006; Perryman, 2009; Ball, 2003), engendering what Power (1997) terms 'pathologies of creative compliance' (cited in Hall and Noyes, 2009, pg.36). To understand this discrepancy between beliefs and practice (Beaver and Weinbaum, 2015) I referred to an Irish study by O'Brien et al. (2019). It showed participants setting improvement targets using school self-evaluation guidelines, but lacking a belief in the target setting process and being confused when using data to set and measure the targets. It was important this project did not suffer a similar fate. For this project to be successful, my leadership of school self-evaluation in this action research study needed to be authentic, involving staff members working in a community of endeavour where they had agency and wanted to do it. This is achieved by ensuring stakeholder input into any decision-making related to the process of school improvement and change (Mortimore et al., 1988; Rozenholtz, 1991; Day and Sammons, 2014).

Some staff members wanted more opportunities to engage in teamwork as part of the school self-evaluation process. However, further conversation on this point revealed genuine concerns among some staff members, that if their voices were not heard and valued enacting school self-

evaluation, this had the potential to be a serious obstacle to the successful implementation of the process in our school. I believe my leadership of school self-evaluation was confounding the problem. I led school self-evaluation with a quick-fix, go-it-alone approach, thereby limiting any possibility of collaborative benefit or meaningful organisational change. Whilst this issue of teachers' voices not being heard in the school self-evaluation process was concerning, I was encouraged by the fact that the observation was given an opportunity to air in an open forum amongst our staff and that the contributor deemed it safe to do so. School staff wanted more opportunities to work as teams in identifying and agreeing on areas that needed attention. This was a clear indication that the assumptions I previously held in relation to the need to lead the school self-evaluation process in hierarchical ways, were incorrect. In fact, the literature shows an internally motivated evaluation culture can occur (Mann and Smith, 2013) if staff are perceived as team members in a school with shared objectives (Potter et al., 2002) rather than individual classroom teachers (Schildkamp et al., 2012). The data showed a desire amongst many school staff to be involved in school improvement initiatives such as school self-evaluation. It was therefore imperative for the successful implementation of school self-evaluation in our school that it was a collegial and collaborative process with involvement in all decision-making. Such interactions had the potential to increase the motivation, interest and positivity of staff towards the school self-evaluation process in our school.

5.3.2 Resourcing school self-evaluation

This theme refers to the daily experience of staff members working in our school. It references the many resources they call upon each day to engage with the school self-evaluation process. These resources cover a wide spectrum, ranging from tangible evaluation frameworks to inner motivation. School self-evaluation is meant to incentivise and motivate school improvement. However, according to the data, some participants held the view that school self-evaluation was causing an unwanted increase in their workload, with no time to do it and this was therefore a disincentive to engage with it. The data seemed to suggest that the work demands placed on them by school self-evaluation was placing extra stress on staff members. This view of school self-evaluation held by staff members was not uncommon in the literature. O'Brien et al. (2015) report that the nature of the activities involved in school self-evaluation seem to be off-putting to teachers, together with the perception that it is time consuming and difficult to carry out. Another study (Hall and Noyes, 2009) shows teachers express concern about the time required to engage in the process and the additional workload. Participants in the study speak about the intensification of their work and pressures on their time. They say the work of school self-

evaluation is an extra burden., adding to their administrative workload. Research suggests that school principals and teachers share a resistance towards the added paperwork that school self-evaluation brings with it (Vanhoof et al., 2009). Ryan et al. (2007), suggest that those promoting school self-evaluation should pay attention to the daily reality of schools in relation to time and resource constraints and the increased workload expected of schools from other various mandates. Against this backdrop, they believe that experience for evaluation need to be realistic. To ensure the success of this project therefore, it was important to provide both clarification and clear expectations regarding the implementation of school self-evaluation by working out practical answers to such pragmatic questions as who, what, where, when and how (O'Brien et al., 2015). Wikeley et al. (2002) suggest that any actions to be undertaken should be coordinated with activities in other policy areas. For example, whilst implementing curricular change, do it through the school self-evaluation process as the vehicle for change. It was imperative for this project therefore, that any undertaking was authentic and genuine and part of the core business of the school.

Another obstacle to the successful implementation of school self-evaluation in our school according to the data, was the lack of tangible support for school staff from the DES. In follow up discussions, staff members expressed a desire for increased supports and a clearer layout from the DES on how to implement school self-evaluation. Discussion during the SWOT analysis revealed an awareness amongst staff members of the existence of DES manuals for schools on how to engage with and implement school self-evaluation. However, it was clear from the discussion that staff members were not overly familiar with the title or contents of these documents. This call for increased support from school staff to implement school self-evaluation in our school was significant. Vanhoof and Petegem (2011) tell us that policy implementation processes are more effective when team members are offered both professional and personal support, for example, continuous professional development, substitute teacher cover, time to meet and talk with each other or a personal copy of the school self-evaluation guidelines etc. The availability or lack of such resources (time, information and support) was informing an emerging pattern regarding the implementation of school self-evaluation within our school.

Upon reflection, I believe this was a failure of leadership from both myself and the DES. With the exception of one advisory visit from the Inspectorate, not one member of my staff has received training in the school self-evaluation process. Between 2012 and 2021, my deputy

principal and I received just four release days to train, in leading and implementing the school self-evaluation process, from the PDST. In addition, it proved very challenging to carve out the time and resources required to engage with and prioritise school self-evaluation, as envisaged by the DES, without disrupting our daily core business of teaching and learning. This was in part due to the simultaneous launch of a range of demanding change and reform initiatives at that time upon schools from various government agencies, coinciding too with a period of significant cutbacks in financial and personnel resources within the system (Coolahan et al., 2017). It was within this context that I naively felt compelled as leader of the school, to take full responsibility for implementing school self-evaluation. This was done in the belief that it would shield my staff from any extra burden to their already busy workload and minimise any negative impact on their morale.

The lack of resources to implement school self-evaluation and the increased workload involved was causing stress to staff members. According to the data, this stress was seen as a significant weakness when trying to implement school self-evaluation in our school. This resonates with Fullan (1993) who believes change in organisations is about change in people, attention to their perceptions of reality and particularly their sense of self. This, Fullan (1993) argues, is the key to successful change agentry. Indeed, Goleman (1996) believes that emotions are the starting point for development and change, due to the recognition that emotions are powerful determinants of 'thinking' (cognitive) processes. Data from the SWOT analysis suggested that there was a "perceived lack of confidence" in doing school self-evaluation among staff members. Probing questions during the discussion revealed that a small number of staff members felt they did not have the necessary skills to carry out school self-evaluation in a competent manner. According to Zeichner and Gore (1990) this is problematic, as school members lacking confidence in their ability to self-evaluate are unlikely to produce keen observations, or unique responses when they encounter problems in the school self-evaluation process. Instead, the response will involve a restricted set of behaviours, based on old patterns of unprocessed emotional distress and confused thinking. To bring about meaningful organisational change through school self-evaluation therefore, staff members must engage in reflection at system level to deal with the emotional dimension of their context (Leitch and Day, 2000). However, research shows this is easier said than done. Expanding policy mechanisms, such as school self-evaluation, have been found to diminish, rather than increase, opportunities for structured reflection (Leitch and Day, 2000).

5.3.3 Leadership and staff motivation

The creation of an internally motivated culture of evaluation within a school is easier said than done according to Karagiorgi et al. (2015). School self-evaluation can only work if team members are positively disposed towards it (MacBeath, 1999). Staff commitment is important to the success of any school self-evaluation initiative (Meuret and Morlaix, 2003). According to Bubb and Earley (2008) winning hearts and minds is crucial in the successful launch of any school improvement initiative. Research shows however, that the necessary starting point of a positive attitude for implementing school self-evaluation is frequently absent (Schildkamp, 2007). According to my reconnaissance data, respondents seemed split on whether this necessary starting point existed in our school or not. A small number of participants listed their motivation and interest in school self-evaluation as a strength in our school. In direct contrast however, a larger majority saw their lack of interest and enthusiasm for school self-evaluation as both a weakness and a threat to its implementation. I was not surprised by this finding, as it was similar to findings in the literature. A study of principals in Holland for example, found they have a generally more positive attitude towards school self-evaluation than that of their teachers (Vanhoof et al., 2009) Similarly, a large survey of primary school principals in Ireland conducted by O'Hara et al. (2016) found that only thirty percent of them felt school selfevaluation was popular with their staff. I found hope however, in the finding that some participants believed there was more potential untapped staff energy and interest in school selfevaluation deep beneath the surface, just waiting to be unleashed, if led well. The literature shows that the quality of a school self-evaluation process is strongly determined by how that evaluation process is carried out (Vanhoof et al., 2011). The attitude of school leadership towards school self-evaluation is significant (Day and Sammons, 2014).

From conversations with my critical friend, she in the first instance observed and referenced my negative attitude towards school self-evaluation. She said it manifested itself through my tone of voice, body language and use of language. She said I got visibly annoyed and angry when talking about the process. To unpack my negative thinking towards the school self-evaluation process, I began to use a reflective journal (extracts can be found in Appendix 17). I noted that I unashamedly did not hide my distain or lack of enthusiasm for school self-evaluation from my staff. When speaking negatively about school self-evaluation in casual conversation with my staff in terms of increased workload etc., I noticed many of them agreed with me. From such self-reflective practice and continuous conversations with my staff had

for the process. This lack of interest and enthusiasm amongst school staff for school selfevaluation was identified by staff members as both a weakness and a threat in the reconnaissance data. According to Leithwood and Mascall (2008) school leaders play a key role in shaping teachers' attitudes, commitment and effectiveness in any school self-evaluation process. This action research project was my attempt to do just that by reflecting on my core values and improving both my attitude towards school self-evaluation and my leadership of the process. A study by Leithwood and Jantzi (2008) found that the more school principals enacted the core leadership practices of (i) building vision and setting directions (ii) understanding and developing people (iii) redesigning the organisation and (iv) managing the teaching and learning programme; the greater their influence on teachers' capacities, motivations and beliefs about the supportiveness of their working conditions. In turn these capacities, motivations and beliefs had a positively significant influence on classroom practice. Going forward therefore, it was imperative, for the success of school self-evaluation as a method for school improvement in our school, that I continuously engaged with my critical friend and reflected on all that I said, did and thought as I led the process.

5.4 Further analysis through the lens of complexity theory

This section will complement the analysis already presented and will comment on the findings through the following four key concepts of complexity theory.

5.4.1 The diversity and dynamic nature of agents

Staff members saw the knowledge and professionalism they brought to self-evaluation as a strength. It was no surprise to find, therefore, that any change in staff was considered by staff members to be a loss to the school self-evaluation process in terms of knowledge and experience. They also said it meant extra work for the remaining school members in retraining and bringing up to speed, new staff members with the process. Typical of a complex system, each member of staff within a school is going to have very diverse experiences of engaging with and implementing school self-evaluation within their school. In fact, data from the SWOT analysis revealed that a small number of staff members still felt that they did not have the necessary confidence or skills to carry out school self-evaluation in a competent manner. Complexity theory would suggest however that such diversity and dynamism among staff members enables a school to continually generate new possibilities and the resilience needed to sustain itself throughout both challenges and losses (Fenwick, 2012). This is problematic however, as the model of school self-evaluation mandated by the DES is a one-size-fits all

approach that seeks input from all school stakeholders as a collective process. The whole school in its entirety is seen as the unit of intervention (Simons, 2013; Chapman and Sammons, 2013). But in reality, the success of the system is very dependent on the capacity of individuals, in a variety of settings with differing skills, knowledge and ability levels (Ryan and Timmer, 2013). School members lacking confidence in their ability to self-evaluate is therefore problematic, as their engagement with the process will involve restricting sets of behaviours based on old patterns of unprocessed emotional distress and confused thinking (Zeichner and Gore, 1990). The necessary starting point of a positive and committed attitude to implementing school self-evaluation is therefore diverse and frequently absent (Schildkamp, 2007). In the data, a large majority of the participants saw their lack of interest and enthusiasm for school self-evaluation as both a weakness and a threat to its implementation which is problematic by itself.

5.4.2 Nested systems, connectedness and patterns of interaction

School staff saw the relationships and the interactions between themselves as a strength. Complexity theory suggests that the more frequent and powerful these interactions are, the more influence they are likely to have on the behaviour of school members (Holland, 1992, 1999; Keshavarz et al., 2010; Fenwick, 2012). Within masses of interaction, the system as a whole can become energised and sensitive to fluctuation, continually operating with unfolding configurations and a multiplicity of possibilities (Osberg, 2008). To ensure the success of school self-evaluation at the whole-school level, therefore, it was important that the participants had plenty of meaningful opportunities to connect and interact with each other. This would suggest that for school self-evaluation to be implemented successfully, it must begin with a 'bottom-up' community approach that involved everyone. This resonates with my findings. The data showed some staff members wanted more opportunities to engage in teamwork as part of the school self-evaluation process. To implement school self-evaluation successfully therefore, it was important to continually give school members adequate time to interact as a collective.

5.4.3 Information flow and feedback loops

Staff members expressed a desire for increased supports to successfully implement school selfevaluation. The availability or lack of resources such as time and information were informing an emerging pattern of discontent amongst school staff with the status quo. This in turn was feeding back into the system and having a negative impact on the implementation of school self-evaluation within our school. Staff members expressed a strong desire for a clearer layout from the DES on how to implement school self-evaluation. This is important as when
information is freely provided, and structures support use of that information, people engage more productively in their work (Bower, 2006). The movement of information, through the interaction of people, lies at the heart of any learning and change process within a complex system, such as a school (Morrison, 2002; Stacey, 2005). To implement school self-evaluation successfully, it was important to resource the project in terms of professional and personal support such as training, access to information, substitute cover and operational manuals.

5.4.4 Change, adaptability, co-evolution, self-organisation, and emergence

Complex systems are always in a state of dynamic interaction, where change is a constant in almost all aspects of school life, occurring with different frequency and intensity (Morrison, 2002; Osberg, 2008; Keshavarz et al. 2010; Fenwick, 2012). When there is a change in conditions, schools can choose to make adaptive changes and evolve to ensure a best fit for their own context (Osberg, 2008). Unity and coherence can sometimes emerge without any imposition or planning (Fenwick, 2012). My findings support this view. Rather than seeing school self-evaluation as a meaningful method of improvement, staff members view school self-evaluation in a negative way as just another area of work that has to be done, adapting a stance of contrived compliance rather than one of educational endeavour (Hall and Noyes, 2009). This sets the challenge of finding ways to change this attitude and emerge with a more meaningful process.

5.5 The next step

Having gathered a detailed picture of the salient issues in the SWOT analysis, it was now imperative to engage school staff in generating solutions and strategies as to how we might collectively overcome them. To do this, we engaged with another facilitation tool called 'Brain Writing 6-3-5'. Use of this tool was discussed and agreed upon by both myself and my critical friend well in advance of the whole staff gathering together. We were quite satisfied that this tool would be both simple and effective for staff to use in generating some ideas to improve the implementation of school self-evaluation in our school. My critical friend and I also liked the fact that it was quick to use. Considering having the time to meet as a whole staff was a rare opportunity, we both felt it was important to optimise the time available to us as much as possible. Brain Writing 6-3-5 is a silent brainstorming process developed by King and Schlicksupp (1998) that anyone can use to identify new ideas or solutions. The goal is to generate as many creative ideas as possible.

This silent approach holds many benefits for a school setting. Staff within any school can be quite diverse in terms of their personality. Some staff members for example, can be extrovert and quite verbal, whilst others can be introvert and quite shy (Kise and Russell, 2008). Those on temporary contracts may have a shorter length of teaching service and are not as confident to speak up, as compared to their counterparts on permanent contracts with many years of experience who can express their point of view with great confidence. In my experience of working in a variety of schools, this diversity can affect the input level of individual staff members at whole school staff meetings (Jennings, 2007; Kise and Russell, 2008). As an approach, Brain Writing 6-3-5 (Hart, 2007) effectively lessens power differentials to create a richer dialogic and inclusive community, ensuring that highly verbal people do not overwhelm quieter members. It enables individuals to see what others have written, allowing creative ideas to be shared for mutual benefit, but in a silent way. The process is conducted in six 'rounds' of five minutes or less using the 'ideas' worksheet pictured below (*Table 5.2*), with teachers working individually and privately at first and then publicly sharing their thoughts.

How can we improve the process of school self-evaluation in our school?				
Participant's name	idea 1	idea 2	idea 3	

Table 5.2Brainwriting 6-3-5 'Idea creation' strategy

The question in this case was 'how can we improve the process of school self-evaluation in our school?' In the first step, each member got a copy of the 'ideas' worksheet (*Table 5.2*). Each staff member was required to fill in their name and three potential ideas within a few minutes. When the allocated time was up, each staff member was required to pass their sheet to the person beside them. They then silently read the three potential ideas from the new worksheet

they just received, taking note of anything that interested them. When the allocated time was up, they again passed this sheet to the person beside them. This process of rotation continued until each member received their own original worksheet back. After all the rotations were completed, each participant was invited to publicly contribute two great ideas from all the worksheets they had read. When all the ideas were recorded, they were then combined and streamlined by discussing the pros and cons of each. The following key ideas emerged from this activity.

There was a desire amongst staff members for clarity and communication on the practicalities of school self-evaluation and continuous professional development on how to implement it correctly in our school. Analysis of the reconnaissance data and the idea generating exercise 'Brain Writing 6-3-5', showed that they wanted to learn and understand more about school self-evaluation and its benefits. I believe my leadership was a barrier due to the uncollaborative, monologic and authoritative approach I had taken to implement school self-evaluation. My original contempt for school self-evaluation and the desire to shield my staff from the extra workload influenced my leadership of the process to go it alone. I did not give my staff any opportunity to take part and learn more about the school self-evaluation process and its purpose. For the last four years, staff went along with this practice. One could argue, that school selfevaluation was being implemented in our school, but not with absolute fidelity to what was originally intended by policymakers. However, this had created a school self-evaluation knowledge gap amongst school staff. A vacuum of knowledge and skills can cause staff members to interpret and adapt policies and guidelines in unpredictable ways to the needs and practices of the organisation (Morrison, 2002; O'Day, 2002; Fidan and Balci, 2007; Ngan et al., 2010; Keshavarz et al., 2010). In such a scenario, school self-evaluation as a policy is interpreted by school staff and gets adapted to their own school context (Hudson et al., 2019). However, this action research intervention was an opportunity to address the issue. It was also an opportunity to invigorate interest, enthusiasm and a positivity for the school self-evaluation process amongst my school staff. A good starting point, according to the staff themselves, was to give them access to information, resources and supports to develop their understanding of school self-evaluation, its purpose and its benefits. They also wanted visual resources to provide clear communication about the school self-evaluation process which were easy to understand, but due to enormous workload pressures, also easy to use. This implied that in any action going forward staff needed engagement with the school self-evaluation literature and some form of training.

There was also a desire amongst staff members to have a coherent whole school approach to conducting school self-evaluation which was simple and efficient to use and produced achievable outcomes. They wanted to ensure that any time spent doing school self-evaluation led to improved outcomes for both teachers and learners. Staff members had very practical ideas on how this could be done. To begin with, staff wanted to administer questionnaires to school stakeholders. Staff seemed to place great value on the involvement and input of other people from the school community into the school self-evaluation process. This was similar to findings from the earlier SWOT analysis where participants saw the relationships and the interactions between them as a strength and powerfully motivating. They therefore wanted more opportunities to collaborate in teams to identify and agree on areas that needed attention. It highlighted a desire amongst staff to move away from the solo approach I had been taking as school leader to implement school self-evaluation. Staff members clearly wanted school self-evaluation to be a form of community endeavour where everyone was involved and had a voice in the process. It indicated school staff assuming a level of agency in the research and working alongside me as co-researchers. I would argue however that it also represented a desire amongst staff for school self-evaluation to be meaningful, and not just a box-ticking exercise. If school self-evaluation was to be successful, it needed to be meaningful by inviting participation.

Another idea from staff to make school self-evaluation more meaningful and successful involved the analysis of pupil's work. The act of analysing pupil's work suggested that staff were keen to focus on areas of teaching and learning that were both practical and meaningful to their everyday work in the classroom. Research in Ireland by O'Brien et al. (2019) shows that if school self-evaluation results in improvement at classroom level, teachers are more likely to value the process.

Finally, staff members desired to have more input into the school self-evaluation process. An analysis of the data showed staff members to be both interested and willing to engage with school self-evaluation if given the time and space to do so. This suggested that time, as a finite resource, could influence staff participation in the school self-evaluation process. The availability of time, or lack thereof, is a recurring theme in the literature (Ryan et al., 2007; Hall and Noyes, 2009; Vanhoof et al., 2009; National Council for Curriculum and Assessment, 2010; O'Brien et al., 2015). Any attempt to improve the process of school self-evaluation in our school must clearly involve creating the time and space for it. Analysis of the idea generation exercise 'Brain Writing 6-3-5', validated this. Staff wished to meet together in a

spirit of collaboration. They wanted to have professional open conversations about the evaluation and improvement of teaching and learning in their school. This was important. Information, knowledge and feedback is passed on when people within a school interact with each other. Stemming from this can be the creation of new learning and new possibilities. The more frequent and powerful the interactions are the stronger the system will be (Holland, 1992, 1999; Keshavarz et al., 2010; Fenwick, 2012). Staff members deemed it important that all whole school decisions adopted a practitioner-based approach, made in a spirit of collaboration as a collective, allowing all voices to be heard. Due to my desire to shield staff from the extra workload involved in school self-evaluation and go it alone however, I had failed to offer staff any opportunity to engage collaboratively as a collective. For this project to succeed staff needed to be allowed to meet collaboratively without distraction. To make this a reality the supports required, such as non-contact classroom time, needed to be in place.

5.5.1 Prioritising our ideas

Sheets containing all their ideas from the session were put on display in the school staff room for reflection and conversation, with the aim of reconvening another group meeting to prioritise, in order of merit, our main ideas. To do this a visual remodelling tool called a prioritisation grid was used (National College, 2012). It is a quick and effective way to generate group consensus about what should be a priority. It is effective in shortlisting a large number of ideas for action into a more manageable number. Each possible action is written on a sticky note and positioned by the group on a grid in terms of the groups desire to do it and the rate of impact it would have on fulfilling their objectives. The rate of desirability is measured on a sliding scale from 'low to high' on a vertical axis. The rate of impact is measured on a sliding scale from 'strong to weak' on a horizontal axis. Any ideas deemed to be high in desirability and strong in impact are placed close to the top right corner of the grid. These ideas are then the chosen priorities for action. The grid can be seen in *Table 5.3*

Table 5.3Idea prioritization strategy

		Prioritization of ideas for action
High Desirability		
Low Desirability		
Low Desirability		
	Weak	Strong Impact
	Impact	

(National College, 2012).

I discussed the technique with my critical friend. She again liked the technique due to its simplicity and the immediacy of its results. However, she did ask what would happen if a large bulk of the ideas from staff found their way to the top right corner as being highly desirable with the potential for a strong impact. What would we do then? I referenced the techniques manual to find the answer. Upon such a scenario arising, it said a limited debate should ensue to attain consensus on having the items distributed across the matrix so that only a few (but it did not specify), fall in the top right-hand corner of the grid. My critical friend and I were confident this could work and concluded that this tool would be most suitable for this initial stage of the project.

A discussion with staff members resulted in the following prioritisation grid (*Table 5.4*). Any ideas which were deemed highly desirable and would have a strong impact were placed in the top right-hand corner by school staff.

	How can we improve the process of school self-evaluation in our school?
High Desirability	A clear agenda of what needs to be done Open communication amongst staff
	Use a chart or poster to clarify thingsMore face-to-face conversation, even a ten-minute chat can
	Bullet point our ideas to keep it simple clarify things Give a clearer understanding of SSE
	Give a handout covering the main points about school self-evaluation for all staff and its purpose with increased support and resources in
	Focus on the positives on the school self-evaluation to continue to encourage development
	Regular short questionnaires about what's working in our school/classrooms and what's not!
	Introduce set timetabled weekly sessions for the specific area being improved/ taughtwith a little reminder on the intercom that morning!
Low Desirability	Different teams Ensure that any focusing on different areas
	Weak Impact Strong Impact

Table 5.4Idea prioritization grid – How can we improve the process of SSE in our school?

The biggest priority for school staff was a desire to have more opportunities for collaborative professional conversations with each other as part of the school self-evaluation process. Staff members felt strongly that this would greatly impact the implementation of school self-evaluation in their school in a positive way. Coupled with this was the aspiration that any time spent doing school self-evaluation needed to improve teaching and learning. This too was a high priority for school staff. Staff wanted more opportunities to meet together as a collective in a spirit of collaboration, to have professional open conversations about the evaluation and improvement of teaching and learning in their school. Staff members deemed it important that all whole school decisions were made in a spirit of collaboration as a collective, with all voices being heard. For this to happen staff needed to be given the supports to meet collaboratively without distraction, such as non-contact classroom time. An analysis of all the reconnaissance data showed that staff viewed time as a valuable commodity. If this time was used to focus on matters of teaching and learning as a collective, they believed it had the strong potential to improve the process of school self-evaluation in the school.

An analysis of the priority matrix also showed that clarity and communication on how school self-evaluation is implemented, was a priority for school staff. They wanted visual resources, such as charts, posters and information sheets to provide clear communication on the practicalities of the school self-evaluation process, which were easy to understand and easy to use. The provision of resources such as this, according to staff, would have an immediate and very strong impact on the implementation of school self-evaluation.

Finally, the desire of staff members for a greater knowledge and understanding of school selfevaluation and its benefits was also a high priority. Reconnaissance data showed that staff members were not familiar with the school self-evaluation literature from the DES. This had created a school self-evaluation knowledge gap amongst school staff. According to staff, a good starting point to address this issue would be to have access to information, resources and supports to develop their understanding of school self-evaluation, its purpose and its benefits. This, they believed, would have a strong and positive impact on the implementation of school self-evaluation in our school. Any action therefore, needed to involve engagement by staff with the school self-evaluation literature and some form of training.

To summarize, the priority issues arising from this reconnaissance were firstly, more opportunities for staff members to meet together as a collective in a spirit of collaboration to have professional open conversations about the evaluation and improvement of teaching and learning in our school. Secondly, the acquisition of visual resources to provide clear communication on the practicalities of the school self-evaluation process, which were easy to understand and easy to use. Lastly, access to information resources and supports to develop staff members' understanding of school self-evaluation, its purpose and its benefits.

Chapter 6 – Action Research Cycle 1

6.1 Introduction

The purpose of this chapter is to describe the first action research cycle in this research. This involved planning and implementing the action steps necessary to improve the implementation of school self-evaluation in our school and monitoring their effects.

6.2 Planning

At a meeting in January 2020 staff members discussed at length the actions steps needed to improve the process of school self-evaluation in our school. The ideas prioritised in the reconnaissance phase were used to frame this action plan.

The priority issues for school staff in the reconnaissance phase were firstly, more opportunities for staff members to meet together as a collective in a spirit of collaboration to have professional open conversations about the evaluation and improvement of teaching and learning in our school. Secondly, the acquisition of visual resources to provide clear communication on the practicalities of the school self-evaluation process, which were easy to understand and easy to use. Lastly, access to information resources and supports to develop staff members' understanding of school self-evaluation, its purpose and its benefits. Discussions during the reconnaissance phase also indicated that for school staff, any implementation strategies needed to lead to improved outcomes for both teachers and learners. There was much sharing of ideas among staff members on how all this could be achieved.

The first priority issue represented a desire amongst school staff for increased opportunities to collaborate and have more input as a collective into the school self-evaluation process. Many ideas were put forward to facilitate this, such as more staff meetings or smaller break-out group meetings. However, one participant argued that it was more important to focus on the actual content of the conversations, rather than the format of the meetings. The group did struggle however, to reach consensus on what the content of these collaborative professional conversations should actually be and what format they should take. As a facilitator of the meeting, I stepped in to remind the group that the focus of the planning meeting was on how to improve the process of school self-evaluation in our school. This intervention seemed to renew energy in the discussion. Eventually, constructive ideas emerged. One in particular proved very promising. Some staff members felt that the only way to really improve our capacity in school self-evaluation was to actually do it. One participant termed it "*learning by*

doing". It was decided by school staff to initiate a school self-evaluation cycle for the current school year, following DES guidelines. Engaging in such a cycle allowed staff members the opportunity to have direct input into the school self-evaluation process and collaborate together in a meaningful way. The DES guidelines (2016) are quite clear that school self-evaluation is primarily about schools taking ownership of their own development and improvement and that professional collaborative review is a key component of the process. The group nominated the in-school management team to lead the action (Principal, deputy principal and two post holders), with all school staff invited to participate. This planned action was labelled "Action 1".

To address the second priority issue consensus was reached on the creation of a school selfevaluation timeline, suitably matched to a school calendar year. The aim was to have a timeline that would communicate to all school staff in a very clear way, what practical school selfevaluation activities should occur at whole school level during each month. Staff members agreed that all school staff should be invited to be involved in its creation and that it should be completed and put on display in the school staff room. This planned action was labelled "Action 2".

The third and final priority issue which emerged during the reconnaissance phase, identified amongst staff members an urgent need for access to information, resources and supports to further develop their understanding of school self-evaluation, its purpose and its benefits. Staff members agreed to create a dedicated school self-evaluation information board for the school staffroom. This would specify for each school term or school year, what our chosen areas of focus were on certain school subjects. The information board would also highlight specific improvement targets and whether they were being met or not. It would also be used to show the results of any staff surveys regarding school self-evaluation or highlight any continuous professional development training opportunities in key target areas. This planned action was labelled "Action 3".

In addition, it was decided by school staff to install school self-evaluation resource trays in the school staffroom. These trays would hold operational manuals, DES guidelines and circulars, newsletters and any available research literature on the implementation of school self-evaluation. Reconnaissance data revealed that staff members were not familiar with the school self-evaluation literature from the DES. This planned action was labelled "Action 4".

The group also asked me, as school principal, to show and talk staff members through the DES school self-evaluation guidelines and quality framework. The aim was to help school staff become more familiar with the contents and give them a good understanding of the purpose and methodology of school self-evaluation. Reconnaissance data revealed this to be a key issue for staff. This planned action was labelled "Action 5".

It was at this point of the proceedings however, that one staff member suggested we should reorder the sequencing of our action steps. They argued that actions two to five should precede "Action 1" and be treated as important preparatory steps for the main task of implementing a school self-evaluation cycle. All staff agreed. This concluded the project's planning phase.

As evident from the planning phase discussion described above, the involvement of dynamic and diverse agents (such as teachers, special needs assistants, principals and special education teachers) in such decision making, makes schoolwork complex, but it also makes our school stronger. Without such variation, the movement and reinterpretation of information would not occur. This lies at the heart of learning and adaption in any complex system (O'Day, 2002; Bower, 2006; Vanhoof and Petegem, 2011; McNaughton et al., 2013; Chapman and Sammons, 2013; Setlalentoa, 2014). Complexity theory was an important guiding factor in my actions throughout this project. For the success of this project for example, it was important my leadership did not treat school staff as recipients of information, but as the source. Interactions, such as the planning meeting described above, needed to be frequent and powerful to influence the actions of staff members in our school. The more our school was open to change and new ideas and trying new things, the closer we would get to improving the school self-evaluation process in our school through increased innovation and creativity.

Keeping the reader in mind, the following is a summary chart (*Table 6.1*) of the agreed actions described above and undertaken as part of this first action research cycle.

How c	an we improv	ve the process of school	l self-evaluati	on in our s	chool?
		Action Plan	l		
What wil	l we do?	Why are we doing it?	Who will do it?	When will we do it?	How will we measure our success? (i.e., what criteria do we believe will show the impact of our work?)
Preparatory step 1	Create a school self- evaluation timeline.	 (i) for clarity on how school self- evaluation is communicated and implemented (ii) to communicate to staff in a clear way 'what' school self-evaluation activities should occur and 'when' during each month of school (iii) so that any time spent doing school self-evaluation is used efficiently to improve teaching and learning 	staff	End of January 2020	A timeline will be created and put on display in the staff room and it will be used in a school self- evaluation cycle
Preparatory step 2	Create a school self- evaluation information board.	(i) for clarity on how school self- evaluation is communicated and implemented (ii) to gain a greater knowledge and understanding of school self- evaluation and its benefits	principal	By Easter 2020	An information board will be attached to a wall in the staff room and labelled as school self- evaluation and it will be used in a

Table 6.1Action research cycle 1: Action plan

					school self- evaluation cycle as part of this project
Preparatory step 3	Create school self- evaluation resource trays.	to gain a greater knowledge and understanding of school self- evaluation and its benefits	principal	By Easter 2020	Resource trays will be attached to a wall in the staff room and labelled as school self- evaluation. They will be stocked with the latest resources and newsletters regarding school self- evaluation and they will be used in a school self- evaluation cycle as part of this project
Preparatory step 4	Show and talk staff members through the DES school self- evaluation guidelines and quality framework.	to become more familiar with their contents and to gain a better understanding of the purpose and methodology of school self- evaluation	principal	By Easter 2020	The principal will meet all staff (individually or in groups) to show and talk them through the DES school self- evaluation guidelines and quality framework

Action	Initiate a	To increase	Led by the	By	The staff
aton 1	school self-	opportunity for	in-school	Easter	will carry
step 1	evaluation	professional	management	2020	out in full, a
	cycle.	collaboration	team but		school self-
			inviting all		evaluation
			stakeholders		cycle using
			to		the DES
			participate		school self-
			(somebody		evaluation
			called this		guidelines,
			<i>"learning by</i>		our new
			doing")		school self-
					evaluation
					timeline,
					resource
					trays and
					information
					board.

After this phase of action planning, I noted in my reflection journal, that during the discussion I observed staff being very concerned with the technical aspects of the school self-evaluation process. Most of the actions listed above, entailed a focus on prescribed procedures, guidelines and information. The exception was 'Action step 1', which put an actual focus on engagement in the process of school self-evaluation itself. However, that said, staff would be following in a technical way the process prescribed by the DES. My concern was that staff members saw school self-evaluation as a form of evidence-based, technical activity for change, with predetermined milestones and fixed end goals, where staff merely played a passive and compliant role. However, this was not to be unexpected. According to McNamara and O'Hara (2012), when the DES provided schools with guidelines and procedures for the formal introduction of school self-evaluation (DES, 2016), no attempt was made within them to define and encourage the conditions necessary to make them possible in practice. Since then, however, the supports and materials available have undergone significant development, but issues still remain (McNamara et al., 2021a). The DES now sends school inspectors to provide professional support to schools on request and has a number of publications and support materials related to school self-evaluation. With the exception of school management teams however, no training in school self-evaluation is provided to entire staff (McNamara et al., 2021). This leaves a gap between the school self-evaluation training needs of schools and the support provided. The school self-evaluation guidelines and procedures, whilst technical, help to fill that gap.

A technical approach to school self-evaluation sees school staff as technicians implementing prescribed procedures to deliver improved outcomes through efficient and effective practice (Bransford and Darling-Hammond, 2005; McNaughton, 2011). It may involve 'technical' rational reflection as means of increasing teachers' efficiency in the delivery of policy (Leitch and Day, 2000) but it presents little perceivable benefit or opportunity for the development of staff members capacities for reflection 'in' or 'on' practice to develop knowledge of practice (Leitch and Day, 2000; Schön, 1983). This was problematic, with potential consequences for the success of this study. I was conscious that for school self-evaluation to be effective, it needs to move beyond a technical interest, and instead engage in critical questioning and reflection by school staff to understand and explain their actions as they institute change to their practice (Sjøbakken and Dobson, 2013; Klenowski and Woods, 2013; Van Der Voort and Wood, 2014).

In keeping with the spirit of this practical action research study to improve our practice, the process of school self-evaluation we implemented was rigorous and critical, asking fundamental questions about what we do, how we do it and most importantly, why we do it. This, according to Leitch and Day (2000), builds problem-solving, self-evaluation capacities and an understanding of effects on teaching and learning, thus encouraging further research, experimentation and ultimately, change (Burton and Bartlett, 2009; McNaughton et al., 2013). As a result, rather than implementing other people's knowledge in practice through prescribed procedures, the intention in this action research was for staff members to become thoughtful and learn from their work, reflecting on both their actions and the consequences (Leitch and Day, 2000).

According to Elliot (1991) "action initiates reflection" (pg.23). The fundamental aim of action research is to investigate and improve practice (Elliot, 1991) through a process of action and reflection (McNiff, 2012). According to Burton and Bartlett (2009) this reflexivity heightens as researchers develop their skills. Therefore, whilst my concerns were legitimate, Ryan et al. (2007) acknowledge that planning and implementing school self-evaluation is multi-faceted and requires considerable expertise. In their study, schoolteachers and principals as novice evaluators, experienced significant challenges with the conceptual and technical aspects of evaluation, suggesting the existence of evaluation knowledge deficits. Another study by Sjøbakken and Dobson (2013) found that school stakeholders during school self-evaluation were not immediately motivated or skilled in writing down their own experiences. But what was even more interesting was the finding that stakeholder engagement with the school elf-evaluation process over a period of time showed a pendulum movement between school self-evaluation school self-evaluation process over a period of time showed a pendulum movement between school self-evaluation school self-evaluation process over a period of time showed a pendulum movement between school self-evaluation school self-evaluation process over a period of time showed a pendulum movement between school self-evaluation school self-evaluation process over a period of time showed a pendulum movement between school self-evaluation school self-evaluation process over a period of time showed a pendulum movement between school self-evaluation process over a period of time showed a pendulum movement between school self-evaluation process over a period of time showed a pendulum movement between school self-evaluation process over a period of time showed a pendulum movement between school self-evaluation process over a period of time showed a pendulum movement period period period period period period period period pe

evaluation as a technical, practical and emancipatory process, suggesting that those engaged with school self-evaluation do not necessarily move in a chronological fashion from technical to practical and practical to emancipatory research, as suggested by Carr and Kemmis (1986), its fluid. Therefore, upon reflection, I felt that staff engaging with a technical perspective towards school self-evaluation at the outset of this action research study was acceptable and part of a longer journey of progression towards more reflexive critical thinking and development.

6.3 Implementation

The following preparatory steps were implemented as part of a suite of resources to be used as part of our planned cycle of school self-evaluation, to improve the process in our school. An evaluation of these actions and their impact on the aim of this research is discussed later in the thesis, as part of an appraisal of our overall engagement in a cycle of school self-evaluation.

6.3.1 Preparatory step 1: Create a school self-evaluation timeline.

In January 2020 staff members came together to discuss and agree on a broad school selfevaluation timeline that would fit our school's annual calendar. The school self-evaluation process envisaged by the DES (DES, 2016), is a six-stage process that facilitates repeated cycles of analysis or a return to a previous stage of the cycle as required. It involves looking at the evidence of pupils' learning, analysing it, interpreting it and acting on it to improve outcomes. The cycle looks as follows:



Figure 6.1 The school self-evaluation cycle

Steps one, two and three of the cycle outline the investigation phase of the process. This begins with the identification of an area for focus and inquiry through teachers' own understanding and knowledge of their school context. This is followed by the gathering of evidence with regard to the area chosen. The evidence is then analysed, and judgements are made regarding the school's current strengths and areas for improvement. These three steps take place in the first year of a four-year cycle. In steps four, five and six schools retain a record of the evaluation and describe the actions for improvement. This school self-evaluation report and improvement plan is then shared with the school community. The improvement plan is put into action and is monitored and evaluated over the course of a three-year implementation.

As a staff we discussed the possibility of mapping the six-stage school self-evaluation model over a calendar year beginning in January, as extra-curricular activity in our school tends to dip during the months of January, February, early March and November. These months are free from seasonal events such as Christmas, Easter and Halloween. They are also outside of busy school periods, such as end of school year activities in May and June and the start of school year activities in September. When we did, we found that it was very workable, potentially giving us the time and space, we needed in the first three stages of the school self-evaluation cycle to engage with the practical tasks of gathering evidence, analysing data and making judgements. We agreed on the following school self-evaluation timeline:

School Self-Evaluation Cycle		
Year 1		Teaching and Learning
January – March	Step 1	Identify focus
March	Step 2	Gather evidence
April – June	Step 3	Analyse and make judgements
September – October	Step 4	Write and Share
November – June	Step 5	Put school improvement plan and digital learning plan into action
Year 2, 3, 4	Step 6	Monitor actions and evaluate impact for beginning of next cycle

Table 6.2School self-evaluation timeline
--

A large copy of the timeline was put on display for all participants to see and use in the school staff room.

6.3.2 Preparatory step 2: Create a school self-evaluation information board.

I purchased a noticeboard specifically for school self-evaluation and organised it to be attached to a wall in the school staff room. Information such as standardised test data, school improvement targets and the new school self-evaluation timeline were displayed there for all staff to see and engage with (*Figure 6.2*).



Figure 6.2 School self-evaluation information board

6.3.3 Preparatory step 3: Create school self-evaluation resource trays.

Preparatory step three entailed the installation of resource trays to hold books, information folders and newsletters relating to school self-evaluation, all in an attempt to help staff have a greater knowledge and understanding of school self-evaluation and its benefits. As the person responsible for the realisation of this action, I ensured the trays were installed in the staff room (*Figure 6.3*). I filled them with the most up-to-date DES literature on school self-evaluation I could find, as well as information on teaching, learning and assessment from the Professional Development Service for Teachers (PDST) and the National Council for Curriculum and Assessment (NCCA).



Figure 6.3 School self-evaluation resource trays

6.3.4 Preparatory step 4: Show and talk staff members through the Department of Education school self-evaluation guidelines and quality framework.

As part of preparatory step four, all staff were to be shown the school self-evaluation circulars and school self-evaluation guidelines from the DES to become familiar with their contents. This was to address the identified need of staff members to have a better understanding of the purpose and methodology of school self-evaluation.

In the spirit of action research, I felt an appropriate approach would be to actively read the DES documents together with each staff member. However, I was aware that meeting staff members on an individual basis had ethical implications regarding power relations which needed further reflection and consideration. I was in the dual role of researcher and school principal. It was important I did not put staff members into a position of vulnerability. For that reason, I discussed with my critical friend, my proposition to meet all staff members on an individual basis. I explained how this approach would allow each staff member and I the time and space to read and discuss in depth together, the DES school self-evaluation documents. I explained further how it would enable me to give my full attention to any individual needs or questions they may have regarding school self-evaluation, in a private, safe and secure way. This, I felt, was important, so as to ascertain as clear a picture as possible, what level of understanding staff members possessed regarding the school self-evaluation process.

As expected, my critical friend questioned if I was putting participants in a vulnerable position. She was concerned for fellow colleagues who may have felt an obligation to take part in this research as it was being carried out by the principal and may be uncomfortable about exposing gaps in their knowledge about school self-evaluation. Whilst I took solace in the fact that this preparatory step was suggested and agreed upon by staff members themselves in the planning phase of this action research cycle, I was aware that they did not suggest this one-to-one approach. My critical friend and I agreed however, that these one-to-one meetings would be beneficial, but with clear steps to be taken, showing an awareness of my ethical responsibilities in this regard and the dual role I was playing as the school principal and an active researcher and the implications of this regarding power and vulnerability.

The individual meetings were scheduled for January 2020. As a precondition, all staff members said they were happy and willing to have this conversation with me. To begin the conversation, I asked what did they know about school self-evaluation. Of the fourteen people I talked to, all

could give me an answer to the purpose of school self-evaluation in terms of its general application as a method to find our weak points as a school and improve on them. I handed them a copy of the school self-evaluation circular and guidelines and the quality framework. We proceeded to actively read through them and discuss the main headings together. Out of the fourteen staff members I engaged with, only eight were familiar with the documents. This can be explained by the fact that these eight staff members were working in the school system for over ten years and therefore had encountered school self-evaluation before in some guise, be it through in-service training or working with the process. For the others this was their first time to explore the documents, having received no exposure to them in teacher training, but more telling, no encouragement to engage with them due to my poor leadership of the process.

This gave credence to the argument that if school self-evaluation was not considered important enough for work at classroom level, it ran the risk of not only being viewed by new teachers as a documentation exercise for the purposes of compliance and accountability (O'Brien et al., 2019), but also as a process in which a negative attitude was totally warranted. A national survey of Irish primary school principals found school self-evaluation to be popular with only thirty percent of staff, with seventy-nine percent of their staff seeking more training in how to do it (O'Hara et al., 2016). This is worrying for the successful implementation of school selfevaluation among new teachers. To improve the implementation of school self-evaluation in our school, it was imperative that any exposure to the process documents was both positive and meaningful. Giving staff members their own time and space to read and navigate their way through these documents in this preparatory step, as compared to doing it in a group type setting, allowed each member a real differentiated approach to learning about school selfevaluation.

At the end of each meeting, I asked the staff member if there was anything they did not understand and if they had any questions. With the exception of a few technical questions about the school self-evaluation process itself, all staff members reported themselves to be now more familiar with the documents and understood their purpose. As a result of this action, I was confident that all staff at least could now identify the school self-evaluation documents and their basic function. However, upon reflection this was a missed opportunity. I did not have any actual hard evidence to demonstrate this assumption. In hindsight, a better strategy would have been to get the first participant I met to explore, discuss and explain the school selfevaluation documents to the next person attending and so on. My role then would have been to gather actual hard evidence that the action target was achieved. However, using a complex adaptive system theory lens, this strategy too would be deemed ineffective as an interaction of any great power, as the movement of the information is limited. In a complex adaptive system, not all information leads to learning and change. A top-down flow of information such as this, whether from researcher to participant or from participant to participant, sees agents within a system as mere recipients of information, instead of a source of valuable information in their own right. I took comfort in the fact however, that staff members would get an opportunity to interact with these documents when the school engaged in a school self-evaluation cycle through this collective action research project. An evaluation of this step, and indeed all the aforementioned preparatory steps, and their impact on the aim of this research are therefore discussed later in the thesis as part of an appraisal of our overall engagement in a cycle of school self-evaluation.

6.3.5 Action step 1: Initiate a school self-evaluation cycle.

6.3.5.1 Implementing the action.

It was decided by staff that this action would be managed by the in-school management team (i.e., the principal, deputy principal and two post holders) for logistical reasons, but that all staff members would be involved. On a practical level, staff members felt that it would be far easier for the in-school management team to regularly come together and coordinate the school self-evaluation cycle rather than the whole staff. Importantly however, staff members were to be consulted at every step of the evaluation process by the in-school management team to ensure a spirit of collaboration, democracy and community endeavour.

In late January 2020 the in-school management team had one of two meetings to work out the strategy for initiating a school self-evaluation cycle in our school. After much discussion and the sharing of ideas, the team decided that the first goal was to facilitate staff members voting on an area of focus that we could evaluate and improve on. Voting sheets were returned by the end of January 2020. The compiled results (*Table 6.3*) were disseminated to all staff members as follows:

Table 6.3	School	self-evaluation	'area of focus	' survev results

	School Self Evaluation Survey <u>compiled results</u>
	(Listed in 1^{st} and 2^{nd} place and 3^{rd} place)
	Teaching and Learning Focus
1.	Handwriting
2.	New books/schemes for English/Gaeilge/Maths
	Oral Irish (+ cumulative phrases/planning in general) / SESE

The in-school management team met again to discuss the results and take action. From the team's deliberations a three-step strategy to initiate a school self-evaluation cycle in our school was discussed, developed and approved by all staff members as follows:

<u>Step 1</u>

The staff perhaps unsurprisingly, identified handwriting as a teaching and learning focus. With the arrival of the new Primary Language Curriculum in 2019, the entire staff had to attend inservice training organised by the PDST. During these in-service days, we were given great opportunities as a staff to review the new language curriculum and our current school practice. With the help of the PDST the teaching and learning of handwriting in our school was an area in which staff members identified gaps in our practice. In the handwriting section of our school's Primary Language Curriculum policy, we had no agreed baseline standard of what good handwriting would look like for each class level. It should be noted at this point of the study, that parents and pupils were not asked to identify an area of school practice that we could evaluate and improve on. We were of course aware that the DES describes school selfevaluation as a collaborative and inclusive process during which school staff, in consultation and with the fullest participation of parents and pupils, engage in reflective enquiry on the work of the school (DES, 2016). However, as a school we felt it was important at this juncture to have the time and space we needed as a staff to learn and improve the process of school selfevaluation in our school. The decision was made therefore not to involve the wider school community at this point in time.

To begin our cycle, we needed to gather data on the teaching and learning of handwriting in our school. We discussed the creation and administration of a survey to parents, pupils and teachers. Through a process of drafting and revision, we crafted three sets of key questions regarding handwriting in our school. One for parents, one for pupils and one for teaching staff. Please refer to Appendix 18 for the survey questions and results. To make sure the data aligned from different perspectives, we asked similar questions to both the parents and pupils and numbered them the same way. We wanted the set of questions to give a broad picture of both the child's and the parents' experience of handwriting. The teacher survey was comprised of eighteen questions broken up into two sections. The first section concerned what the teaching and learning of handwriting looked like in our school at that current moment in time. The second section asked teachers what the teaching and learning of handwriting 'should' look like in our school into the future and to give some specifics. We felt this approach would allow us to see, in a broad sweep, how handwriting was being taught in the school, but also allow individual teachers then to break free of any group think or indoctrinated school policy and suggest some alternative progressive practice ideas. We also included the current section on handwriting from our school's Primary Language Curriculum policy and asked teaching staff which parts they would like to retain. The teaching staff survey was administered in hard copy due to the small scale in numbers.

The parent's survey was administered online. Out of a total of one hundred and thirty-eight families we received thirty-one responses. It is hard to explain the low response. A key factor may be the effect digital technology has had on devaluing the importance of handwriting quality and ability for these older children and their families. Due to the technologies available, handwriting may no longer be seen as too important (Hensher, 2013; Stevenson and Just, 2014; Ates, 2018; Taneri and Akduman, 2018). A child can just as easily type and print their work. Whatever the reason for such a low response, it suggested that the teaching and learning of handwriting may not have been as high a priority for pupils' families as it was for school staff.

Further analysis showed the majority of responses in our survey coming from more junior classes such as senior infants to 3rd class. In my experience of engaging with parents in parent teacher meetings, handwriting in this younger age cohort seems to be something more tangible and valued by parents. They see its value due to the positive or negative effect it can have on both their child's development and ability to express their thoughts, ideas and knowledge in written form. The handwriting ability of older siblings too would often be used as a benchmark to gauge the quality of their younger child's handwriting.

Next, we surveyed the pupils. Assessing pupils is part of daily teaching and learning in Irish primary schools and is considered best practice (NCCA, 2007). This survey did not therefore

overburden pupils. All the work carried out by pupils in relation to this project was carried out during school hours. The survey did not take from the normal activity of a school day and the pupils' right to an education. The whole process and the purpose of the survey was clearly explained to all pupils in a differentiated and age-appropriate way. Pupils were kept together as part of their normal class grouping and all activities involved in the survey were carried out in line with Child Protection best practice guidelines (DES, 2016) and the Data Protection Act (Government of Ireland, 2018). In line with best teaching practice, we were at all times vigilant for signs of stress or tension in pupils caused by engagement in the survey. Regular reminders were also given by staff members to pupils, about communicating any concerns or tensions they had during the survey. Systems were in place for pupils to bring these concerns to school staff, for example raising a hand to talk to the teacher. All pupils were treated fairly, sensitively, with dignity and an ethic of respect and freedom from prejudice.⁵

In conclusion, all surveys from parents, pupils and teachers were completed and gathered up by late February for analysis by staff. Data is the raw material of school self-evaluation. An understanding of data in terms of definition, usage, problems, solutions and guidelines is essential in implementing a school self-evaluation process (Chapman and Sammons, 2013). The capacity of staff to learn and develop skills in using data this way was essential to the continued effective implementation of school self-evaluation in our school (Schildkamp et al., 2013; Faddar et al., 2018). In this first step, staff members identified a focus area for improvement and gathered data to investigate it through the creation and administration of a survey instrument to parents, pupils and teachers. These are key skills in the school self-

⁵ With the help of our two special education teachers, both myself and the deputy principal arranged a timetable for senior infants, 2nd class, 4th class and 6th pupils to take the survey. In the school library we set up the necessary computers, each connected to the internet. One group at a time was escorted to the school library. Each child was assigned a computer. They were given help to access the online pupil survey and then asked to fill it in. All pupils were given the same survey. If any issues arose understanding the survey, the deputy principal, the two special education teachers and I were on hand if they needed help. For senior infant pupils each question was read aloud and explained by the special education teachers. A pause was then given after each question to allow the special education teachers assist each senior infant in submitting their answer.

evaluation process described by the DES guidelines (2016). This learning for staff was invaluable to our aim of improving the implementation of school self-evaluation in our school. It also allowed all staff the opportunity to work together in a collaborative and meaningful way. All decisions were made in a spirit of improvement and community endeavour. In other words, not only were we effective in gathering the data as a collective, but as a collective we took the opportunity to learn and develop from the experience itself (Timperley, 2013).

Step 2

In mid-January 2021, school staff discussed the process of collecting handwriting samples for a collective review of pupils' handwritten work (DES, 2016). It was felt by us all that collecting handwriting samples from every pupil would simply take too long. In addition, an analysis of such a broad sample would also be too time consuming. For the purposes of this evaluation, we only required a simple snapshot of current handwriting standards in our school. We therefore agreed on collecting a broad, but small sample for ease of analysis. We adopted this approach knowing that this method of evaluation is recommended and described in the DES school self-evaluation guidelines (2016).

We collected samples from the same classes who completed the pupil online survey i.e., senior infants, 2nd class, 4th class and 6th class. From each of these classes it was decided to ask each teacher to use the Primary Language Curriculum pupil handwriting progression continua (NCCA, 2019) to select three samples, each showing the top, middle and bottom of handwriting standards within their class. Gathering, recording, interpreting, using and reporting information on children's work in this way, is part of regular daily assessment in Irish primary schools and is considered best practice (NCCA, 2007). In some cases, the teacher photocopied from a child's copybook. In other cases, the teacher just asked the selected classes to copy something out into their copy book in their best handwriting. This allowed a collection of twelve samples to be gathered from the entire pupil cohort by early February. In this second step, staff members gained valuable experience in building on existing evidence from a range of sources. In this case, work samples from different pupil age groups. These are key skills in the school self-evaluation process described by the DES guidelines (2016).

Step 3

Gathering data is not in itself self-evaluation, but more so how it is used is critical (Chapman and Sammons, 2013). We therefore organised class teachers to meet in small working groups to analyse all the data gathered from steps one and two. By using (analysing, interpreting etc.)

the data, it was turned into information to evaluate our functioning at whole school level to foster school improvement. Findings from the reconnaissance phase showed staff seeking more opportunities for collaborative professional conversations. This group meeting was an opportunity to address some of that perceived need. Staff members were being given the opportunity to evaluate the teaching and learning of handwriting in our school and agree on any necessary improvement plan. The logistics and proposed content of the meeting was as follows:

• Meeting A

Staff members were split into three working groups as follows:

Group 1: Jr/Sr/1st

Group 2: 2nd/3rd/4th

Group 3: 5th/6th

Research suggests that groups which are assigned tend to perform better than self-selected groups (Felder and Brent, 2001). Research also suggests that groups of three or four members tend to work best (Csernia et al., 2002). This was our first time to try this approach of 'break out' groups in our school. Normally we would just meet as a whole staff. According to Beebe and Masterson (2003), there are both advantages and disadvantages to working in groups. However, by understanding the benefits and potential pitfalls, a school can capitalise on the virtues of group work and minimise the obstacles that hinder success (Burke, 2011).

As this was a collaborative and dialogical process it was important to get the groups talking and debating with each other right from the start. Instructions for the meeting were therefore kept as simple as possible. The group's objective was to simply read, discuss and take notes on (i) all the survey data gathered from parents, pupils and staff on the teaching and learning of handwriting in our school to date as it pertained to their class grouping (ii) the twelve handwriting samples gathered by teachers from their pupils and (iii) the comparisons between our current approach as compared to the description of good and effective practice provided in the new Primary Language Curriculum (DES, 2019). These notes would serve as a record of their thought process and reasoning to share with future full gatherings of staff. Successful group work is easier if group members know what the final product is supposed to be (Burke, 2011; Davis, 1993). The use of working groups as an evaluation tool is recommended by the DES in their school self-evaluation guidelines (2016). It was an opportunity for staff to be part of a working group engaged in professional collaboration to evaluate and improve matters relating to the teaching and learning of handwriting in our school, emerging from a review of pupils' work and surveys administered to parents, pupils and staff.

• Meeting B

Staff members decided to come together as a collective to compare their group notes. Having analysed the data, staff members reported they now had a good general picture of where the teaching and learning of handwriting was in our school. The salient issues identified were a lack of uniformity and consistency in the teaching and learning of handwriting across all classes in terms of letter formation, correct writing implements, pencil grip and copybook presentation. I proposed to staff that maybe now it was time to try and begin the process of mapping that collective knowledge into some form of collaborative consensus about what good effective teaching and learning of handwriting could and should look like in our school. I asked staff to consider using a project management technique called '5W2H' (Tague, 2005). I explained to staff that the process involved all of us collectively, creating a set of key questions to ask ourselves about the teaching and learning of handwriting in our school, based on who, what, when, where, why and how. This approach was in keeping with the action research methodology of the study, based on active participation, open-ended objectives, high levels of commitment and active learning (Bell et al., 2004) with the expectation that any improvement strategies were developed by the staff members themselves as a form of collaborative intervention (Reason and Bradbury, 2001; Bana, 2010; Klenowski and Woods, 2013; Walters, 2014; Jacobs, 2016).

By the end of the meeting, staff created a list of four key questions relating to the teaching and learning of handwriting in our school (Appendix 19). Each member of staff took a copy of these key questions and agreed to attend another working group meeting to answer them.

• Meeting C

Staff members again asked to be split into the same three working groups as before. The purpose of this final group meeting was to reflect on and discuss the key questions asked at our last whole school meeting about the teaching and learning of handwriting. It was decided by staff members that each working group would answer each key question on a large sheet of

paper in the context of their class grouping. As before, these sheets would serve as a record of their progress to share with future full gatherings of staff.

• Meeting D

To conclude the process of formulating a school handwriting policy, we decided to bring the staff back together as a whole school. They were keen to look at the answers given by each working group to the key questions asked regarding what could and should the teaching and learning of handwriting look like in our school. Staff members wanted to look at the answers given on each group's sheet, in an attempt to debate the commonalities and gain consensus. Agreed answers were written onto a large blank answer sheet. This method allowed our collective knowledge to be shared in a quick and easy manner. It also allowed everyone as a collective to debate the differences and to see the commonalities in our attempt to construct and reach whole school consensus on an agreed handwriting policy for our school.

Using the key questions as our guide, we collectively reached consensus on a draft handwriting policy (Appendix 20) detailing what the teaching and learning of handwriting 'would' look like in our school going forward, as described below. For the next two years we planned to put our handwriting policy into action and monitor its impact. More importantly however, as part of this research project we had completed key stages of a school self-evaluation cycle.

6.3.5.2 Monitoring implementation of the action and its effects.

An important step in the Elliot (1991) action research cycle is to both monitor and evaluate the implementation of any actions taken. To monitor the implementation and effects of this plan therefore, I administered survey 3 (appendix 21). Section 'A' compromised twenty-five Likert-scale questions and three yes/no question. Section 'B' comprised six open-ended questions. The anonymous survey was administered to staff in a paper and pen format and received twelve responses out of fourteen.

The quantitative data gathered from section 'A' was illustrated in numerical graph format (bar chart) for analysis (Dixon & Woolner, 2012) (Appendix 22). In relation to the cycle of school self-evaluation, we had just engaged with as part of the first action phase, the data showed that six staff members agreed and two strongly agreed that they had the necessary skills required to carry it out. The data showed nine staff members to be comfortable engaging in self-evaluation to improve outcomes for their pupils, as the evaluation was something they all did on a regular

basis as part of daily practice. This suggested that the root of any problems experienced in 'doing' self-evaluation seemed to lay elsewhere.

The data showed a belief amongst the majority of school staff that they needed more training in how to conduct school self-evaluation. This resonated with earlier data which too showed a feeling amongst the majority of staff members that they needed more training in school self-evaluation. Data from survey 3 however, showed the majority of staff believing they had the necessary skills required to carry out school self-evaluation. I interpreted this to be a desire for training in how the DES wanted school staff to do school self-evaluation. This may also have been due to the fact that the majority of staff members according to the data, found any resources provided by the DES on how to conduct school self-evaluation, hard to understand.

In addition, the data clearly showed that the approach being mandated by the DES was causing an increase in workload and anxiety for all staff members, bar one. The end result according to the data, was a process which was both meaningless and unpopular with staff members.

An initial analysis of the data gathered from section 'B' showed half of staff members being unable or choosing not to name any DES school self-evaluation policy documents or resources (*Table 6.4*). This was despite the preparatory work we had undertaken before our first action step.

Any useful policy documents that you know of regarding SSE?				
S3 Respondent 1	No			
S3 Respondent 2	(No answer)			
S3 Respondent 3	None			
RS3espondent 4	None			
S3 Respondent 5	(No answer)			
S3 Respondent 6	SSE guidelines			
S3 Respondent 7	SSE guidelines			
S3 Respondent 8	SSE guidelines (Purple book)			
S3 Respondent 9	(No answer)			
S3 Respondent 10	SSE guidelines			
S3 Respondent 11	SSE guidelines			

Table 6.4Useful documents regarding SSE according to staff members

S3 Respondent 12	LAOS – looking at our schools
	School self-evaluation guidelines

This lack of knowledge may have been a factor in participants having divided opinions on the purpose of school self-evaluation, namely whether it was a process of improvement or accountability, as can be seen from some of the responses in *Table 6.5*

Table 6.5The purpose of SSE according to staff members – sample survey results

What do you think is the purpose of SSE?	
S3 Respondent 1	Give a sense of agency to schools to work on what actually needs <u>improvement</u>
S3 Respondent 4	I think it's a <u>tick the box exercise</u> and I am not really sure to be honest

However, as can be seen in *Table 6.6*, the majority of staff saw the potential for school selfevaluation to improve the school.

Table 6.6The positive aspects of SSE according to staff members – sample survey answers

What are the positive aspects of SSE as a government policy?		
S3 Respondent 5	As a government policy the positive aspects are; it allows for a <u>whole school review</u> of our own personal teachings and going forward <u>delivering the best possible</u> <u>outcomes</u> for our learners	
S3 Respondent 6	It is a positive way of improving a school , the teaching, learning and overall management	
S3 Respondent 12	It allows schools the <u>autonomy</u> to choose areas of most concern to them and their organisation. It acknowledges the inbuilt desire of the vast majority of teachers <u>to</u> <u>improve their schools teaching and</u> <u>learning.</u>	

However, the data clearly showed concerns amongst staff members regarding the practical implementation of school self-evaluation. As can be seen from the sample responses below in

Table 6.7, resourcing issues, regarding time, information and support were still causing stress amongst some staff.

 Table 6.7
 The negative aspects of SSE according to staff members – sample survey answers

What are the negative aspects of SSE as a government policy?		
S3 Respondent 2	Very time consuming: lots of abstract, repetitive 'splitting of hairs'; <u>quite</u> <u>meaningless</u> exercise at times	
S3 Respondent 3	Additional paperwork on already stretched staff	
S3 Respondent 6	Added workload and time required to do it	
S3 Respondent 7	Time consuming– takes a lot of meetingtimes.Increased paperworkequalsincreased stresswhich affects morale.	

In fact, these issues were causing a variety of emotions amongst some staff to the process in general, as can be seen from some of the responses in *Table 6.8*

Table 6.8Emotions felt by staff members when they think of SSE

What emotions come to mind when you think of SSE?	
S3 Respondent 1	<u>Calm</u> , content, mostly positive, somewhat <u>inattentive and detached</u> . Irritated (with stop-start, training followed by strike) <u>unenthusiastic but willing to participate</u> , have input, follow instructions
S3 Respondent 9	Frustration , dissatisfaction, <u>disappointment</u> , annoyance

In conclusion, the data showed staff members having varied opinions on the school selfevaluation process in general, as is evident in *Table 6.9*

Table 6.9Staff members opinions, thoughts, ideas, observations and comments in relationto SSE – sample survey answers

Please describe any other opinions, thoughts, ideas, observations or comments you have in relation to your experience of SSE to date in your educational career		
S3 Respondent 1	Happy to participate in SSE but also happy to be told what to do at times if it saves on time/discussion/meetings	
S3 Respondent 2	I find it quite tiresome in general	
S3 Respondent 3	SSE does not seem to be a 'whole school process'	
S3 Respondent 6	I think it's great for management to involve everyone in the process	
S3 Respondent 9	<u>SSE is added pressure</u> from the Department.	
S3 Respondent 11	I feel as <u>though it is seen as an approach of</u> <u>something that just 'has to be done' as a</u> <u>school approach overall, rather than knowing</u> <u>the specific benefits overall for teaching and</u> <u>learning at a smaller scale.</u>	

I found the data gathered from section 'B' to be extremely rich. However, I was quite surprised to see that despite all the preparatory work and the actions we had taken as a collective to date, half of school staff were still unable to name any DES school self-evaluation policy documents or resources. Staff were also unclear as to the purpose of school self-evaluation despite having spent time together examining these policy documents and actively engaging in an action cycle. In essence, even though staff members had worked together in the first action phase to evaluate the school's handwriting policy, the reconnaissance data was showing that significant concerns and confusion still remained. One staff member claimed that they did not know what benefit school self-evaluation actually had for teaching and learning but as it was government policy it just had to be done regardless. In addition, in our action plan we agreed to give ourselves the necessary time and space outside of the classroom to meet and collaborate regarding the school self-evaluation process, yet staff members still claimed to be stressed in terms of workload and lack of time, claiming it to be quite a tiresome process and indeed, added pressure. In fact, one respondent wanted to be told what to do if it saved time on such discussions and meetings. So, what exactly was going on here? To be honest I didn't know. After all the work we had done

collectively in creating and implementing our action plan I did not expect these negative outcomes.

6.4 Conclusion

Without data to compare what our actions have accomplished to the performance of the prior state we have little or no foundation for decision making or improvement (Hart, 2007). As part of a second phase of reconnaissance therefore, I decided to collect data using both a quantitative and qualitative survey approach triangulated with semi-structured focus group interviews. Using more than one method such as this in a study, gives a more comprehensive and trustworthy picture than one which does not (De Vos et al., 2007; Cohen et al., 2011; Babbie, 2012). We rely on the perceptions of participants, who can provide insightful information on the topic under review (MacBeath, 1999).

An analysis of this reconnaissance data showed staff members acknowledged the benefits of school self-evaluation in improving their school and were favourable to the increased levels of collaboration, autonomy, decision-making, reflection and discussion it brought. However, they said the increase in their workload due to school self-evaluation and the lack of resources such as time, information and support to carry it out, was causing them continued stress, thereby negatively impacting their attitude towards the process. The findings also showed that amongst staff members, there was no collective understanding of school self-evaluation, its purpose or how it differed to other evaluation activities undertaken in our school. For example, staff members believed that engaging with the school self-evaluation process was equivalent to doing the same job as the inspectorate. They saw it as an added thing that they were being forced to do in their already busy workload. This lack of clarity was negatively impacting their attitude towards the process and causing staff members to comply with school self-evaluation in a contrived way as a form of resistance.

Chapter 7 – Second Phase of Reconnaissance

7.1 Introduction

The second cycle of the Action Research began with a reconnaissance phase to help us as a school explain any shortcomings in the implementation of our action plan. This reconnaissance also helped us to guide and shape the direction of our future and subsequent action phases. Babbie (2012) believes the activities of analysis, interpretation and presentation are the culminating activities of qualitative inquiry. This involves the categorising, ordering, manipulating and summarising of the data to obtain answers (DeVos et al., 2007). We needed to collect data to find out what our actions had accomplished compared to the performance of the prior state. Without this data we had no foundation for making decisions regarding any further necessary action. As part of a second phase of reconnaissance, four staff members agreed to be interviewed and audio recorded. One was a special education teacher; the other three were a junior infant teacher, a 4th class teacher and a 2nd class teacher. Some researchers advocate undertaking two or three rounds of interviews (Seidman, 2006). Therefore, the interviews were held over two days, with the participants interviewed in pairs. The interviews followed the protocol described in section 4.7.1. The information collected from the survey and interviews yielded rich amounts of both quantitative and qualitative data. In order to ensure credibility, a rigorous and systematic approach was used to reduce and interpret the data. These processes are described and explained in greater detail in the following chapter. To begin I detail the methodological approach and design decisions taken.

7.2 Reconnaissance

Reconnaissance data from the first phase of action research showed the lack of interest and enthusiasm for school self-evaluation amongst the majority of staff members to be both a weakness and a major threat to its successful implementation. To address this and other concerns relating to time, information and support, an action plan was implemented by school staff. However, despite all the preparatory work and the actions we had taken as a collective to date, half of school staff were still unable to name any DES school self-evaluation policy documents or resources. Staff were also unclear as to the purpose of school self-evaluation despite having spent time together examining these policy documents and actively engaging in an action cycle. In essence, even though staff members had worked together in the first action phase to evaluate the school's handwriting policy, the reconnaissance data was showing that significant concerns and confusion still remained.
I arranged a meeting with my critical friend to discuss these issues. We both agreed that something more was going on here. We had created an action plan as a collective to improve the implementation of school self-evaluation in our school, but an initial analysis of the data showed the plan had not worked for all staff members. We both felt that more reconnaissance and fact finding was necessary to explain this effect. To further interpret the findings yielded in the survey my critical friend and I discussed gathering reconnaissance data using a different method. I therefore consulted the literature on research methodology (Briggs et al., 2012). At a later meeting I proposed to my critical friend that an interview of staff members may be suited to understanding the lived experience of staff as they implement school self-evaluation in our school, but more importantly the meaning they make of that experience (Coleman, 2012). My critical friend agreed. We therefore formulated and constructed a set of eleven interview questions (Appendix 23) to probe more deeply into the survey data gathered and co-construct the experiences of these staff members and their relationship with school self-evaluation to date.

Over two days, four staff members were interviewed in two groups of two. The interviews followed the protocol described in section 4.7.1. All went according to schedule. After the interviews were completed, I transcribed them into written form and gave the transcripts to the interviewees for their approval. Once approval was given, I subjected both the transcripts and the gathered survey data to simultaneous thematic analysis (Floyd, 2012; Braun and Clarke, 2019). For a full worked example of the thematic analysis applied to the survey and interviews please refer to Appendix 24.

These interviews proved fruitful, as findings later showed the emergence of new data pointing to a feeling amongst staff members that by being mandated to implement school selfevaluation, they were being forced to do the work of the inspectorate. So, whilst the data later showed staff members to have an appetite for school self-evaluation in principle, they seemed to have no desire for, and were resistant to, the version outlined by the DES.

7.3 Findings and discussion

In an attempt to dig deeper and clarify and expand on data gathered in the reconnaissance phase of this second action research cycle, we carried out two semi-structured focus group interviews with four consenting staff members. From the data collected, I generated two key themes to frame a thematic analysis. The first, entitled 'Attitudes to school self-evaluation', is about the people tasked with implementing school self-evaluation to achieve the end goal of improved outcomes in teaching and learning. The second theme 'Leadership and staff motivation', is about the important role school leadership plays in motivating and sustaining staff to engage meaningfully with the school self-evaluation process. What follows are the findings from our two focus group interviews and a further re-examination of our staff survey.⁶

7.3.1 Attitudes to school self-evaluation

This first theme to emerge from the data analysis was about the people tasked with implementing the school self-evaluation process, to achieve the end goal of improved outcomes in teaching and learning. It describes the daily experiences of school members working in our school as they implement school self-evaluation and all that entails physically, mentally and emotionally. Based on these experiences, it details the deep held views of staff members towards school self-evaluation and how this impacts their engagement with the process.

Staff members acknowledged the potential of school self-evaluation to improve our school. Similar to the findings from action research cycle one, school improvement was desirable amongst school staff. It was an end goal they believed in working towards in order to achieve an outcome of improvement. The data showed the majority of school staff (83%) were committed to the principle of school improvement and saw it as a worthwhile endeavour.

This view was also echoed by respondents in the focus group interviews as follows:

"To improve the learning process for our children...to improve our practice and...to help...children...achieve more or achieve their potential" (interviewee C)

"I think the intention is very good... I think it has to be a positive thing you know because we're always trying to improve, it's good to keep looking at ourselves and looking at our practice" (interviewee D)

Reconnaissance data showed staff members acknowledging positive outcomes due to their engagement with this first phase of action research. They expressed an appreciation for the

⁶ The findings were sent to all staff members in early September 2020 via email for comment and further analysis, with a due date of late September 2020. Due to a global pandemic and national emergency health measures, it was no longer recommended to gather as a group for meetings We were therefore unable to meet in person. All correspondence was therefore via email. I have included their contributions in sections 7.3.1 and 7.3.2.

increased autonomy, participation, reflection, discussion and collaboration it allowed. The time spent by school staff interacting with each other, discussing DES policy documents and actively doing evaluation, seemed to help staff members gain confidence in integrating the process into their daily practice.

"It allows for a whole school review of our own personal teachings and going forward delivering the best possible outcomes for our leaners" (S3 respondent 5)

The literature underlines the important contribution that such strong collegial relationships can make to positive school improvement and change (Rozenholtz, 1991; Day and Sammons, 2014). It was interesting to note that some respondents saw this process of identifying weaknesses and putting plans into place as part of their normal everyday activities within the classroom anyway. Staff members saw the school self-evaluation process as both supportive and complimentary to their professional daily practice.

"Because in our job, that's what we do every day. You do that day to day in your own classroom and in your setting that we're watching, and seeing what children need to work on" (interviewee B)

"it's happening anyway, on a classroom level...on a band level, within classes...teachers talk about things...we know the areas of strengths and weaknesses...and look at our own test results, or their own assessment, and see where a class needs to go" (interviewee A)

"We don't reflect on it. We don't stop and say...this is evaluation...That's what teaching is though isn't it? You're improving all the time" (interviewee D)

School staff saw the positive contribution that their collective engagement with school selfevaluation made at both classroom and whole school level.

"Strength can become a whole school thing...for example, with the handwriting; if people start to implement changes within their rooms, then that's shown throughout the school, whether that's in displays or handwriting competitions...becomes a whole school result" (interviewee A)

"So far, it's been classroom based and I'd say it has led to improvements...There's a very supportive environment here in the school" (interviewee D)

"Any good school with a positive atmosphere, this happens, you see we've a little weakness, it's mentioned at a staff meeting and we do something to change it" (interviewee C) The willingness of staff to positively engage with school self-evaluation increased when they were given the autonomy to choose the focus of improvement.

"Only thing that would encourage enthusiasm is that if it's something you would personally do next year, like the handwriting, I do feel positive about that next year, I feel good that we'll all be doing the same thing...I wouldn't feel...negative about this because it is...actually going to be used next year" (interviewee C)

"I think if it's interesting, I have the enthusiasm, it depends on what the topic is, what the area is. There are certain subject areas I'd be much more interested in than others" (interviewee B)

"I would need a lot of encouragement and a lot of support... and enthusiasm and interest in what the area is" (interviewee A)

In addition, they recognised the benefits school self-evaluation brought to our whole school and themselves in terms of internal accountability.

"I think there's an accountability as well, when it's a whole school thing" (interviewee B)

"Nobody wants to be seen to be letting the side down, and you want to pull your weight, if it's on a whole school level. So, it does impact your classroom teaching" (interviewee A)

"We're all professionals...everyone here is here with the best of intentions...works hard and wants the best...for our students..., pride in our school, we want the school to be the best that it can be...staff are willing and are enthusiastic and care enough about the school as a whole" (interviewee A)

The literature shows that an internally motivated evaluation culture can occur (Mann and Smith, 2013) if staff are perceived as team members in a school with shared objectives (Potter et al. 2002) rather than individual classroom teachers (Schildkamp et al., 2012). Every opportunity must be used to show the positive benefits that collective actions taken at whole school level can have for classroom practice and vice versa.

To increase the motivation, interest and positivity of staff towards the school self-evaluation process, the first action plan gave increased opportunities for staff members to collaborate together and to be involved in all decision-making. The data showed a general consensus among school staff, that collaborating as a collective and playing a fully active part in the school self-evaluation and decision-making process, had a positive effect on their experience

of 'doing' school self-evaluation in our school. Increased opportunities for collective discussion and decision-making seemed to have a positive impact on attitudes towards school self-evaluation amongst the majority of staff. Ninety percent of staff said they felt involved in the school self-evaluation process and that this was a contributing factor to any success.

"Great for management to involve everyone in the process" (S3 respondent 6)

"I think it's important that everyone has an active part and everyone probably feels that they have a voice...I think with the school self-evaluation that we've been doing, everybody has been involved, everybody has been asked an opinion, whether they want to or not...I think from that regard, it's been operating very successfully where everyone has been involved in the process" (interviewee B)

To ensure success, it was therefore imperative that the implementation of school self-evaluation in our school, was both a collegial and collaborative process with staff involvement in all decision-making.

According to some respondents, the promotion and growth of reflection and discussion amongst staff members was seen as a positive and motivating factor to engage with the school self-evaluation process.

"Working in smaller teams, and then feeding back – probably like we did with the handwriting. I did feel that was helpful...kind of funnelling it through, so it's not just one big staff meeting and everyone's sitting there...the voices who are confident in speaking probably, and probably the people who have been doing it longer feel a little bit more confident in doing it...I think for younger teachers, it's quite difficult in that setting to put themselves forward, even though they might have a lot of newer ideas, maybe, if they can bring fresh things to the table" (interviewee B)

According to O'Day (2002), information, knowledge and feedback are passed on when people interact with each other. This increases their exposure to raised levels of variety, innovation and creativity. The more frequent and powerful these interactions are, the stronger a school will be in its ability to learn, adapt and evolve.

"I like the group thing ... that we can all you know share ideas" (interviewee D)

To ensure the success of school self-evaluation going forward therefore, it was important to ensure that staff members had plenty of meaningful opportunities to interact, discuss and reflect on teaching and learning matters collectively, with an input into all decision-making relating to improvement and change at whole school level (Leitch and Day, 2000).

In terms of practical supports, time out from the classroom to engage with the school elfevaluation process was seen as very positive

"One thing that's very practical is the time out that we had from the classroom to be able to talk in focus groups about...the handwriting...it's very hard as you now if you're trying to do that at lunchtime or after school. Like we were given time and plenty of time to, so that helped to engage and we could all agree in our group" (interviewee D)

Ryan et al. (2007) suggests that those promoting school self-evaluation should pay attention to the daily reality of schools in relation to time and resource constraints and the increased workload. Vanhoof and Petegem (2011) tell us that policy implementation processes are more effective when team members are offered both professional and personal support. To ensure the success of the project going forward therefore, there was a need for clear and practical expectations regarding the implementation of school self-evaluation. It was also necessary to give adequate time for school members to talk and reflect collectively when implementing the process in our school. In addition, resourcing the project in terms of professional and personal support such as training, access to information, substitute cover and operational manuals, was imperative.

However, as Keshavarz et al. (2010) show in their study of Australian primary schools as complex systems, planned interventions do not always lead to predictable and intended outcomes. When time was carved out for staff to engage in the process in a more meaningful way as a collective, this caused problems for some participants. There was a feeling of it being valuable time wasted

"An awful lot of investment for very little, a very specific little area" (interviewee D)

"I know there's an effort to have every voice heard...there was so much talk and discussion and meetings to get a few, quite basic points...so I guess that's the part of it that I have an instinct to resist...it's just the time and all the democracy...On the flip side, if we weren't given all the time, we'd all be moaning wouldn't we" (interviewee C)

"There were times where I didn't think we all needed to be part of it, that you...might have got that bit done...sometimes, too many cooks can slow a whole process down" (interviewee D)

"Happy to be told what to do if it saves on time, discussion, meetings" (S3 respondent 1)

Time is a valuable and finite resource in schools (NCCA, 2010). According to the literature and the findings of this study, school staff are of the belief that there is insufficient time to do all that is required of them such as implementing curriculum, meeting the needs of learners, communicating with parents and working collaboratively together to deliver positive change as a staff (Hall and Noyes, 2009; Cambridge Primary Review, 2009; NCCA, 2010; O'Brien et al., 2015; McNamara et al., 2021a). For that reason, staff members prioritise and make productive choices (O'Day, 2002). This may explain the negativity of staff members to the discussions and meetings described above. It is important to give attention to these beliefs as they can influence how school staff receive, interpret and enact policies (McGrath et al., 2017; Lambert and O'Connor, 2018). According to Skerritt et al. (2021), it is simply not possible for school staff as a collective unit to respond to school self-evaluation in coherent and consistent ways; Instead, the complex and intricate policy work of various actors combine to make school self-evaluation happen.

However, despite the generally positive shifts in outlook as described above, the majority of staff members, seemed to have a negative attitude towards the school self-evaluation process. Our survey showed a strongly held belief amongst participants that school self-evaluation was still not popular with ninety percent of their colleagues. The survey gave interesting insights into staff members' relationship with the school self-evaluation process. Eighty percent described their own attitude to school self-evaluation as indifferent, with only twenty percent claiming it to be positive. Despite the action plan, staff still felt stressed, under resourced and overworked when engaging in the school self-evaluation process. One respondent commented "I can understand, that it is perhaps, adding more work, to already very busy teachers" (S3 respondent 5). The interplay of other components within a school such as a perceived lack of time, training opportunities and resources can have the potential to inform the emerging collective behaviour among staff members within that school system. (Keshavarz et al., 2010). Feelings of stress and workload pressure can become the dominant view, with knock-on effects for motivation and enthusiasm. An emerging pattern such as this has the potential to inform the entire behaviour of school staff and affect the potential success of school self-evaluation in a school (O'Day, 2002; Keshavarz et al. 2010). I discussed the survey and interview data with my critical friend. She described the findings best as a love-hate relationship. Those who felt positive towards the process were at the same time not enthusiastic about it.

"Positive without being enthusiastic ... I don't enjoy the process ... I don't enjoy that kind of work in terms of policies ... and paperwork ... it's not something I would enjoy ... I have to do with

work...I'd have no feelings of excitement...my lack of enthusiasm about the whole thing...I'm very happy to do whatever" (interviewee D)

Despite the general negativity it seemed our staff did have an appetite for school self-evaluation and acknowledged its benefits for improving our school. Even though it wasn't popular with staff members, they could clearly see its merits when approached in the collaborative manner undertaken in the first action phase. This was a noticeable change in attitude following the implementation of the first set of action strategies. However, they seemed to have no desire for the version of school self-evaluation outlined by the DES. At the onset, this research was very much motivated by my own values of authenticity and collaboration being denied in practice as I led school self-evaluation. The desire to take ownership of the problem and implement change came from me initially and not my school staff. Since school self-evaluation was not an issue raising any concern for staff, it came as no surprise that they were unenthusiastic to engage with it and own it. However, research shows the important role school staff play in implementing change in schools (Garner, 2010; Thornburg and Mungai, 2011). This action research aimed to address that issue, affording them the time and space necessary to own the process and make it their own. According to the literature however, implementing change in schools is a complex process (Austin and Harkins, 2008; Guhn, 2009; Given et al., 2010; Craig, 2012; Hannay and Earl, 2012; Williams et al., 2012), even when staff members are provided with the time, and supports they need to do it well (Pescarmona, 2010). Balancing the needs of staff members while driving change is tricky and very difficult, as staff may truly believe they have no reason to change (Kise and Russell, 2008).

In our attempt to improve the implementation of school self-evaluation in our school for example, the request for extra supports from staff to do it, such as time and resources, were delivered. However, despite all the actions taken, the negative attitude of staff members to the version of school self-evaluation put forward by the DES, had not changed, signalling a strong resistance from staff members to implementing the process. Like Barger and Kirby (1995), Clancy (1997) found that resistance to change increases when needs are not met and that leaders in general, fail to recognise and deal effectively with these needs. The solution, according to Kise (2004), lies in providing staff members with the motivation to change their practice, through structuring events and situations to produce changes in their beliefs (cited in Kise and Russell, 2008). This solution was to prove very fruitful in this research, as will be explained later in the thesis.

One respondent believed that the aims of school self-evaluation for a school were "very long term" (S3 respondent 1). This was problematic they contended, where "staff turnover is high" (S3 respondent 1). In fact, of the fourteen staff members involved in this action research project, at the time of writing, five have since moved on.

"I'm always a little nervous about plans that operate over a few years...you've a changeover of teachers, you could have a new principal, maternity leaves, you've everything and those things get lost and so I love when something can be implemented between September and June and done and it's there...Anything that's a little more long term I am starting to lose enthusiasm for because I've started to realise they just fade away...if you say for a year, 1, 2, 3 and 4, we're going to implement this...it starts of great but inevitably people come and go and puff...we forget...but the intention is good" (interviewee C)

The findings overall, suggested that staff members did not care too much about the school selfevaluation process as envisaged by the DES. It was seen as irrelevant and a "quite meaningless exercise at times" (S3 respondent 2) containing "lots of abstract repetitive splitting of hairs" (S3 respondent 2) but a requirement "Acceptance...acquiescence, resignation" (interviewee D)

"I do very little taking out school self-evaluation documents and reading them" (interviewee B)

On reading these findings one staff member commented, "I was surprised that there were some negative views towards school self-evaluation" (S3 respondent 13).

Such diverse views are typical of a complex adaptive system, demonstrating in very simple terms that a 'one size fits all' rule or solution may not meet the needs of an entire school community (Ngan et al., 2010; Chapman and Sammons, 2013). However, complexity theory suggests that this diversity of views amongst people within a school, are essential if it is to learn, adapt and evolve (Morrison, 2002; Keshavarz et al., 2010; Chapman and Sammons, 2013; Fidan and Balci, 2017). In a study of educational reform in the United States of America, O'Day (2002) states that *"without variation among the agents of a system or in the surrounding environment, there is little information on which to act"* (pg.6). However, Keshavarz et al. (2010) warn that it would be a failure not to investigate and address the misgivings of people within a system, as their concerns can emerge as the dominant pattern, informing the entire behaviour of people within a school (Keshavarz et al., 2010; O'Day, 2002). The fact that some school staff found school self-evaluation both meaningless and irrelevant was undoubtingly going to have major consequences for the successful implementation of school self-evaluation

in our school. MacBeath (1999) argues that school self-evaluation can only work if team members are positively disposed towards it. For the successful implementation of school self-evaluation in our school therefore, it was imperative to address this issue. We needed to take action to further build on the positive gains amassed in the first phase of this action research by continuing to support staff in this work, highlighting the very positive and real benefits to their daily practice. This required a different approach.

Seventy percent of respondents felt they needed more training on how to conduct school selfevaluation from the DES, with only ten percent saying the process of school self-evaluation was easy to understand

"I find it hard to engage with" (interviewee D).

"Attitudes are one of confusion to it...a lot of confusion...wasn't much clear guidance in the first place" (interviewee B)

"I think what they're supposed to be doing mostly, across the board is the big part of the problem...The main thing schools are missing is just guidance...if they were to give very clear guidelines on what they want to see" (interviewee A)

O'Brien et al. (2019) reports that the levels of support provided to Irish school in 2012 to formally implement school self-evaluation was relatively generous in comparison with other European countries at that time. In 2015 the European Commission (2015) indicated that Ireland was one of eleven countries that had five or more different types of supporting measures in place to help schools implement school self-evaluation. One support included an advisory visit to schools by the inspectorate. In the context of our school, this was not looked on favourably by school staff.

"A long time ago, I just remember the cold in the library, that's what I remember... 'pick a very small area' ... that's all I remember...I wouldn't be keen to get more training on it really" (interviewee D)

Upon reflection therefore, school staff had minimal contact with the policymakers responsible for the school self-evaluation policy that they were tasked to implement (with the exception of the principal and one other member of staff). The only training given to staff was through DES guidelines (DES, 2016), in-service feedback from two members of staff, an inspectorate-built website and an advisory visit with a school inspector (O'Brien et al., 2019) which the data showed was not meaningful for some.

In essence, the view of our staff was the DES was delivering as the knowledgeable expert, with staff members playing a passive role. The movement of information in this way was limited. The top-down flow of policy information from the DES, treated staff members as mere recipients of information, instead of valuable sources of information in their own right. This phenomenon is discussed in a study by O'Day (2002). She argues that the nature and quality of information must be adequate and fit for purpose if an organisation is to learn from it. In her study of American educational reform, O'Day (2002) found that a top-down transmissive model of information flow, can result in schools reacting to directions imposed from above and outside the school in a form of compliance, more so than a meaningful reflection on internal practice. In such an act of compliance, she concludes that the information received by the school may not lead to learning and change. This resonates with international and Irish research indicating that the existence of a driving force for collaborative internal school improvement is questionable, other than in the area of simple compliance with rules (McNamara and O'Hara, 2012). Hall and Noyes (2009) argue that being engaged in such an activity that is more about compliance than educational endeavour, makes teachers feel oppressed and professionally compromised. Other teachers see it as a threat to their autonomy in the classroom (Vanhoof et al., 2009). My findings from the reconnaissance data support this view.

My findings showed there to be a real disconnect between school staff on the ground and DES policy.

"As an SNA it seems irrelevant to me" (S3 respondent 3)

"My own attitude is, I suppose one of indifference...on board with doing whatever I have to do, but I'm not over interested in it" (interviewee B)

"a kind of apathy towards it" (interviewee C)

"Not realistic" (S3 respondent 5)

Be that as it may, it is assumed by the DES that school staff will drive and steer the school on a predetermined route to the destination of school improvement, using available resources. Indeed, they are mandated to do so. However as described earlier, whilst there was an acknowledgement amongst staff members of the potential benefits of school self-evaluation, seventy percent of staff members felt the DES approach to school self-evaluation was meaningless, irrelevant and out of touch with reality on the ground. So, despite staff members having just created a suite of resources and engaged in a school self-evaluation cycle, it seemed staff were merely complying with their mandated duties in a contrived way, just to satisfy an external governing body.

"It's not an area of personal interest to me but I still think it's a fairly good idea for schools to do...it's...what schools do...all the time seeing how we can improve things, but...I wouldn't get excited about it" (interviewee C)

In a complex adaptive system such as a school, government mandates are interpreted by the individual staff members within them (O'Day, 2002; Skerritt et al., 2021). This produces unpredictable thoughts and actions by school staff as they engage with and implement government policy. (Morrison, 2002; Keshavarz et al., 2010; Skerritt et al., 2021). An analysis of the thoughts and actions of staff members in the reconnaissance data showed the majority of school staff struggled to adapt to the external government mandate of school self-evaluation. On a whole school level, the collective attitude of staff members towards school self-evaluation seemed generally negative. However, the data showed this not to be the case when participants were driving their own self-evaluation processes on a daily basis within their own classrooms. In this context, teachers seemed comfortable engaging in the self-evaluation process in their own setting, i.e., observing and questioning, creating their own resources, interpreting documents and theorising their own practice (Simons, 2013). The data showed them to be comfortable using it each day to drive towards improved outcomes for all the pupils within their classroom. If staff members were 'doing' self-evaluation each day in their own classroom, then why wasn't this being replicated at whole school level? I concluded that the root of the problems experienced in 'doing' self-evaluation at whole school level seemed to lay elsewhere.

Specifically, an analysis of the data suggested that school staff felt they were doing the work of the inspectorate when 'doing' school self-evaluation. They also felt accountable to the inspectorate for this work. In simple terms, staff felt mandated to use school self-evaluation as a method for school improvement, but at the same time their engagement and results with it were open to public scrutiny for accountability purposes. In reality, school staff are externally accountable for their efforts to improve their school (DES, 2016). This tension between improvement and accountability seemed to be problematic. Staff members felt the school self-evaluation process was something imposed on them from outside as an accountability exercise for the inspectorate rather than an improvement process that they do for themselves. In essence, a "box ticking exercise" (S3 respondent 1).

"It is seen as an approach of something that just 'has to be done' as a school approach, rather than knowing the specific benefits overall for teaching and learning" (interviewee C)

School staff also felt they were being 'forced' to do the job of the inspectorate, viewing school self-evaluation as being that of inspection by stealth.

"It forces them to get on board with it and I guess then you need proof in the form of paper work" (interviewee C)

"Rather than, the department coming in and doing their WSE and deciding, this area, you need to improve, and this is what we want you to do. They were just saying, okay schools, you need to do this yourselves" (interviewee A)

One respondent even said the purpose of school self-evaluation was to "*cut down on unnecessary investigations*" and to "*take pressure of the department*" (S3 respondent 1)

This view amongst staff of being forced to be accountable and give assistance to an external agency such as the inspectorate, did not arise in the first phase of the research. This was a noticeable change in response following the implementation of the first set of action strategies. The fore-fronting of this tense relationship between inspection and school self-evaluation is highlighted by O'Day (2002). She believes that the pattern of interaction between inspection and school self-evaluation is contradictory. School improvement, she argues, is premised on a flat command structure aimed at fostering growth. Members within the school are seen as a source of information, with valuable insights into their own unique setting and context (O'Day, 2002). These members are seen to be proactive as a collective in fostering improvement within their school through meaningful reflection on internal practices (DES, 2016).

In contrast however, O'Day believes (2002) school accountability is premised on a hierarchal structure, aimed at fostering compliance. Members within the school are seen as recipients of externally mandated '*one size fits all*' rules and programs, delivered in a '*top-down*' flow of information (pg.16-17). School staff interpret and adapt these directions to their own organisation in unpredictable ways (Keshavarz et al., 2010).

One participant commented, "perhaps...when the process becomes more familiar it will be a more 'liked journey'...perhaps the department's guidelines could be presented to staff differently" (interviewee A)

I agreed with this view. For the success of this project, school self-evaluation needed to be presented differently to staff. Action had to be taken to show school staff that externally mandated school self-evaluation was basically the same in operation as internal classroom self-evaluation but on a macro level. We needed to show the commonalities between the two. We needed to demonstrate that the practical tasks of staff members 'doing' self-evaluation in their classroom were basically the same as the externally mandated school self-evaluation they felt forced to use. They were more similar in operation than not.

More importantly, action was required to show staff that school self-evaluation was not a form of inspection. Although staff members were introduced to school self-evaluation in the first phase through various collective actions, the reconnaissance data showed there to be no collective understanding of what school self-evaluation was. At individual level, staff members had diverse interpretations as to the purpose of school self-evaluation. These ranged from being one of improvement, identifying weakness, whole school planning, autonomy, professional development, recognising strengths and accountability. This is typical of a complex system, where diverse members within an organisation, can have diverse interpretations of an external mandate (Keshavarz et al., 2010). To ensure the successful implementation of school selfevaluation in our school we needed to give staff a cohesive understanding of where school selfevaluation fitted alongside all the other evaluation activities undertaken in our school. Although they were doing self-evaluation all the time in their own classrooms, they saw it as an added thing to do in their already busy workload. One participant commented "although we have always been self-evaluating our teaching, subconsciously we now feel it's an extra 'job' or task to complete on top of our everyday responsibilities" (interviewee B). It seemed they could not see the overlap between the two.

7.3.2 Leadership and staff motivation

This second theme that emerged in the reconnaissance data was the important role school leadership plays in motivating and sustaining staff to engage meaningfully with the school self-evaluation process.

Self-reflection in the first cycle of action research showed me to have implemented school selfevaluation in a structure of hierarchal compliance. I led school self-evaluation with a quickfix, go-it-alone approach to shield my staff from any burden, limiting any possibility of collaborative benefit or meaningful organisational change. The voice of school staff was simply not sought. In this action plan however, staff members wanted to play a fully active part in the school self-evaluation process, in partnership with school leadership. Over the course of this action project my leadership of school self-evaluation was changing towards one of community endeavour, where all staff members had agency in the process. Staff members were invited to engage in a collaborative cycle of evaluation, with involvement in all decision making through a suite of supports. According to the reconnaissance data, this had a positive effect.

"I think with the school self-evaluation that we've been doing, everybody has been involved, everybody has been asked an opinion, whether they want to or not...I think from that regard, it's been operating very successfully where everyone has been involved in the process" (interviewee B)

School staff believed that responsibility for the success of any school self-evaluation process lay in the approach taken by me as the senior school leader. They contended that school leaders must use their skillset to facilitate opportunities for the motivation of all staff members to actively participate and collaborate together in school self-evaluation.

"Whether it's principal and vice-principal, the main push comes from them, but that everybody needs to be on board...everything that happens in a school, the staff have to be a team...once staff are involved and know what's happening...this is where we're going with it, this is the direction we're going to take...the push comes from management, from leadership... I need that driver, that motivation" (interviewee A)

The literature suggests that principals have a key role to play in setting direction and creating a positive school culture, including a proactive mindset, and supporting and enhancing the staff motivation and commitment needed to foster improvement (Day and Sammons, 2014). School leaders play a key role in shaping teachers' attitudes, commitment and effectiveness in any school self-evaluation process (Leithwood et al., 2008). Going forward therefore, it was imperative for the success of school self-evaluation as a method for school improvement in our school, that I, as school principal, continue to reflect on all that I said, did and thought as I led the process.

There was also general consensus among staff members that school leaders should be responsible in resourcing and coordinating the process *"you're down to who keeps everyone together, and rounds it up, and mediates...I suppose the leadership has to come from the top"* (interviewee A). Fidan and Ozturk (2015) believe that a flat and flexible hierarchical structure such as this, where hierarchical structures are maintained to carry out functions like coordination on the one hand and horizontal structures are allowed in order to conduct complex functions such as communication, innovation and creativity on the other, could be more

successful in creating more space for innovation. This comes from the school leader facilitating participants to self-organise in order for them to find the best way to implement school self-evaluation successfully in their own school (Morrison, 2002).

For the success of this project, my leadership of school self-evaluation needed to continue to resource and coordinate staff members working in a community of endeavour, where they had full agency and input into any decision-making related to the process of school improvement and change (Day and Sammons, 2014). Such interaction had the potential to increase the motivation, interest and positivity of staff members towards the school self-evaluation process and in turn, improve its implementation in our school.

7.4 Further analysis through the lens of complexity theory

This section will complement the analysis already presented and will comment on the findings through the following four key concepts of complexity theory.

7.4.1 The diversity and dynamic nature of agents

In a complex adaptive system such as a school, government mandates are interpreted by each individual staff member (O'Day, 2002; Skerritt et al. 2021). This produces unpredictable thoughts and actions by school staff as they engage with and implement government policy (Morrison, 2002; Keshavarz et al., 2010; Skerritt et al., 2021). According to Skerrit and colleagues (2021), it is simply not possible for school staff as a collective unit to respond to school self-evaluation in coherent and consistent ways. Each staff member within a school is going to have very diverse views on all aspects of school self-evaluation and its implementation within their school. This is problematic, however, as the model of school self-evaluation mandated by the DES is a one-size-fits all approach that seeks input from all school stakeholders as a collective process. The whole school in its entirety is seen as the unit of intervention (Simons, 2013; Chapman and Sammons, 2013). But in reality, the success of school self-evaluation as a government policy is very dependent on the attitude and capacity of individuals in each of their own school settings (Ryan and Timmer, 2013). In my findings, eighty percent described their own attitude to school self-evaluation as indifferent, with only twenty percent claiming it to be positive. Such diverse views are typical of a complex adaptive system, demonstrating in very simple terms that a 'one-size-fits all' rule or solution may not meet the needs of an entire school community (Ngan et al., 2010; Chapman and Sammons, 2013). There were divided opinions on the purpose of school self-evaluation, namely whether it was a process of improvement or accountability. Staff felt mandated to use school selfevaluation as a method for school improvement, but at the same time, their engagement and results with it are open to public scrutiny for accountability purposes. This tension between improvement and accountability seemed to be problematic. Staff members felt the school self-evaluation process was something imposed on them from outside as an accountability exercise for the inspectorate rather than an improvement process that they do for themselves. Interviews unearthed a perception held by some staff members of a convergent relationship between the role of the inspectorate and school self-evaluation and accountability and improvement. Action was required to show that the practical tasks of 'doing' self-evaluation in classrooms were basically the same as the externally mandated school self-evaluation they felt forced to use and that school self-evaluation was not a form of inspection. To ensure the successful implementation of school self-evaluation in our school, the staff needed a cohesive understanding of where school self-evaluation fitted alongside all the other evaluation activities undertaken in our school.

7.4.2 Nested systems, connectedness and patterns of interaction

Complexity theory would suggest that members in a school community need to develop an internal interactive capacity to meet any external challenge. Within masses of interaction, the system as a whole can become energised and sensitive to fluctuation, continually operating with unfolding configurations and a multiplicity of possibilities (Osberg, 2008). This suggests that for school self-evaluation to be successfully implemented, it must begin with a 'bottomup' community approach. An internally motivated evaluation culture can occur (Mann and Smith, 2013) if staff are perceived as team members in a school with shared objectives (Potter et al., 2002) rather than individual classroom teachers (Schildkamp et al., 2012). The more staff members are facilitated to engage in collaborative professional conversations regarding school self-evaluation, the more such interactions have the potential to increase the motivation, interest and positivity of staff towards the school self-evaluation process, thereby improving its implementation and potential success. However, I led school self-evaluation in a structure of hierarchal compliance, with a quick-fix, go-it-alone approach to shield my staff from any burden, limiting any possibility of collaborative benefit or meaningful organisational change. The voice of school staff was simply not sought. In this action plan however, staff members played a fully active part in the school self-evaluation process, in partnership with school leadership. My findings suggested that this approach had a very positive influence on the outcome of initiating and implementing a school self-evaluation cycle in our school. The data showed a general consensus among school staff, that collaborating as a collective and playing

a fully active part in the school self-evaluation and decision-making process, had a positive effect on their attitude towards and their experience of 'doing' school self-evaluation in our school.

7.4.3 Information flow and feedback loops

According to complexity theory, information, knowledge and feedback are passed on when people interact with each other (O'Day, 2002). This lies at the heart of any learning and change process within a complex system, such as a school education system (Morrison, 2002). However, using this theory O'Day (2002) identifies major flaws with school self-evaluation as a government mandated policy. School improvement, she argues, is premised on a flat command structure aimed at fostering growth. Members within the school are seen as a source of information, with valuable insights into their own unique setting and context (O'Day, 2002). In contrast, school accountability is premised on a hierarchal structure, aimed at fostering compliance, where members are seen as recipients of externally mandated 'one-size-fits all' rules and programs, delivered in a 'top-down' flow of information (O'Day, 2002). This topdown flow of policy information from the DES as the knowledgeable expert treats staff members as mere recipients of information playing a passive role, instead of valuable sources of information in their own right. Such a top-down transmissive model of information flow, can result in schools reacting to directions imposed from above and outside the school in a form of compliance, more so than a meaningful reflection on internal practice (O'Day, 2002; Hall and Noyes, 2009; McNamara and O'Hara, 2012). In Ireland, this has been compounded even further by the lack of a loop mechanism for school staff to feedback to the policymakers responsible for school self-evaluation policy that they were tasked to implement. Contact between the two has been minimal. The only training given to staff has been through DES guidelines (DES, 2016), in-service feedback from two members of staff, an inspectorate-built website and an advisory visit with a school inspector (O'Brien et al., 2019) which data showed was not meaningful for some. According to complexity theory, staff members receiving information in such a passive and complaint way, may not lead to learning and change (Morrison, 2002; Bower, 2006). To ensure the success of school self-evaluation, it was important to ensure that staff members had plenty of meaningful opportunities to interact, discuss and reflect on teaching and learning matters collectively, with an input into all decisionmaking relating to improvement and change at whole school level (Leitch and Day, 2000). In addition, resourcing the project in terms of professional and personal support such as training, access to information, substitute cover and operational manuals, was imperative.

7.4.4 Change, adaptability, co-evolution, self-organisation and emergence

Implementing change in schools is a complex process (Austin and Harkins, 2008; Guhn, 2009; Given et al., 2010; Craig, 2012; Hannay and Earl, 2012; Williams et al., 2012). In complex systems such as schools, planned interventions do not always lead to predictable and intended outcomes (Keshavarz et al., 2010). When there is a change in conditions, school members can make adaptive changes without any imposition or planning and emerge unified with a coherent best fit for their context (Osberg, 2008; Fenwick, 2012). The findings showed that the staff did have an appetite for school self-evaluation and acknowledged its benefits for improving our school. Even though it wasn't popular, they could clearly see its merits when approached in a collaborative manner. This was a noticeable change in attitude following the implementation of the first set of action strategies. Yet despite this, seventy percent of the staff members felt the DES approach to school self-evaluation was meaningless, irrelevant, and out of touch with reality on the ground. The fact that some school staff found school self-evaluation both meaningless and irrelevant was undoubtingly going to have major consequences for the successful implementation of the process in our school. In addition, the interplay of other components within our school, such as the perceived lack of time, training opportunities, and resources, was informing the emerging collective behaviour among staff members and negatively affecting the potential success of school self-evaluation in our school (Keshavarz et al., 2010). Staff members were prioritising and making productive choices by complying with their mandated duties in a contrived way, just to satisfy an external governing body (O'Day, 2002). To address these concerns, the staff members organised extra support for themselves, such as time and resources. However, true to the unpredictable nature of a complex system, despite all the actions taken, the data showed the negative attitude of the staff members to the version of school self-evaluation put forward by the DES, did not change. The staff emerged strongly resistant to implementing the process. In complex systems, such as schools, planned interventions do not always lead to predictable and intended outcomes (Keshavarz et al., 2010). Some staff members even went as far as to say that the time carved out for them to engage more meaningfully in the process as a collective, was actually valuable time wasted. Implementing change in schools is a complex process, even when staff members are provided with the time, and support they need to do it well (Austin and Harkins, 2008; Guhn, 2009; Pescarmona, 2010; Given et al., 2010; Craig, 2012; Hannay and Earl, 2012; Williams et al., 2012). A school's general behaviour therefore, is the result of the interplay of multiple factors,

and is accordingly an emergent phenomenon that is not easily or fully predictable (Osberg, 2008).

7.5 Summary of findings

To summarise, staff members acknowledged the benefits of self-evaluation in improving their school. However, staff reported a lack of resources in the practical implementation of the school self-evaluation approach mandated by the DES such as time, information and support. This caused an increase in workload and anxiety for school members and divided opinions on the purpose of school self-evaluation, namely whether it was a process of improvement or accountability. This was having a negative impact on their interest and enthusiasm for the process. To ensure the success of the project going forward therefore, there was a need for clear and practical expectations regarding the implementation of school self-evaluation. It was also necessary to give adequate time for school members to talk and reflect collectively as they implemented the process in our school. In addition, resourcing the project in terms of professional and personal support such as training, access to information, substitute cover and operational manuals, was imperative.

To improve the implementation of school self-evaluation in our school, staff members tackled the issues of time, information and support in an action plan, in the belief that this would increase the level of interest and enthusiasm for school self-evaluation amongst staff members. As a result, the evidence showed a positive shift in outlook amongst staff members to certain aspects of the process, such as increased levels of collaboration, autonomy, decision-making, reflection and discussion. However, the data showed school staff to be complying with the external mandate of school self-evaluation in a somewhat contrived way.

Interviews unearthed a perception held by some staff members of a convergent relationship between the role of the inspectorate and school self-evaluation and accountability and improvement. Staff members engaged in school self-evaluation believed they were doing the job of the inspectorate. They saw it as an added thing to do in their already busy workload. The reconnaissance data showed this perception to be problematic as it was having an impact on their attitude towards the implementation of the school self-evaluation process in our school. Our action plan involved making staff members familiar with DES school self-evaluation guidelines. The guidelines (DES, 2016) clearly described school self-evaluation and external evaluation as two different, but complementary processes, as they both focus on improvement. The guidelines stated that the school self-evaluation process gave schools a means of identifying and addressing priorities, whilst external evaluation gave the inspectorate a means to inspect and assess the school's teaching and learning practices, taking into account the schools' engagement with self-evaluation and its outcomes. However, based on the data gathered from staff in this reconnaissance phase, it was clear that some school members saw both processes as interchangeable. They saw no difference between external and internal evaluation. They were seen as one and the same. All the inner beliefs, attitudes and emotions associated by most staff members with inspection were applied to school self-evaluation. This was problematic as both international and Irish research show school staff in general, having a negative disposition towards school inspection (O'Connor, 2001; Oluwatomi, 2007; Dillon, 2012; Ofsted, 2019). Also problematic, was the absence amongst staff of any collective understanding of what school self-evaluation was. The actions taken to clearly outline and define the distinct purpose of school self-evaluation to staff members, had clearly not succeeded.

Going forward, it was important to show staff members that school self-evaluation was so much more than a form of inspection. I needed to develop with staff a cohesive understanding of where school self-evaluation fitted alongside all the other evaluation activities undertaken in our school and to demonstrate how it was basically similar in operation at macro level, as the internal classroom self-evaluation they were doing at micro level in their daily practice.

It was also important to maintain the interest and positivity of staff members to collaborate together and be involved in all decision-making. My findings suggested that the nature and strength of interactions between staff members had a very positive influence on the outcome of initiating and implementing a school self-evaluation cycle in our school. Interaction such as this seemed to increase the flow of information, knowledge and feedback between school staff, raising their exposure to variety, innovation and creativity, in turn creating new possibilities and opportunities for learning, adaption and evolution. It was in these forums that staff members explored diverse individual interpretation, in attempts to reach a consensus of interpretation at whole school level. Complexity theory would suggest that members in a school community need to develop this internal interactive capacity to meet any external challenge. For this to become a reality however, school leaders need to resource and coordinate this community of collaborative endeavour, giving all staff members full agency and input into any decision-making related to engage in collaborative professional conversations regarding school self-evaluation, the more such interactions have the potential to increase the motivation,

interest and positivity of staff towards the school self-evaluation process, thereby improving its implementation and potential success. It was important therefore, that there were continued opportunities for staff to interact meaningfully, to collectively discuss and reflect on teaching and learning. It was imperative for the successful implementation of school self-evaluation in our school, that it was a collegial and collaborative process with input from all staff into any decision-making. Every opportunity needed to be taken to show the positive benefits that such collective actions taken at whole school level had for classroom practice and vice versa.

7.6 The next step

In my review of the literature, I came upon an article by Mutch (2013) which described a process undertaken in New Zealand to develop a collaborative school evaluation approach which balanced the strengths and weaknesses of both internal and external perspectives. At its core was self-evaluation. Similar to Ireland's approach to school self-evaluation (DES, 2016), it provided schools in New Zealand with an externally mandated self-evaluation framework. Mutch (2013) argues however, that the methodology of evaluation was unfamiliar and imposed on schools and in the end required more supports to enable the schools undertake the process. More interestingly however, was Mutch's (2013) contention that for school self-evaluation to succeed, there was a need to resolve the tensions between improvement and accountability. This resonated with the findings in this action research study.

To improve the implementation of school self-evaluation, a collaborative project involving a series of workshops was launched in 2008 between the New Zealand school inspectorate and school stakeholders to enhance school self-evaluation. School stakeholders came away with three tasks from these workshops. The first was to come to an agreed statement with their staff and school community of what self-evaluation meant in the context of their school. The second was to collate and organise all the evaluative activities that were already undertaken in their school in an attempt to show that self-evaluation was not a new added-on activity. It existed in much of the work already being done by schools. The third was to graphically organise their evaluation plan in a coherent way to portray an internal-external continuum, at the same time showing an accountability and improvement focus to explain to staff why they were doing the evaluation activities that they were doing. It also allowed schools to discuss where they thought much of their evaluation energy as a school staff was being spent and where they might need to refocus. In my experience, these activities had never been done in the roll-out of school self-evaluation in Irish primary schools. The only direct contact staff members have had with the

creators of the school self-evaluation policy they have been tasked to implement has been through DES guidelines, an inspectorate-built website, an advisory visit from a school inspector and in-service feedback from two members of staff. In a national survey of Irish primary school principals (O'Hara et al., 2016), approximately seventy-eight percent said both they and their staff needed more training on how to conduct school self-evaluation. No opportunity was ever afforded to school staff to train together and learn how to 'do' school self-evaluation, with the exception of two members of staff (O'Brien et al. 2019). Nor was an opportunity afforded to staff to collectively discuss and reflect on 'why' we do school self-evaluation and where it fits within our current school practice. Findings from this action research have shown this non-contact approach to be problematic.

As a result of the process described in Mutch (2013), an OECD report (Nusche et al., 2012) found that the earlier apprehension of 'inspection' by schools in New Zealand, was removed or at least attenuated and that the generally positive response to evaluation by school staff could be explained by its non-threatening nature. Before I gave the article to staff for analysis and comment, I discussed it with my critical friend. She found the article easy to read, easy to understand and very relevant. She felt it would be quite accessible and user friendly for school staff. She said it would be important to keep this in mind going forward, considering that staff members had only just begun to settle into a new school year with their pupils amidst a plethora of local pandemic restrictions. Staff members as a result were both anxious and extremely busy. According to my critical friend, any actions going forward in this research project, had to be mindful of that fact and not over burden staff.

A copy of the article by Mutch (2013) was sent to staff via email for comment and analysis, with a due date of late September 2020. At a follow up meeting with staff members, three strong ideas emerged from their engagement with the article. This feedback formed the basis of a possible action plan. Unsurprisingly, their ideas resembled those described in the article i.e., to develop a collaborative school evaluation approach for our school, which defined and balanced the relationship between internal and external perspectives. The two main ideas were as follows:

1. As a school we should create an agreed definition of what school self-evaluation means to us in the context of our own school.

In this action, staff were opening our school to change. If our school remained resistant to an external force, such as DES school self-evaluation policy, our opportunity to experience

variety, innovation and creativity would be limited (Fidan and Balci, 2017). The diversity of our staff and their diverse interpretation of that policy were there to be explored (Cilliers, 1998; Morrison, 2002). Complexity theory suggests that such diversity and dynamism among members puts the system in a state of uncertainty, continually operating with unfolding configurations and a multiplicity of possibilities (Osberg, 2008). This generates the resilience a system needs to sustain itself throughout both challenges and losses. As diverse and dynamic individuals interact within a system, information and resources are exchanged. The more frequent and powerful these interactions are, the more influence they are likely to have on the behaviour of school members and in turn, collective behaviour patterns at whole-school level (Holland, 1992, 1999; Keshavarz et al., 2010; Fenwick, 2012). It is here that new relationships and new structures are forged causing the school to evolve (Morrison, 2002). This suggested that for school self-evaluation to be implemented successfully it must begin with a 'bottomup' community approach. This diversity gave our school great dynamism (Morrison, 2002; Keshavarz et al, 2010; Fenwick, 2012; Chapman and Sammons, 2013; Fidan and Balci, 2017). As leader of the school, it was important that I now stepped back and allowed staff to interact and self-organise in a community of collaborative endeavour, with full agency and input into all decision-making related to the creation of a shared definition of school self-evaluation (Osberg, 2008; Day and Sammons, 2014; Fidan & Öztürk, 2015). This interaction would motivate the increased sharing of information and interpretation among staff members. It would stimulate new possibilities in our collective attempt to reach whole school consensus on what school self-evaluation means in the context of our own school (Leitch and Day, 2000; O'Day, 2002).

2. As a school we should list and organise into a graph all the general activities that are undertaken in our school for accountability and improvement, acknowledging why we do them and for whom

According to complexity theorists, everyone within our school community is interconnected within a nested network system (Holland, 1998; Cilliers, 2001; Keshavarz et al., 2010; Fenwick, 2012). Within this system there are other nested sub-systems such as individuals, classes and year groups. On a macro level however, our school is also nested within supra systems such as the DES. As part of this particular supra system our school is in a continuous state of dynamic interaction where change is a constant, occurring with different frequency and intensity (Morrison, 2002; Osberg, 2008; Keshavarz et al., 2010; Fenwick, 2012). When there is a change in policy, such as school self-evaluation, schools can choose to make adaptive

changes and evolve to ensure a best fit. School members as a diverse, but yet dependent collective, can randomly organise themselves and emerge with the best fit for their context (Osberg, 2008; Fenwick, 2012). If on the other hand however, a school chooses not to listen, adapt or contribute to policy, it runs the risk of becoming stagnant and ineffective as an organisation (Morrison, 2002). To adapt and survive as a school, with the ability to feedback into and influence the evolution and emergence of future policy (Osberg, 2008), we needed to know our context and acknowledge our place within a nested system. Making a list of all the evaluation activities undertaken in our school, knowing why we do them and for whom, represented the need of our school to be able to communicate and explain, in graph format, our evaluation plan, to all those we are interconnected with as part of a nested network.

Chapter 8 – Action Research Cycle 2

8.1 Introduction

The purpose of this chapter is to describe action research cycle two in this research. This involved planning and implementing the action steps necessary to improve the implementation of school self-evaluation in our school.

8.2 Planning

At an online meeting in early October 2020, staff members discussed at length the action steps needed to improve the implementation of school self-evaluation in our school. Findings from the review (i.e., reconnaissance) of the first phase of action research showed staff members to have diverse opinions on the purpose of school self-evaluation and its place in relation to all the other evaluation activities undertaken in our school.

Two ideas emerged in the reconnaissance phase to frame this action plan and address these issues. The first idea generated by staff members represented their desire to create an agreed statement of what school self-evaluation meant within the context of their school. To implement school self-evaluation successfully, staff members needed clarity on the practicalities of school self-evaluation and what was expected of them (McNamara and O'Hara, 2005; Ryan et al., 2007; Schildkamp, 2007; McNamara and O'Hara, 2012; O'Brien et al., 2015). This required an unambiguous definition of school self-evaluation that was meaningful, free of confusion, easy to work with and both agreeable and understood by staff members. The hope here was this would help school staff to take ownership of the process and engage more effectively with it (MacBeath, 1999; Vanhoof and Petegem, 2009; Vanhoof et al., 2009; O'Brien et al., 2015). However, findings from the last reconnaissance phase showed staff members having no collective understanding of school self-evaluation or how it differed in relation to other evaluation activities undertaken in their school. At an individual level, the evidence showed staff members to have diverse interpretations regarding the purpose of school self-evaluation. These ranged from being one of improvement, identifying weaknesses, whole school planning, autonomy, professional development to recognising strengths and accountability. Whilst it is typical for diverse members within an organisation such as a school to have diverse interpretations of an external mandate (Keshavarz et al., 2010), it is problematic when a government expects their policies to be performed exactly as they were designed (Skerritt et al., 2021). The findings from this action research study resonated with Terhart's

(2013) claim that the culture and convictions of school staff and the culture and convictions of reformers remain significantly apart.

One of the reasons for this according to the literature is that the language and theoretical thinking used in evaluation initiatives can prevent the very people who are responsible for school self-evaluation in the field, from engaging with it (Simons, 2013). However, this problem can be addressed. One prerequisite is a strong staff commitment to school self-evaluation (OECD, 2013). Empirical evidence tells us that policy implementation processes are far more effective if they involve clear communication (MacBeath, 1999) and involve shared objectives amongst staff (Potter et al., 2002). This secures a commitment and a positive attitude amongst staff members to school self-evaluation as a meaningful exercise (Schildkamp, 2007; Vanhoof et al., 2009; O'Brien et al., 2015). This encourages both staff ownership and staff engagement. For these reasons, it was imperative for the improvement and overall success of our school self-evaluation practice and to ensure staff commitment, that staff members as a collective came to a shared understanding, using clear and simple language, of what school self-evaluation meant for our school.

There was much sharing of ideas among staff members on how this could be done. Agreement was finally reached amongst school staff that the best way to do this was for each staff member to source and share with each other, any information they could find which described best practice and gave clarity on the practicalities of school self-evaluation, and what was expected of them. Using this knowledge, we aimed to reach consensus on a definition of school self-evaluation specific to our own school context. Staff members agreed that all school staff should be invited to be involved in the creation of this statement and that it should be completed by the end of October 2020. This planned action was labelled 'Action 1'.

Findings from the last reconnaissance phase also showed that staff members had no clear understanding of how school self-evaluation differed in relation to other evaluation activities undertaken in our school, such as inspection. This was problematic as international research shows most school staff in general to have a negative bias towards school inspection (O'Connor, 2001; Oluwatomi, 2007; Dillon, 2012; Ofsted, 2019). Findings from the first phase of this action research showed this bias having a negative impact on staff member's attitudes towards the school self-evaluation process.

However, this problem can be addressed according to Wikeley et al. (2002), if school selfevaluation is coordinated with other school activities and this is communicated effectively to everyone (MacBeath, 1999). For this reason, it was imperative for the improvement and authenticity of our school self-evaluation practice that staff acknowledged all the evaluation activities undertaken in our school and understood the part they played in them.

Staff members shared many ideas on how we could do this and at one of our regular meetings consensus was finally reached on systematically listing all of the evaluation activities undertaken in our school. This list would categorise and portray all the activities into a coherent diagram showing who carries them out, why and for whom. This diagram would then be used to communicate and explain to everyone, in as clear and efficient a way as possible, all the evaluation activities undertaken in our school. This planned action was labelled 'Action 2'. Staff members agreed that all school staff should be invited to be involved in the creation of this diagram and that it was to be completed by January 2021. This concluded the project's planning phase. Keeping the reader in mind, the following is a summary chart of the agreed actions undertaken as part of this second action research cycle.

How can we improve the process of school self-evaluation in our school?					
		Action P	lan		
What will we do?	Action number	Why are we doing it?	Who will do it?	When will we do it?	How will we measure
					our success? (i.e., what criteria do we believe will show the impact of our work?)
Create an agreed definition of school self-evaluation.	1	to increase the flow of information and shared interpretation between members within our own context, with the aim of reaching consensus	Principal and staff	Immediately	An agreed statement of what school self- evaluation means in our school will be adopted into our school self- evaluation policy.

Table 8.1Action research cycle 2: Action plan

List and organise into a graph all the general activities undertaken in our school for accountability and improvement.	2	to better understand how school self- evaluation differs in relation to other evaluation activities	Principal and staff	By January 2021	All the evaluative activities already undertaken in our school will be graphically illustrated and adopted
accountability and		differs in			in our school
improvement.					
					-
		undertaken			into our
		in our school			school self-
		and to			evaluation
		communicate			policy.
		and explain			
		in as clear			
		and efficient			
		a way as			
		possible our			
		evaluation			
		activities to			
		everyone			

8.3 Implementation

8.3.1 Action step 1: Create an agreed definition of school self-evaluation.

8.3.1.1 Implementing the action.

Earlier reconnaissance data showed a wide diversity of opinion on the topic of school selfevaluation amongst staff members. To reach agreement on any definition of school selfevaluation, we needed a mechanism that allowed any collective decisions to be made in an effective and efficient way (Kacprzyk et al. 2020). I asked staff members to consider using a decision-making technique called 'multi-voting' (Tague, 2005). Multi-voting is a group decision-making technique that allows the most popular options from a list to be selected, in order to get an idea about the consensus of the group. I saw the technique being used at a school leadership training seminar in NUI Maynooth to reduce a long list of items to a more manageable number, by means of a structured series of votes. The result is a shorter list, identifying what is important to the group. Multi-voting, in general, allows the items that are favoured by all, to rise to the top (Tague, 2005). This approach was in keeping with the action research methodology of the project, based on active participation, open-ended objectives and high levels of commitment and active learning (Bell et al., 2004), with the expectation that any improvement strategies would be developed by the school members themselves as a form of collaborative intervention (Reason and Bradbury, 2001; Bana, 2010; Klenowski and Woods, 2013; Walters, 2014; Jacobs, 2016). To begin, various definitions of school self-evaluation were gathered by school staff as a community task from various sources including books, journals, DES policy documents and the internet. They are all listed with their source in Appendix 25. A total of nineteen definitions of school self-evaluation were gathered by staff members. The plan was for staff members to reduce this list of nineteen definitions to three definitions via online multi-voting. The top definitions of school self-evaluation would then be combined into one definition by consensus.

The nineteen definitions were compiled into an online survey (Appendix 26) and sent to staff in early October 2020, with a due date of mid-October 2020. Staff were requested to select from the list the three definitions of school self-evaluation that they most agreed with. Staff were not required to justify their decisions. The survey received fourteen responses out of fourteen. The content of the online survey and the full results can be seen in Appendix 27.

At this stage we did not engage in any analysis or discussion of the choices made by staff members. Our main focus was to narrow this list of definitions down by identifying and assigning priority to the highest ranked items, with a high degree of staff agreement. The top three votes were shared amongst five definitions. These top ranked definitions were compiled into an online survey one last time and sent to staff with a due date of late October 2020. Staff were asked to select the one definition of school self-evaluation that they most agreed with. The survey received twelve responses out of fourteen. Two staff members from the original fourteen participants were now on leave and this made engagement with the project more difficult for them. The content of the online survey and the results can be seen in Appendix 28.

An analysis of the definitions chosen by staff showed they had both differences and similarities. Some gave a narrow and brief definition of school self-evaluation, whilst others were lengthier and broader in scope. For the purposes of analysis, I underlined and extracted every key word and phrase. I then assigned these key words and phrases a description title, each presenting and encapsulating a recurring aspect of school self-evaluation. These descriptions were then analysed and combined into overarching themes (see *Table 8.2*).

Table 8.2	Definitions of school self-evaluation thematic analys	ic
1 ubic 0.2	Definitions of school self evaluation mematic analys	10

Definitions of School Self-Evaluation Thematic Analysis					
Definition (With extracted key words and phrases in bold)		Themes	Percentage of staff votes in agreement	Overall ranking	
Α	"SSE is considered as an <u>internal</u> and formative evaluation which is based on a <u>collection of evidence</u> . Or in other words the schools have to base their <u>judgments</u> on all the evidence gathered to identify the effectiveness of the implementation of the <u>schools' programs</u> SSE stresses that it is a <u>reflection</u> <u>process</u> by the schools on their own practice. SSE needs to be operated in a <u>systematic and transparent</u> manner in order to achieve its aims to <u>improve the</u> <u>students' achievement</u> and enhance the schools' <u>professional and</u> organizational learning" (Hamzah and Tahir, 2013, pg.51)	 Internal Collection of evidence Making judgements and decisions Goals, targets, strategies and programs Reflection on practice Systematic and transparent Pupil development Teacher development School development Improvement 	17%	3	
В	"School self-evaluation (SSE) is an <u>internal process</u> which aims to ensure <u>quality</u> , <u>improve the teaching–</u> <u>learning process</u> and increase <u>school</u> <u>performance</u> " (Hofman et al., 2009, cited in O'Brien et al., 2019, pg.2)	 Internal Quality Teacher development Pupil development School development Improvement 	0%	4	

С	"School Self-Evaluation is,	Internal	58%	1
	by definition, something that	Making		
	schools do to themselves, by	judgements		
	themselves and for	• •		
	<u>themselves</u> '. School Self-	and decisions		
	Evaluation (SSE) involves	• Teacher		
	examining teaching and	development		
	learning strategies, the	 Pupil 		
	performance and	development		
	development culture and	• School		
	other aspects of school	development		
	operations so they can be	-		
	strengthened and supported	Whole school		
	to <i>improve student</i>	community		
	outcomes. It also provides	Reflection on		
	an opportunity for the <u>whole</u>	practice		
	school community, including	• Improvement		
	learners, parents and all	Cycle of		
	staff, <u>to reflect</u> on the learner			
	outcomes in light of their	planning		
	goals, targets and key <u>improvement</u> strategies from			
	the previous <i>planning cycle</i> "			
	the previous <u>planning cycle</u>			
	(Swaffield and MacBeath, 2005, pg.239; Smith, 2012, cited in Setlalentoa, 2014, pg.525)			
D	<i>"a procedure involving</i>	Systematic	25%	2
	systematic information	and	2370	2
	gathering that is initiated by	transparent		
	the school itself and intends	Collection of		
	to assess the functioning of	evidence		
	the school and the	School		
	attainment of its educational	development		
	<u>goals</u> for purposes of	• Goals, targets,		
	supporting <i>decision making</i>	strategies and		
	and learning and for	programs		
	fostering <u>school</u>	1 0		
	improvement as a whole"	Making		
	(Schildkamp, 2007, pg.4)	judgements		
	(Semieramp, 2007, pg.+)	and decisions		
		 Internal 		
		Internal		
		 Improvement 		

Е	"Self-evaluation is a process	• Reflec	ction on 0%	4
	of reflection on practice,	practio	ce	
	made <u>systematic and</u>	• Syster	matic	
	transparent, with the aim of	and		
	<u>improving pupil,</u>	transp	parent	
	professional and	• Pupil		
	organisational learning"	develo	opment	
	(MacBeath, 2005, pg.4)	• Teach	ler	
		develo	opment	
		• Schoo	ol	
		develo	opment	
		Impro	ovement	

All the definitions chosen by staff, displayed strong references to school self-evaluation being a systematic internal process of improvement, which used reflection and judgement to develop pupils, teachers and the school, as a whole. Interestingly, the word 'quality', appeared only once in the definition sharing the lowest ranking. The literature shows quality can be multifaceted, contested and never fully representable (Stake and Schwandt, 2006) and therefore, difficult to define. Also of note were the strong references to the technical aspects of school self-evaluation in the definitions ranked second and third i.e., the systematic process of collecting evidence, making decisions, planning and setting targets and implementing them. This was significant and obviously an important issue for staff, in terms of school selfevaluation practice. It suggested that school staff wanted clarity on the practicalities of school self-evaluation and what was expected of them. Another salient aspect was the reference to 'community', which was mentioned in the definition with the most votes. Staff members seemed to favour a definition of school self-evaluation in which the school community played an active role, with potential development benefits for all. This was significant, and up until this research, the school community had only played a passive role in the process, due to my assumed hierarchal leadership style.

As can be seen from Table 8.2 above, definition 'C' gained fifty-eight percent of the votes. However, that said, this definition was not agreeable to approximately forty percent of respondents. There seemed to be no definitive winner. This was problematic. I discussed the results with my critical friend. We both agreed that the exercise had been very worthwhile in encouraging staff to actively engage and take ownership of the process. However, the end goal of the exercise was to give clarity to staff members on the practicalities of school selfevaluation and what was expected of them, by creating an unambiguous definition of the process that was meaningful, free of confusion, easy to work with and both agreeable, and understood by all staff. We felt it would be therefore foolish to ignore over forty percent of the school staff. One disadvantage of multi-voting as a decision-making method, according to the literature, is that it does not guarantee consensus (Tague, 2005). My critical friend and I both agreed that we had gone as far as we could go using the multi-voting approach. An online meeting was organised to update staff of this outcome. Staff members agreed that it would be better at this point to take a more nuanced approach, and amalgamate the themes which resonated with school staff, into a bespoke definition agreeable to all.

The plan was to use the definition with the most votes and its key themes, as the bedrock of our agreed definition of school self-evaluation. In addition, strong key themes from other definitions were to be interwoven to give it strength. Whilst the goal was to come to an agreeable statement that was meaningful, clear, user-friendly, flexible and well documented for school staff, equally it needed to be reliable and valid, grounded in theory and best practice from scholarly research (Chapman and Sammons, 2013). It was important that we got the balance right between what we felt were good practical solutions and the evidence-based descriptions of best practice described in research literature (Haroon et al., 2015). This was important to assist staff members in making better decisions and achieving better learning outcomes for both them and their pupils. We know that studies into the effects of school selfevaluation have shown it to improve school quality in terms of improved pupil achievement levels (Campbell and Levin, 2009; McNaughton et al., 2012) but that this is dependent on school staff being committed and positively disposed towards it (MacBeath, 1999; Meuret and Morlaix, 2003). However, if staff members are included and involved as a team (Schildkamp et al., 2012; Mann and Smith, 2013) in all decision-making (Rozenholtz, 1991; Day and Sammons, 2014) and school self-evaluation is implemented with clear communication (MacBeath, 1999) and shared objectives amongst staff (Potter et al., 2002), studies show this to encourage both staff engagement and ownership of the process. Therefore, it was imperative, for the improved implementation of school self-evaluation in our school, to get both the definition and the process of reaching consensus on it right. This entailed a step-by-step process of drafting, ensuring there were appropriate references to all key words and phrases and continuous feedback from staff members on each draft. The final draft agreed by staff was as follows:

"School Self-Evaluation is a <u>cyclical</u> procedure we do to ourselves, by ourselves and for ourselves. It provides an opportunity for the <u>whole school community</u>, including learners, parents and all staff to reflect on our practice and <u>systematically gather information</u> to examine, <u>assess and make decisions</u> on our teaching and learning strategies, the performance and development culture and other aspects of our school's operations, so that they can be strengthened and supported to foster <u>quality</u>, <u>improved teaching and learning</u> and <u>enhanced</u> <u>professional and organisational performance</u>".

This definition made strong references to all the aspects of school self-evaluation identified by staff members as important. It described a cyclical and systematic internal process of quality improvement, involving the whole community, based on gathering evidence, reflection and making judgements to develop pupils, staff and the school as a whole. It also stated school self-evaluation was being done to improve quality within the school in terms of pupil, staff and school outcomes. Lastly, it outlined that active responsibility for school self-evaluation lay with the whole community. This was a welcome move away from it being my sole responsibility as principal of the school, where everyone else played a passive role in the process. The literature shows that such an approach encourages ownership and engagement in the process amongst staff (Parr and Timperley, 2013; Chapman and Sammons, 2013). This shift from me leading school self-evaluation, to them leading it and taking ownership as a community, was important and represented a significant and strategic shift in emphasis over the course of this action research.

This draft definition was sent to staff for final sign off in early November 2020 in the form of an online survey for their approval and feedback (Appendix 29). The survey asked staff members for their thoughts on the definition and if they agreed with it. The survey also asked staff members had they any comments, thoughts or ideas on the way we reached the definition. The survey received seven responses. I raised my concern at this low response with my critical friend. She believed it was due to the Covid-19 global pandemic. All staff were religiously following and monitoring public health safety guidelines in their classrooms. As a result, staff members were both nervous and exhausted on a daily basis. Upon reflection, it was therefore no surprise to me that at that specific point in time, this study had lost some of its urgency. Another factor at play was staff leave. During the research for example, four members of staff went on leave and did not return. However, they all asked to be kept informed of progress and responded with feedback, when possible, but not always. From this point on in the study, I deemed it important to continuously monitor the response rate and consult with my critical friend and staff members to check for any concerns or issues.

Out of seven respondents, six agreed with the definition. One respondent however took issue with the phrase "we do to ourselves, by ourselves and for ourselves" (S6 respondent 2). In the opinion of this respondent, school self-evaluation was a process "driven by an external force" (S6 respondent 2) which schools had no choice but to submit to. This was true. In 2012, school self-evaluation was put on a formal footing by the DES as a government mandated, whole school development process. Prior to this schools were only required to produce general written reports and establish and maintain systems to assess the efficiency and effectiveness of its operations, but with no requirement for annual productivity targets or specific outputs (O'Brien et al., 2015). The respondent also referred with fondness to this more informal arrangement saying "schools by their nature are centres of continued analysis, with a view to enhancing outcomes, but in a more informal fashion" (S6 respondent 2).

This more informal approach is supported by Brady (2016) who argues that the government mandated school self-evaluation guidelines introduce a level of prescription never before seen in an Irish school improvement process. She argues that the 2012 document is so prescriptive that it leaves very little room for the teachers own self to be present in the evaluation or for their own criteria for good teaching to be developed or even considered, making what is supposed to be a spontaneous and self-generated process into a bureaucratic, externally produced, standard driven procedure. Staff members in this study saw the DES as an external force making them do added work in the form of inspecting their own school. This was problematic. The prevalence of this perception amongst several staff was having a negative impact on their attitude towards school self-evaluation and its successful implementation. It was important that our collectively agreed definition acknowledged school self-evaluation was not a form of inspection by stealth driven by an external force.

After much reflection and consultation with my critical friend and the teacher who raised the concern, it was decided therefore, to replace the phrase with the name of our school. This, we felt, would better reflect school self-evaluation being an internally owned process in our school driven by the staff within. This was a significant and important step forward in creating within school staff a sense of ownership of the school self-evaluation process. The opportunities afforded to staff members in our action plan to interact in a spirit of collegiality and
collaboration, and to collectively discuss and make decisions regarding a shared meaning of school self-evaluation, was bearing fruit.

This draft definition (*Table 8.3*) was compiled into an online survey (Appendix 30) and distributed to staff in mid- November 2020, with a due date of late November 2020. The survey received nine responses.

Table 8.3An agreed definition of what school self-evaluation means to us for our school

An agreed definition of what "School Self-Evaluation" means to us for our school

"In [our school name] school self-evaluation is a cyclical procedure, which provides an opportunity for the whole school community, including learners, parents and all staff to reflect on our practice and systematically gather information to examine, assess and make decisions on our teaching and learning strategies, the performance and development culture and other aspects of our school's operations, so that they can be strengthened and supported to foster quality, improved teaching and learning and enhanced professional and organisational performance"

Out of nine respondents, all nine agreed with the final draft definition. Eight respondents had no further comments, thoughts or ideas on the agreed definition and wrote '*pass*'.

All staff members were involved in the creation of this collective definition to address the absence amongst school staff of a collective understanding of what school self-evaluation was. This first action was enacted by staff members to improve the implementation of school self-evaluation in our school. It aimed to give clarity to staff members on the practicalities of school self-evaluation and what was expected of them, by creating an unambiguous definition of school self-evaluation that was meaningful, free of confusion, easy to work with and both agreeable and understood by all staff.

8.3.1.2 Monitoring implementation of the action and its effects.

The process devised by staff to reach this definition was generally well received. One respondent found the process "to be clear" (S7 respondent 6). In terms of giving clarity on how things are done for example, one respondent said "the definition is clear and understood" (S7 respondent 4), whilst another stated "I think the above statement gives a very clear definition of what school self-evaluation means to us and takes the mystery out of it" (S7 respondent 1). This clarity was important. For example, one respondent believed that having such an agreed definition "is essential for the development and improvement of the school" (S7 respondent 5). However, it must be acknowledged that one respondent in feedback said that 'parents' were

not specifically mentioned (S7 respondent 6). Yet in the statement they were. This begged the question, was the statement read with vigour by staff members or was this just a simple oversight. Whilst the agreed definition was a tool to provide staff members with clarity on the practicalities of school self-evaluation and what was expected of them, its usefulness going forward would need to be measured against the success of any future implementation of school self-evaluation in our school (Ryan et al., 2007; Vanhoof and Petegem, 2009). In terms of ownership, staff members positively embraced the opportunity to give their input into an agreed definition of what school self-evaluation meant to them in their own school. One respondent said "it enabled everyone to voice an opinion without getting distracted or some voices taking over" (S7 respondent 8) whilst another stated that the agreed definition was reached "by coming together as a whole school staff to organise and improve on strategies to best suit our role in the school and the future learning of the children" (S7 respondent 5). The opportunities given to staff members for whole school reflection and collective input into all decision-making regarding the creation of a meaningful definition seemed to have a positive impact on their attitude towards school self-evaluation too. One respondent stated "the manner in which the statement has been reached allows for all voices to be heard, ultimately promoting buy-in from all" (S7 respondent 1).

8.3.2 Action step 2: List and organise into a graph all the general activities undertaken in our school for accountability and improvement.

8.3.2.1 Implementing the action.

To list and organise into a graph all the general activities undertaken in our school for accountability and improvement, we used the four-quadrant evaluation framework (*Figure 8.1*), described earlier in Mutch (2013), where each evaluation activity is placed within one of the four following quadrants based on their purpose (i.e., improvement or accountability) and who conducted it (i.e., internally or externally):



Figure 8.1 Four-quadrant evaluation diagram

Quadrant 1: External accountability relates to school evaluations that are externally mandated, implemented and reported upon e.g., audit, assurance, compliance.

Quadrant 2: Internal accountability is where evaluations are selected and conducted by the school for internal or external accountability purposes e.g., self-assessment, self-accountability, self-inspection.

Quadrant 3: Internal improvement is where formative evaluations are selected and conducted by the school for the purposes of improvement. e.g., self-evaluation, self-review, action research.

Quadrant 4: External improvement is where a developmental or improvement focused school evaluation is conducted by an outside agency or person. e.g., external evaluation, external development review.

Mutch (2013) describes in great detail how this four-quadrant framework was used to list and categorise all the evaluation activities undertaken by schools each year in New Zealand. It was an attempt in New Zealand to show that school self-evaluation was not a new 'added-on' activity, but that it existed in much of the work already being done in their schools. According to Valli and Buese (2007), many teachers express concern at the time required to engage in

school self-evaluation and the additional workload involved. They say the work of school selfevaluation is an extra burden adding to their workload (Hall and Noyes, 2009). According to Vanhoof et al. (2009) if staff members experience school self-evaluation as something imposed on them, this can contribute to it being seen as more of an obligation than anything else, the principal objective being compliancy rather than improvement. To prevent such confusion, my staff members wanted to see how school self-evaluation differed from all the other evaluation activities undertaken in their school, understanding why they were doing it and for whom. This was important, as O'Brien et al. (2015) like Vanhoof et al. (2009) believe, one precondition which favours a successful school self-evaluation is achieving an awareness that it is both meaningful and worthwhile.

To begin, the basis of each quadrant was compiled into an online survey and sent to staff in mid-November 2020, with a due date of late November 2020. Staff were requested to answer four questions. If they could not think of anything to contribute, they had to write the word '*pass*'. The survey received nine responses. The content of the online survey and the results can be seen in Appendix 31. What follows is a sample (*Table 8.4*).

Table 8.4Evaluation activities undertaken in our school and why, according to ourschool staff

Question 1		at you can think of that our school engages ted externally for accountability purposes? g please just write 'pass')
	Respondent	Answer
	4	Pass
	5	 Inspectorate advisory teams / individual enhancing progress of whole school development PDST supports guidance Comparing national figures of child development / statistics to current school levels etc.
	7	 Teaching and learning School self-evaluation Special Education Needs and inclusion Mentoring and Droichead process Monitoring attendance Curricular changes Active and green school flags

	 Vetting procedures Employment
	Employment Data protoction CDPP
	Data protection – GDPR
	Health and safety Child anote stigger and see Iferry
	• Child protection and welfare
	• NEPS
	• Anti-bullying procedures
	• Legislation
8	FSSU
9	Credit union quiz Pass
-	
•	can think of that our school engages in that
Question 2 we select and conduct for self	• • •
(If you cannot think of anything	piease just write pass)
Respondent	Answer
4	Pass
6	• Individual teacher self-reflection
	Continuous Professional Development
	Whole School Self-Evaluation
	• Staff meetings
	• Great communication between
	colleagues
	Planning
7	• Assessment of and for learning,
What are all the activities the	teaching and learning activities
	at you can think of that our school engages for school self-improvement purposes?
(If you cannot think of anything	
Respondent	Answer
3	• Handwriting – uniformity of
	approach and practice
	• Píosa as Gaeilge performed by each
	class at assembly (pre COVID)
6	Continuous Professional
	Development
	• Whole school self-evaluation
	• Self-reflection and review of practice
	• Open minded to ideas for
	improvement
	improvementTeam spirit among staff
8	improvement

Question 4		can think of that our school engages in that xternally for improvement purposes? please just write 'pass')
	Respondent	Answer
	1	Pass
	2	CoursesTraining programmesUpskilling
	9	• Courses carried out in the Education Centre in order to upskill ourselves

As can be seen from the table above, staff were able to list many of the evaluation activities undertaken in our school. However, when categorising these activities as having an accountability or an improvement focus, they experienced difficulties. Staff members were both unclear and confused regarding whether the focus of school self-evaluation was one of improvement or accountability. For example, four out of the nine respondents believed school self-evaluation was conducted for accountability purposes (respondents 3,5,6,7). One respondent for example said DES inspection was an activity conducted externally for accountability purposes but at the same time also said it was conducted externally for improvement purposes (respondent 6). The majority of staff members did not associate school self-evaluation with activities that were conducted for improvement. When asked for activities that our school conducts for self-accountability purposes, two respondents listed school selfevaluation (respondents 5,6). A fourth respondent listed whole school development (respondent 3). When asked for activities that our school conducts for school improvement purposes, only one of the nine respondents listed school self-evaluation (respondent 6). This blurred line between improvement and accountability was problematic as international research shows most school staff in general to have a negative bias towards accountability mechanisms, such as school inspection (O'Connor, 2001; Olutwatomi, 2007; Dillon, 2012; Ofsted, 2019). This was causing a diverse range of responses amongst staff towards school self-evaluation, ranging from resistance to a contrived compliance. When combined with what was my hierarchal leadership of the process, the overall outcome was one of an ineffective, unauthentic and fabricated improvement process, to satisfy the inspectorate.

To improve the implementation of school self-evaluation in our school, this issue needed to be addressed. It all seemed to stem from staff members having a limited understanding of how school self-evaluation differed in relation to other evaluation activities undertaken in their school. A meeting was organised to discuss these results with school staff and to begin the process of rectifying the problem. After the discussion, staff members collectively began to place each evaluation activity they listed into a group based on its purpose (i.e., improvement or accountability) and who conducts it (i.e., internally or externally) using the four-quadrant evaluation framework described in Mutch (2012) as our guide. In feedback at the end of the meeting, staff agreed that a clear graphic device to explain this list of evaluation activities to everyone would be invaluable. New people join our school each year, mainly due to staff taking leave, resignations or retirements. There was of course also the impact of Covid-19 on staffing, which proved especially challenging in the conduct of this action research. With such an annual turnover of school staff, it was important to have mechanisms in place to sustain any improvement made in implementing school self-evaluation. This graphic diagram would allow us to communicate and explain all the evaluation activities undertaken in our school in as clear and efficient a way as possible. It would also help people to see how school self-evaluation differed from all these other activities, why they were doing it and for whom. Showing school self-evaluation as a meaningful and worthwhile endeavour in this way, according to the literature, is favourable to its successful implementation (Vanhoof et al., 2009; O'Brien et al., 2015). This was important to ensure the continued successful implementation of school selfevaluation in our school going forward. Together as a staff we engaged in a process of filtering and organising the list of evaluation activities into a coherent diagram that could be explained to everyone, which was approved by staff members. The completed diagram can be seen in Figure 8.2 below.

Quadrant 2: Internal accountability is where evaluations an by the school for internal or external accountability numbers. The		
	evaluations are selected and conducted purposes. These might be self- ia or indicators provided by the external usually summative and provide a ent, self-accountability, self-inspection	Cluadrant 3: Internal improvement is where formative evaluations are selected and conducted by the school for the purposes of improvement. Much of a school's day-to-day business and medium to long term strategic decision-making is in this quadrant. The scope can vary from teacher reflection through to school-wide self-review. It can include students setting and monitoring their own progress through to school-wide analysis of student results to inform planning and resource allocation e.g., self-evaluation, self-review, action research Such as: Handwriting – uniformity of approach and practice School subject discussion to identify what is working well and areas to improve on Share information from CPD with each other
 Staff meetings and communication Teacher Self-reflection Policy Planning Curriculum Planning Quadrant 1: External accountability relates to school evaluations that are externall mandated, implemented and reported upon. These focus more on compliance, audit, or assurance. These evaluations have their place in providing summative, objective, external scrutiny unencumbered by internal history, politics or perceptions e.g., audit, assurance, 	o school evaluations that are externally focus more on compliance, audit, or oviding summative, objective, external or perceptions e.g., audit, assurance,	Researching and assessing data such as achievement statistics School self-evaluation Reflection and review of our practice Parent and pupil input Parent and pupil input Cluadrant 4. External improvement is where a developmental or improvement focused school evaluation is conducted by an outside agency or person. For example, an engagement with consultants, advisors, professional developers or the inspectorate to conduct an analysis of a current situation so that future actions can be determined e.g., external evaluation,
compliance Such as: DES Inspection Comparing national figures and statistics of child development SEN and inclusion Monitoring attendance Curriculum change Vetting procedures Employment compliance Data Protection Health and Safety (building regulations, fire etc.) Child protection and welfare Anti-Bullying procedures Any relevant legislation FSSU	Such Extremely and Extremely a	external development review Such as: Such as: Suppliers of new schemes coming in to allow us examine and discuss materials and resources Upskilling Training programmes Courses CPD PDST advisory visits Whole school evaluation

IN LOD OLIVER LEDIT

Figure 8.2 Diagram of all the evaluation activities undertaken in our school

8.3.2.2 Monitoring implementation of the action and its effects.

The diagram was sent via email to staff members for final sign of in the form of an online survey for their approval and feedback (Appendix 32). the survey asked for their thoughts on the diagram and if they agreed with it. They survey also asked had they any comments, thoughts or ideas on the wat we reached this diagram.

Two respondents said the quadrant diagram gave a clear picture of all the evaluation activities undertaken in the school and their primary focus.

"The quadrant format is very clear" (S9 respondent 1)

"It was well put together and made everything look much clearer when it was broken down into each quadrant" (S9 respondent 2)

One respondent found that the quadrant diagram helped them to better understand the purpose of specific evaluation processes.

"I thought there were good points made as to what these evaluative processes are actually for" (S9 respondent 1)

The same respondent goes on to specifically mention their attitude to whole school evaluation as being *"really negative...I tend to think of it as something we do for outside people"* (S9 respondent 1). However, the quadrant diagram helped to *"point out the benefits"* (S9 respondent 1). This was a significant movement in attitude towards seeing school self-evaluation as a positive process for our school, and an important step to ensure staff members engaged with it in an authentic and uncontrived way.

Interestingly, at the conclusion of the first phase of action research, when asked to describe what emotions came to mind when they thought about school self-evaluation, the same respondent said "confusion, apathy, fear, bewilderment and trepidation", (interviewee C) describing school self-evaluation as "a little distasteful...not very appealing to me" (interviewee C).

At the conclusion of this second phase however, when speaking of the quadrant diagram, the same respondent said *"if it's intended as a motivational tool, I would deem it a success! … If it's designed to encourage engagement with the SSE and WSE process, I would also say it's successful"* (S9 respondent 1).

Whilst the diagram was a tool to show staff members what evaluation activities are undertaken in our school, who carries them out, why and for whom, its creation represented a significant shift in building a whole staff commitment to school self-evaluation and their recognition of its professional value.

8.4 Findings and discussion

The findings showed these actions had a positive impact on the attitude of school staff towards school self-evaluation, by not only giving staff members greater clarity about the process and their role within it, but also a greater sense of ownership. The process of reducing nineteen definitions to one definition, also revealed one interesting finding. Out of a possible forty-two votes (14 respondents x 3 votes each = 42 votes), the definition of school self-evaluation offered to primary schools by the DES, received just one vote. This resonated with earlier findings that seventy percent of staff members found school self-evaluation, as presented by the DES meaningless, irrelevant and confusing. Emerging negative attitudes such as these have the potential to inform the entire behaviour of people within a school and the potential success of its school self-evaluation process (MacBeath, 1999; O'Day, 2002; Keshavarz et al., 2010).

The definition of school self-evaluation agreed upon by staff in this study showed staff members were positively disposed towards a cyclical and systematic internal process of quality improvement involving the whole community, based on gathering evidence, reflection and making judgements to develop pupils, teachers and the school as a whole.

An analysis of the definition of school self-evaluation, offered by the DES (2016) as "School self-evaluation is a <u>collaborative</u>, inclusive, and collective process of <u>internal</u> school review. An <u>evidence-based</u> approach, it involves <u>gathering information</u> from a range of sources, and then <u>making judgements</u>. All of this with a view to bring about <u>improvements in students</u>' <u>learning</u>" (pg.10), similarly makes reference to school self-evaluation being a collaborative internal process based on gathering evidence and making judgements to develop pupils. Interestingly however, it makes no reference to teacher and school development or the concept of community. There is also no mention of it being an ongoing continuous process to ensure quality.

The disconnect between school staff on the ground and DES policy had real implications for the successful implementation of school self-evaluation in our school. Staff members seemed to favour a definition of school self-evaluation which valued the school community as a whole and offered potential development benefits for all within it (e.g., pupils, parents, staff, locality). This was further borne out by participants voting favourably to insert our school's name in the agreed final definition. This resonated with research by O'Brien et al. (2019) which showed that teachers were more likely to value school self-evaluation if the process results in improvements at classroom level. To improve the implementation of school self-evaluation in our school therefore, it was important that every opportunity was taken to show the positive benefits that collective actions had at both whole school and classroom level. However, it is important to state that whilst there was a focus on classroom improvement at the outset of this research, as the work proceeded it became clear there were real and significant issues regarding staff negativity towards school self-evaluation, as well as issues regarding their sense of ownership and value of the process.

For example, findings showed that whilst staff were able to list many evaluation activities that were undertaken in our school, they experienced some difficulties when categorising these activities as having an accountability or an improvement focus. The DES guidelines (2016) clearly state that school self-evaluation is focused on improvement. In the findings here however, four out of nine respondents believed school self-evaluation was conducted for accountability purposes (S8 respondents 3,5,6,7). This resonated with data from the reconnaissance phase where school staff felt they were doing the job of the inspectorate.

"Rather than the department coming in and doing their whole school evaluation and deciding, this area, you need to improve, and this is what we want you to do. They were just saying, okay schools, you need to do this yourselves" (interviewee A)

One respondent even said the purpose of school self-evaluation was to "*cut down unnecessary investigations*" and "*to take pressure of the department*" (S3 respondent 1).

One respondent said DES inspection was an activity conducted externally for accountability purposes but also at the same time said it was conducted externally for improvement purposes (S8 respondent 6). The findings also showed that school staff did not associate school self-evaluation with activities that were conducted for improvement. When asked for activities that our school conducts for self-accountability purposes, two respondents listed school self-evaluation (S8 respondents 5,6). A fourth respondent listed whole school development (S8 respondent 3)

When asked for activities that our school conducts for school improvement purposes, respondents replied

"Regular meetings to discuss the areas in which we are performing well...and identifying the areas upon which we can improve to ensure we are improving as a school" (S8 respondent 9)

"Discuss areas of improvement...discuss what is working well...focus on areas in need of enhancement" (S8 respondent 5)

"School subject based discussion and meetings to act on and improve on" (S8 respondent 3)

However, only one of the nine respondents listed school self-evaluation (S8 respondent 6).

This suggested that staff members in general, did not see school self-evaluation as a process of improvement and didn't identify it as such. So, what did staff identify as a process of improvement? Interestingly, when asked for activities that our school engages in that are conducted externally for improvement purposes, one respondent named 'Department of Education inspections'. This blurring of lines between improvement and accountability was problematic and was negatively affecting their attitude towards the school self-evaluation process and its potential successful implementation. The findings showed however, that when staff members listed, categorised and portrayed all the evaluation activities undertaken in our school into a clear and coherent diagram using the four-quadrant framework described by Mutch (2013), it had a positive impact on their attitude towards the school self-evaluation process.

For school self-evaluation to be implemented successfully therefore, it must have clarity of purpose for school staff (Leitch and Day, 2000). Policy implementation is more effective if staff members are motivated (Mann and Smith, 2013) and have agency in the process working as a team (Rosenholtz, 1991; Schildkamp et al., 2012), with clear communication (MacBeath, 1999), shared objectives and expectations (Potter et al., 2002; Ryan et al., 2007; Vanhoof and Petegem, 2009) and input into all decision-making (Day and Sammons, 2014). Making school self-evaluation more meaningful in this way helps to secure commitment and a positive attitude from those implementing it (Schildkamp, 2007; Vanhoof et al., 2009; O'Brien et al., 2015), vital preconditions to the successful implementation of school self-evaluation in any school (; MacBeath, 1999; Meuret and Morlaix, 2003; Schildkamp, 2007; Bubb and Early, 2008). The information, knowledge and feedback shared between staff members in such an environment, increases their exposure to raised levels of variety, innovation and creativity, leading to new possibilities and a continuous evaluation of how things are done, both of which are vital to organisational improvement (Morrison, 2002; O'Day, 2002; Keshavarz et al., 2010; Chapman and Sammons, 2013; Fidan and Balci, 2017). Strong interactive relationships such as these,

make a positive contribution to school improvement and change (Rozenholtz, 1991; Day and Sammons, 2014), proving that an internally motivated evaluation culture can occur (Mann and Smith, 2013) if staff are perceived as team members in a school with shared objectives (Potter et al. 2002). According to my findings, school self-evaluation to be implemented successfully must also be a collegial and collaborative process of community endeavour, where staff members have continued meaningful opportunities to interact, discuss and reflect collectively on the distinct purpose of school self-evaluation within their own school, with full agency and input into all decisions and actions necessary regarding its implementation.

8.5 Further analysis through the lens of complexity theory

This section will complement the analysis already presented and will comment on the findings through the following four key concepts of complexity theory.

8.5.1 The diversity and dynamic nature of agents

The 'one-size-fits' all model of school self-evaluation mandated by the DES did not resonate with the diverse and dynamic range of staff members within our school. Implementing school self-evaluation successfully is complex due to this diversity and dynamism (Keshavarz et al., 2010; Fidan and Balci, 2017). On this basis, a big question for this study was, how could we reach beyond the collective level of our school, to mobilise the 'change' needed at individual level, to implement school self-evaluation successfully.

Coming to an agreed definition of school self-evaluation by the staff had a positive impact on the attitude of school staff towards the process, by not only giving the staff members greater clarity about the process within their own context and their role within it, but also a greater sense of ownership. It resonated with them and was more meaningful, in comparison to the definition of school self-evaluation offered by the DES (2016). As a school principal, I too personally moved to see school self-evaluation as a form of self-improvement that allowed us the freedom to collectively take ownership of our school's own processes, motivating us to increase our knowledge and skills in teaching and learning. All these developments had a positive impact on the implementation of school self-evaluation in our school at an individual level.

Due to a diverse range of knowledge, staff members also experienced however, some difficulties categorizing activities undertaken in our school as having accountability or an improvement focus. The findings showed, however, that when staff members listed, categorized, and portrayed all the evaluation activities undertaken in our school into a clear and

coherent diagram using the four-quadrant framework described by Mutch (2013), it had a positive impact on their attitude towards the school self-evaluation process. To ensure the successful implementation of school self-evaluation in our school, the staff needed a clear understanding of what school self-evaluation was and where it fitted alongside all the other evaluation activities undertaken in our school.

8.5.2 Nested systems, connectedness and patterns of interaction

Complexity theory contends that in the right conditions, interactions amongst members within a system have the power to alter behaviour patterns at individual level and in turn collective behaviour patterns at whole school level (Holland, 1992, 1999; Keshavarz et al., 2010; Fenwick, 2012). My findings support this view. When school self-evaluation was implemented as a collegial and collaborative process of community endeavour, where the staff members have continued meaningful opportunities to interact, discuss, and reflect collectively on the distinct purpose of school self-evaluation within their own school, it made a positive contribution to school improvement and change.

8.5.3 Information flow and feedback loops

Policies get adapted to local contexts by the people within them and often times they are not exactly what was intended (McNamara et al., 2002; Meuret and Morlaix et al., 2003; Fullan et al., 2006; Howlett and Raynor, 2006; Ngan et al., 2010; Terhart, 2013; McNamara and Nayir, 2015; Hudson et al., 2019). The disconnect between school staff on the ground and DES policy had real implications for the successful implementation of school self-evaluation in our school. Staff members seemed to favour a definition of school self-evaluation which valued the school community as a whole and offered potential development benefits for all within it. The blurring of lines between improvement and accountability among staff members was problematic and was negatively affecting their attitude towards the school self-evaluation process and its potential successful implementation. However, the increased flow of information, knowledge and feedback shared between staff members raised their exposure to heightened levels of variety, innovation, and creativity, leading to new possibilities and an evaluation of how things are done. For example, when staff members spent the time and effort to unpack and analyse various definitions of school self-evaluation and re-write the definition in their own words, it resonated with them and was more meaningful. Coming to our own agreed definition of school self-evaluation had a positive impact on the attitude of school staff towards the process.

The findings also showed that when staff members themselves listed and categorised all the evaluation activities undertaken in our school into a diagram under the headings of eternal accountability, internal accountability, internal improvement and eternal improvement, it gave us all clarity and a better understanding regarding the name and purpose of each evaluation activity. When structures are placed to support the provision, movement and use of information, people engage more productively in their work (Bower, 2006). The movement of information, through the interaction of people, lies at the heart of any learning and change process within a complex system, such as a school (Morrison, 2002; Stacey, 2005). Our actions helped develop a positive attitude towards the process, showing school self-evaluation to be a non-threatening, non-judgmental and formative process of improvement, with a positive focus on our own efforts at improvement and good practice.

8.5.4 Change, adaptability, co-evolution, self-organisation and emergence

Fidan and Ozturk (2015) believe that a flat and flexible hierarchical structure can be more successful in creating more space for innovation in a complex system. My actions as school leader facilitated the participants to interact and self-organise in order for them to find the best way to implement school self-evaluation successfully in our school (Morrison, 2002). Complexity theory contends that such multiple interactions among diverse school members can allow unity and coherence to emerge without any imposition or planning (Fenwick, 2012). Consequently, some staff members reported that the actions we had taken as a collective, would motivate them and encourage them, to engage with the school self-evaluation process in a more successful way going forward. This was a significant movement in attitude towards seeing school self-evaluation as a positive process, and an important step in ensuring staff members engaged with and emerged from the process in a more authentic and uncontrived way.

8.6 Summary of findings

This second phase of action research revealed three key findings. The first finding showed that school staff were very receptive to being offered the opportunity to come to an agreed definition of what school self-evaluation meant to them for their school. It helped them to better understand both the purpose and operation of school self-evaluation within their own context. It resonated with them and was more meaningful, in comparison to the definition of school self-evaluation offered by the DES (2016). The findings showed it made a difference by not only giving staff members greater clarity about school self-evaluation and their role within it,

but also a greater sense of ownership of the process. This had a positive impact on their attitude towards school self-evaluation.

The definition of school self-evaluation agreed upon by staff showed staff members were positively disposed towards a cyclical and systematic internal process of quality improvement involving the whole community, based on gathering evidence, reflection and making judgements to develop pupils, teachers and the school as a whole. Staff members seemed to favour a definition of school self-evaluation which valued the school community as a whole and offered potential development benefits for all within it (e.g., pupils, parents, staff, locality). This was further borne out by staff members voting favourably to insert our school's name into the final agreed definition.

Based on these findings therefore, to improve the implementation of school self-evaluation, the process needs to be a collegial and collaborative process of community endeavour, where staff members have meaningful opportunities to interact, discuss and reflect collectively on the distinct purpose of school self-evaluation within their own school, with full agency and input into all decisions and actions necessary regarding its implementation.

The second key finding showed that when staff members themselves listed and categorised all the evaluation activities undertaken in our school into a diagram under the headings of external accountability, internal accountability, internal improvement and external improvement, it gave us all clarity and a better understanding regarding the name and purpose of each evaluation activity. They said the diagram helped us see the benefits of school self-evaluation and would therefore motivate and encourage us all, as a community, to engage with the school self-evaluation process in a more successful way going forward. This was a significant movement in attitude towards seeing school self-evaluation as a positive process, and an important step in ensuring staff members engaged with it in an authentic and uncontrived way. Based on these findings therefore, to improve the implementation of school self-evaluation, school staff must be allowed to list and categorise what evaluation activities are undertaken in their school, who carries them out and for whom. Our actions helped develop a positive attitude towards the process of improvement, with a positive focus on our own efforts at improvement and good practice.

The third and final key finding showed that my professional practice as a school leader, underwent change as a result of my own self-development during this study. The findings revealed learning for me on both a personal and professional level as a member of the school staff. At the beginning of this project, the focus was on my actions as leader of the school selfevaluation process. At the outset, I felt school self-evaluation was an unwelcome addition to my staff's workload, more attuned to a policy of control and compliance than improvement. My experience of this influenced both my negative attitude and hierarchical approach to school self-evaluation. However, the action cycles expanded concentrically to include the actions of my school staff. Through my study and self-reflection of values and this action research, I personally moved to see school self-evaluation as a form of self-improvement. To increase the potential success of school self-evaluation, this action research revealed the necessity for an active and engaged leadership that worked closely in more low-rise ways with staff members in a spirit of community endeavour, rather than a form of leadership based on hierarchy. This was a significant shift, as a principal's leadership can influence a school's handling of evaluation, whether it is perceived as a threat or an opportunity for development (MacBeath, 2008). As a result of this study, I concluded that school self-evaluation allowed us the freedom as a school staff, to collectively take ownership of our school's own processes, and motivated us to increase our knowledge and skills in teaching and learning. This change in my thinking as principal, had a positive impact on the implementation of school self-evaluation in our school.

Chapter 9 – Conclusion and Recommendations

9.1 Introduction

In this chapter I summarise the work presented in the thesis which investigated school selfevaluation in practice, examining its implementation by school staff in a primary school. The purpose of this qualitative study was to improve the school self-evaluation process in my school and how I, as school leader in collaboration with my school staff, could create the optimal conditions necessary for its successful implementation, whilst also addressing any barriers which may have existed. The conclusions drawn from this study derive from the interpretation of the findings from the surveys, the interviews, the documents and the literature review and they form the basis for the recommendations provided. I outline the contribution of the study to the aims of the research, as well as its contribution to knowledge and the methodological field, followed by its limitations. Finally, I identify aspects of school self-evaluation for possible future research.

9.2 Summary of findings

School self-evaluation is a form of internal evaluation which has become a key mechanism for school improvement and accountability to guarantee educational quality (Hopkins, 2005; MacBeath, 2006; McNamara and O'Hara, 2008; Vanhoof et al., 2011; DES, 2012; OECD, 2013). Since 2012, all schools in Ireland are formally required to engage in school self-evaluation (DES, 2012). This thesis seeks to augment the existing body of research relating to school self-evaluation at implementation level in an Irish school setting. The research question guiding this study was how can we improve the implementation of school self-evaluation in our school.

The study demonstrated from a methodological perspective that the adoption of an action research approach enabled us as a collective to reflect on and examine in a systematic and careful way our implementation of school self-evaluation. This practice-based approach facilitated learning during each cycle of the study from the perspective of staff members. Taking this approach enhanced the value of the study, so that it was more than a technical exercise. It was also offering evidence on how school self-evaluation mechanisms, as well as improvement strategies could be designed and developed for use in primary schools.

From a theoretical perspective, to answer the research question and help school leaders to be more aware of the optimal conditions and potential barriers to successfully implementing school self-evaluation in their school, staff members responses about their experience with school self-evaluation were analysed through the conceptual lens of complexity theory. This study makes a beneficial and distinctive contribution to knowledge by drawing complexity theory and action research together, to explore how the model can be applied in practical terms to the implementation of school self-evaluation in an Irish rural primary school, at local level by school staff.

To begin, this study found that similar to research by O'Brien et al. (2015) school selfevaluation seemed to be 'built-on' rather than 'built-in' to the accepted core business of our school. However, school staff saw the relationships and the interactions between them whilst they engaged in school self-evaluation as a strength. Staff members reported the interactions between them to be powerfully motivating in sparking their interest to be involved in school activities concerning teaching and learning. Anything that threatened these relationships or their focus on teaching and learning was seen as a weakness. Similar to research by O'Day (2002), this study found that the more staff members were facilitated to exchange information, knowledge and feedback in collaborative professional conversation and meaningfully involved in any decision-making regarding school self-evaluation, this increased their motivation, interest and positivity towards the process. This in turn had a very positive influence on the outcome of initiating and implementing a school self-evaluation cycle. In addition, this study supported findings in other research which showed an internally motivated evaluation culture can occur, if staff work as team members with shared objectives and input into any decisionmaking related to the process of school improvement (Mortimore et al., 1988; Rozenholtz, 1991; Potter et al., 2002; Schildkamp et al., 2012; Mann and Smith, 2013; Parr and Timperley, 2013; Jones and Harris, 2014; Day and Sammons, 2014).

However, numerous researchers have suggested that school staff are resistant to school selfevaluation due to a perception of it being time consuming, difficult to carry out and an increase in their workload (Valli and Beuse, 2007; Hall and Noyes, 2009; Vanhoof et al., 2009; O'Brien et al., 2015). The results of this study support these conclusions. Staff members reported the school self-evaluation process to be under resourced, time consuming and an increase to their workload. This negatively affected their motivation and enthusiasm for it. Similar to the conclusions drawn in other research (O'Day, 2002; Keshavarz et al., 2010), the data from this study showed these issues caused unwanted stress for staff members and negatively informed their collective attitude and behaviour towards the process. The data also showed there to be a lack of self-confidence in doing school self-evaluation among staff members. Numerous research has pointed to the importance of providing support and clear expectations to school staff if school self-evaluation is to be implemented successfully, such as effective leadership, continuous professional development, substitute teacher cover, time to meet and talk with each other and answers to pragmatic process questions such as who, what, where, when and how (O'Day, 2002; Ryan et al., 2007; Hall and Noyes, 2009; Vanhoof and Petegem, 2011; Vanhoof et al., 2011; Mutch, 2013; O'Brien et al., 2015). The data from this study confirmed these conclusions. Three actions taken as school principal had a positive impact on the implementation of school self-evaluation in our school. Firstly, when I led the school self-evaluation process adhering to my core values of collaboration and authenticity, the motivation, interest and positivity of staff members towards the process increased. Secondly, the same effect was also found, when I used every opportunity to show the positive benefits school self-evaluation and our collective actions had on classroom practice. Finally, when I resourced the process in terms of professional and personal supports, there was a positive shift in outlook amongst staff members to certain aspects of the school self-evaluation process, such as increased levels of collaboration, autonomy, decision-making, reflection and discussion.

The study demonstrated therefore, that school leaders play a key role in shaping staff members' attitudes, commitment and effectiveness in any school self-evaluation process. A hierarchical 'top-down' approach to leading the process fails, because it forces people to change what they do, without having input or ownership into the change process. For school self-evaluation to succeed, this study revealed the necessity for an active, engaged leadership that works closely, in low-rise ways with all school staff, rather than a form of leadership that views the process as something to be tightly controlled or delegated. It is important that school leaders, in as far as is possible, resource the process, with plenty of opportunities for staff members to collectively talk, reflect and make decisions in a spirit of collegiality and collaboration. Equally, it shows the importance of school leaders being positive towards school self-evaluation and continuously reflecting on all that they say, do and think as they lead the process.

This study expanded the school self-evaluation literature however by moving beyond school self-evaluation as a technical exercise to be executed and asked staff members to answer the important question of, why are we doing school self-evaluation? My analysis of the data revealed an absence amongst staff members of any collective understanding of what school self-evaluation was. The findings showed staff members to have diverse opinions on the purpose of school self-evaluation and its place in relation to all the other evaluation activities undertaken in our school. This supports findings in other research which shows diverse

members within an organisation can be both unpredictable and diverse in their reactions and interpretations of an external mandate (Morrison, 2002; O'Day, 2002; Fidan and Balci, 2007; Keshavarz et al., 2010; Ngan et al., 2010). To secure the positive commitment of school staff in implementing school self-evaluation, various studies point to the importance of clear communication (MacBeath, 1999; Leitch and Day, 2000; Meuret and Morlaix, 2003; McNamara and O'Hara, 2005; Ryan et al., 2007; Schildkamp, 2007; Bubb and Earley, 2008; Vanhoof et al., 2009; O'Brien et al., 2015; Vanhoof and Petegem, 2009; McNamara and O'Hara, 2012; Simons, 2013; Terhart, 2013; Fullan, 2015). The results of this study support this conclusion, finding that the lack of clarity amongst staff members regarding the purpose of school self-evaluation had a negative impact on their attitude towards it. In addition, the findings showed a real disconnect between school staff and DES policy regarding school selfevaluation, similar to conclusions drawn from other research (O'Brien et al., 2015; McNamara and O'Hara, 2012). Data from this study showed seventy percent of staff members found school self-evaluation, as presented by the DES, meaningless, irrelevant and confusing. Indeed, out of a possible forty-two votes, the definition of school self-evaluation offered to primary schools by the DES, received just one vote.

However, the findings of this study confirm conclusions drawn in other current research which show that when staff collectively engage in a meaningful process of coming to an agreeable and clear definition of school self-evaluation relevant to their own school context, it makes a difference by not only giving staff members greater clarity about school self-evaluation and their role within it, but also a greater sense of ownership of the process (Mutch, 2013). In addition, my analysis of the data revealed that school staff were very receptive to being offered the opportunity to come to an agreed definition of school self-evaluation for their school, as it helped them to better understand both the purpose and operation of the process within their own school and make it more meaningful. This was found to have a positive impact on their attitude towards the school self-evaluation process. For the potential success of school self-evaluation to be realised, this study has demonstrated therefore, the importance of giving staff members the opportunity to collectively engage in a meaningful process of coming to an agreeable and clear definition of school self-evaluation which they own. This provided them with a clear and shared sense of purpose in the process.

Regardless of having a collectively agreed definition of school self-evaluation however, my analysis of the data showed school staff to be complying with the external mandate of school self-evaluation in a contrived way, similar to conclusions drawn in other research (; O'Day,

2002; Ball, 2003; Webb, 2006; Vanhoof et al. 2009; Perryman, 2009; NcNamara and O'Hara, 2012; Beaver and Weinbaum, 2015; O'Brien et al., 2019). My findings also support the views of other research which has shown school staff being able to name most of the evaluation activities undertaken in their school but struggling to identify if they have an accountability or improvement focus (Mutch, 2013). This distinction matters, as international research shows most school staff in general to have a negative bias towards accountability mechanisms (O'Connor, 2001; Olutwatomi, 2007; Dillon, 2012; Ofsted, 2019). However, my analysis of the data from this study showed the majority of staff members to be both unclear and confused as to whether school self-evaluation was a process of improvement or accountability, with no clear understanding of how it differed in relation to other evaluation activities undertaken in our school, such as inspection. Just under half of staff members saw school self-evaluation as an accountability exercise by an external source, where school self-evaluation was something 'imposed' on them from outside as a means of judgement on their school.

In addition to these findings, an analysis of the data unearthed a perception held by some staff members of a convergent relationship between the role of the inspectorate and school self-evaluation and accountability and improvement. Whilst the findings of this study do not confirm or refute what other researchers have found in regard to a negative bias towards school inspection amongst school staff in general (O'Connor, 2001; Oluwatomi, 2007; Dillon, 2012; Ofsted, 2019), this study was distinctive in that it demonstrated a belief amongst staff members that their engagement in the school self-evaluation process was equivalent to doing the job of the inspectorate. Staff members saw both the process of external inspection and self-evaluation as interchangeable with no difference between them, seeing them as both one and the same. All the inner beliefs, attitudes and emotions associated by most staff members with inspection were being applied to school self-evaluation. My analysis of the data revealed that this was having a negative impact on their attitude towards the implementation of school self-evaluation.

Other research however has suggested that when school staff are given the opportunity to list, categorise and portray every general activity undertaken in their school for accountability and improvement, into a clear and coherent diagram explaining who carries them out, for whom and why, highlighting its benefits, it gave staff members greater clarity and a more cohesive understanding regarding the name and purpose of each evaluation activity undertaken in the school (Mutch, 2013). The results of this study support this conclusion. In addition to this finding, my analysis of the data also showed that such an exercise significantly motivated and

encouraged staff members to engage with the school self-evaluation process in a more positive, authentic and uncontrived way as a process of improvement. Similar to other research, this study found that producing the necessary evidence to change the inner beliefs of staff members provided them with the motivation to change their practice (Kise, 2004). This study has demonstrated that to increase the potential success of school self-evaluation, it is important that staff members understand its purpose in relation to all other evaluation activities undertaken in their school.

9.3 The next step

These findings were sent to staff members and my critical friend for feedback and comment. At a follow up online meeting of school staff, it was suggested by some members that we should identify another area of focus for improvement and implement another cycle of school self-evaluation, employing all the new resources at our disposal, which we had developed in the study to date. These included the school self-evaluation timeline which we created in the first phase, showing what had to be done and when. It also included from the second phase, our recently created definition of school self-evaluation and our explanatory diagram of all the evaluation activities undertaken in our school.

At the end of this second cycle of school self-evaluation it would be interesting to survey staff to see if there were any further movements in attitude towards the process. Throughout this study, complexity theory has shown us that outcomes are not always predictable and therefore further reconnaissance and investigation may be required. Whilst another phase of action research was beyond the time constraints of this study, the question that needed to be asked was, 'does school staff defining the process of school self-evaluation and how it differs from other forms of evaluation undertaken in the school, improve the potential success of the process?'

Staff also suggested that we revisit our school's handwriting policy and assess if the policy is meeting its aims of

- developing and improving fine motor skills in the early years
- ensuring good letter formation from Junior Infants to Sixth Class
- encouraging all children to use the correct pencil grip
- ensuring legibility and neat presentation of all written work
- helping to develop speed, accuracy and writing fluency

- ensuring uniformity and consistency in letter formation from class to class
- providing children with a relevant life skill
- helping in the improvement of spelling

This study will be of interest to researchers from the fields of school evaluation and school leadership. From a school evaluation perspective, this work provides the reader with a deeper understanding of school self-evaluation in practice. The various cycles of study which are presented, address a gap in the literature by demonstrating and analysing the implementation of school self-evaluation in practice by school staff in an Irish primary school (McNamara et al., 2021). The study expands our knowledge and understanding of the conditions which enable school staff to implement the process of school self-evaluation in a successful and meaningful way. At a school leadership level, this thesis examined the impact of values and reflective practice on a school principal's ability to lead the school self-evaluation process. For these reasons, this research and its findings will be disseminated in multiple ways. I will write and submit an article to the Irish National Teachers Organisation (INTO) outlining the research and its findings for potential publication in its monthly magazine for primary school teachers in Ireland. In addition, the same article will be submitted to the Irish Primary Principals Network (IPPN) for potential publication in 'Leadership+', their monthly magazine for primary school principals in Ireland. I will also submit a paper on my research for potential presentation to both the Community of Action Research Network (CARN) and the Network for Educational Action Research in Ireland (NEARI). Finally, it is my intention to offer to present my work at any of IPPN's county or national meetings.

9.4 Limitations

The limitations of this study were related to the research design, including the challenges of conducting insider research and limited transferability due to a small sample size and a single research site. In this study, each of the cycles were conducted with a cohort of school staff in a medium sized rural primary school, with a maximum potential sample size of fourteen people in any of the cycles. Steps were taken to ensure the dependability of both the data and my conclusions. However, the transferability of the results and the potential to extrapolate them to a wider context was limited by the sample population and size. The fourteen participants represented a range of class levels and came from diverse backgrounds, but half of them have all worked in this same school over the past ten years, and therefore, their experiences with

implementing change may reflect the particulars of this school and how it initiates and supports reform for improvement.

My role as an insider was a potential limitation because of my dual role as principal in the participating school. There is always the danger of significant power differentials emerging and threats that (a) school staff feel compelled to engage because of my role and (b) see it as a way for me to get them to take on work that I am doing. Being principal of the school for over eight years I experienced the same reforms in school improvement policy that participants described in the surveys and interviews conducted as part of this research. As such, the potential existed for a biased interpretation of the data. In order to address this potential bias, I used specific strategies such as reflexivity to improve the trustworthiness of this study by maintaining a reflective journal. I also paid careful attention to statements and attitudes that were discrepant from my own experiences in order to ensure that I fully represented them in the analysis and discussion of the findings. By using the strategy of member checking, I confirmed that transcripts were accurate and that respondents agreed with my description and analysis of the findings. My critical friend too played a very important role at salient moments throughout the research. The thesis is replete with many examples of moments where conversations with my critical friend formatively impacted my thinking and subsequent actions. In addition, participants self-selected, raising the possibility that they were motivated by a desire to express strongly held views which may not have been typical of the views held by their peers. It must be acknowledged that different participants would likely have expressed different perceptions, drawing on their own experiences of and attitudes towards school selfevaluation. The intention of this study was to conduct an in-depth pragmatic, action research study of the implementation of school self-evaluation in an Irish primary school setting, supported by school leadership. However, larger scale studies may be needed if claims are to be made about the viability of the approaches and mechanisms used in wider contexts.

Due to the action research approach adopted for this study, the concern I had regarding my leadership of the school self-evaluation process was chosen as the focus of research so that cycles could be implemented, evaluated and changes made in practice. The result of this is that the study was context specific with insights relevant to the issues pertinent to my school. It focused on the implementation of school self-evaluation in practice, examining its implementation by school staff in my school. In the nature and spirit of action research, a wider study with a more diverse range of staff from alternative educational settings may yield different results and experiences.

Another limitation is the fact that data for this study was gathered at a fixed point in time following the experiences of staff members engaging with the school self-evaluation process and so its findings are limited by that. However, action research, by its nature does not come to an abrupt end and there is always a next step to take as listed in section 9.3. Noteworthy too of course, was the challenge of conducting insider research during the worst pandemic in our lifetime and the many challenges it posed for our school.

Finally, academic literature on the implementation of school self-evaluation in Irish schools is growing (McNamara and O'Hara, 2005; McNamara and O'Hara, 2006; McNamara et al., 2008; McNamara and O'Hara, 2012; O'Brien et al., 2014; O'Brien et al., 2015; Gustafsson et al., 2015; Brown et al., 2017; O'Brien et al., 2017; Brown et al., 2018; O'Brien et al. 2019). As research in the area progresses, new alternative themes for study may emerge which were not addressed in this thesis. However, while the above limitations have been noted, the purpose of this study was to carry out an action research inquiry into the implementation of school self-evaluation in a natural context based within my own school environment. It is my hope that that this research will contribute to a deeper understanding of school self-evaluation in Ireland and be a spur to further research.

9.5 Recommendations for future work

Throughout the cycles of this research thesis, many questions were raised and answered. However, as is the nature of any research, the results identified and posed many new questions for future research. Throughout the study I was concerned that the views and experiences of my staff as they implement school self-evaluation would be at the core of the research, determining the conclusions I would draw. The centrality of their voices to the discussion in this report indicates how successfully I have addressed that concern. The themes which have emerged from their experiences have identified a number of areas for further research as follows:

- More school-based action research is needed to tease out the complexities, difficulties and opportunities that school self-evaluation presents. From such research, school leaders will be better placed to initiate and sustain more effective and practicable methods of self-evaluation in their schools.
- 2. The small size of this study enabled me to gain a deeper insight into the implementation of school self-evaluation in a primary school. However, it would be valuable to carry out the research across a greater number of schools to strengthen the validity of the

findings. It would allow stronger inferences to be drawn from the conclusions and the capacity to make generalisations from the findings.

- 3. Throughout this study, complexity theory has shown that outcomes are not always predictable and therefore further reconnaissance and investigation may be required. Whilst another phase of action research was beyond the time constraints of this study, further investigation into staff members collectively defining school self-evaluation and how it differs from other forms of evaluation undertaken in their school, would be worthwhile, in order to evaluate its impact, if any, to the potential success of the process.
- 4. One of the limitations of the study was that an evaluation of the actions taken and their impact on teaching and learning outcomes was beyond the scope of the research. A longitudinal study to evaluate the impact of the process on the performance of both pupils and staff over a number of years would be very worthwhile.
- 5. This study focused on obtaining the views of school staff implementing school selfevaluation in a primary school. Insights from teaching staff were sought through interviews. The views of other school personnel, particularly those of special needs assistants, is largely unexplored and would add significantly to further developing an understanding of the process in schools.
- 6. In section 4.6 I outline how action research underpins the model of school self-evaluation espoused in Irish primary schools. Action research is based on a 'bottom-up' collaborative approach to change, putting key stakeholders at the centre of all efforts to improve in a spirit of partnership. Yet in a contradiction, schools are mandated to engage with school self-evaluation in a 'top-down' edict from the DES. In addition, the inspectorate are key stakeholders tasked to advise and oversee the implementation of this action research-based process. Yet as key stakeholders and partners, they are generally absent in the practical implementation of the process itself (McNamara et al., 2021a). This is contradictory to the participative collaborative spirit of action research and school self-evaluation. According to complexity systems theory, these connections and relationships are critical to a school's success. This warrants further discussion and research into revising or reforming the role that Ireland's Inspectorate plays in school self-evaluation as an action research methodology.

9.6 Final thoughts

In many educational systems, school self-evaluation has become a key mechanism for school improvement and accountability and a necessary mechanism to manage changes in the school organisation (Hopkins, 2005; MacBeath, 2006; McNamara and O'Hara, 2008; Vanhoof et al., 2011; DES, 2012). Many education systems have recently applied newly developed evaluation methods at school level in the form of internal or self-evaluation (Nevo, 2010), evident by the fact that it is now compulsory in two thirds of European education systems (European Commission/EACEA/Eurydice, 2015). Research indicates that the quality of a school self-evaluation process and its impact on pupil learning outcomes is strongly determined by how that evaluation process is carried out (Vanhoof et al., 2011). In this endeavour, principals play a key role in setting the direction and creating a positive school culture, including a proactive mindset, and supporting and enhancing the motivation and commitment needed from staff to foster improvement (Day and Sammons, 2014). In this area I have focused my research.

The study showed a positive change in staff members' attitudes, commitment, and capacity to implement school self-evaluation when the school leadership itself was reflexive and positive towards the process and provided plenty of opportunities for staff members to collectively talk, reflect and make decisions regarding the process in a spirit of collegiality and collaboration.

The study would urge school leaders to be cautious however of school staff complying with school self-evaluation in a contrived way. The research showed the process of improvement and change needed to be at a substantive and deep level rather than a simple overlay of existing practices and beliefs. The study unearthed an assumption amongst staff members that their engagement in school self-evaluation was equivalent to doing the job of the inspectorate. A negative bias towards inspection was in turn having a negative impact on their attitude towards school self-evaluation. Yet, the study showed school staff continuing to comply with the process in a contrived way. However, when school leadership (i) provided opportunities for staff members to collectively come to an agreeable and clear definition of school self-evaluation which they owned and (ii) produced evidence to show them its purpose in relation to all other evaluation activities undertaken in the school, it significantly motivated and encouraged school staff to engage with school self-evaluation in a more positive and meaningful way as a process of improvement. Whether this actually led to an improvement in pupil outcomes however, is beyond the scope of this study, as it will take time for the results of these processes to become apparent.

There is still much work to be done, and room for development in the area. School selfevaluation requires further investigation in practice in order to bridge the gap between policy and implementation, as described in the literature (McNamara et al., 2002; Meuret and Morlaix, 2003; Howlett and Raynor, 2006; ; Fullan et al., 2006; Ngan et al., 2010; McNamara and O'Hara, 2012; Terhart, 2013; McNamara and Nayir, 2015; O'Brien et al., 2015; Hudson et al., 2019; Skerritt et al., 2021) and achieve its stated aim of contributing to school improvement.

Bibliography

Al-Haddad, S. and Kotnour, T. (2015). 'Integrating the Organizational Change Literature: A Model for Successful Change'. Journal of Organizational Change Management 28 (2): 234–62. <u>https://doi.org/10.1108/JOCM-11-2013-0215</u>.

Alhadeff-Jones, M. (2008). 'Three Generations of Complexity Theories: Nuances and Ambiguities'. *Educational Philosophy and Theory* 40 (1): 66–82. https://doi.org/10.1111/j.1469-5812.2007.00411.x.

Alise, M.A., and Teddlie, C. (2010). 'A Continuation of the Paradigm Wars? Prevalence Rates of Methodological Approaches Across the Social/Behavioural Sciences'. Journal of Mixed Methods Research 4 (2): 103–26. <u>https://doi.org/10.1177/1558689809360805</u>.

Almalki, S. (2016). 'Integrating Quantitative and Qualitative Data in Mixed Methods Research—Challenges and Benefits'. Journal of Education and Learning 5 (3): 288. <u>https://doi.org/10.5539/jel.v5n3p288</u>.

Anderson, G.L., and Herr, K. (1999). 'The New Paradigm Wars: Is There Room for Rigorous Practitioner Knowledge in Schools and Universities?' Educational Researcher 28 (5): 12–40. https://doi.org/10.3102/0013189X028005012.

Anderson, J., and Boyle, C. (2020). "Good" Education in a Neo-Liberal Paradigm: Challenges, Contradictions and Consternations'. In *Inclusive Education: Global Issues and Controversies*, 35–57. BRILL. <u>https://doi.org/10.1163/9789004431171</u>.

Anderson, S., Leithwood, K. and Strauss, T. (2010). 'Leading Data Use in Schools: Organizational Conditions and Practices at the School and District Levels'. Leadership and Policy in Schools 9 (3): 292–327. <u>https://doi.org/10.1080/15700761003731492</u>.

Anh, V.T.K. (2018). 'Evaluation Models in Educational Program: Strengths and Weaknesses'. VNU Journal of Foreign Studies 34 (2): 140–50. <u>https://doi.org/10.25073/2525-2445/vnufs.4252</u>.

Argyris, C., and Schon, D. (1996). Organisational Learning. Reading, MA: Addison-Wesley.

Artino, A.R., La Rochelle, J.S., Dezee, K.J. and Gehlbach. H. (2014). 'Developing
Questionnaires for Educational Research: AMEE Guide No. 87'. Medical Teacher 36 (6):
463–74. <u>https://doi.org/10.3109/0142159X.2014.889814</u>.

Ary, D., Jacobs, L.C., Razavieh, A. and Ary, D. (2010). Introduction to Research in Education. 8th ed. Belmont, CA: Wadsworth.

Ates, H.K. (2018). 'Problems of Gifted and Talented Students Regarding Cursive Handwriting: Parent Opinions'. Edited by Ahmet Cezmi Savaş. European Journal of Educational Research 7 (2): 295–301. <u>https://doi.org/10.12973/eu-jer.7.2.295</u>.

Atienza, C.M.R. (2017). 'Building Organizational Identity: An Insider Action Research from a Founder's Viewpoint'. Systemic Practice and Action Research 30 (6): 569–92. https://doi.org/10.1007/s11213-017-9410-2.

Auemaneekul, N. (2010). 'The Continuum from Action Research to Community Based Participation Research'. Journal of Public Health 40 (1): 86–100.

Austin, M. and Harkins, D. (2008). Shifting spaces and emerging voices: Participation, support, and conflict in one school administrative team. Early Education & Development, 19(6), 907-940.

Avis, J. (2003). 'Re-Thinking Trust in a Performative Culture: The Case of Education'. Journal of Education Policy 18 (3): 315–32. <u>https://doi.org/10.1080/02680930305577</u>.

Aziz, S.M.M., and Rehman, Z. (2018). 'Implementation of CIPP Model for Quality Evaluation at School Level: A Case Study'. Journal of Education and Educational Development 5 (1): 189. <u>https://doi.org/10.22555/joeed.v5i1.1553</u>.

Babbie, E. (2012). The Practice of Social Research. 13th ed. New York: Wadsworth Company.

Bailey, K.D. (1994). Methods of Social Research. 4th ed. New York: Toronto: New York: Free Press; Maxwell Macmillan Canada; Maxwell Macmillan International.

Ball, S., and Goodson, F. (1985). Teachers' Lives and Careers. Sussex: The Falmer Press.

Ball, S.J. (2003). 'The Teacher's Soul and the Terrors of Performativity'. Journal of Education Policy 18 (2): 215–28. <u>https://doi.org/10.1080/0268093022000043065</u>.

Bambino, D. (2002). 'Critical Friends.' Educational Leadership 59 (6): 25.

Bana, Z. (2010). "Great Conversation" for School Improvement in Disadvantageous Rural Contexts: A Participatory Case Study'. Educational Action Research 18 (2): 213–37. https://doi.org/10.1080/09650791003741111. Barber, M., and Moursehead. M. (2009). 'Shaping the Future: How Good Education Systems Can Become Great in the Decade Ahead'. Report on the International Education Roundtable, July 7th.

Barber, M. (1997). The Learning Game: Arguments for an Education Revolution. Updated ed. London: Indigo.

Barger, N.J. and Kirby, L.K. (1995). The Challenge of Change in Organizations: Helping Employees Thrive in the New Frontier. 1st ed. Palo Alto, Calif: Davies-Black Pub.

Barnett, B.G., Shoho, A.R. and Oleszewski, A.M. (2012). 'The Job Realities of Beginning and Experienced Assistant Principals'. Leadership and Policy in Schools 11 (1): 92–128. https://doi.org/10.1080/15700763.2011.611924.

Bassey, M. (1990). 'On the Nature of Research in Education (Part 2)'. Research Intelligence 37: 39–44.

Basten, D. and Haamann, T. (2018). 'Approaches for Organizational Learning: A Literature Review'. SAGE Open 8 (3): 215824401879422. <u>https://doi.org/10.1177/2158244018794224</u>.

Beaver, J.K. and Weinbaum, E.H. (2015). 'State Test Data and School Improvement Efforts'. Educational Policy 29 (3): 478–503. <u>https://doi.org/10.1177/0895904813510774</u>.

Beebe, S. A., & Masterson, J. T. (2003). Communicating in small groups. Pearson Education Inc. Boston: Massachusetts.

Bell, J., and Woolner, P. (2012). 'Developing and Using Questionnaires'. In Research Methods in Educational Leadership and Management, edited by A.R.J. Briggs, M. Coleman, and M. Morrison, 266–80. London: SAGE Publishing.

Bell, J., Cheney, G., Hoots, C., Kohrman, J.S., Stidham, L. and Traynor, S. (2004). 'Comparative Similarities and Differences between Action Research, Participative Research, and Participatory Action Research'. <u>https://arlecchino.org/ildottore/mwsd/group2final-</u> <u>comparison.html</u>.

Bell, L., and Stevenson, H. (2006). Education Policy - Process, Themes and Impact. London: Routledge.

Bell, M. (2005). Peer Observation Partnerships in Higher Education. 1st ed. New South Wales: Higher Education Research and Development Society of Australia. Benton, T. and Craib, I. (2001). Philosophy of Social Science: The Philosophical Foundations of Social Thought. Traditions in Social Theory. Houndmills, Basingstoke, Hampshire; New York: Palgrave.

Berryman, M. (2013). 'Leaders' Use of Classroom Evidence to Understand, Evaluate and Reform Schooling for Indigenous Students'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 123–35. Emerald Group Publishing. https://doi.org/10.1108/S1474-7863(2013)0000014009.

Biesta, G. (2009). 'Good Education in an Age of Measurement: On the Need to Reconnect with the Question of Purpose in Education'. Educational Assessment, Evaluation and Accountability 21 (1): 33–46. <u>https://doi.org/10.1007/s11092-008-9064-9</u>.

Biesta, G. (2010). 'Pragmatism and the Philosophical Foundations of Mixed Methods Research1'. In SAGE Handbook of Mixed Methods in Social & Behavioural Research, by Abbas Tashakkori and Charles Teddlie, 95–118. 2455 Teller Road, Thousand Oaks California 91320 United States: SAGE Publications, Inc. <u>https://doi.org/10.4135/9781506335193.n4</u>.

Biesta, G. and Osberg, D. (2010). 'Complexity, Education and Politics from the Inside-Out and the Outside-In: An Introduction'. In *Complexity Theory and the Politics of Education*, 1–3. Leiden, The Netherlands: Brill. <u>https://doi.org/10.1163/9789460912405_002</u>.

Biesta, G. (2011). 'Disciplines and Theory in the Academic Study of Education: A Comparative Analysis of the Anglo-American and Continental Construction of the Field'. Pedagogy, Culture & Society 19 (2): 175–92. https://doi.org/10.1080/14681366.2011.582255.

Biesta, G., Allan, J. and Edwards, R. (2011). 'The Theory Question in Research Capacity Building in Education: Towards an Agenda for Research and Practice'. British Journal of Educational Studies 59 (3): 225–39. <u>https://doi.org/10.1080/00071005.2011.599793</u>.

Biesta, G. (2004). 'Education, Accountability, and the Ethical Demand: Can the Democratic Potential of Accountability Be Regained?' Educational Theory 54 (3): 233–50. https://doi.org/10.1111/j.0013-2004.2004.00017.x.

Blok, H., Sleegers, P. and Karsten, S. (2005). School Self-Evaluation in Primary Education. Amsterdam: SCO-Kohnstamm Instituut. Blok, H., Sleegers, P. and Karsten, S. (2008). 'Looking for a Balance between Internal and External Evaluation of School Quality: Evaluation of the SVI Model'. Journal of Education Policy 23 (4): 379–95. <u>https://doi.org/10.1080/02680930801923773</u>.

Bloomberg, L.D. and Volpe, M. (2008). Completing Your Qualitative Dissertation: A Roadmap from Beginning to End. Los Angeles: Sage Publications.

Bollen, R., Creemers, B., Hopkins, D., Lagerweij, N., Reynolds, D. and Stoll. L. (2005). Making Good Schools: Linking School Effectiveness and Improvement. 0 ed. Routledge. https://doi.org/10.4324/9780203975916.

Bonner, M., Koch, T. and Langmeyer, D. (2004). 'Organizational Theory Applied to School Reform: A Critical Analysis'. School Psychology International 25 (4): 455–71. https://doi.org/10.1177/0143034304048779.

Borden, L.M., and Perkins, D.F. (1999). 'Assessing Your Collaboration: A Self-Evaluation Tool'. Journal of Extension 37 (2): 78–83.

Bower, D.F. (2006). 'Sustaining School Improvement'. Complicity: An International Journal of Complexity and Education 3 (1). <u>https://doi.org/10.29173/cmplct8744</u>.

Boyatzis, R. (1998). Transforming Qualitative Information: Thematic Analysis and Copde Development. Thousand Oaks, CA: SAGE Publications.

Brady, A.M. (2016). 'The Regime of Self-Evaluation: Self-Conception for Teachers and Schools'. British Journal of Educational Studies 64 (4): 523–41. https://doi.org/10.1080/00071005.2016.1164829.

Bransford, J. and Darling-Hammond, L. (2005). Preparing teachers for a changing world: What teachers should learn and be able to do. Indianapolis, IN: Jossey-Bass, An Imprint of Wiley.

Braun, V. and Clarke, V. (2019). 'Thematic Analysis'. In Handbook or Research Methods in Health Social Sciences, 843–60. New Jersey: Springer.

Braun, V. and Clarke, V. (2006). 'Using Thematic Analysis in Psychology'. Qualitative Research in Psychology 3 (2): 77–101. <u>https://doi.org/10.1191/1478088706qp0630a</u>.

Breacháin, A.Ó, and O'Toole, L. (2013). 'Pedagogy or Politics: Cyclical Trends in Literacy and Numeracy in Ireland and Beyond'. Irish Educational Studies 32 (4): 401–19. https://doi.org/10.1080/03323315.2013.851441.

Breen, R.L. (2006). 'A Practical Guide to Focus-Group Research'. Journal of Geography in Higher Education 30 (3): 463–75. <u>https://doi.org/10.1080/03098260600927575</u>.

Brezicha, K., Bergmark, U. and Mitra, D.L. (2015). 'One Size Does Not Fit All: Differentiating Leadership to Support Teachers in School Reform'. Educational Administration Quarterly 51 (1): 96–132. <u>https://doi.org/10.1177/0013161X14521632</u>.

Briggs, A., Coleman, M. and Morrison, M. (2012). Research Methods in Educational Leadership and Management.

https://search.ebscohost.com/login.aspx?direct=true&scope=site&db=nlebk&db=nlabk&AN =1069103.

Brisson-Banks, C.V. (2010). 'Managing Change and Transitions: A Comparison of Different Models and Their Commonalities'. Library Management 31 (4/5): 241–52. https://doi.org/10.1108/01435121011046317.

British Educational Research Association [BERA] and Ethical Guidelines for Educational Research (2018), fourth edition, London. n.d. Ethical Guidelines for Educational Research, Fourth Edition, London. 4th ed. London.

Brough, S. and Tracey, S. (2013). 'Changing the Professional Culture of School Review: The Inside Story of ERO'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 96–107. Emerald Group Publishing.

Brown, G. (2013). 'AsTTle - A National Testing System for Formative Assessment. How The National Testing Policy Ended up Helping Schools and Teachers.' In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 43–56. Emerald Group Publishing.

Brown, M., McNamara, G., O'Brien, S., Skerritt, C., O'Hara, J., Faddar, J., Cinqir, S., Vanhoof, J., Figueiredo, M. and Kurum, G. (2020a). 'Parent and Student Voice in Evaluation and Planning in Schools'. Improving Schools 23 (1): 85–102. https://doi.org/10.1177/1365480219895167. Brown, Martin, Gerry McNamara, Shivaun O'Brien, Craig Skerritt, and Joe O'Hara. (2020b). 'Policy and Practice: Including Parents and Students in School Self-Evaluation'. *Irish Educational Studies* 39 (4): 511–34. <u>https://doi.org/10.1080/03323315.2020.1814839</u>.

Brown, M., McNamara, G. and O Hara, J. (2016a). 'Quality and the rise of value-added in education: The case of Ireland'. Policy Futures in Education, 14 (6):810-829 https://doi.org/10.1177%2F1478210316656506

Brown, M., McNamara, G., O'Hara, J. and O'Brien, S. (2016b). 'Exploring the Changing Face of School Inspections'. Eurasian Journal of Educational Research 16 (66): 1–35. https://doi.org/10.14689/ejer.2016.66.1.

Brown, M., McNamara., O Hara, J and O'Brien, S. (2017). 'Inspectors and the Process of Self-Evaluation in Ireland' In: Jacqueline Baxter (eds). School inspectors: Operational challenges in National Policy Contexts. London: Springer.

Brown, M., McNamara, G., O'Hara, J., O'Brien, S. and Faddar, J. (2018) 'Integrated coprofessional evaluation? Converging approaches to school evaluation across frontiers'. Australian Journal of Teacher Education, 43 (12):76-90. <u>https://doi.org/10.14221/ajte.2018v43n12.6</u>

Brydon-Miller, M., Prudente, M. and Aguja, S. (2017). 'Educational Action Research as Transformative Practice'. In The BERA/SAGE Handbook of Educational Research: Two Volume Set, by Dominic Wyse, Neil Selwyn, Emma Smith, and Larry Suter, 435–51. 1 Oliver's Yard, 55 City Road London EC1Y 1SP: SAGE Publications Ltd. https://doi.org/10.4135/9781473983953.n22.

Bubb, S., and Earley, P. (2008). From Self-Evaluation to School Improvement: The Importance of Effective Staff Development. Reading: CFBT Education Trust. http://www.leeds.ac.uk/educol/documents/168444.htm.

Burke, A. (2011). 'Group Work: How to Use Groups Effectively'. The Journal of Effective Teaching 11 (2): 87.

Burton, D.M., and Bartlett, S. (2009). Key Issues for Education Researchers. Book, Whole. London: SAGE Publications. <u>https://doi.org/10.4135/9781446269480</u>.

Bush, T. (2011). Theories of Educational Leadership and Management. 4th ed. London: SAGE Publications.
Butler, H. (2011). The Critical Friend: Facilitating Change and Wellbeing in School Communities. Camberwell, Victoria, Australia: ACER Press.

Cambridge Primary Review (2009). Towards a new Primary Curriculum. Part 1: Past and Present. The University of Cambridge.

Campbell, C. and Levin, B. (2009). 'Using Data to Support Educational Improvement'. Educational Assessment, Evaluation and Accountability 21 (1): 47–65. <u>https://doi.org/10.1007/s11092-008-9063-x</u>.

Carney, S. (2003). 'Globalisation, Neo-Liberalism and the Limitations of School Effectiveness Research in Developing Countries: The Case of Nepal'. Globalisation, Societies and Education 1 (1): 87–101. <u>https://doi.org/10.1080/1476772032000061833</u>.

Carr, W. (2006). 'Education without Theory'. British Journal of Educational Studies 54 (2): 136–59. <u>https://doi.org/10.1111/j.1467-8527.2006.00344.x</u>.

Carr, W. and Kemmis, S. (1986). Becoming Critical: Education, Knowledge, and Action Research. London; Philadelphia: Falmer Press.

Carroll, M. (2018). 'Understanding Curriculum: An Actor Network Theory Approach'. Studies in Self-Access Learning Journal, September, 247–61. https://doi.org/10.37237/090302.

Chapman, C. and Sammons, P. (2013). School Self-Evaluation for School Improvement: What Works and Why? University of Glasgow and University of Oxford: CFBT Education Trust.

Chapman, C. (2008). 'Towards a Framework for School-to-School Networking in Challenging Circumstances'. Educational Research 50 (4): 403–20. https://doi.org/10.1080/00131880802499894.

Charmaz, K. (2006). Constructing Grounded Theory. London; Thousand Oaks, Calif: Sage Publications.

Check, J.W. and Schutt, R.K. (2012). Research Methods in Education. Thousand Oaks, Calif: Sage Publications.

Chu, D., Strand, R. and Fjelland. R. (2003). 'Theories of Complexity'. Complexity 8 (3): 19–30. <u>https://doi.org/10.1002/cplx.10059</u>.

Cilliers, P. (2001). 'Boundaries, Hierarchies and Networks in Complex Systems'. *International Journal of Innovation Management* 05 (02): 135–47. https://doi.org/10.1142/S1363919601000312.

Clancy, S.G. (1997). 'STJs and Change: Resistance, Reaction, or Misunderstanding?' In Developing Leaders: Research and Applications in Psychological Type and Leadership Development, edited by C. Fitzgerald and L.K. Kirby, 415–38. Palo Alto, Calif: Consulting Psychologists Press.

Clarke, P., Ainscow, M. and West, M. (2009). 'Learning from Difference'. In, edited by A. Harris and J. Chrispeels, 77–89. London: Routledge.

Cochran-Smith, M. and Lytle, S.L. (1999). 'Relationships of Knowledge and Practice: Teacher Learning in Communities'. Review of Research in Education 24: 249. https://doi.org/10.2307/1167272.

Coe, R. (2009). 'School Improvement: Reality and Illusion'. British Journal of Educational Studies 57 (4): 363–79.

Coghlan, D. (2019). Doing Action Research in Your Own Organisation. London: SAGE Publications.

Cohen, L., Manion, L. and Morrison, K. (2011). Research Methods in Education. 7th ed. New York: Routledge.

Cohen, Louis, Manion, L. and Morrison, K. (2007). Research Methods in Education. 6th ed. London: Routledge.

Coleman, J.S. (1995). 'Equality of Educational Opportunity (COLEMAN) Study (EEOS), 1966: Version 3'. ICPSR - Interuniversity Consortium for Political and Social Research. https://doi.org/10.3886/ICPSR06389.V3.

Coleman, M. (2012). 'Interviews'. In Research Methods in Educational Leadership and Management, edited by A.R.J. Briggs, M. Coleman, and M. Morrison, 3rd ed., 250–65. London: SAGE Publishing.

Cooksey, R.W., and McDonald, G. (2011). Surviving and Thriving in Postgraduate Business Research. Prahran, Vic.: Tilde University Press.

Coolahan, J., Drudy, S., Hogan, P., Hyland, A. and McGuinness, S. (2017). Towards a Better Future: A Review of the Irish School System. Ballinglanna, Glounthaune, Co. Cork: IPPN.

Corbett, J., and Redding, S. (2017). Using Needs Assessments for School and District Improvement: A Tactical Guide. Washington: Council of Chief State School Officers.

Costa, A., and Kallick, B. (1993). 'Through the Lens of a Critical Friend'. Educational Leadership 51 (2): 49–51.

Costley, C., Elliott, G.C. and Gibbs, P. (2010). Doing Work Based Research: Approaches to Enquiry for Insider-Researchers. Book, Whole. Los Angeles: Sage.

Coghlan, D. (2019). Doing Action Research in Your Own Organisation. 5th edition. Thousand Oaks, CA: SAGE Publications.

Craig, I. (2017). Toxic Leadership. London: Bloomsbury.

Craig, C. (2012). "Butterfly under a pin": An emergent teacher image amid mandated curriculum reform. The Journal of Educational Research, 105(2), 90-101.

Creemers, B.P.M. and Reezigt, G. (2005). 'Linking School Effectiveness and School Improvement: The Background and Outline of the Project'. School Effectiveness and School Improvement 16 (4): 359–71. <u>https://doi.org/10.1080/09243450500234484</u>.

Creemers, B.P.M. and Kyriakides, L. (2010). 'Using the Dynamic Model to Develop an Evidence-Based and Theory-Driven Approach to School Improvement'. Irish Educational Studies 29 (1): 5–23. https://doi.org/10.1080/03323310903522669.

Creswell, J.W. (1994). Research Design: Qualitative & Quantitative Approaches. Thousand Oaks, Calif: Sage Publications.

Creswell, J.W. (2014). Research Design: Qualitative, Quantitative and Mixed Methods Approaches. 4th ed. Thousand Oaks, California: SAGE Publishing.

Cromien, S. (2000). 'Review of Department's Operations, Systems and Staffing Needs'. Dublin: Department of Education and Science.

Csernica, J., Hanyka, M., Hyde, D., Shooter, S., Toole, M. and Vigeant, M. (2002). Practical guide to teamwork, version 1.1. College of Engineering, Bucknell University.

Cunningham, R. (2001). 'Complexity Theory and School Improvement: Some Possible Connections'. In. Edinburgh University.

http://www.leeds.ac.uk/educol/documents/00003214.doc.

Cunningham, J. (2009). Unlikely Radicals: Irish Post-Primary Teachers and the Asti, 1909-2009. Book, Whole. Cork: Cork University Press.

Dahler-Larsen, P. (2006). Evaluation Culture: A Concept Is Created. Odense: Syddansk Universitetsforlag.

Government of Ireland (2018) Data Protection Act

Davies, L. (2001). 'School Effectiveness: Fracturing the Discourse'. British Educational Research Journal 27 (4): 509–10.

Davies, D., Rudd, P. and National Foundation for Educational Research in England and Wales. (2001). Evaluating School Self-Evaluation. Slough: NFER.

Day, C. (1993). 'Reflection: A Necessary but Not Sufficient Condition for Professional Development'. British Educational Research Journal 19 (1): 83–93.

Day, C. and Sammons, P. (2014). Successful School Leadership. Berkshire: Education Development Trust. <u>http://www.educationdevelopmenttrust.com/our-research-and-insights/research/successful-school-leadership</u>.

De Coster, I., Birch, P., Birch, S. and Colclough, O. (2015). 'Assuring Quality in Education: Policies and Approaches to School Evaluation in Europe'. Luxembourg: Publication's office of the European Union.

De Crauwe, A. and Naidoo, J. (2004). School Evaluation for Quality Improvement. Paris, France: UNESCO: International Institute for Educational Planning.

De Vos, A.S., Strydom, H., Fouche, C.B. and Delport, C.S.L. (2007). Research at Grassroots for Social Sciences and Human Service Professions. 3rd ed. Pretoria: Van Schaik.

Demetriou, D. and Kyriakides, L. (2012). 'The Impact of School Self-Evaluation upon Student Achievement: A Group Randomisation Study'. Oxford Review of Education 38 (2): 149–70. <u>https://doi.org/10.1080/03054985.2012.666032</u>. Demski, D. (2014). 'Which Data Do Principals and Teachers Use to Inform Their Practice? Evidence From Germany with a Focus on the Influence of School Culture'. In Using Data in Schools to Inform Leadership and Decision Making, edited by A.J. Bowers, A.R. Shoho, and B.G. Barnett, 121–50. Charlotte, NC: Information Age.

Denzin, N.K. and Lincoln, Y.S. (2018). The Sage Handbook of Qualitative Research. Fifth. Book, Whole. Los Angeles: Sage.

Department of Education and Science (1999), 'Department of Education and Science Annual Report 1999'. Department of Education and Science. <u>Department of Education and Science</u> <u>Annual Report 1999 (edepositireland.ie)</u>

Department of Education and Skills. (2012). Circular 0039/2012 – Implementation of School Self-Evaluation. Dublin: Department of Education and Skills.

Department of Education and Skills. (2012). An Introduction to School Self-Evaluation of Teaching and Learning in Primary Schools. Dublin: Inspectorate.

Department of Education and Skills. (2012). School Self-Evaluation Guidelines for Primary Schools. Dublin: Inspectorate.

Department of Education and Skills. (2013). 'A Guide to Whole School-Evaluation – Management, Leadership and Learning in Primary Schools. Evaluation Support and Research Unit Inspectorate

Department of Education and Science. (2003). 'Looking at Our School: An Aid to Self-Evaluation in Second Level Schools'. Dublin: Stationery Office.

Department of Education and Science. (2004). A Brief Description of the Irish Education System. Dublin: Communications Unit.

Department of Education and Skills. (2011). Literacy and Numeracy for Learning and Life: The National Strategy to Improve Literacy and Numeracy among Young People 2011-2020. Dublin: Department of Education and Skills.

Department of Education and Skills. (2015). 'Advancing School Autonomy in the Irish School System'. Consultation Paper.

Department of Education and Skills. (2016). Chief Inspector's Report: January 2013-July 2016. Dublin: The Inspectorate.

Department of Education and Skills. (2016). Looking at Our School 2016: A Quality Framework for Primary Schools. Dublin: The Inspectorate.

Department of Education and Skills. (2016). School Self-Evaluation Guidelines Primary 2016-2020: Inspectorate Guidelines for Schools. Dublin: The Inspectorate.

Department of Education and Skills. (2018). Main Features of Budget 2019: Education and Skills, Information Note. Dublin: Stationery Office.

Department of Education and Skills. (2019). Primary Language Curriculum. Dublin: Department of Education and Skills.

Devos, G. (1998). 'Conditions and Caveats for Self-Evaluation: The Case of Secondary Schools'. Presented at the Annual Meeting of the American Educational Research Association, San Diego, USA.

Dewey, J. (1922/2008). Education as Engineering. Reproduced in *Journal of Curriculum Studies* 41(1): 1-5

Dewey, J. (1933). How We Think. New York: Heath and Co.

Dewey, J. (1938/1998). The Problem of Logical Subject Matter. In Hickman, L.A. and Alexander, T.M. *The Essential Dewey, Volume 2: Ethics, Logic, Psychology*. Bloomington: Indiana University Press.

Dillman, D.A., Smyth, J.D. and Christian, L.M. (2014). Internet, Phone, Mail, and Mixed-Mode Surveys: The Tailored Design Method. 4th edition. Hoboken: Wiley.

Dillon, S. (2012). 'The Way I See It Is... Whole-School Evaluation in Irish Post-Primary Schools from the Perspectives of Principals, Teachers, Parents and Students'. Doctor of Education thesis, Dublin: Dublin City University.

Dixon, P, and Woolner, P. (2012). 'Quantitative Data Analysis: Using SPSS'. In Research Methods in Educational Leadership and Management, edited by A.R.J. Briggs, M. Coleman, and M. Morrison, 340–62. SAGE Publications.

DuFour, R. (2004). 'What Is a Professional Learning Community?' Educational Leadership, Schools as Learning Communities, 61 (8): 6–11.

Duignan, P. (2010). Educational Leadership - Key Challenges and Ethical Tensions. New York: Cambridge University Press.

Durning, S.J. and Artino, A.R. (2011). 'Situativity Theory: A Perspective on How Participants and the Environment Can Interact: AMEE Guide No. 52'. Medical Teacher 33 (3): 188–99. <u>https://doi.org/10.3109/0142159X.2011.550965</u>.

Dussault, M. and Barnett, B.G. (1996). 'Peer-assisted Leadership: Reducing Educational Managers' Professional Isolation'. Journal of Educational Administration 34 (3): 5–14. https://doi.org/10.1108/09578239610118848.

Dyson, L. (2018). 'Drawing on Theoretical Knowledge to Build Evaluation Capacity'. Evaluation Matters—He Take Tow Tea Aromatawai 4 (July): 1. https://doi.org/10.18296/em.0032.

Dyson, L. (2020). 'Walking on a Tightrope: Agency and Accountability in Practitioner Inquiry in New Zealand Secondary Schools'. Teaching and Teacher Education 93 (July): 103075. <u>https://doi.org/10.1016/j.tate.2020.103075</u>.

Earl, L.M. (2013). 'Looking Forward: Evaluation in New Zealand Education'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 147–58. Emerald Group Publishing.

Earl, L.M. and Katz, S. (2006). Leading Schools in a Data-Rich World: Harnessing Data for School Improvement. Thousand Oaks, Calif: Corwin Press.

Earley, P. and Greany, T. (eds.) (2017). 'Toxic Leadership'. In School Leadership and School Reform, 182–90. London: Bloomsbury.

Earley, P. and Bubb, S. (2013). 'A Day in the Life of New Headteachers: Learning from Observation'. Educational Management Administration & Leadership 41 (6): 782–99. https://doi.org/10.1177/1741143213494189.

Educational Research Centre (2013) Gerry Shiel, Brían Merriman, Judith Cosgrove, Gráinne Moran, Ireland) St. Patrick's College, Dublin, and R. Perkins. 2013. Learning for Life: The Achievements of 15-Year-Olds in Ireland on Mathematics, Reading Literacy and Science in PISA 2012.

Ehrensal, P. (2015). 'Waiting for Super leader: Leadership as Anti-Resource Discourse'. Journal of Educational Administration and History 47 (1): 68–83. https://doi.org/10.1080/00220620.2015.974147. Elliot, J. (1991). Action Research for Educational Change. Buckingham: Open University Press.

Emstad, A.B. (2011). 'The Principal's Role in the Post-Evaluation Process. —How Does the Principal Engage in the Work Carried out after the Schools Self-Evaluation?' Educational Assessment, Evaluation and Accountability 23 (4): 271–88. <u>https://doi.org/10.1007/s11092-011-9128-0</u>.

Ennis, R. and Philosophy Documentation Centre. (1991). 'Critical Thinking: A Streamlined Conception'. Teaching Philosophy 14 (1): 5–24. <u>https://doi.org/10.5840/teachphil19911412</u>.

European Commission/EACEA/Eurydice. (2015). Assuring Quality in Education: Policies and Approaches to School Evaluation in Europe. Eurydice Report. Luxembourg: Publications Office of the European Union.

Everson, K.C. (2017). 'Value-Added Modelling and Educational Accountability: Are We Answering the Real Questions?' Review of Educational Research 87 (1): 35–70. https://doi.org/10.3102/0034654316637199.

Faddar, J., Vanhoof, J. and De Maeyer, S. (2017). 'School Self-Evaluation Instruments and Cognitive Validity. Do Items Capture What They Intend To?' *School Effectiveness and School Improvement* 28 (4): 608–28. <u>https://doi.org/10.1080/09243453.2017.1360363</u>.

Faddar, J., Vanhoof, J. and De Maeyer, S. (2017). 'Instruments for School Self-Evaluation:
Lost in Translation? A Study on Respondents' Cognitive Processing'. Educational
Assessment, Evaluation and Accountability 29 (4): 397–420. <u>https://doi.org/10.1007/s11092-017-9270-4</u>.

Faddar, J., Vanhoof, J. and De Maeyer, S. (2018). 'School Self-Evaluation: Self-Perception or Self-Deception? The Impact of Motivation and Socially Desirable Responding on Self-Evaluation Results'. School Effectiveness and School Improvement 29 (4): 660–78. https://doi.org/10.1080/09243453.2018.1504802.

Fadhel, K. (2002). 'Positivist and Hermeneutic Paradigm, A Critical Evaluation under Their Structure of Scientific Practice'. The Sosland Journal, 21–28.

Fancy, H. (2007). 'Schooling Reform: Reflections on the New Zealand Experience'. In International Handbook of School Effectiveness and Improvement, edited by T. Townsend, 1:325–38. Netherlands: Springer. Farrell, C. and Morris, J. (2003). 'The 'Neo-Bureaucratic' State: Professionals, Managers and Professional Managers in Schools, General Practices and Social Work'. Organization 10 (1): 129–56. https://doi.org/10.1177/1350508403010001380.

Farley-Ripple, E.N. and Buttram, J.L. (2014). 'Developing Collaborative Data Use through Professional Learning Communities: Early Lessons from Delaware'. Studies in Educational Evaluation 42 (September): 41–53. <u>https://doi.org/10.1016/j.stueduc.2013.09.006</u>.

Felder, R.M. and Brent, R. (2001). Effective strategies for cooperative learning. Journal of Cooperation & Collaboration in College Teaching, 10(2), 69–75.

Fenstermacher, G.D. (1994). 'The Knower and the Known: The Nature of Knowledge in Research on Teaching'. Review of Research in Education 20: 3–56.

https://doi.org/10.2307/1167381.Fenwick, Tara. 2012. 'Complexity Science and Professional Learning for Collaboration: A Critical Reconsideration of Possibilities and Limitations'. *Journal of Education and Work* 25 (1): 141–62. https://doi.org/10.1080/13639080.2012.644911.

Fidan, T, and Öztürk, L. (2015). 'The Relationship of the Creativity of Public and Private School Teachers to Their Intrinsic Motivation and the School Climate for Innovation'. Procedia-Social and Behavioural Sciences, no. 195: 905–14.

Fidan, T and Balci, A. (2017). 'Managing Schools as Complex Adaptive Systems: A
Strategic Perspective'. International Electronic Journal of Elementary Education 10 (1): 11–
26. <u>https://doi.org/10.26822/iejee.2017131883</u>.

Fink, D. (2005). Leadership for Mortals - Developing and Sustaining Leaders of Learning. London: Paul Chapman Publishing.

Finn, C. (2012). 'Understanding PISA and What It Tells Us About Educational Standards in Ireland'. Dublin: National Economic and Social Council.

Floyd, A. (2012). 'Narrative and Life History'. In Research Methods in Educational
Leadership and Management, edited by A.R.J. Briggs, M. Coleman, and M. Morrison, 223–
35. SAGE Publications.

Fludernik, M. ed. (2012). Beyond Cognitive Metaphor Theory: Perspectives on Literary Metaphor. Routledge. <u>https://doi.org/10.4324/9780203815809</u>.

Franco, L. A. and Nielsen, M.F. (2018). 'Examining Group Facilitation in Situ: The Use of Formulations in Facilitation Practice'. Group Decision and Negotiation 27 (5): 735–56. https://doi.org/10.1007/s10726-018-9577-7.

Freddano, M. and Siri, A. (2012). 'Teacher Training for School Self-Evaluation'. Procedia -Social and Behavioural Sciences 69 (December): 1142–49. https://doi.org/10.1016/j.sbspro.2012.12.044.

Freiberg, H.J. (2005). School Climate: Measuring, Improving and Sustaining Healthy Learning Environments. 0 ed. Routledge. https://doi.org/10.4324/9780203983980.

Frye, A.W. and Hemmer, P.A. (2012). 'Program Evaluation Models and Related Theories: AMEE Guide No. 67'. Medical Teacher 34 (5): e288–99. https://doi.org/10.3109/0142159X.2012.668637.

Fullan, M. (2015). Leading in a Culture of Change. 5th ed. London: Routledge.

Fullan, M. (2001). Leading in a Culture of Change. 1st ed. San Francisco: Jossey-Bass.

Fullan, M. (2016). The NEW Meaning of Educational Change. Fifth edition. New York, NY: Teachers College Press.

Fullan, M., Hill, P.W. and Crévola, C. (2006). Breakthrough. Thousand Oaks, Calif.: [Melbourne, Vic]: [Toronto]: Corwin Press; NSDC; Ontario Principals' Council.

Fullan, M. (1993). Change forces: Probing the depths of educational reform. London: Falmer Press.

Garner, P. (2010). Emotional competence and its influences on teaching and learning. Educational Psychology Review, 22(3), 297-321

Garvin, D., Edmondson, A. and Gino, F. (2008). 'The Learning Organisation Survey: Is Yours a Learning Organisation?' Harvard Business Review, no. March: 109–16.

Gibson, J.E. (2012). 'Interviews and Focus Groups with Children: Methods That Match Children's Developing Competencies'. Journal of Family Theory & Review 4 (2): 148–59. https://doi.org/10.1111/j.1756-2589.2012.00119.x.

Gilmour, L. (2002). Collins Dictionary & Thesaurus. Glasgow: HarperCollins.

Given, H., Kuhn, L., LeeKeenan, D., Mardell, B., Redditt, S. and Twombly, S. (2010). Changing school culture: Using documentation to support collaborative inquiry. Theory Into Practice 49(1), 36-46.

Glanz, J. (2014). Action Research: An Educational Leader's Guide to School Improvement. New York: Rowman and Littlefield.

Gleeson, J. (2004). 'Cultural and Political Contexts of Irish Post Primary Curriculum: Influences, Interests and Issues'. In In Curriculum and Ideology: Irish Experiences, International Perspectives, edited by C Sugrue, 101–40. Dublin: Liffey Press.

Gleeson, J. (2010). Curriculum in Context: Partnership, Power and Praxis in Ireland. Oxford; New York: Peter Lang.

Gleeson, J. and Ó Donnabháin, J. (2009). 'Strategic Planning and Accountability in Irish Education'. Irish Educational Studies 28 (1): 27–46. https://doi.org/10.1080/03323310802597291.

Glenn, M. (2011). 'Developing Holistic Practice through Reflection, Action and Theorising'.
Educational Action Research 19 (4): 489–502.
https://doi.org/10.1080/09650792.2011.625694.

Goddard, W. and Melville, S. (2011). Research Methodology. Kenwyn, South Africa: Juta & Co.

Goldstein, H. (2011). Multilevel Statistical Models. Hoboken, N.J.: Wiley. http://rbdigital.oneclickdigital.com.

Goleman, D. (1996). Emotional Intelligence. London: Bloomsbury.

Government of Ireland. (1998). 'Education Act'. Dublin: Stationery Office.

Government of Ireland. (2018). Project Ireland: National Development Plan 2018-2027. Dublin: Department of Public Expenditure and Reform.

Goyal, N. and Howlett, M. (2019). 'Combining Internal and External Evaluations within a Multilevel Evaluation Framework: Computational Text Analysis of Lessons from the Asian Development Bank'. Evaluation 25 (3): 366–80. <u>https://doi.org/10.1177/1356389019827035</u>.

Grant, C. and Osanloo, A. (2014). 'Understanding, Selecting and Integrating a Theoretical Framework in Dissertation Research: Creating the Blueprint for Your "House". Administrative Issues Journal 4 (2): Article 4.

Greenwood, D. J. and Levin, M. (2006). Introduction to Action Research: Social Research for Social Change. Book, Whole. Thousand Oaks: SAGE Publications.

Gronn, P. (2016). 'Fit for Purpose No More?' Management in Education 30 (4): 168–72. https://doi.org/10.1177/0892020616665062.

Groves, R.M., Fowler, F.J.J., Couper, M.P., Lepkowski, J.M., Singer, E. and Tourangeau, R. (2009). Survey Methodology. 2nd ed. Wiley Series in Survey Methodology. Hoboken, N.J: Wiley.

Guest, G., MacQueen, K.M. and Namey, E.E. (2012). Applied Thematic Analysis. Los Angeles: Sage Publications.

Guhn, M. (2009). Insights from successful and unsuccessful implementations of school reform programs. Journal of Educational Change, 10(4), 337-363.

Gunter, H., Hall, D. and Bragg, J. (2013). 'Distributed Leadership: A Study in Knowledge Production'. Educational Management Administration & Leadership 41 (5): 555–80. https://doi.org/10.1177/1741143213488586.

Gurr, D. and Huerta, M. (2013). 'The Role of the Critical Friend in Leadership and School Improvement'. Procedia - Social and Behavioural Sciences 106 (December): 3084–90. https://doi.org/10.1016/j.sbspro.2013.12.356.

Guskey, T.R. (2002). 'Does It Make a Difference? Evaluating Professional Development.' Educational Leadership: Journal of the Department of Supervision and Curriculum Development 59 (6): 45–51.

Gustafsson J., Ehren, M., Conyngham, G., McNamara, G., Altrichter, H., and O'Hara, J. (2015) 'From inspection to quality: Ways in which school inspection influences change in schools'. *Studies in Educational Evaluation*, 47 :47-57

Hall, C. and Noyes, A. (2009). 'School Self-evaluation and Its Impact on Teachers' Work in England'. Research Papers in Education 24 (3): 311–34. https://doi.org/10.1080/02671520802149873. Hall, G.E. and Hord, S.M. (2015). Implementing Change: Patterns, Principles and Potholes.4th ed. New Jersey: Pearson.

Hannay, L. and Earl, L. (2012). School district triggers for reconstructing professional knowledge. Journal of Educational Change, 13(3), 311-326.

Hammersley, M. (2013). The Myth of Research-Based Policy & Practice. http://public.ebookcentral.proquest.com/choice/publicfullrecord.aspx?p=1191082.

Hammick, M., Olckers, L. and Campion-Smith, C. (2009). 'Learning in Interprofessional Teams: AMEE Guide No 38'. Medical Teacher 31 (1): 1–12. https://doi.org/10.1080/01421590802585561.

Hamzah, M.I.M. and Tahir, H.B.M. (2013). 'A Glimpse into School Self-Evaluation in Malaysia: Are We Doing The Right Things? Or Are We Doing the Things Right?' Asian Social Science 9 (12): p50. <u>https://doi.org/10.5539/ass.v9n12p50</u>.

Hanlon, G. (1998). 'Professionalism as Enterprise: Service Class Politics and the Redefinition of Professionalism'. Sociology 32 (1): 43–63. https://doi.org/10.1177/0038038598032001004.

Haque, S. (2015). 'Organizational Theories: A Comparative Analysis of Key Perspectives of Rational, Natural and Open System Views of Organization'. Case Studies Journal 4 (9). http://www.casestudiesjournal.com/Volume%204%20Issue%209%20Paper%205.pdf.

Hargreaves, A. (1998). 'The Emotional Politics of Teaching and Teacher Development: With Implications for Educational Leadership'. International Journal of Leadership in Education 1 (4): 315–36. <u>https://doi.org/10.1080/1360312980010401</u>.

Haroon, M., Ranmal, R., McElroy, H. and Dudley, J. (2015). 'Developing Clinical
Guidelines: How Much Rigour Is Required?' Archives of Disease in Childhood - Education
& Practice Edition 100 (2): 89–96. <u>https://doi.org/10.1136/archdischild-2013-305195</u>.

Harris, A. (2013). Distributed Leadership Matters. London: Corwin Press.

Harris, A. (2002). School Improvement: What's in It for Schools? What's in It for Schools? London; New York: Routledge/Falmer.

Harris, A. and Chrispeels, J. eds. (2009). Improving Schools and Educational Systems: International Perspectives. Contexts of Learning. Classrooms, Schools and Society. London; New York: Routledge.

Hart, N. (2007). Facilitator Tool Kit - A Guide for Helping Groups Get Results. University of Wisconsin: Office of Quality Improvement.

Hendriks, M., Doolaard, S. and Bosker, R.J. (2002). 'Using School Effectiveness as a Knowledge Base for Self-Evaluation in Dutch Schools: The ZEBO-Project'. In School Improvement through Performance Feedback, edited by A.J. Visscher and R. Coe, 114–42. Lisse: Swets and Zeitlinger.

Hensher, P. (2013). The Missing Ink: The Lost Art of Handwriting. 1st American pbk. ed. New York: Faber and Faber.

Hersey, P., Blanchard, K. and Johnson, D. (2015). Management of Organisational BehaviourLeading Human Resources. 10th ed. India: Pearson India education Services Ltd.

Hibberts, M.F. and Johnson, R.B. (2012). 'Mixed Methods Research'. In Research Methods in Educational Leadership and Management, edited by A.R.J. Briggs, M. Coleman, and M. Morrison, 3rd ed., 122–39. London: SAGE Publishing.

Hislop, H. (2013). 'Applying an Evaluation and Assessment Framework: An Irish Perspective'. Presented at the Irish Presidency of the Council of the European Union, Presidency Conference: Better Assessment and Evaluation to Improve Teaching and Learning, Dublin, March 20.

http://52.31.79.233/media/eupresidency/content/documents/Applying-an-Evaluation-and-Assessment-Framework---An-Irish-Perspective-(Dr-Harold-Hislop).pdf.

Hislop, H. (2017). 'A Co-Professional Approach to Inspection for Accountability and Improvement'. The Irish Journal of Education / Iris Eireannach an Oideachais 42: 3–27.

Hitchcock, G. and Hughes, D. (1995). Research and the Teacher: A Qualitative Introduction to School-Based Research. 2nd ed. London; New York: Routledge.

Hofman, R.H., Gray, J.M. and Hofman, W.H. (2010). 'Institutional Contexts and International Performances in Schooling: Comparing Patterns and Trends Over Time in International Surveys'. European Journal of Education 45 (1): 153–73. Hofman, R.H., Dijkstra, N.J. and Adriaan Hofman, W.H. (2009). 'School Self-Evaluation and Student Achievement'. School Effectiveness and School Improvement 20 (1): 47–68. https://doi.org/10.1080/09243450802664115.

Hofman, R.H., Dukstra, N.J. and Adriaan Hofman, W.H. (2005). 'School Self-Evaluation
Instruments: An Assessment Framework'. International Journal of Leadership in Education 8
(3): 253–72. https://doi.org/10.1080/13603120500088802.

Hogan, P. (2010). The New Significance of Learning - Imaginations Heartwork. London: Routledge.

Holland J.H. (1992). Adaptation in natural and artificial systems: an introductory analysis with applications to biology, control, and artificial intelligence. Cambridge, Mass: MIT Press.

Holland J.H. (1999). Emergence: from chaos to order. Reading, Mass: Perseus Books.

Holland, J.H. (2006). 'Studying Complex Adaptive Systems'. *Journal of Systems Science and Complexity* 19 (1): 1–8. <u>https://doi.org/10.1007/s11424-006-0001-z</u>.

Hopkins, D. (2001). School Improvement for Real. London: Routledge Falmer.

Hopkins, D., Ainscow, M. and West, M. (1994). School Improvement in an Era of Change. London: Cassell.

Hopkins, D. (ed.) (2005). The Practice and Theory of School Improvement. Dordrecht: Springer Netherlands. <u>https://doi.org/10.1007/1-4020-4452-6</u>.

Hopkins, E., Hendry, H., Garrod, F., McClare, S., Pettit, D., Smith, L. Burrell, H. and Temple, J. (2016). 'Teachers' Views of the Impact of School Evaluation and External Inspection Processes'. Improving Schools 19 (1): 52–61. https://doi.org/10.1177/1365480215627894.

Hord, S. (2008). Leading Professional Learning Communities. London: SAGE Publications.

Hord, S. (2012). Reclaiming Our Teaching Profession. New York: Teachers' College Press.

Howlett, M. and Rayner, J. (2006). 'Understanding the Historical Turn in the Policy Sciences: A Critique of Stochastic, Narrative, Path Dependency and Process-Sequencing Models of Policy-Making over Time'. Policy Sciences 39 (1): 1–18. https://doi.org/10.1007/s11077-005-9004-1. Hoy, W.K., Miskel, C.J. and Tarter, C.J. (2012). Educational Administration: Theory, Research, and Practice. 9th ed. New York: McGraw-Hill Humanities/Social Sciences/Languages.

Hudson, B., Hunter, D. and Peckham, S. (2019). 'Policy Failure and the Policy-Implementation Gap: Can Policy Support Programs Help?' Policy Design and Practice 2 (1): 1–14. https://doi.org/10.1080/25741292.2018.1540378.

Humada-Ludeke, A. (2013). The Creation of a Professional Learning Community for School Leaders: Insights on the Change Process from the Lens of the School Leader. Rotterdam: Sense Publishers.

Hyden, L.C. and Bulow, P.H. (2003). 'Who's Talking: Drawing Conclusions from Focus Groups-Some Methodological Considerations'. International Journal of Social Research Methodology 6 (4): 305–21. https://doi.org/10.1080/13645570210124865.

In Praxis Inc. (2006). Professional Learning Communities: An Exploration. Canada: Alberta Education.

Ingram, D., Louis, K.S. and Schroeder, R.J. (2004). 'Accountability Policies and Teacher Decision Making: Barriers to the Use of Data to Improve Practice'. Teachers College Record 106 (6): 1258–87. <u>https://doi.org/10.1111/j.1467-9620.2004.00379.x</u>.

'International Early Learning and Child Well-Being Study Assessment Framework'. (2021). OECD Education Working Papers 246. Vol. 246. OECD Education Working Papers. https://doi.org/10.1787/af403e1e-en.

Jacobs, S.D. (2016). 'The Use of Participatory Action Research within Education-Benefits to Stakeholders'. World Journal of Education 6 (3): p48. <u>https://doi.org/10.5430/wje.v6n3p48</u>.

Jaffer, K. (2010). 'School Inspection and Supervision in Pakistan: Approaches and Issues'. PROSPECTS 40 (3): 375–92. <u>https://doi.org/10.1007/s11125-010-9163-5</u>.

Jansen, J.D. (1995). 'Effective Schools?' Comparative Education 31 (2): 181–200. https://doi.org/10.1080/03050069529100.

Janssens, F., and van Amelsvoort, G. (2008). 'School Self-Evaluations and School Inspections in Europe: An Exploratory Study'. Studies in Educational Evaluation 34 (1): 15– 23. <u>https://doi.org/10.1016/j.stueduc.2008.01.002</u>. Jennings, M. (2007). Leading Effective Meetings, Teams, and Work Groups in Districts and Schools. Alexandria, VA: Association for Supervision and Curriculum Development.

Jimerson, J.B. and McGhee, M.W. (2013). 'Leading Inquiry in Schools: Examining Mental Models of Data-Informed Practice'. Current Issues in Education (Tempe, Ariz.) 16 (1).

Jimerson, J.B. (2014). 'Thinking about Data: Exploring the Development of Mental Models for "Data Use" among Teachers and School Leaders'. Studies in Educational Evaluation 42 (September): 5–14. <u>https://doi.org/10.1016/j.stueduc.2013.10.010</u>.

Johnson, R. B. and Onwuegbuzie, A.J. (2004). 'Mixed Methods Research: A Research Paradigm Whose Time Has Come'. Educational Researcher 33 (7): 14–26. https://doi.org/10.3102/0013189X033007014.

Johnston, R. and Badley, G. (1996). 'The Competent Reflective Practitioner'. Innovation and Learning in Education 2: 4–10.

Jones, M. and Harris, A. (2014). 'Principals Leading Successful Organisational Change: Building Social Capital through Disciplined Professional Collaboration'. Edited by Dr Kadir Beycioglu and Associate Professor Yasar Kondakci. Journal of Organizational Change Management 27 (3): 473–85. <u>https://doi.org/10.1108/JOCM-07-2013-0116</u>.

Jones, P. and Bradbury, L. (2018). Introducing Social Theory. Third Edition. Cambridge; Medford, MA: Polity Press.

Kacprzyk, J., Merigó, J., Nurmi, H. and Zadrozny, S. (2020). 'Multi-Agent Systems and Voting: How Similar Are Voting Procedures'. In Information Processing and Management of Uncertainty in Knowledge-Based Systems, 1237:172–84. Cham, Switzerland: Springer.

Kacprzyk, J., Merigó, J., Nurmi, H. and Zadrożny, S. (2020). 'Multi-Agent Systems and Voting: How Similar Are Voting Procedures'. In Information Processing and Management of Uncertainty in Knowledge-Based Systems, edited by Marie-Jeanne Lesot, Susana Vieira, Marek Z. Reformat, João Paulo Carvalho, Anna Wilbik, Bernadette Bouchon-Meunier, and Ronald R. Yager, 1237:172–84. Communications in Computer and Information Science. Cham: Springer International Publishing. <u>https://doi.org/10.1007/978-3-030-50146-4_14</u>. Karagiorgi, Y., Afantiti-Lamprianou, T., Alexandrou-Leonidou, V., Karamanou, M. and Symeou, L. (2018). "Out of the Box" Leadership: Action Research towards School Improvement'. Educational Action Research 26 (2): 239–57. https://doi.org/10.1080/09650792.2017.1310052.

Karagiorgi, Y., Nicolaidou, M. Yiasemis, C. and Georghiades, P. (2015). 'Emergent Data-Driven Approaches to School Improvement: The Journey of Three Schools through Self-Evaluation'. Improving Schools 18 (1): 69–82. <u>https://doi.org/10.1177/1365480214566914</u>.

Keeves, J.P. (ed.) (1997). Educational Research, Methodology and Measurement: An International Handbook. 2nd ed. Resources in Education. New York, N.Y: Pergamon.

Kember, D., Ha, T.S., Lam, B.H., Lee, A., NG, S., Yan, L. and Yum, J.C.K. (1997). 'The Diverse Role of the Critical Friend in Supporting Educational Action Research Projects'. Educational Action Research 5 (3): 463–81. <u>https://doi.org/10.1080/09650799700200036</u>.

Kemmis, S. and McTaggart, R. (eds) (1988) The Action Research Planner (second edition). Geelong, Victoria: Deakin University Press.

Keshavarz, N., Nutbeam, D., Rowling, L. and Khavarpour, F. (2010). 'Schools as Social Complex Adaptive Systems: A New Way to Understand the Challenges of Introducing the Health Promoting Schools Concept'. Social Science & Medicine 70 (10): 1467–74. https://doi.org/10.1016/j.socscimed.2010.01.034.

Kincheloe, J. (2012). Teachers as Researchers (Classic Edition): Qualitative Inquiry as a Path to Empowerment. Book, Whole. Florence: Routledge. https://doi.org/10.4324/9780203801550.

Kirwan, L. (2015). 'Mathematics Curriculum in Ireland: The Influence of Pisa on the Development of Project Maths'. International Electronic Journal of Elementary Education 8 (2): 317–32.

Kise, J. and Russell, B. (2008). Differentiated School Leadership - Effective Collaboration, Communication and Change through Personality Type. London: SAGE Publications.

Kivunja, C. and Kuyini, A.B. (2017). 'Understanding and Applying Research Paradigms in Educational Contexts'. International Journal of Higher Education 6 (5): 26. https://doi.org/10.5430/ijhe.v6n5p26. Klenowski, V. and Woods, A. (2013). 'Exploring the Possibilities and Methodological Challenges of Evaluation Practice: A Democratic Question'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 201–16. Emerald Group Publishing.

Kools, M. and Stoll, L. (2016). 'What Makes a School a Learning Organisation?' OECD Education Working Papers 137. Vol. 137. OECD Education Working Papers. https://doi.org/10.1787/5jlwm62b3bvh-en.

Krosnick, J.A. and Presser, S. (2010). 'Question and Questionnaire Design'. In Handbook of Survey Research, edited by P.V. Marsden and J.D. Wright, 2nd ed., 263–314. Bingley, UK: Emerald Group Publishing.

Krueger, R.A. (1997). Developing Questions for Focus Groups. Book, Whole. Thousand Oaks: SAGE Publications.

Krueger, R.A. and Casey, M.A. (2009). Focus Groups: A Practical Guide for Applied Research. 4th ed. Book, Whole. Los Angeles: SAGE.

Krueger, R.A. and Casey, M.A. (2015). Focus Groups: A Practical Guide for Applied Research. 5th ed. Book, Whole. Thousand Oaks, California: SAGE.

Kvale, S. and Brinkmann, S. (2009). Interviews: Learning the Craft of Qualitative Research Interviewing. 2nd ed. Los Angeles: Sage Publications.

Kyriakides, L. and Campbell, R.J. (2004). 'School Self-Evaluation and School Improvement: A Critique of Values and Procedures'. Studies in Educational Evaluation 30 (1): 23–36. https://doi.org/10.1016/S0191-491X(04)90002-8.

Ladd, V.J. (2009). 'School Nurses: Positive Deviant Leaders in the School Setting'. The Journal of School Nursing 25 (1): 6–14. <u>https://doi.org/10.1177/1059840508327628</u>.

Ladden, B. (2015). 'School Evaluation: An Exploration of the Impact of Evaluation Processes on the Staff in an Irish Post-Primary School'. Professional Doctorate in Education, Dublin City University.

Lai, M. (2013). 'A Thousand Flowers Blooming: The Implications of School Self-Review for Policy Developers'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 57–68. Emerald Group Publishing.

Lai, M.K., Wilson, A., McNaughton, S. and Hsiao, S. (2014). 'Improving Achievement in Secondary Schools: Impact of a Literacy Project on Reading Comprehension and Secondary School Qualifications'. Reading Research Quarterly 49 (3): 305–34. <u>https://doi.org/10.1002/rrq.73</u>.

Lai, M. and Kushner, S. (2013). 'Introduction to the New Zealand Case'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 29. Emerald Group Publishing.

Lam, T. and Bengo, P. (2003). 'A Comparison of Three Retrospective Self-Reporting
Methods of Measuring Change in Instructional Practice'. American Journal of Evaluation 24
(1): 65–80. <u>https://doi.org/10.1177/109821400302400106</u>.

Lambert, K. and O'Connor, J. (2018). 'Breaking and Making Curriculum from inside "Policy Storms" in an Australian Pre-service Teacher Education Course'. *The Curriculum Journal* 29 (2): 159–80. https://doi.org/10.1080/09585176.2018.1447302.

Lambrechts, W. and Hindson, J. (2016). Research and Innovation in Education for Sustainable Development: Exploring Collaborative Networks, Critical Characteristics and Evaluation Practices. Environment and School Initiatives – ENSI.

Lave, J. and Wenger, E. (1991). Situated Learning: Legitimate Peripheral Participation. Book, Whole. Cambridge [England]; New York; Cambridge University Press.

Le Fevre, D. (2010). 'Changing Tack: Talking about Change Knowledge for Professional Learning'. In Weaving Evidence, Inquiry and Standards to Build Better Schools, edited by H. Timperley and J. Parr, 71–91. Wellington: NZCER Press.

Leitch, R. and Day, C. (2000). 'Action Research and Reflective Practice: Towards a Holistic View'. Educational Action Research 8 (1): 179–93. https://doi.org/10.1080/09650790000200108.

Leithwood, K., Day, C., Sammons, P., Harris, A. and Hopkins, D. (2006). Successful School Leadership: What It Is and How It Influences Pupil Learning. Nottingham: University of Nottingham.

Leithwood, K. and Jantzi, D. (2008). 'Linking Leadership to Student Learning: The Contributions of Leader Efficacy'. Educational Administration Quarterly 44 (4): 496–528. https://doi.org/10.1177/0013161X08321501. Leithwood, K. and Mascall, B. (2008). 'Collective Leadership Effects on Student Achievement'. Educational Administration Quarterly 44 (4): 529–61. https://doi.org/10.1177/0013161X08321221.

Lesot, M.J. (ed.) (2020). Information Processing and Management of Uncertainty in Knowledge-Based Systems: 18th International Conference, IPMU 2020, Lisbon, Portugal, June 15-19, 2020: Proceedings. Part 1. Communications in Computer and Information Science 1237. Cham, Switzerland: Springer.

Lessard, S., Bareil, C., Lalonde, L., Duhamel, F., Hudon, E., Goudreau, J. and Lévesque, L. (2015). 'External Facilitators and Interprofessional Facilitation Teams: A Qualitative Study of Their Roles in Supporting Practice Change'. Implementation Science 11 (1): 97. https://doi.org/10.1186/s13012-016-0458-7.

Liamputtong, P. (2019). Handbook of Research Methods in Health Social Sciences. New York, NY: Springer Berlin Heidelberg.

Lieberman, A. and Miller, L. (2014). 'Unpacking Professional Learning Communities: Getting from Here to There'. In Reach the Highest Standard in Professional Learning -Learning Communities, edited by S. Hord and P. Roy, 1–17. London: SAGE Publications.

Lincoln, Y.S. and Guba, E.G. (1994). Naturalistic Inquiry. Newbury Park, CA: Sage Publications.

Lofthouse, R., Hall, E. and Wall, K. (2012). "Practitioner Research." In Research Methods in Educational Leadership and Management, edited by A. Briggs, M. Coleman, and M. Morrison, 170–187. London: Sage.

Loughran, J.J. (1996). Developing Reflective Practice: Learning about Teaching and Learning through Modelling. London: Falmer Press.

Louis C., Manion, L. and Morrison, K. (2018). Research Methods in Education. Vol. Eighth edition. New York: Routledge.

https://login.jproxy.nuim.ie/login?url=https://search.ebscohost.com/login.aspx?direct=true&d b=nlebk&AN=1614634&site=ehost-live.

Lynch, K., Grummell, B. and Devine, D. (2015). New Managerialism in Education. Hampshire: Palgrave MacMillan. MacBeath, J. (1999). Schools Must Speak for Themselves - The Case for School Self-Evaluation. London: Routledge.

MacBeath, J. (2005). Background, Principles and Key Learning in Self-Evaluation: A Guide for School Leaders. Nottingham: National College for School Leadership.

MacBeath, J. (2006). School Inspection and Self-Evaluation: Working with the New Relationship. 1st ed. London: Routledge.

MacBeath, J. (2008). 'Stories of Compliance and Subversion in a Prescriptive Policy Environment'. Educational Management Administration & Leadership 36 (1): 123–48. https://doi.org/10.1177/1741143207084064.

MacBeath, J. (2008). 'Leading Learning in the Self-Evaluating School'. School Leadership & Management 28 (4): 385–99. <u>https://doi.org/10.1080/13632430802292332</u>.

MacBeath, J. (2011). Learning In and Out of School: The Selected Works of John MacBeath. Book, Whole. Abingdon, Oxon: Routledge. <u>https://doi.org/10.4324/9780203154618</u>.

MacBeath, J., Laveault, D., Nusche, D. and Santiago, P. (2012). OECD Reviews of Evaluation and Assessment in Education: New Zealand 2011. OECD Reviews of Evaluation and Assessment in Education. OECD Publishing. <u>https://doi.org/10.1787/9789264116917-en</u>.

MacBeath, J. and McGlynn, A. (2002). Self-Evaluation: What's in It for Schools? What's in It for Schools? London: Routledge Falmer.

MacGilchrist, B. (2000). 'Improving Self-Improvement?' Research Papers in Education 15 (3): 325–38. <u>https://doi.org/10.1080/02671520050128795</u>.

Mackenzie, N. and Knipe, S. (2006). 'Research Dilemmas: Paradigms, Methods and Methodology'. Issues in Educational Research 16 (2): 193.

Mann, M.J. and Smith, M.L. (2013). 'Promoting Rigor-in-Practice through School Self-Evaluation: A Middle School's Experience with Model Development, Implementation, and Evaluation'. RMLE Online 37 (4): 1–15. <u>https://doi.org/10.1080/19404476.2013.11462107</u>.

Marion, R. (1999). The Edge of Organization: Chaos and Complexity Theories of Formal Social Systems. Thousand Oaks, Calif: Sage Publications.

Marsh, J.A. and Farrell, C.C. (2015). 'How Leaders Can Support Teachers with Data-Driven Decision Making: A Framework for Understanding Capacity Building'. Educational Management Administration & Leadership 43 (2): 269–89. https://doi.org/10.1177/1741143214537229.

Mathews, B.P., Ueno, A., Kekale, T., Repka, M., Lopez Pereira, Z. and Silva, G. (2001). 'Quality Training: Needs and Evaluation-Findings from a European Survey'. Total Quality Management 12 (4): 483–90. <u>https://doi.org/10.1080/09544120123778</u>.

McCleskey, J.A. (2014). 'Situational, Transformational, and Transactional Leadership and Leadership Development'. Journal of Business Studies Quarterly 5 (4): 117–30.

McGrath, G., Alfrey, L. and Jeanes, R. (2017). 'Community Connections in Health and Physical Education: A Reflection of Stakeholder Experiences'. *Asia-Pacific Journal of Health, Sport and Physical Education* 8 (2): 101–14. https://doi.org/10.1080/18377122.2017.1304157

McIntyre, D. (2005). 'Bridging the Gap between Research and Practice'. Cambridge Journal of Education 35 (3): 357–82. <u>https://doi.org/10.1080/03057640500319065</u>.

National College (2012) Facilitator Toolkit, Nottingham, National College for School Leadership.

McNamara, G. and O'Hara, J. (2005). 'Internal Review and Self-Evaluation — the Chosen Route to School Improvement in Ireland?' Studies in Educational Evaluation 31 (4): 267–82. https://doi.org/10.1016/j.stueduc.2005.11.003.

McNamara, G. and O'Hara, J. (2006). 'Workable Compromise or Pointless Exercise: School-Based Evaluation in the Irish Context'. Educational Management Administration & Leadership 34 (4): 564–82. <u>https://doi.org/10.1177/1741143206068218</u>.

McNamara, G. and O'Hara, J. (2008). 'The Importance of the Concept of Self-Evaluation in the Changing Landscape of Education Policy'. Studies in Educational Evaluation 34 (3): 173–79. <u>https://doi.org/10.1016/j.stueduc.2008.08.001</u>.

McNamara, G. and O'Hara, J. (2012). 'From Looking at Our Schools (LAOS) to Whole School Evaluation -Management, Leadership and Learning (WSE-MLL): The Evolution of Inspection in Irish Schools over the Past Decade'. Educational Assessment, Evaluation and Accountability 24 (2): 79–97. <u>https://doi.org/10.1007/s11092-012-9143-9</u>. McNamara, G., O'Hara, J. and Ní Aingléis, B. (2002). 'Whole-School Evaluation and Development Planning: An Analysis of Recent Initiatives in Ireland'. Educational Management & Administration 30 (2): 201–11. https://doi.org/10.1177/02611X02030002510.

Mc Namara, G., O Hara, J. and Boyle, R. (2008). 'Influences Shaping National Evaluation Policies: the case of Ireland'. The Evaluator, 14 :15-19

McNamara, G., O'Hara, J., Boyle, R. and O'Sullivan, C. (2009). 'Developing a Culture of Evaluation in the Irish Public Sector: The Case of Education'. Evaluation 15 (1): 101–12. https://doi.org/10.1177/1356389008095492.

McNamara, G., O'Hara, J., Lisi, P.L. and Davidsdottir, S. (2011). 'Operationalising Self-Evaluation in Schools: Experiences from Ireland and Iceland'. Irish Educational Studies 30 (1): 63–82. https://doi.org/10.1080/03323315.2011.535977.

McNamara, G., Brown, M., Gardezi, S., O'Hara, J., O'Brien, S. and Skerritt, C. (2021a). 'Embedding Self-Evaluation in School Routines'. *SAGE Open* 11 (4): 215824402110525. <u>https://doi.org/10.1177/21582440211052552</u>.

McNamara, G., Skerritt, C., O'Hara, J., O'Brien, S. and Brown, M. (2021b) 'For Improvement, Accountability, or the Economy? Reflecting on the Purpose(s) of School Self-Evaluation in Ireland'. *Journal of Educational Administration and History*, October, 1–16. <u>https://doi.org/10.1080/00220620.2021.1985439</u>.

McNaughton, S. (2011). Designing better schools for culturally and linguistically diverse children: A science of performance model for research. New York and London: Routledge.

McNaughton, S., Lai, M., Jesson, R. and Wilson, A. (2013). 'Evaluation in Effective Research-Practice Partnerships'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 69–80. Emerald Group Publishing.

McNaughton, S., Lai, M. and Hsiao, S. (2012). 'Testing the Effectiveness of an Intervention Model Based on Data Use: A Replication Series across Clusters of Schools'. School Effectiveness and School Improvement 23 (2): 203–28. https://doi.org/10.1080/09243453.2011.652126.

McNiff, J. (2017). Action Research - All You Need to Know. London: SAGE Publications.

McNiff, J. and Whitehead, J. (2002). Action Research: Principles and Practice. 2nd ed. Book, Whole. London; New York; Routledge Falmer.

Mennin, S. (2010). 'Self-Organisation, Integration and Curriculum in the Complex World of Medical Education: Self-Organisation, Integration and Curriculum'. Medical Education 44 (1): 20–30. <u>https://doi.org/10.1111/j.1365-2923.2009.03548.x</u>.

Mertens, D.M. (2014). Research and Evaluation in Education and Psychology: Integrating Diversity with Quantitative, Qualitative, and Mixed Methods.

Mertler, C.A. (2011). Action Research: Improving Schools and Empowering Educators. London: SAGE Publications.

Meuret, D. and Morlaix, S. (2003). 'Conditions of Success of a School's Self-Evaluation: Some Lessons of a European Experience'. School Effectiveness and School Improvement 14 (1): 53–71. <u>https://doi.org/10.1076/sesi.14.1.53.13867</u>.

Miles, M.B., Huberman, A.M. and Saldaña, J. (2014). Qualitative Data Analysis: A Methods Sourcebook. Third edition. Thousand Oaks, California: SAGE Publications, Inc.

Miller (2018). Nature of School Leadership: Palgrave Macmillan. https://link.springer.com/book/10.1007/978-3-319-70104-2.

Ministry of Education. (2003). 'April Quarterly: Planning for Better Student Outcomes'. Wellington, New Zealand: Ministry of Education.

Morçöl, G. and Wachhaus, A. (2009). 'Network and Complexity Theories: A Comparison and Prospects for a Synthesis'. Administrative Theory & Praxis 31 (1): 44–58. https://doi.org/10.2753/ATP1084-1806310103.

Morgan, D.L. (1988). Focus Groups as Qualitative Research. Qualitative Research Methods, v. 16. Newbury Park, Calif: Sage Publications.

Morrison, K. (2002). School Leadership and Complexity Theory. London: Routledge Falmer.

Mortimore, P., Sammons, P., Stoll, L., Lewis, D. and Ecob, R. (1988). School Matters: The Junior Years. Somerset: Open Books.

Muijs, D., and Reynolds, D. (2002). Effective Teaching. London: Chapman.

Muijs, D. (2012). "Surveys and sampling." In Research Methods in Educational Leadership and Management, edited by A. Briggs, M. Coleman, and M. Morrison, 140-154. London: Sage.

Mulford, B. (2003). 'OECD Commissioned Report. "School Leaders: Challenging Roles and Impact on Teacher and School Effectiveness". Australia: Faculty of Education: University of Tasmania.

Mulford, B. (2005). 'Accountability Policies and Their Effects'. In International Handbook of Educational Policy, edited by Nina Bascia, Alister Cumming, Amanda Datnow, Kenneth Leithwood, and David Livingstone, 13:281–94. Springer International Handbooks of Education. Dordrecht: Springer Netherlands. <u>https://doi.org/10.1007/1-4020-3201-3_14</u>.

Murphy, G. (2019). 'A Systematic Review and Thematic Synthesis of Research on School Leadership in the Republic of Ireland: 2008-2018'. Journal of Educational Administration 57 (6): 675–89.

Murphy, M. (2019). 'Ever Greater Scrutiny: Researching the Bureaucracy of Educational Accountability'. In Education Governance and Social Theory: Interdisciplinary Approaches to Research, edited by A. Olmedo and A. Wilkins, 193–207. London: Bloomsbury.

Mutch, C. (2013). 'Developing a Conceptual Framework for School Review'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 82–95. Emerald Group Publishing.

National Health and Medical Research Council (Australia), Australian Research Council, and Universities Australia. (2018) National Statement on Ethical Conduct in Human Research 2007 (Updated 2018).

National Research Council. (2003). Strategic Education Research Partnership. Edited by M.S. Donovan, A.K. Wigdor, and C.E. Snow. Washington, D.C.: National Academies Press. https://www.nap.edu/catalog/10670/strategic-education-research-partnership.

National Union of Teachers. (2005). NUT Education Statement: Meeting the Needs of All Children and Young People; Bringing Down Barriers. London: NUT.

Nayir, F. and McNamara, G. (2015). 'The Increasingly Central Role of School Self-Evaluation in Inspection Systems Across Europe: The Case of Ireland.' Turkish Journal of Education 3 (17343): 0. <u>https://doi.org/10.19128/turje.181075</u>. National Council for Curriculum and Assessment (NCCA), (2010). 'Curriculum Overload in Primary Schools – An overview of national and international experiences' Dublin. <u>Microsoft</u> <u>Word - FINAL_ncca_curr_overload_litreview_110210.doc</u>

National Council for Curriculum and Assessment (2007). Assessment Guidelines for Schools. Dublin

National Council for Curriculum and Assessment (2019). New Primary Language Curriculum. Dublin

Neil, P. and Johnston, J. (2005). 'An Approach to Analysing Professional Discourse in a School Self-Evaluation Project'. Research in Education 73 (1): 73–86. https://doi.org/10.7227/RIE.73.6.

Neuman, W. L. (2006). Social Research Methods: Qualitative and Quantitative Approaches. 6th ed. Boston: Pearson/AandB.

Nevo, D. (2010). 'Internal and External Evaluation'. In International Encyclopaedia of Education, edited by P. Peterson, E. Baker, and B. McGaw, 781–85. Oxford: Elsevier.

Ngan, M.Y., Lee, J.C.K. and Brown, G.T.L. (2010) 'Hong Kong principals' perceptions on changes in evaluation and assessment policies: they're not for learning'. Asian journal of educational research and synergy, 2(1), 36-46.

Nixon, Iain, KSA Partnership, and Higher Education Academy (Great Britain). (2008). Work-Based Learning: Impact Study. York: Higher Education Academy.

Noddings, N. (2003). Caring: A Feminine Approach to Ethics & Moral Education. 2nd ed. Book, Whole. Berkeley: University of California Press.

Noffke, S.E. and Somekh, B. (eds.) (2013). The SAGE Handbook of Educational Action Research. London: SAGE.

Nusche, D. and OECD (eds.) (2012). OECD Reviews of Evaluation and Assessment in Education: New Zealand 2011. OECD Reviews of Evaluation and Assessment in Education. Paris: OECD Publishing.

Nuttall, D.L., Goldstein, H., Prosser, R. and Rasbash, J. (1989). 'Differential School Effectiveness'. International Journal of Educational Research 13 (7): 769–76. https://doi.org/10.1016/0883-0355(89)90027-X. O'Brien S., McNamara G., O'Hara J. (2014) 'Critical facilitators: External supports for selfevaluation and improvement in schools'. *Studies in Educational Evaluation*, 43 :169-177

O'Brien, S., McNamara, G. and O'Hara, J. (2015). 'Supporting the Consistent Implementation of Self-Evaluation in Irish Post-Primary Schools'. Educational Assessment, Evaluation and Accountability 27 (4): 377–93. <u>https://doi.org/10.1007/s11092-015-9218-5</u>.

O'Brien, S., McNamara, G., O Hara, J., and Brown, M. (2017) Evaluation: Int. Journal of Theory, Research & Practice, 23 (1):61-79 <u>https://doi.org/10.1177%2F1356389016684248</u>

O'Brien, S., McNamara, G., O'Hara, J. and Brown, M. (2019). 'Irish Teachers, Starting on a Journey of Data Use for School Self-Evaluation'. Studies in Educational Evaluation 60 (March): 1–13. <u>https://doi.org/10.1016/j.stueduc.2018.11.001</u>.

O'Connor, P.P. (2001). 'The Policy, Process and Impact of Whole School Inspection at Primary Level in The Republic of Ireland from The Perspective of Some Inspectors and Teachers'. <u>https://doi.org/10.21954/OU.RO.00004DE2</u>.

O'Day, J. (2002). 'Complexity, Accountability, and School Improvement'. Harvard Educational Review 72 (3): 293–329. <u>https://doi.org/10.17763/haer.72.3.021q742t8182h238</u>.

OECD. (2009). PISA 2009 Assessment Framework: Key Competencies in Reading, Mathematics and Science. OECD - Programme for International Student Assessment. Paris: OECD.

OECD. (2013). 'Equity, Excellence and Inclusiveness in Education Policy Lessons from Around the World' By Andreas Schleicher in Synergies for Better Learning: An International Perspective on Evaluation and Assessment. OECD Reviews of Evaluation and Assessment in Education. OECD. <u>https://doi.org/10.1787/9789264190658-en</u>.

OECD. (2013). PISA 2012 Results: What Students Know and Can Do, Student Performance in Mathematics, Reading and Science. Vol. 1. Paris: OECD Publishing.

OECD. (2013). Synergies for Better Learning: An International Perspective on Evaluation and Assessment. OECD Reviews of Evaluation and Assessment in Education. OECD. https://doi.org/10.1787/9789264190658-en.

OECD. (2016). What Makes a School a Learning Organisation? A Guide for Policy Makers, School Leaders and Teachers. Paris: OECD Publishing.

Ofsted. (2006). 'Best Practice in Self-Evaluation'. Her Majesty's Inspectorate. http://dera.ioe.ac.uk/5988/1/Best%20practice%20in%20selfevaluation%20A%20survey%20of%20schools,%20colleges%20and%20local%20authorities

%20(PDF%20format).pdf.

Ofsted. (2013). 'Preparing a School Self-Evaluation Summary'. Her Majesty's Inspectorate. https://webarchive.nationalarchives.gov.uk/20120920182801/https://www.nsomerset.gov.uk/NR/rdonlyres/DA788F74-A2F1-4440-8FE9-B2DAF62B6AB7/0/GuidanceforSchoolSelfEvaluation.pdf.

Ofsted. (2019). 'Teachers' Awareness and Perceptions of Ofsted Teacher Attitude Survey 2019 Report'. Ofsted.

O'Hara, J., McNamara, G., Brown, M. and O'Brien, S. (2016). Notes on a Small Island: National Survey Results of Principals Attitudes Towards School Self-Evaluation (2014-2015). Dublin: (EQI) Centre for Evaluation, Quality and Inspection.

Oluwatomi, A.M. (2007). 'Teacher Attitude to Inspectors and Inspection: Quality Control'. International Journal of Educational Research 3 (2): 203–15.

Oolbekkink-Marchand, H.W., van der Steen, J. and Nijveldt, M. (2014). 'A Study of the Quality of Practitioner Research in Secondary Education: Impact on Teacher and School Development'. Educational Action Research 22 (1): 122–39. https://doi.org/10.1080/09650792.2013.854175.

Osberg, D. (2008). 'The Logic of Emergence: An Alternative Conceptual Space For Theorizing Critical Education'. *Journal of the Canadian Association for Curriculum Studies* 6 (1). <u>https://go.exlibris.link/qfyD7yZx</u>.

O'Sullivan, H. (2011). 'Leading and Managing Professional Learning in Schools - Chapter 8'. In Leading and Managing Schools, edited by H. Sullivan and J. Burnham. London: SAGE Publications.

Pal Baliyan, S. (2014). 'Evaluation of Onion Varieties for Productivity Performance in Botswana'. World Journal of Agricultural Research 2 (3): 129–35. <u>https://doi.org/10.12691/wjar-2-3-8</u>. Parr, J.M. and Timperley, H.S. (2013). 'Evaluation as a Double-Edged Sword: Building Schools' Evaluative Capability While Evaluating Their Efforts in Raising Achievement'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 108–22. Emerald Group Publishing.

Pasmore, W.A. (2011). 'Tipping the Balance: Overcoming Persistent Problems in
Organizational Change'. In Research in Organizational Change and Development, edited by
Abraham B. (Rami) Shani, Richard W. Woodman, and William A. Pasmore, 19:259–92.
Emerald Group Publishing Limited. <u>https://doi.org/10.1108/S0897-3016(2011)0000019011</u>.

Pasmore, W.A. and Woodman, R.W. (2017). 'The Future of Research and Practice in Organizational Change and Development'. In Research in Organizational Change and Development, edited by Abraham B. (Rami) Shani and Debra A. Noumair, 25:1–32. Emerald Publishing Limited. <u>https://doi.org/10.1108/S0897-301620170000025001</u>.

Patton, M.Q. (2002). Qualitative Research and Evaluation Methods. 3rd ed. Book, Whole. Thousand Oaks, Calif: Sage Publications.

Perryman, J. (2009). 'Inspection and the Fabrication of Professional and Performative Processes'. Journal of Education Policy 24 (5): 611–31. https://doi.org/10.1080/02680930903125129.

Pescarmona, I. (2010). Complex instruction: Managing professional development and school culture. Intercultural Education, 21(3), 219-227.

Phelps, R. and Hase, S. (2002). 'Complexity and Action Research: Exploring the Theoretical and Methodological Connections'. Educational Action Research 10 (3): 507–24. https://doi.org/10.1080/09650790200200198.

Plowright, D. and Godfrey, R. (2008). 'School Self-Evaluation: Can Head Teachers Meet the New Challenge?' International Studies in Educational Leadership 36: 35–58.

Potter, D., Reynolds, D. and Chapman, C. (2002). 'School Improvement for Schools Facing Challenging Circumstances: A Review of Research and Practice'. School Leadership & Management 22 (3): 243–56. <u>https://doi.org/10.1080/1363243022000020381</u>.

Powell, T.C. (2001). 'Competitive Advantage: Logical and Philosophical Considerations'. Strategic Management Journal 22 (9): 875–88. <u>https://doi.org/10.1002/smj.173</u>.

Power, M. (1997). The Audit Society: Rituals of Verification. Oxford, [England]; New York: Oxford University Press.

Priestley, M. and Biesta, G. (2014). *Reinventing the Curriculum: New Trends in Curriculum Policy and Practice*.

Punch, K. (2005). Introduction to Social Research: Quantitative and Qualitative Approaches.2nd ed. London; Thousand Oaks, Calif: SAGE.

Rauch, F., Zehetmeier, S. and Posh, P. (2019). 'Educational Action Research'. In Action Learning and Action Research: Genres and Approaches, edited by Outruns Zuber-Skerritt and Lesley Wood, 111–26. Emerald Publishing Limited. <u>https://doi.org/10.1108/978-1-78769-537-520191012</u>.

Ravitch, S.M. and Wirth, K. (2007). 'Developing a Pedagogy of Opportunity for Students and Their Teachers: Navigations and Negotiations in Insider Action Research'. Action Research 5 (1): 75–91. <u>https://doi.org/10.1177/1476750307072878</u>.

Reason, P. and Bradbury, H. (eds.) (2001). The Sage Handbook of Action Research. 1st ed. London: SAGE Publications.

https://books.google.ie/books/about/Handbook_of_Action_Research.html?id=v5TQfaZqZxE C&redir_esc=y.

Reeves, S., Peller, J., Goldman, J. and Kitty, S. (2013). 'Ethnography in Qualitative Educational Research: AMEE Guide No. 80'. Medical Teacher 35 (8): e1365–79. https://doi.org/10.3109/0142159X.2013.804977.

Rhodes, C. (1998). 'Book Reviews: Chris Argyris and Donald A. Schon (1996): Organizational Learning II: Theory, Method and Practice Reading, MA: Addison-Wesley, XXIX + 305 Pp, \$41.95 ISBN 0-201-62983-6'. Asia Pacific Journal of Human Resources 36 (1): 107–9. https://doi.org/10.1177/103841119803600112.

Robinson, V. and Timperley, H. (2013). 'School Improvement through Theory Engagement'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 163–77. Emerald Group Publishing Limited. <u>https://doi.org/10.1108/S1474-7863(2013)0000014010</u>.

Robson, C. (2002). Real World Research: A Resource for Social Scientists and Practitioner-Researchers. 2nd ed. Book, Whole. Oxford, Kamden, Mass; Blackwell Publishers.

Roby, D. (2011). 'Teacher Leaders Impacting School Culture'. Education 131 (4): 782-90.

Rosenholtz, S.J. (1991). Teacher's Workplace: The Social Organization of Schools. Research on Teaching Series. New York: Teachers College Press.

Ryan, K.E., Chandler, M. and Samuels, M. (2007). 'What should School-Based Evaluation Look Like?' Studies in Educational Evaluation 33 (3–4): 197–212. https://doi.org/10.1016/j.stueduc.2007.07.001.

Ryan, K.E. and Timmer, J. (2013). 'New Zealand Style School Review: A View from Outside'. In A Developmental and Negotiated Approach to School Self-Evaluation, 14:195–206. Emerald Group Publishing Limited. <u>https://doi.org/10.1108/S1474-7863(2013)0000014012</u>.

Sahlberg, P. (2009). Learning first: School accountability for a sustainable society. In K. D. Gariepy, B. L. Spencer, & J. C. Couture. 2009. 'Learning First: School Accountability for a Sustainable Society'. In Educational Accountability: Professional Voices from the Field, edited by K.D. Gariepy, B.L. Spencer, and J.C. Couture, 1–22. Rotterdam: Sense Publishers.

Samier, E.A. (2018). 'Education in a Troubled Era of Disenchantment: The Emergence of a New Zeitgeist'. Journal of Educational Administration and History 50 (1): 41–50. https://doi.org/10.1080/00220620.2017.1399865.

Sammons, P., Hillman, J. and Mortimore, P. (1995). 'Key Characteristics of Effective Schools: A Review of School Effectiveness Research'. London: Ofsted. <u>https://files.eric.ed.gov/fulltext/ED389826.pdf</u>.

Sammons, P., Thomas, S. and Mortimore, P. (1997). Forging Links: Effective Schools and Effective Departments. London: Paul Chapman Publishing.

Sampayo, S. (2015). 'Closing the Gaps in Professional Development: A Tool for School-Based Leadership Teams'. Electronic Theses and Dissertations: University of Central Florida. <u>https://stars.library.ucf.edu/etd/723?utm_source=stars.library.ucf.edu%2Fetd%2F723&utm_medium=PDF&utm_campaign=PDFCoverPages</u>.

Sandars, J. (2009). 'The Use of Reflection in Medical Education: AMEE Guide No. 44'. Medical Teacher 31 (8): 685–95. <u>https://doi.org/10.1080/01421590903050374</u>.

Scheerens, J. (1990). 'School Effectiveness Research and the Development of Process Indicators of School Functioning'. *School Effectiveness and School Improvement* 1 (1): 61– 80. https://doi.org/10.1080/0924345900010106. Scheerens, J. (2000). Improving School Effectiveness. Fundamentals of Educational Planning 68. Paris: UNESCO, International Institute for Educational Planning.

Schein, E.H. (2013). Humble Inquiry: The Gentle Art of Asking Instead of Telling. San Francisco: Berrett-Koehler Publishers, Inc.

Schildkamp, K. (2007). 'The Utilisation of a Self-Evaluation Instrument for Primary Education'. PhD Thesis, Enschede: University of Twente.

https://research.utwente.nl/en/publications/the-utilisation-of-a-self-evaluation-instrument-forprimary-educa.

Schildkamp, K., Lai, M. and Earl, L.M. (eds.) (2013). Data-Based Decision Making in Education: Challenges and Opportunities. Studies in Educational Leadership, volume 17. Dordrecht; London; New York: Springer.

Schildkamp, K., Poortman, C. and Handelzalts, A. (2016). 'Data Teams for School Improvement'. School Effectiveness and School Improvement 27 (2): 228–54. https://doi.org/10.1080/09243453.2015.1056192.

Schildkamp, K., Vanhoof, J., van Petegem, P. and Visscher, A. (2012). 'The Use of School Self-Evaluation Results in the Netherlands and Flanders'. British Educational Research Journal 38 (1): 125–52. <u>https://doi.org/10.1080/01411926.2010.528556</u>.

Schildkamp, K. and Visscher, A. (2010). 'The Utilisation of a School Self-evaluation Instrument'. Educational Studies 36 (4): 371–89. https://doi.org/10.1080/03055690903424741.

Schildkamp, K. and Visscher, A. (2013). 'Data-Centred School Self-Evaluation in the Netherlands: Characteristics and Prerequisites'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 186–200. Emerald Group Publishing.

Schon, D. (1983). The Reflective Practitioner: How Professionals Think in Action. London: Temple Smith.

School Development Planning Initiative. (2003). School Development Planning Initiative: National Progress Report 2002. Dublin: Department of Education and Science Stationary Office.

School for Education Research and Engagement, Nelson Mandela Metropolitan University, South Africa, Geoffrey van Der Voort, Lesley Wood, and COMBER, Faculty of Education Sciences, North-West University, Potchefstroom Campus. (2014). 'Assisting School Management Teams to Construct Their School Improvement Plans: An Action Learning Approach'. South African Journal of Education 34 (3): 1–7. https://doi.org/10.15700/201409161046.

Scribner, J.P., Cockrell, K.S., Cockrell, D.H. and Valentine, J.W. (1999). 'Creating Professional Communities in Schools through Organizational Learning: An Evaluation of a School Improvement Process'. Educational Administration Quarterly 35 (1): 130–60. https://doi.org/10.1177/0013161X99351007.

Seidman, I. (2006). Interviewing as Qualitative Research: A Guide for Researchers in Education and the Social Sciences. 3rd ed. New York: Teachers College Press.

Senge, P.M. (2006). The Fifth Discipline: The Art and Practice of the Learning Organization. Rev. and Updated ed. London: Random House Business Books.

Setlalentoa, W.N. (2014). 'School Self Evaluation: How Involved Are Educators in the Process?' International Journal of Educational Sciences 7 (3): 525–33. https://doi.org/10.1080/09751122.2014.11890214.

Shulman, L. (1987). 'Knowledge and Teaching: Foundations of the New Reform'. Harvard Educational Review 57 (1): 1–23. <u>https://doi.org/10.17763/haer.57.1.j463w79r56455411</u>.

Shulman, L.S. (1986). 'Those Who Understand: Knowledge Growth in Teaching'. Educational Researcher 15 (2): 4–14. <u>https://doi.org/10.3102/0013189X015002004</u>.

Silverman, D. (2005). Doing Qualitative Research: A Practical Handbook. 2nd ed. London; Thousand Oaks, Calif: Sage Publications.

Simons, H. (2013). 'Walking Backwards into the Future'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 9–13. Emerald Group Publishing.

Simons, H. (2013). 'Enhancing the Quality of Education through School Self-Evaluation'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 14–27. Emerald Group Publishing.

Singleton, R. and Straits, B.C. (2010). Approaches to Social Research. 5th ed. New York: Oxford University Press.

Sivesind, K., Skedsmo, G. and Langfeldt, G. (2006). Educational Management. Oslo: Cappelen akademisk forlag.

Sjøbakken, O.J. and Dobson, S. (2013). 'School Self-Evaluation in the Longer Time Scale. Experiences from a Small Scandinavian State'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 171–85. Emerald Group Publishing.

Skerritt, C., O'Hara, J., Brown, M., McNamara, G. and O'Brien, S. (2021). 'Enacting School Self-Evaluation: The Policy Actors in Irish Schools'. *International Studies in Sociology of Education*, February, 1–23. <u>https://doi.org/10.1080/09620214.2021.1886594</u>.

Smith, I. (2011). 'Organisational Quality and Organisational Change: Interconnecting Paths to Effectiveness'. Library Management 32 (1/2): 111–28. https://doi.org/10.1108/01435121111102629.

Smith, M.D. (2012). 'School Self-Evaluation: A Jamaican Perspective'. Journal of Third World Studies 29 (1): 137–54.

Smith, W.C. (2014). 'The Global Transformation Toward Testing for Accountability'. Education Policy Analysis Archives 22 (December): 116. https://doi.org/10.14507/epaa.v22.1571.

Smith, W.C. and Benavot, A. (2019). 'Improving Accountability in Education: The Importance of Structured Democratic Voice'. Asia Pacific Education Review 20 (2): 193–205. https://doi.org/10.1007/s12564-019-09599-9.

Smithson, J. (2000). 'Using and Analysing Focus Groups: Limitations and Possibilities'. International Journal of Social Research Methodology 3 (2): 103–19. https://doi.org/10.1080/136455700405172.

Somekh, B. (2006). Action Research: A Methodology for Change and Development. Doing Qualitative Research in Educational Settings. Maidenhead: Open University Press.

Somekh, B. and Zeichner, K. (2009). 'Action Research for Educational Reform: Remodelling Action Research Theories and Practices in Local Contexts'. Educational Action Research 17 (1): 5–21. <u>https://doi.org/10.1080/09650790802667402</u>.

Soobrayan, V. (2003). 'Ethics, Truth and Politics in Constructivist Qualitative Research'. Westminster Studies in Education 26 (2): 107–23. https://doi.org/10.1080/0140672030260204. Stacey, R.D. (2005). *Experiencing Emergence in Organizations: Local Interaction and the Emergence of Global Pattern*. Book, Whole. London; New York; Routledge. https://go.exlibris.link/5R1fWzK8.

Stake, R. (1978). 'More Subjective!' Presented at the AERA, Division D 'Should Educational Evaluation be more Objective or more Subjective? Mimeo, University of Illinois, Champaign, IL.

Stake, R.E. and Schwandt, T.A. (2006). 'On Discerning Quality in Evaluation'. In The Sage Handbook of Evaluation, edited by I.F. Shaw, J.C. Greene, and M.M. Mark, 405–18. London: SAGE Publishing.

Stake, R.E. (2004). Standards-Based & Responsive Evaluation. Thousand Oaks, Calif: Sage.

Stevenson, N.C. and Just, C. (2014). 'In Early Education, Why Teach Handwriting Before Keyboarding?' Early Childhood Education Journal 42 (1): 49–56. https://doi.org/10.1007/s10643-012-0565-2.

Stoll, L. and Louis, K.S. (eds.) (2007). Professional Learning Communities - Divergence, Depth and Dilemmas. Berkshire: Open University Press.

Stoll, L., Bolam, R., McMahon, A., Wallace, W. and Thomas, S. (2006). 'Professional Learning Communities: A Review of the Literature'. Journal of Educational Change 7 (4): 221–58. https://doi.org/10.1007/s10833-006-0001-8.

Stufflebeam, D. and Zhang, G. (2017). The CIPP Evaluation Model - How to Evaluate for Improvement and Accountability. New York: The Guilford Press.

Styhre, A. (2002). 'The Edge of Organization: Chaos and Complexity Theories of Formal Social Systems, Russ Marion; Thousand Oaks, London and Sage, New Delhi 1999'. Scandinavian Journal of Management 18 (1): 127–30. <u>https://doi.org/10.1016/S0956-5221(01)00020-3</u>.

Sugrue, C. (2004). Curriculum and Ideology: Irish Experiences, International Perspectives. Book, Whole. Dublin: Liffey Press.

Sugrue, C. (2009). 'Performativity and Professionalism: Irish Primary Principals' Experience of Building Leadership Capacity'. European Educational Research Journal 8 (3): 372–86. https://doi.org/10.2304/eerj.2009.8.3.372.
Swaffield, S. (2004). 'Critical Friends: Supporting Leadership, Improving Learning'. Improving Schools 7 (3): 267–78. <u>https://doi.org/10.1177/1365480204049340</u>.

Swaffield, S. and MacBeath, J. (2005). 'School Self-evaluation and the Role of a Critical Friend'. Cambridge Journal of Education 35 (2): 239–52. https://doi.org/10.1080/03057640500147037.

Tabrizi, S. and Rideout, G. (2016). 'Styles of Educational Leadership for Modernist and Postmodernist Approaches'. International Journal for Infonomics 9 (4): 1239–46. https://doi.org/10.20533/iji.1742.4712.2016.0152.

Tague, N.R. (2005). The Quality Toolbox. 2nd ed. Milwaukee, Wis: ASQ Quality Press.

Taneri, P.O. and Akduman, N. (2018). 'Termination of the Teaching of the Continuous Cursive Handwriting in Schools'. Iliria International Review 8 (1). https://doi.org/10.21113/iir.v8i1.391.

Taylor, D. and Hamdy, H. (2013). 'Adult Learning Theories: Implications for Learning and Teaching in Medical Education: AMEE Guide No. 83'. Medical Teacher 35 (11): e1561–72. https://doi.org/10.3109/0142159X.2013.828153.

Teddlie, C. and Reynolds, D. (eds.) (2000). The International Handbook of School Effectiveness Research. London; New York: Falmer Press.

Teodorovic, J. (2009). 'School Effectiveness: Literature Review'. Zbornik Instituta Za Pedagoska Istrazivanja 41 (1): 7–24. <u>https://doi.org/10.2298/ZIPI0901007T</u>.

Terhart, E. (2013). 'Teacher Resistance against School Reform: Reflecting an Inconvenient Truth'. School Leadership & Management 33 (5): 486–500. https://doi.org/10.1080/13632434.2013.793494.

The Health Foundation. (2010). Complex Adaptive Systems. London: The Health Foundation.

Thornburg, D. & Mungai, A. (2011). Teacher empowerment and school reform. Journal of Ethnographic and Qualitative Research, 5(4), 205-217

Thrupp, M. (2001). 'Sociological and Political Concerns about School Effectiveness Research: Time for a New Research Agenda'. School Effectiveness and School Improvement 12 (1): 7–40. <u>https://doi.org/10.1076/sesi.12.1.7.3459</u>. Timperley, H.S. (2013). 'The New Zealand Educational Context: Evaluation and Self-Review in a Self-Managing System'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 30–42. Emerald Group Publishing.

Tolofari, S. (2005). 'New Public Management and Education'. *Policy Futures in Education* 3 (1): 75–89. <u>https://doi.org/10.2304/pfie.2005.3.1.11</u>.

Trafford, V. and Leshem, S. (2008). Stepping Stones to Achieving Your Doctorate. Open University Press.

Turner, J.R., Baker, R. and Morris, M. (2018). 'Complex Adaptive Systems: Adapting and Managing Teams and Team Conflict'. In Organizational Conflict, edited by Ana Alice Vilas Boas. InTech. <u>https://doi.org/10.5772/intechopen.72344</u>.

Tymms, P. B. (1992). 'The Relative Effectiveness of Post-16 Institutions in England (Including Assisted Places Scheme Schools)'. British Educational Research Journal 18 (2): 175–92. <u>https://doi.org/10.1080/0141192920180206</u>.

UNESCO. (2004). 'Understanding education quality'. In: UNESCO, Education for all: The quality imperative (pp. 27–37). Paris: UNESCO.

Valli, L. and Buese, D. (2007). 'The Changing Roles of Teachers in an Era of High-Stakes Accountability'. American Educational Research Journal 44 (3): 519–58. https://doi.org/10.3102/0002831207306859.

Van Bruggen, J. (2010). Inspectorates of Education in Europe; Some Comparative Remarks about Their Tasks and Work. Brussels: SICI.

Van Der Voort, G. and Wood, L. (2014). 'Assisting School Management Teams to Construct Their School Improvement Plans: An Action Learning Approach'. South African Journal of Education 34 (3): 1–7. <u>https://doi.org/10.15700/201409161046</u>.

Van der Westhuizen, P. C. (ed.) (2014). Schools as Organisations. 4th ed. Hatfield, South Africa: Van Schaik Publishers.

Van Petegem, P. (1998). 'Towards a Stimulating Policy for Self-Evaluation in Schools'. In Educational Research Association. San Diego, USA. https://files.eric.ed.gov/fulltext/ED419840.pdf. Van Petegem, P. (2005). Designing School Policy: School Effectiveness Research as a Starting Point for School Self-Evaluation. Leuven: Acco.

Van Petegem, P. (1998). 'Towards a Stimulating Policy for Self-Evaluation in Schools'.

Vanhoof, J. and Van Petegem, P. (2012). 'The Process and Results of School Self-Evaluation Through the Eyes of Experts: A Delphi Study'. International Studies in Educational Administration 40 (1): 47–61.

Vanhoof, J., De Maeyer, S. and Van Petegem, P. (2011). 'Variation in the Conduct and the Quality of Self-evaluations: A Multi-level Path Analysis'. Educational Studies 37 (3): 277–87. <u>https://doi.org/10.1080/03055698.2010.506326</u>.

Vanhoof, J. and Van Petegem, P. (2007). 'Matching Internal and External Evaluation in an Era of Accountability and School Development: Lessons from a Flemish Perspective'. Studies in Educational Evaluation 33 (2): 101–19.

https://doi.org/10.1016/j.stueduc.2007.04.001.

Vanhoof, J. and Van Petegem, P. (2013). 'Judging and Explaining the Quality of School Self-Evaluations: Indicators and Findings on Meta-Evaluation from a Flemish Perspective'. In Advances in Program Evaluation, edited by Mei Lai and Saville Kushner, 217–29. Emerald Group Publishing.

Vanhoof, J. and Van Petegem, P. (2010). 'Evaluating the Quality of Self-Evaluations: The (Mis)Match between Internal and External Meta-Evaluation'. Studies in Educational Evaluation 36 (1–2): 20–26. <u>https://doi.org/10.1016/j.stueduc.2010.10.001</u>.

Vanhoof, J. and Van Petegem, P. (2011). 'Designing and Evaluating the Process of School Self-Evaluations'. Improving Schools 14 (2): 200–212. https://doi.org/10.1177/1365480211406881.

Vanhoof, J., Van Petegem, P. and De Maeyer, S. (2009). 'Attitudes towards School Self-Evaluation'. Studies in Educational Evaluation 35 (1): 21–28. https://doi.org/10.1016/j.stueduc.2009.01.004.

Vaughn, S., Schumm, J.S. and Sinagub, J.M. (1996). Focus Group Interviews in Education and Psychology. Thousand Oaks: Sage Publications.

Walters, D. (2014). 'Grounded Practice: Putting the "Self' Back into Self-Evaluation'. Educational Action Research 22 (1): 72–92. <u>https://doi.org/10.1080/09650792.2013.851612</u>.

Webb, P.T. (2006). 'The Choreography of Accountability'. Journal of Education Policy 21 (2): 201–14. https://doi.org/10.1080/02680930500500450.

Wenger, E. (1998). Communities of Practice. Cambridge: Cambridge University Press.

Wenger, E. and Lave, J. (1999). Situated Learning - Legitimate Peripheral Participation. Cambridge: Cambridge University Press.

Wenger, E. and Wenger, B. (2015). 'Learning in Landscapes of Practice'. In Learning in Landscapes of Practice: Boundaries, Identity, and Knowledgeability in Practice-Based Learning, edited by E. Wenger, M. Fenton-O'Creevy, S. Hutchinson, C. Kubiak, and B. Wenger, 13–30. London; New York: Routledge.

Westraad, S. and GM South Africa Foundation. (2011). Changing Schools in Challenging Contexts. Port Elizabeth: GM South Africa Foundation.

Whitehead, J. and McNiff, J. (2006). Action Research Living Theory. London: SAGE Publications.

Wikeley, F., Stoll, L. and Lodge, C. (2002). 'Effective School Improvement: English Case Studies'. Educational Research and Evaluation 8 (4): 363–85. https://doi.org/10.1076/edre.8.4.363.8818.

Williams, R., Brien, K., LeBlanc, J. (2012). Transforming schools into learning organizations: Supports and barriers to educational reform. Canadian Journal of Educational Administration and Policy, 134, 1-32.

Wood, L. and Govender, B. (2013). "You Learn from Going through the Process": The Perceptions of South African School Leaders about Action Research'. Action Research (London, England) 11 (2): 176–93. <u>https://doi.org/10.1177/1476750313479411</u>.

Wood, P. and Butt, G. (2014). 'Exploring the Use of Complexity Theory and Action Research as Frameworks for Curriculum Change'. Journal of Curriculum Studies 46 (5): 676–96. <u>https://doi.org/10.1080/00220272.2014.921840</u>.

Woods, P. (2006). Successful Writing for Qualitative Researchers. 2nd ed. London; New York: Routledge.

Wood, L. and Damons, B. (2013). Developing Virtuous Leaders: An Action Research Approach to Improving School Leadership in a South African Context. In Value and Virtue in Practice-Based Research, Conference Proceedings of Value and Virtue Conference, edited by J. McNiff. York: York St John University, July 4–6.

Wrigley, T. (2013). 'Rethinking School Effectiveness and Improvement: A Question of Paradigms'. Discourse: Studies in the Cultural Politics of Education 34 (1): 31–47. https://doi.org/10.1080/01596306.2012.698862.

Wrigley, T. and Wormwell, L. (2016). 'Infantile Accountability: When Big Data Meet Small Children'. Improving Schools 19 (2): 105–18. <u>https://doi.org/10.1177/1365480216651520</u>.

Wroe, A. and Halsall, R. (2001). 'School Self-Evaluation: Measurement and Reflection in the School Improvement Process'. Research in Education 65 (1): 41–52. https://doi.org/10.7227/RIE.65.4.

Wyse, D., Selwyn, N., Smith, E. and Suter, L.E. (2016). The BERA/SAGE Handbook of Educational Research. Book, Whole. London: SAGE Publications.

Zhang, G., Zeller, N., Griffith, R., Metcalf, D., Williams, J., Shea, C. and Misulis, K. (2011).
'Using the Context, Input, Process, and Product Evaluation Model (CIPP) as a
Comprehensive Framework to Guide the Planning, Implementation, and Assessment of
Service-Learning Programs'. Journal of Higher Education Outreach and Engagement 15: 57– 84.

Definitions of School Self-Evaluation found in a review of the literature

"a systematic process, largely initiated by the school itself, where participants describe and evaluate the functioning of the school for the purposes of making decisions or undertaking initiatives in the context of (aspects of) overall school (policy) development"

(Van Petegem, 2005, pg.104; Vanhoof and Van Petegem, 2010, pg.20)

"SSE can be described as a process, in large part initiated by the school, whereby highly eligible participants systematically describe and judge the functioning of the school in order to make decisions or adopt initiatives within the framework of school development"

(Vanhoof and Petegem, 2010, cited in Faddar et al., 2017, pg.398)

"SSE is considered as an attempt involving various changes made by the schools in order to ensure a better teaching and learning environment, improve students' academic achievement and also strengthen the schools' ability to implement those changes"

(Hopkins, 2005, cited in Hamzah and Tahir, 2013, pg.50)

"SSE is considered as an internal and formative evaluation which is based on a collection of evidence. Or in other words the schools have to base their judgments on all the evidence gathered to identify the effectiveness of the implementation of the schools' programs...... SSE stresses that it is a reflection process by the schools on their own practice. SSE needs to be operated in a systematic and transparent manner in order to achieve its aims to improve the students' achievement and enhance the schools' professional and organizational learning"

(Hamzah and Tahir, 2013, pg.51)

"School self-evaluation (SSE) is an internal process which aims to ensure quality, improve the teaching–learning process and increase school performance"

(Hofman et al., 2009, cited in O'Brien et al., 2019, pg.2)

"School Self-Evaluation is, by definition, something that schools do to themselves, by themselves and for themselves'. School Self-Evaluation (SSE) involves examining teaching and learning strategies, the performance and development culture and other aspects of school operations so they can be strengthened and supported to improve student outcomes. It also provides an opportunity for the whole school community, including learners, parents and all staff, to reflect on the learner outcomes in light of their goals, targets and key improvement strategies from the previous planning cycle"

(Swaffield and MacBeath, 2005, pg.239; Smith, 2012, cited in Setlalentoa, 2014, pg.525)

"To arrive at an appraisal of their current functioning (strengths and weaknesses) as a point of departure for a plan or vision for the future. Self-evaluation is a procedure which is initiated and carried out by the school in order to describe and evaluate its own functioning"

(Vanhoof et al., 2009, pg.21) (Blok et al., 2005, pg.3)

"The primary purpose of the school is student learning. Evaluation, a tool that promotes and supports both the individual and the system, can be very helpful......to have a sound basis for decision-making about potential developments to increase student learning. It can provide the school with clear guidelines and parameters to follow in order to make positive changes that facilitate this learning.... Moreover, the school can develop and improve through a reflective practice which can help stimulate the decision-making processes"

(Dahler-Larsen, 2006; Sivesend et al., 2006, cited in Emstad, 2011, pg.271-272)

"a situated analysis of educational quality"

(Simons, 2013, pg.9)

"School self-evaluation is a process of conceiving, collecting, analysing and communicating information to: (i) inform decision-making within a school (ii) ascribe value or worth (iii) establish public confidence in the school (iv) demonstrate professional self-accountability (v) to meet the purposes of accountability, development, knowledge and encourage creativity in schools"

(Simons, 2013, pg.16-17)

"Evaluation for accountability purposes involves reporting on goals and standards (including checking on compliance matters) while an improvement focus involves assisting schools to improve through their self-review"

(Mutch, 2013, pg.83)

"Self-review involves investigating evidence about student outcomes and current ways of doing things to find out where improvement is needed"

(Ministry of Education, 2003, cited in Mutch, 2013, pg.84)

"a procedure involving systematic information gathering that is initiated by the school itself and intends to assess the functioning of the school and the attainment of its educational goals for purposes of supporting decision making and learning and for fostering school improvement as a whole"

(Schildkamp, 2007, pg.4)

"a process mainly initiated by the school to collect systematic information about the school functioning, to analyse and judge this information regarding the quality of the school's education and to make decisions that provide recommendations"

(Devos, 1998, pg.51-52, cited in Demetriou et al., 2012, pg.150)

"Teachers and schools are encouraged to formulate their own development plan, to discuss their own perceived strengths and weaknesses, thus fostering collegiality and cohesiveness in terms of the school's own mission and aspirations for improvement"

(Brady, 2016, pg.524)

"School self-evaluation is a process by which members of staff in a school reflect on their practice and identify areas for action to stimulate improvement in the areas of pupil and professional learning"

(Chapman and Sammons, 2013, pg.2)

"Self-evaluation is a process of reflection on practice, made systematic and transparent, with the aim of improving pupil, professional and organisational learning"

(MacBeath, 2005, pg.4)

"School self-evaluation is a collaborative, inclusive, and reflective process of internal school review. An evidence-based approach, it involves gathering information from a range of sources, and then making judgements. All of this with a view to bring about improvements in students' learning."

(DES, 2016, pg.10)

"Its primary goal is to help schools to maintain and improve through critical self-reflection. It is concerned to equip teachers with the know-how to evaluate the quality of learning in their classrooms so that they do not have to rely on an external view, yet welcome such a perspective because it can enhance and strengthen good practice"

Survey

Thank You for taking the time to complete this questionnaire regarding:

School Self-Evaluation

Your help is most appreciated.

- DES = Department of Education and Skills
- SSE = School Self-Evaluation

Instructions:

In each question, please tick the box that feels most right to you and please fill in as best you can the description boxes. If you need more space, please write your answer on another piece of paper with the question number beside it.

- 1. The existing resources provided by the Department of Education and Skills (DES) are useful for School Self-Evaluation (SSE)
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 2. More resources are required from the DES on how to conduct SSE
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **3.** Rather than each school spending time and resources developing their own internal evaluation procedures, schools should be provided with a generic set of tools to assist with the implementation of SSE
 - □ Disagree Strongly
 - □ Disagree

- □ Indifferent
- □ Agree
- □ Agree Strongly
- 4. Staff at this school have the necessary skills required to carry out SSE
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 5. Principals and Deputy Principals need more training on how to conduct SSE
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 6. Teachers need more training on how to conduct SSE
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 7. Staff at this school have the capacity to analyse quantitative data
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 8. Staff at this school have the capacity to analyse qualitative data
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly

- **9.** The Principal and Deputy Principal of this school have the necessary training required to carry out peer review (teacher observation)
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **10.** Teachers of this school have the necessary training to carry out peer review (teacher observation)
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **11.** The Board of Management of this school have the necessary skills required to carry out evaluation and planning duties required of Board of Managements.
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **12.** Results from externally devised standardised tests (e.g., literacy and numeracy tests) should be used as part of the self-evaluation process of schools.
 - **Disagree Strongly**
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 13. Peer review is used as part of the SSE process in this school
 - □ Yes
 - □ No
- 14. Does your school have a set of procedures for carrying out SSE?
 - □ Yes
 - □ No

- 15. Does your school have an SSE policy?
 - □ Yes
 - □ No

16. The process of SSE is easy to understand

- □ Disagree Strongly
- □ Disagree
- □ Indifferent
- □ Agree
- □ Agree Strongly
- 17. The SSE guidelines developed by the DES are easy to understand
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 18. The Principal and Deputy Principal conduct SSE on a regular basis in this school
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly

19. Teachers conduct SSE on a regular basis in this school

- □ Disagree Strongly
- □ Disagree
- □ Indifferent
- □ Agree
- □ Agree Strongly

20. SSE involves all staff

- □ Disagree Strongly
- □ Disagree
- □ Indifferent
- □ Agree
- □ Agree Strongly
- 21. SSE reports should be published on the internet

- □ Disagree Strongly
- □ Disagree
- □ Indifferent
- □ Agree
- □ Agree Strongly
- 22. SSE results in better management
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 23. SSE results in better teaching and learning
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **24.** SSE places a lot of stress on staff
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **25.** SSE increases staff morale
 - **Disagree Strongly**
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **26.** SSE takes up a lot of time
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly

27. SSE is popular with the majority of staff in this school

- □ Disagree Strongly
- □ Disagree
- □ Indifferent
- □ Agree
- □ Agree Strongly

28. How would you describe your attitude to SSE in general?

- □ Positive
- □ Negative
- □ Indifferent
- □ Other

If 'Other' please describe:

29. Please describe briefly, in the box below, what you think is the purpose of SSE

30. Please list, in the box below, any useful policy documents that you know of regarding SSE

31. Please describe in the box below, what in your opinion are the positive aspects of SSE as a government policy

32. Please describe below, what in your opinion are the negative aspects of SSE as a government policy

33. In the box below, please describe any other opinions, thoughts, ideas, observations or comments you have in relation to your experience of SSE to date in your educational career. *(if any)*

Definitions of School Self-Evaluation

A definition of School Self-Evaluation for our school

Please click the three definitions of School Self-Evaluation below which you most agree with:

- □ "a systematic process, largely initiated by the school itself, where participants describe and evaluate the functioning of the school for the purposes of making decisions or undertaking initiatives in the context of (aspects of) overall school (policy) development"
- □ "SSE can be described as a process, in large part initiated by the school, whereby highly eligible participants systematically describe and judge the functioning of the school in order to make decisions or adopt initiatives within the framework of school development"
- □ "SSE is considered as an attempt involving various changes made by the schools in order to ensure a better teaching and learning environment, improve students' academic achievement and also strengthen the schools' ability to implement those changes"
- □ "SSE is considered as an internal and formative evaluation which is based on a collection of evidence. Or in other words the schools have to base their judgments on all the evidence gathered to identify the effectiveness of the implementation of the schools' programs...... SSE stresses that it is a reflection process by the schools on their own practice. SSE needs to be operated in a systematic and transparent manner in order to achieve its aims to improve the students' achievement and enhance the schools' professional and organizational learning"
- □ "School self-evaluation (SSE) is an internal process which aims to ensure quality, improve the teaching–learning process and increase school performance"
- □ "School Self-Evaluation is, by definition, something that schools do to themselves, by themselves and for themselves'. School Self Evaluation (SSE) involves examining teaching and learning strategies, the performance and development culture and other aspects of school operations so they can be strengthened and supported to improve student outcomes. It also provides an opportunity for the whole school community, including learners, parents and all staff, to reflect on the learner outcomes in light of their goals, targets and key improvement strategies from the previous planning cycle"

- □ "To arrive at an appraisal of their current functioning (strengths and weaknesses) as a point of departure for a plan or vision for the future. Self-evaluation is a procedure which is initiated and carried out by the school in order to describe and evaluate its own functioning"
- □ "The primary purpose of the school is student learning. Evaluation, a tool that promotes and supports both the individual and the system, can be very helpful......to have a sound basis for decision-making about potential developments to increase student learning. It can provide the school with clear guidelines and parameters to follow in order to make positive changes that facilitate this learning.... Moreover, the school can develop and improve through a reflective practice which can help stimulate the decision-making processes"
- □ "a situated analysis of educational quality"
- "school self-evaluation is a process of conceiving, collecting, analysing and communicating information to: (i) inform decision-making within a school (ii) ascribe value or worth (iii) establish public confidence in the school (iv) demonstrate professional self-accountability (v) to meet the purposes of accountability, development, knowledge and encourage creativity in schools"
- □ "Evaluation for accountability purposes involves reporting on goals and standards (including checking on compliance matters) while an improvement focus involves assisting schools to improve through their self-review"
- □ "Self-review involves investigating evidence about student outcomes and current ways of doing things to find out where improvement is needed "
- □ "a procedure involving systematic information gathering that is initiated by the school itself and intends to assess the functioning of the school and the attainment of its educational goals for purposes of supporting decision making and learning and for fostering school improvement as a whole"
- □ "a process mainly initiated by the school to collect systematic information about the school functioning, to analyse and judge this information regarding the quality of the school's education and to make decisions that provide recommendations"

- □ "Teachers and schools are encouraged to formulate their own development plan, to discuss their own perceived strengths and weaknesses, thus fostering collegiality and cohesiveness in terms of the school's own mission and aspirations for improvement"
- □ "School self-evaluation is a process by which members of staff in a school reflect on their practice and identify areas for action to stimulate improvement in the areas of pupil and professional learning"
- □ "Self-evaluation is a process of reflection on practice, made systematic and transparent, with the aim of improving pupil, professional and organisational learning"
- □ "School self-evaluation is a collaborative, inclusive, and reflective process of internal school review. An evidence-based approach, it involves gathering information from a range of sources, and then making judgements. All of this with a view to bring about improvements in students' learning."
- □ "Its primary goal is to help schools to maintain and improve through critical selfreflection. It is concerned to equip teachers with the know-how to evaluate the quality of learning in their classrooms so that they do not have to rely on an external view, yet welcome such a perspective because it can enhance and strengthen good practice"

Definitions of School Self-Evaluation

A definition of School Self-Evaluation for our school

Please click the one definition of School Self-Evaluation below which you most agree with:

- "SSE is considered as an internal and formative evaluation which is based on a collection of evidence. Or in other words the schools have to base their judgments on all the evidence gathered to identify the effectiveness of the implementation of the schools' programs...... SSE stresses that it is a reflection process by the schools on their own practice. SSE needs to be operated in a systematic and transparent manner in order to achieve its aims to improve the students' achievement and enhance the schools' professional and organizational learning"
- □ "School Self-Evaluation is, by definition, something that schools do to themselves, by themselves and for themselves'. School Self Evaluation (SSE) involves examining teaching and learning strategies, the performance and development culture and other aspects of school operations so they can be strengthened and supported to improve student outcomes. It also provides an opportunity for the whole school community, including learners, parents and all staff, to reflect on the learner outcomes in light of their goals, targets and key improvement strategies from the previous planning cycle"
- □ "School self-evaluation (SSE) is an internal process which aims to ensure quality, improve the teaching–learning process and increase school performance"
- □ "Self-evaluation is a process of reflection on practice, made systematic and transparent, with the aim of improving pupil, professional and organisational learning"
- □ "a procedure involving systematic information gathering that is initiated by the school itself and intends to assess the functioning of the school and the attainment of its educational goals for purposes of supporting decision making and learning and for fostering school improvement as a whole"

Definitions of School Self-Evaluation

A definition of School Self-Evaluation for our school

The following definition of school self-evaluation has come from a process of reducing nineteen definitions down to five definitions using your votes. The key words and phrases from the final five definitions have been interwoven into the one definition below. Please read the definition of school self-evaluation offered below and then answer the questions which follow. Thank you.

An agreed statement of what "School Self-Evaluation" means to us for our school

"School Self-Evaluation is a cyclical procedure we do to ourselves, by ourselves and for ourselves. It provides an opportunity for the whole school community, including learners, parents and all staff to reflect on our practice and systematically gather information to examine, assess and make decisions on our teaching and learning strategies, the performance and development culture and other aspects of our school's operations, so that they can be strengthened and supported to foster quality, improved teaching and learning and enhanced professional and organisational performance"

- 1. Do you agree with this statement?
 - □ Yes
 - □ No
 - □ Not sure
- 2. Have you any comments, thoughts or ideas on this statement?
- **3.** Have you any comments, thoughts or ideas on the way we reached this statement? (*i.e.*, *involving all teaching staff, voting from a list of definitions, combining the common words etc.*)

Definitions of School Self-Evaluation

A definition of School Self-Evaluation for our school

The following definition of school self-evaluation has come from a process of reducing nineteen definitions down to five definitions using your votes. The key words and phrases from the final five definitions were interwoven into the one definition and then adjusted based on your feedback. Please read the definition of school self-evaluation offered below and then answer the questions which follow. Thank you.

An agreed statement of what "School Self-Evaluation" means to us for our school

"In [our school name] school self-evaluation is a cyclical procedure, which provides an opportunity for the whole school community, including learners, parents and all staff to reflect on our practice and systematically gather information to examine, assess and make decisions on our teaching and learning strategies, the performance and development culture and other aspects of our school's operations, so that they can be strengthened and supported to foster quality, improved teaching and learning and enhanced professional and organisational performance"

- 4. Do you agree with this statement?
 - □ Yes
 - □ No
 - \Box Not sure
- **5.** Have you any further comments, thoughts or ideas on the agreed statement? (*if you cannot think of anything please just write 'pass'*)

Question 1	What are all the activities that you can think of that our school engages in that are driven or conducted externally for accountability purposes? (if you cannot think of anything please just write 'pass')
Question 2	What are all the activities you can think of that our school engages in that we select and conduct for self-accountability purposes? (if you cannot think of anything please just write 'pass')
Question 3	What are all the activities that you can think of that our school engages in that we select and conduct for school self-improvement purposes? (if you cannot think of anything please just write 'pass')
Question 4	What are all the activities you can think of that our school engages in that are conducted or supported externally for improvement purposes? (if you cannot think of anything please just write 'pass')

A graphic diagram to communicate and explain <u>all the evaluation activities undertaken in our school</u>

The following diagram has come from a process of listing all the evaluation activities undertaken in our school based on its purpose (i.e., improvement or accountability) and who conducts it (i.e., internally or externally). Please read the diagram and then answer the questions which follow. Thank you.

A graphic diagram to communicate and explain all the evaluation activities			
undertaken in our school			
6. Do you agree with this diagram?			
\Box Yes			
\Box No			
\Box Not sure			
7. Have you any further comments, thoughts or ideas on the agreed diagram?			
(if you cannot think of anything please just write 'pass')			

Focus Group Interview Questions

- 1. What is your attitude to School Self-Evaluation? Why do you think this is?
- 2. What skills do you think are needed to engage or partake in School-Self Evaluation?
- 3. What emotions come to your mind when you think of School Self-Evaluation?
- 4. Who should be responsible for School Self-Evaluation in a school?
- 5. What in your opinion is the purpose of School Self-Evaluation?
- 6. Does School Self-Evaluation, in your opinion, bring improvement at school level or classroom level? Or both? And if so...How?
- 7. In your opinion, should School Self-Evaluation be aimed at school level improvement or classroom level improvement? Or both? Why do you think this?
- 8. From your teaching career experience to date, has the School Self-Evaluation process led to an improvement at school level that you can describe to us?
- 9. From your teaching career experience to date, has the School Self-Evaluation process led to an improvement in classroom teaching and learning that you can describe to us?
- 10. Of the two outcomes above, was one of more value than the other? If so...Why? If not...Why?
- 11. How do you feel about your capacity to engage with School Self-Evaluation? What is enabling this? (e.g., training, leadership, enthusiasm, school climate) What is disabling this? (e.g., time, resources, training, skill set, your ability, apathy)

Author	Vinny's Reflective Journal				
Whitehead	Action	Reflection		Significance	New Action
					(Leads you back to
					the start/Left side
					of the grid)
McNiff		Thoughts/Thinking		What I learned	
Schein		Observation	Reaction	Judgement	Intervention
(taken form		(what did you observe?	(how did you	(what was your	(what did you do
Coghlan)		Can you describe it?)	react? What feelings were	judgement about	about it? How did
			aroused in	what happened? What thoughts or	you intervene? Remember: doing
			you?)	evaluations did	nothing or
			you.y	the event	remaining silent is
				trigger?)	also an
				55 /	intervention)
					This box is crucial
					Vinny
					(i) to show how you're learning from
					one episode of
					action reflection
					feeds back into new
					practice. (ii) to show how
					your learning
					influences new
					actions
					 (iii) to show you taking action based
					on your thinking.
					You could regard the
					new action as the
					beginning of a new cycle. This process
					emphasises the
					cyclical nature of
					'action-reflection' (A
					core aspect of action
					research)

Unary's Reflective Journal Significance New Action Mehilf Thoughts/Thinking Reaction Significance New Action Schlein Observation Reaction Modernet Itervention (what diagram Observation Reaction New Action Itervention (what diagram Observation Reaction New Action Itervention (what diagram Observation Reaction New Action Itervention (what diagram Observation Reaction New Sourd New Action (what diagram Observation Reaction New Sourd New Sourd New Sourd (what diagram Constance New Sourd New Sourd New Sourd New Sourd (what diagram Fearance New Sourd New Sourd New Sourd New Sourd (what diagram New Sourd New Sourd New Sourd New Sourd New Sourd (what diagram New Sourd Ne
Whitehead Action Reflection Significance New Action McNiff Thoughts/Thinking Reaction What learned Lead sou back to the start/left side of the gid) McNiff Thoughts/Thinking Reaction What learned Intervention (What any our back to the start/left side of the gid) Observation (Mod di you did was your used the was your used the was your to the word did the event in ger?) Intervention Coghian) (Wat Sour the
Minimetry Deservation (what did you abserve?) Coghian) Reaction (what did you abserve?) Con you describe it?) Preaction (what did you abserve?) react States Intervention (what did you do you?) Image: Schein Coghian) Image: Schein (what did you abserve?) Con you describe it?) Reaction (what did you abserve?) react States Intervention (what did you do dout it? How did you intervent?) Image: Schein Coghian) Image: Schein (what did you abserve?) Con you describe it?) Image: Schein (what did you abserve?) react Schein you?) Image: Schein (what did you do you intervent?) Image: Schein (what did you abserve?) Image: Schein (what did you abserve?) Image: Schein (what did you do you?) Image: Schein (what did you do you intervent?) Image: Schein (what did you abserve?) Image: Schein (what did you abserve?) Image: Schein (what did you do you intervent?) Image: Schein (what did you abserve?) Image: Schein (what did you abserve?) Image: Schein (what did you do you?) Image: Schein (what did you abserve?) Image: Schein (what did you abserve?) Image: Schein (what did you do you intervent?) Image: Schein (what did you abserve?) Image: Schein (what did you abserve?) Image: Schein (what did you abserve?) Image: Schein (what did you abserve?) Image: Schein (what did you abserve?) Image: Schein (what did you abserve?) Image: Schein (what did you abserve?)
Schein (taken form Coghian) Deservation (what dd you observe? Con you describe h?) Con you? Con you describe h?) Con you desc
Facus J was J really J learned Viny relaxed the second of the stand of the stand relaxed to the second of the stand of the second of the sec
JI F. LUNDS 95 LIC

State State		Vinny's Refle	ctive Journal		New Action
Whitehead	Action	Reflection		Significance	(Leads you back to the start/Left side of the grid)
McNiff		Thoughts/Thinking		What I learned	Intervention
Schein (taken form Coghlan)		Observation (what did you observe? Can you describe it?)	Reaction (how did you react? What feelings were aroused in you?)	Judgement (what was your judgement about what happened? What thoughts or evaluations did the event trigger?)	(what did you do about it? How did you intervene? Remember: doing nothing or remaining silent is also an intervention)
	we met as a whole staff	I was very excited for this meeting. This was apother offermunity to address as a forther together	I feel it went well. I enjoyed it. I was tivel it	He stoff the stoff the stoff the operated the operated	This box is crucial y Vinny (i) to show how your learning from one episode of action reflection feeds back into new practice. (ii) to show how y your learning influences new actions
	conduct a swot analysis of the	and laerning ssues. The bethe scribe happily. Again this ulanced methe	after it.	discussion and deligte distory by participation	(iii) to show you taking action based on your thinking. You could regard the new action as the beginning of a new cycle. This process
	T+L of handnoriting in our school.	time and space to share katersh and observe end fister more carefally and		and their SNOT and/sis responses. Jam improving	emphasises the cyclical nature of 'action-reflection' (A core aspect of action research)
		the debate the debate was tweld. shif seeme enthisigstic.	Phys.	M Listening Skills as I centainly Picked MP an how Den Norman	As part of the A.R. c; cle feedba stage, it will be inportant that
	2	The Jogen havi to then not ness simple and herefore did not inkinge, an	Hel F	view and intermet data defferen That is	I analyse the Jata unbia 14. whilst acham my cens.
	h C	momentium. nowledge was perty being shared purt	/	I would have notice I er picked up	herei 2
		Colled.		n before. As a trach, recearch of	tioner skenner can analf
			4	interpret	regle

			Vinny's Refle	ctive Journal	Ci-line Mineman	New Action	
¢.	Whitehead	Action	Reflection	1013	Significance	(Leads you back to the start/Left side of the grid)	
	McNiff		Thoughts/Thinking		What I learned	Intervention	
	Schein (taken form Coghlan)		Observation (what did you observe? Can you describe it?)	Reaction (how did you react? What feelings were aroused in you?)	Judgement (what was your judgement about what happened? What thoughts or evaluations did the event trigger?)	(what did you do about it? How did you intervene? Remember: doing nothing or remaining silent is also an intervention)	
		Duning Jan 27th - Jon 31st I showed and snas the sst	I fett it was important to most each penter individually to allow Phem ask individual questions in a sole + secure	I found this, action great as a secure and a leader has their first time the set the	tatter of fitte confused as rupon afledor, this was a, poor way to explore the documents.	This box is crucial Vinny (i) to show how your learning from one episode of action reflection feeds back into new practice. (ii) to show how your learning influences new actions	
		circular, guide likes and LAOS. This was done on an individual basis. His was Action Step (D. Sach	environment i.e. not in front of anyone. It also allonged ne ene and to reach and con firm consensus on the sst focus teny in a multication	AR HIS NOT	Elivergina as a knowledgeble experiential the scott member was in a progive tole. This in essence was what I south	(iii) to show you taking action based on your thinking. You could regard the new action as the beginning of a new cycle. This process emphasises the cyclical nature of 'action-reflection' (A core aspect of action research) The beauty	Coursing for
		member was net individually with sub cover, del by the D.r.+ sets	ray and uncerf day group thin Jensurel ellicat Callanel all menifers seemed most were surgerized ab		To charge in juncertain The A.R. Have are that the purpose of the obversation was to sindly introduce the documents. St ping pop interded to	Reflection Alfred Alfred Alfred Alfred Mart J My judg an more Story L	tas of the reing a f e providence it and my leades tent by easined and encodes I auguse of elations fine eing where eing where
		eek.	tephal bounerts re SSE from The Dept.	*	be an in-s training se. I took and in the fail full start of these du	ervice as a ssion pro- port, s that a S tervion l numents	l research actitister egarding e have alg earned th mportant
					He schol The schol in an sse through collectiv projecto	tengaged cyde this o phis o	i critical friend con play in tea saues bef motionact talen.

Phase 1 – Becoming familiar with the data

I immersed myself in this data to become familiar with the depth and breadth of its content. This involved repeated reading of the data in an active way by making notes and searching for meanings and patterns.

Phase 2 – Data coding

This phase involved the production of initial codes from the data. This initial coding was undertaken by using gerunds and keeping the codes active and as close to the original statements as possible (Charmaz, 2006, pg.42-71). To begin, I underlined what I felt was every key word and phrase in both the survey data and the interview transcripts. I then combined every underlined word and phrase into colour coded group. For example, any words or phrases associated with making improvements were colour coded pink, whilst any words or phrases associated with paperwork were colour coded dark green etc., for example:

	Survey part 1 (multiple-choice questions)
Question 1	 INPUT: The <u>existing resources</u> provided by the DES are <u>useful</u> for SSE (<u>80% indifferent</u>, 10% agree)
Question 2	 INPUT: <u>More resources are required</u> from the DES on how to conduct SSE (<u>70% yes</u>, 20% indifferent)
Question 3	 INPUT: <u>Schools should be provided with generic tools for SSE</u> rather than creating their own (<u>yes 60%</u>, indifferent 20%, no 10%)
Question 4	 INPUT: <u>Staff have the necessary skills</u> to carry out SSE (<u>80%</u> <u>yes</u>, 10% indifferent, 10% no)
Question 5	 INPUT: <u>Management needs more training</u> on how to conduct SSE (<u>yes 50%, indifferent 50%)</u> indifferent result may be due <u>to power relations</u>
Question 6	 INPUT: <u>Staff need more training</u> on how to conduct SSE (<u>70%</u> <u>yes</u>, 20% indifferent, 10% no)

Question 7	 INPUT: <u>Staff can analyse quantitative data (90% yes</u>, 10% indifferent)
Question 8	 INPUT: <u>Staff can analyse qualitative data (90% yes</u>, 10% indifferent)
Question 9	• INPUT: School <u>management have</u> the necessary <u>training</u> <u>required to</u> carry out <u>peer review</u> (teacher observation) (<u>40% yes</u> , 30% no, 30% indifferent)
Question 10	• INPUT: <u>Staff have</u> the necessary <u>training to</u> carry out <u>peer</u> <u>review</u> (teacher observation) (<u>60% no</u> , 10% yes, 30% indifferent)
Question 11	 INPUT: <u>BOM have the skills to do SSE</u> (<u>no 50%</u>, indifferent 30%, agree 20%)
Question 12	 INPUT: <u>Standardised test should be used as part of SSE (90% yes</u>, 10% no)
Question 13	 PROCESS: <u>Peer review is</u> used as <u>part of the SSE process</u> in this school (<u>100% no</u>)
Question 14	 PROCESS: has the <u>school a set of procedures for doing SSE</u> (60% yes, 10% no, indifferent 30%)
Question 15	 PROCESS: has the school an SSE policy (90% yes, 10% no)
Question 16	 PROCESS: The process of SSE is easy to understand (no 40%, indifferent 50%, yes 10%)
Question 17	 PROCESS: The <u>SSE guidelines</u> developed by the DES re <u>easy to</u> <u>understand</u> (<u>no 80%</u>, 10% indifferent, 10% yes)
Question 18	 PROCESS: School <u>management conduct SSE on a regular basis</u> in this school (90% yes, 10% indifferent)
Question 19	 PROCESS: <u>Teachers conduct SSE on a regular basis in this</u> <u>school (90% yes</u>, 10% indifferent)
Question 20	• OUTPUT: <u>SSE involves all staff (90% yes</u> , 10% indifferent)
Question 21	 PROCESS: <u>SSE reports</u> should be <u>published on the internet (no</u> <u>90%</u>, indifferent 10%)

Question 22	 OUTCOMES: <u>SSE results in better management</u> (no 30%, indifferent 50%, <u>ves 20%</u>)
Question 23	 OUTCOMES: <u>SSE results in better teaching and learning (yes 60%</u>, indifferent 20%, 20% no)
Question 24	 UNINTENDED CONSEQUENCES: SSE places a lot of <u>stress</u> on staff (<u>ves 90%</u>, 10% no)
Question 25	 UNITENDED CONSEQUENCES: SSE <u>increase staff morale</u> (<u>no</u> <u>90%</u>, 10% yes)
Question 26	 UNITENDED CONSEQUENCES: SSE <u>takes up a lot of time</u> (<u>yes 90%</u>, 10% indifferent)
Question 27	 UNINTENDED CONSEQUENCES: SSE is popular with the majority of staff in our school (no 90%, 10% indifferent)
Question 28	 UNINTENDED CONSEQUENCES: How would you describe your <u>attitude to SSE in general (positive 20%</u>, indifferent 80%)

Survey part 2 (open-ended questions)			
Question 29: What do you think is the purpose of SSE?			
S3 Respondent 1	Give schools a sense of accountability		
	Give a sense of agency to schools to work on what actually needs improvement		
	Cut down on unnecessary investigations/ revisions in schools where they aren't needed		
	Take pressure off Department		
S3 Respondent 2	To look at teaching and learning on our school: identify opportunities for improvement in practice and results		
S3 Respondent 3	Another box ticking exercise		
S3 Respondent 4	I think it's a <u>tick the box exercise</u> and I am not really sure to be honest		
S3 Respondent 5	A whole school plan to evaluate what areas we as a school need to improve on and start to facilitate in our daily school lives, for both teachers and learners alike		

S3 Respondent 6	For a school to identify an area of weakness
	and put a plan into action of how to improve
	the area
S3 Respondent 7	To identify areas of weakness and improve as
	needed. Can be helpful regarding whole-school
	planning
S3 Respondent 8	Illustrate good practice. Identify areas that
	need improvement. Schools to identify and
	develop their own improvement areas.
S3 Respondent 9	(no answer)
S3 Respondent 10	To improve all aspects of the school to lead to
	enhanced learning
S3 Respondent 11	To identify an area of weakness to improve on
	as a school. Target specific areas – improved
	outcomes for the better of each pupil and
	<mark>school</mark> as a whole
S3 Respondent 12	The purpose of SSE is to enable schools to
	critically evaluate teaching and learning within
	their organisation and to put improvement
	plans in place based on their findings
Question 30: Any useful policy do	cuments that you know of regarding SSE?
S3 Respondent 1	No
S3 Respondent 1 S3 Respondent 12	No (no answer)
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3	No (no answer) None
S3 Respondent 1 S3 Respondent 12	No (no answer)
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5	No (no answer) None None (no answer)
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4	No (no answer) None None
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6	No (no answer) None None (no answer) SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 8	No (no answer) None None (no answer) SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9	No (no answer) None (no answer) SSE guidelines (no answer) (no answer)
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10	No (no answer) None (no answer) SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines (SSE guidelines SSE guidelines (SSE guidelines SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines (Purple book) (no answer) SSE guidelines SSE guidelines SSE guidelines LAOS – looking at our schools
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 9 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positive	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines re aspects of SSE as a government policy?
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positive S3 Respondent 1	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines re aspects of SSE as a government policy? Allows schools to make practical changes
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 9 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positive	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines re aspects of SSE as a government policy? Allows schools to make practical changes Allows time for staff to reflect and be objective
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positive S3 Respondent 1	No (no answer) None None (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines re aspects of SSE as a government policy? Allows schools to make practical changes Allows time for staff to reflect and be objective and professional as we review all aspects of
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positive S3 Respondent 1	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines re aspects of SSE as a government policy? Allows schools to make practical changes Allows time for staff to reflect and be objective
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positive S3 Respondent 1	No (no answer) None None (no answer) SSE guidelines SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines re aspects of SSE as a government policy? Allows schools to make practical changes Allows time for staff to reflect and be objective and professional as we review all aspects of

S3 Respondent 4	As an SNA I don't really know the positive aspects of SSE, as my role and SSE have no relevance to each other
S3 Respondent 5	As a government policy the positive aspects are; it allows for a whole school review of our own personal teachings and going forward delivering the best possible outcomes for our
S3 Respondent 6	learners It is a positive way of <mark>improving a school, the</mark>
	teaching, learning and overall management
S3 Respondent 7	It does improve learning and management within a school. Provides opportunities for reflection and discussion.
S3 Respondent 8	Improvement of learning methodologies and strategies. Identify areas for development. Taking ownership of curriculum
S3 Respondent 9	(no answer)
S3 Respondent 10	Improvement of teaching, learning methodologies and enhancement of engagement
S3 Respondent 11	Improvement in learning overall. More focus on teaching areas of weakness. Methodologies and skills improved. Specific and easy to back by data and show improvements. SMART goals - attainable
S3 Respondent 12	All schools in the country are expected to partake. It allows schools the <u>autonomy</u> to choose areas of most concern to them and their organisation. It acknowledges the inbuilt desire of the vast majority of teachers to improve their schools teaching and learning.
Question 32: What are the negative aspects of	
S3 Respondent 1	Policy includes very long term aims e.g., in 2023 we'll look at, stage 2 of our English policy happens next yearetc. In that time teachers and principals change, move, retire, take leave. Would prefer one item done and dusted each year entirely
	Also, we had just got into it when it was <mark>paused</mark> for strike which was demoralising and <u>took the</u> energy out of the whole process.
S3 Respondent 2	Very time consuming; lots of abstract, repetitive 'splitting of hairs'; <u>quite meaningless</u> exercise at times
S3 Respondent 3	Additional paperwork on already stretched staff
S3 Respondent 4	Paper work. Time consuming
S3 Respondent 5	The negative aspects of SSE as a government policy are; paperwork to follow paperwork, stresses on teachers to over perform beyond their means where they are already over- stretched. Scare tactics used by inspectors to

	deliver outcomes by over worked and
	underappreciated teachers and not being
	realistic.
S3 Respondent 6	Added workload and time required to do it
S3 Respondent 7	Time consuming – takes a lot of meeting times. Increased paperwork equals increased stress which affects morale.
S3 Respondent 8	Negative stress, time for analysis (timetabling), time consuming, lot of meeting time, takes away from classroom planning time, increased paperwork
S3 Respondent 9	(no answer)
S3 Respondent 10	Time consuming, morale, work overload limits time for planning, stress
S3 Respondent 11	Time consuming. Unnecessary stress/ staff morale. Taking away from teacher planning/class time
S3 Respondent 12	Continuous CPD regarding the process for teachers new to the profession has not been a feature. It is a (formal) addition to an already full workload.
Ouestion 33: What emotions come to m	aind when you think of SSE?
Question 33: What emotions come to m	nind when you think of SSE?
S3 Respondent 1	hind when you think of SSE? Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow instructions
S3 Respondent 1 S3 Respondent 2	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow
S3 Respondent 1	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow
S3 Respondent 1 S3 Respondent 2	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow
S3 Respondent 1 S3 Respondent 2 S3 Respondent 3	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow instructions Overwhelmed, stressedwhere do I find the time? on one hand but on the other hand a feeling of achievement and happy knowing that I was involved in the teamwork to try and make
S3 Respondent 1 S3 Respondent 2 S3 Respondent 3 S3 Respondent 4	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow instructions Overwhelmed, stressedwhere do I find the time? on one hand but on the other hand a feeling of achievement and happy knowing that I was involved in the teamwork to try and make the school better Inspired, motivated, determined but at the
S3 Respondent 1 S3 Respondent 2 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow instructions Overwhelmed, stressedwhere do I find the time? on one hand but on the other hand a feeling of achievement and happy knowing that I was involved in the teamwork to try and make the school better Inspired, motivated, determined but at the

S3 Respondent 9	Frustration, dissatisfaction, disappointment, annoyance
S3 Respondent 10	
S3 Respondent 11	At Dept. level Vague, unclear, uncertainty, bewilderment, complex, time consuming, work overload, unsupported, lack of communication, under resourced, inconsistent, lack of staff morale, paperwork but with the possibility at our own level of Potential, achievement, success, positivity, focus, guidance, support, attainable, boost in staff morale when progress and figures in areas improve
S3 Respondent 12	Don't have any strong emotions positive or negative really
-	ther opinions, thoughts, ideas, observations or comments
	ence of SSE to date in your educational career
S3 Respondent 1	Happy to participate in SSE but also happy to be told what to do at times if it saves on time/discussion/meetings
S3 Respondent 2	find it quite tiresome in general and would welcome a <u>directive from the Dept. of Ed</u> <u>outlining what to do/how and when in each</u> <u>subject –</u> "there's great freedom in having no choice"
S3 Respondent 3	SSE does not seem to be a 'whole school process' and as an SNA it seems irrelevant to me
S3 Respondent 4	I don't have any thoughts other than SNAs are not involved in SSE and I don't know whether or not it would be beneficial for an SNA to be included or not
S3 Respondent 5	I do feel it would be <u>more beneficial to discuss</u> pressing matters regarding <u>planning</u> , schemes etc. going forward <u>rather than policy.</u>
S3 Respondent 6	I think it's great for management to involve
-------------------	--
55 Respondent o	everyone in the process (and that this will
	always draw things out and make it hard to
	please everyone) At the end of the day <u>once</u>
	the target subject area improves that's the
	main thing.
S3 Respondent 7	(no answer)
S3 Respondent 8	(no answer)
S3 Respondent 9	SSE is added pressure from the Department.
	Taking time out to discuss items with staff
	could be more <mark>valuable if a sub was provided to</mark>
	ensure work is still covered as teachers (SET
	and class) return
S3 Respondent 10	(no answer)
S3 Respondent 11	Aware of SSE in a whole school approach and
	its benefits. Unaware of resources/further
	information to enhance the benefits of SSE. I
	feel as though it is seen as an approach of
	something that just 'has to be done' as a school
	approach overall, rather than knowing the
	specific benefits overall for teaching and
	learning at a smaller scale.
RS3 respondent 12	When SSE was first introduced, the
•	inspectorate made themselves available to
	their own schools to explain the process to staff
	as a whole. There was no compulsion to invite
	the inspector to deliver the seminar, but in our
	school, we did and it was most beneficial in
	terms of the clarity it brought.
	terms of the clarity it brought.

The colour coded words and phrases were then combined into one master document for ease of thematic analysis. They were then each assigned an alphanumeric code and description title. This removed the need for colour coding. Identifying the codes and matching them with the actual data extracts that demonstrate that code is an important part of phase two.

Actual data extract quotes	Code	Description title
Purpose/Reason		
Accountability	R1	Accountability
Box ticking exercise		
• Tick the box exercise		
Improvement	R 2	Improvement
• Improvement in teaching and learning		
• A need to improve a school for both		
teachers and learners		
To improve		
Improve		

• Identify areas that need improvement		
• Improve the school to enhance		
learning		
• Improve outcomes for the better of		
each pupil and the whole school		
Improvement plans based on findings		
• To evaluate for improvement	R3	Evaluate (weakness
• Identify an area of weakness		and good practice)
• Identify areas of weakness		
• Illustrate good practice		
• Identify area of weakness		
• Critically evaluate teaching and		
learning		
• Put a plan into action	R4	Planning
Whole school planning		
Improvement plans		
Positive aspects of SSE		
Practical changes	P1	Practical
To reflect	P2	Reflect and discuss
Reflection and discussion		
Professional	P3	Professional
• As an SNA I am unaware of SSE	P4	SNA
• As an SNA my role and SSE have no		
relevance to each other		
Whole school review	P5	Whole school
• Focus on teaching in areas of		review and work on
weakness		weaknesses
Delivering the best possible outcomes	P6	Outcomes
• Improving the teaching, learning and	P7	Improvement
management of a school		
• Improvement of learning		
• Improve learning and management		
• Improvement of teaching and learning		
• Improvement in learning		
• To improve the school's teaching and		
learning		
Taking ownership	P8	Autonomy
Autonomy		
Engagement	P9	Engagement
2100000000		

		(with teaching and learning improvement)
Negative aspects of SSE	NT1	The laws to wa
• Very long term aims	N1	Too long term
• Staff turnover is high. Would prefer		
one item done and dusted each year	NO	No
• Roll out of SSE was paused and took	N2	
the energy out of the process		momentum/energy in the process
Very time consuming	N3	No time
 Time consuming 	110	
 Time consuming Time required to do it 		
 Time required to do th Time consuming 		
 Time consuming Time consuming 		
Time consuming		
Time consuming		
Abstract and quite meaningless	N4	Meaningless
Over stretched staff	N5	Overstretched staff
Over stretched staff to over perform		
beyond their means		
Additional paperwork	N6	More paper work
Paperwork		
Paperwork		
Increased paperwork		
Increased paperwork		
Staff stress	N7	More stress
Increased stress		
• Negative stress		
Unnecessary stress		
• stress		
Over worked	N8	Over worked staff
Added workload		
• Addition to an already full workload		
Not realistic	N9	Not realistic
Affects morale	N10	Lowers morale
Morale		
Morale		
• Takes away from classroom planning	N11	Takes from
time		classroom teaching
		and planning time

• Taking away from teacher		
planning/class time		
Other comments		
Happy to participate in SSE	OC 1	Participate yes!
 Happy to be told what to do if it saves on time/discussion/meetings Would like a directive from Dept. saying what to do/how and when in each subject 	OC 2	Tell me what to do
• I find it quite tiresome in general	OC3	Tiresome process
 As an SNA it seems irrelevant to me SNAs are not involved in SSE. I don't know whether or not it would be beneficial for an SNA to be included 	OC4	Irrelevant
• It would be more beneficial to discuss planning rather than policy	OC5	Teaching and learning planning more than policy please
• Great for management to involve everyone in the process	OC6	Involve everyone
• Once the target subject area improves that's the main thing	OC7	Once target area improves
• SSE is added pressure	OC8	Pressure
• Valuable if a sub was provided to ensure work is still covered as teachers return from meetings	OC9	Proper sub cover
• It is seen as an approach of something that just 'has to be done' as a school approach, rather than knowing the specific benefits overall for teaching and learning at a smaller scale.	OC10	SSE has to be done but not sure why
• When SSE was first introduced, we invited the inspector to deliver the seminar, most beneficial in terms of the clarity it brought	OC11	Partnership with inspectorate would be good
Emotions		
 Confused Frustrated Calm Annoyed 	E1	All references to emotions
Leadership		

• <i>it's the principal or the whole staff,</i>	L1	All references
decide what the school needs to		specific to
improve		leadership

As can be seen from the example above, identifying the codes and matching them with the actual data extract quote that demonstrate that code was an important part of phase two.

With these new alphanumeric codes, I now turned my attention to coding the illustrative graph data from the questionnaire

Survey part 1 (multiple-choice questions)		
Question 1	 INPUT: The <u>existing resources</u> provided by the DES are <u>usefu</u>l for SSE (<u>80%</u> <u>indifferent</u>, 10% agree) 	N4Anti OC11
Question 2	 INPUT: <u>More resources are required</u> from the DES on how to conduct SSE (<u>70% yes</u>, 20% indifferent) 	N9OC9OC11
Question 3	 INPUT: <u>Schools should be provided with</u> <u>generic tools for SSE</u> rather than creating their own (<u>ves 60%</u>, indifferent 20%, no 10%) 	N9N6
Question 4	 INPUT: <u>Staff have the necessary skills</u> to carry out SSE (<u>80% yes</u>, 10% indifferent, 10% no) 	P3Anti OC11
Question 5	 INPUT: <u>Management needs more training</u> on how to conduct SSE (<u>ves 50%,</u> <u>indifferent 50%)</u> <u>indifferent result may</u> <u>be due to power relations</u> 	 N4 N9 P3 P8 P9 L1 OC11
Question 6	 INPUT: <u>Staff need more training</u> on how to conduct SSE (<u>70% yes</u>, 20% indifferent, 10% no) 	 N4 N9 P3 P8 P9 OC11

Question 7	 INPUT: <u>Staff can analyse quantitative</u> <u>data (90% yes</u>, 10% indifferent) 	• P3
Question 8	 INPUT: <u>Staff can analyse qualitative data</u> (90% yes, 10% indifferent) 	• P3
Question 9	 INPUT: School <u>management have</u> the necessary <u>training required to</u> carry out <u>peer review</u> (teacher observation) (<u>40%</u> <u>yes</u>, 30% no, 30% indifferent) 	 N3 N5 N6 N8 N9
Question 10	• INPUT: <u>Staff have</u> the necessary <u>training</u> <u>to</u> carry out <u>peer review</u> (teacher observation) (<u>60% no</u> , 10% yes, 30% indifferent)	 N3 N5 N6 N8 N9
Question 11	 INPUT: <u>BOM have the skills to do SSE</u> (<u>no 50%</u>, indifferent 30%, agree 20%) 	N4N9
Question 12	 INPUT: <u>Standardised test should be used</u> as part of SSE (90% yes, 10% no) 	 P5 P1 R3
Question 13	 PROCESS: <u>Peer review is</u> used as <u>part of</u> <u>the SSE process</u> in this school (<u>100% no</u>) 	 N3 N5 N6 N8 N9
Question 14	 PROCESS: has the school a set of procedures for doing SSE (60% yes, 10% no, indifferent 30%) 	• R4 • P3
Question 15	 PROCESS: has the <u>school an SSE policy</u> (90% yes, 10% no) 	 P3 R4 OC10
Question 16	 PROCESS: The process of SSE is easy to understand (no 40%, indifferent 50%, ves 10%) 	N4OC11
Question 17	 PROCESS: The <u>SSE guidelines</u> developed by the DES re <u>easy to understand</u> (<u>no</u> <u>80%</u>, 10% indifferent, 10% yes) 	N4OC11

	•	
Question 18	 PROCESS: School <u>management conduct</u> <u>SSE on a regular basis in this school</u> (90% yes, 10% indifferent) 	 P3 R3 P8 P9 L1 OC10
Question 19	 PROCESS: <u>Teachers conduct SSE on a</u> <u>regular basis in this school (90% yes</u>, 10% indifferent) 	 P3 R3 P8 P9 OC10
Question 20	• OUTPUT: <u>SSE involves all staff (90% yes</u> , 10% indifferent)	 P3 P9 R4 OC6 Anti OC9 OC10
Question 21	 PROCESS: <u>SSE reports</u> should be <u>published on the internet (no 90%</u>, indifferent 10%) 	• R1
Question 22	 OUTCOMES: <u>SSE results in better</u> <u>management</u> (no 30%, indifferent 50%, <u>yes</u> <u>20%</u>) 	 N1 N9 L1
Question 23	 OUTCOMES: <u>SSE results in better</u> <u>teaching and learning (yes 60%</u>, indifferent 20%, 20% no) 	• P7
Question 24	 UNINTENDED CONSEQUENCES: SSE places a lot of <u>stress</u> on staff (<u>yes 90%,</u> 10% no) 	N7E1OC8
Question 25	• UNITENDED CONSEQUENCES: SSE increase staff morale (no 90%, 10% yes)	N10E1
Question 26	• UNITENDED CONSEQUENCES: SSE <u>takes up a lot of time</u> (<u>yes 90%</u> , 10% indifferent)	N3N1

Question 27	 UNINTENDED CONSEQUENCES: SSE is popular with the majority of staff in our school (<u>no 90%</u>, 10% indifferent) 	N4OC3OC10
Question 28	 UNINTENDED CONSEQUENCES: How would you describe your <u>attitude to SSE in general (positive 20%</u>, indifferent 80%) 	 N4 E1 OC10

My attention now turned to the interview transcriptions. I proceeded to underline what I felt was every key word and phrase. For ease of analysis, I also assigned each interviewee's voice a colour code of red, green, yellow and purple. These key words and phrases were then combined into one master interview document for ease of analysis. The alphanumeric codes were then applied to each keyword and phrase.

Focus Group Interview One	
 attitudes are one of confusion to it 	 N4 E1 OC10
• it was taken off the tablehard to get that momentum back, to get back into the swing of it.	 OC10 N2 E1 OC3
 a lot of confusion wasn't much clear guidance in the first place 	 N4 E1 OC10
 something very broad, I know is what we were told at one stage, and don't go too specific well, how broad do we go, how specific do we get. 	 N1 N4 OC10
 I think the level of engagement from school to school varies greatly, just from speaking to my own friends 	 N9 N4 N1 P9 OC10
 I think, what they're supposed to be doing mostly, across the board is the big part of the problem 	 OC10 N4
 one other thing that was being thrown at schools from the department, as if there wasn't really enough already. 	 N8 N6 N5
 Rather than, the Department coming in and doing their WSE and deciding, this area, you need to improve, and this is what we want you to do. They were just saying, okay schools, you need to do this yourselves. 	 N8 R2 R1 Anti OC11

 own attitude is, I suppose one of indifference, 	• N4
	• E1
	• OC10
• there's so much going on between new language	• N8
curriculum already being talks of a maths	• OC8
one, different programmes being introduced the	
whole time, So, you're constantly running	
around, doing this, doing that, and then I just feel	
it's one extra thing being added into the equation.	
• on board with doing whatever I have to do, but I'm	• N4
not over interested in it.	• R1
	• OC10
	• OC2
	• P3
	• P9
	• E1
• to identify areas of weakness	• R3
	• P5
	• OC7
• it's the principal or the whole staff, decide what the	• R2
school needs to improve	• P7
	• L1
	• OC6
 what the school needs to improve 	• R2
	• P7
	• OC7
• for a particular two years	• N1
	• R4
 pick something and improve it. 	• R2
	• P7
	• OC7
 schools are doing anyway, because in our job, that's 	• R2
what we do every day.	• R3
	• R4
	• P3
	• P9
	• P8
	• P2
 getting everyone on the same – singing off the 	• R4
same hymn sheet, school-wide,	• P6
	• OC6
 You do that day to day in your own classroom and 	• P8
in your setting that we're watching, and seeing	• P9
what children need to work on.	• P3
	• R2
	• R3
	• R4

	• P2
 I do see the benefit of it having one specific thing 	• P5
that the school overall has a focus on,	• R4
	• R2
	• P5
	• P7
 school self-evaluation then is more aimed at 	• R2
school level, or is it aimed at classroom level	• R2 • P5
	• F5
aimed at both, but	- D 9
• it's happening anyway, on a classroom level on	• P8
a band level, within classes, teachers talk about	• P9
things we know the areas of strengths and	• P3
weaknessesand look at our own test results,	• R2
or their own assessment, and see where a class	• R3
needs to go	• R4
	• P2
 it pulls all that together and gives focus to the 	• R4
school of where to go.	• P5
	• OC6
• starts at class level, individual classes,, strength	• R4
can become a whole school thing. For example, with	• P5
the handwriting; if people start to implement	• OC6
changes within their rooms, then that's shown	
throughout the school, whether that's in displays or	
handwriting competitionsbecomes a whole	
school result	
 anything we've certainly looked at would have been 	• R4
areas that even we would all have said, oh	• P9
yeah, that's definitely an area. And I think that's	• P8
what got is started on the handwriting, was that	• P5
what got is started on the handwriting, was that	• P2
• we had all identified that ourselves. So, I suppose	• R2
you're first identifying things from your own	• R2 • R3
experience, before you look at it as a whole school	• R4
thing	• P5
	• P8
	• P9
• all have our own little zone, then when	• R4
we come together then in staff meetings or in	• OC6
discussions, then it becomes a whole school issue	
 certainly, anything that we've looked at, 	• R2
has been something that certainly I,	• R3
in my own experience, would have flagged in my	• R4
own classes as areas that would have needed	• P3
things.	• P9
	• P8

 it forces you,
around to doing it.it does force you to do• R1something about an issue that exists.• P7especially the handwriting I think I would always• P9have been something that was always on the back• OC7burner,I must try and improve• OC10that.• It's like everything else, you've only so many hours• N3in the week, and trying to get everything done, that• OC8suppose• It's probably brings it to the forefront and makes• R4you actually focus on it a bit more than you would• R2have done otherwise.• P7• what are we actually doing well first, with a• R2
 something about an issue that exists
 especially the handwriting I think I would always have been something that was always on the back burner,
 have been something that was always on the back burner,
burner,I must try and improve that.• OC10• it's like everything else, you've only so many hours in the week, and trying to get everything done, that I suppose• N3
 that. it's like everything else, you've only so many hours in the week, and trying to get everything done, that I suppose this probably brings it to the forefront and makes you actually focus on it a bit more than you would have done otherwise. P7 P9 OC7 what are we actually doing well first, with a
 it's like everything else, you've only so many hours in the week, and trying to get everything done, that I suppose this probably brings it to the forefront and makes you actually focus on it a bit more than you would have done otherwise. P7 P9 OC7 what are we actually doing well first, with a
in the week, and trying to get everything done, that I suppose • OC8 • this probably brings it to the forefront and makes you actually focus on it a bit more than you would have done otherwise. • R4 • P7 • P9 • OC7 • What are we actually doing well first, with a
 I suppose this probably brings it to the forefront and makes you actually focus on it a bit more than you would have done otherwise. P7 P9 OC7 what are we actually doing well first, with a
 this probably brings it to the forefront and makes you actually focus on it a bit more than you would have done otherwise. P7 P9 OC7 what are we actually doing well first, with a
you actually focus on it a bit more than you would have done otherwise. • P7 • P9 • OC7 • what are we actually doing well first, with a • R2
you actually focus on it a bit more than you would have done otherwise. • P7 • P9 • OC7 • what are we actually doing well first, with a • R2
 have done otherwise. P7 P9 OC7 what are we actually doing well first, with a R2
 P9 OC7 what are we actually doing well first, with a R2
 OC7 what are we actually doing well first, with a R2
what are we actually doing well first, with a P2
positive stand first. Would you ever see it like • P7
that? I always think of, what do we need
to improve
we're so paperwork focused and paperwork driven N6
at the minute,
you're constantly thinking, of inspectors N7 N10
and WSEs that the focus • N10
suppose the focus externally would be on areas to • R1
improve. It's very rarely they come in and say, well, • R2
you did great on such a thing, you know. Then tend • R3
to come in and they will focus on the weaknesses, • P5
and they will pull you up on things • E1
Anti OC11
 a positive start,but if you started with it Anti OC11
said, you're doing these things really, really • N10
<mark>they'd still be saying, but you need to</mark> • R2
improve. • R1
• P5
 If we were to choose an area that's really strong, R2
but step it up one more notch <mark>, would</mark> • R4
probably change people's attitudes • P7
• P9
• E1
 with that mind set. You're kind of automatically, R2
looking to improve something, you're • R3
thinking of it as an area that you're falling down on. • P5
Maybe the attitudes would be different
board if you came in looking from a point of view • P7

of, we're doing really well in this, so let's try to	• P9
maintain that, or keep developing it.	• E1
• a few years ago, in specific areas of maths, and part	• R3
of it was to raise the overall score, say in maths and	• P5
standardised tests, and we put focus on the areas	• P7
that would have been traditionally weaker. And we	• P9
definitely did see a huge improvement At	• P1
school level, we did see a big	• OC7
improvement as a result when you're	
focused on something, you get more results.	
 we'd said aside time every week to do a little bit on 	• R2
that Kind of stuck it to the more	• P1
practical areas of maths <mark>,every week</mark>	• P7
we'd take one day and touch on thatit	• P9
helped to focus everyone <mark>,</mark> if	• OC7
you touched on it throughout the term every week,	
you were actually teaching it properly and coming	
back and revisiting things	
improvement?Absolutely, definitely.	
 from a school level, 	• R1
pick it up in the area, or I need to focus on this.	• R4
nobody wants to be seen to be	• P5
letting the side down, and you want to pull your	• P8
weight, if it's on a whole school level. So, it does	• P9
impact your classroom teaching,	
• I think there's an accountability as well, when it's a	• R1
whole school thing.	• P8
	• P9
	• OC6
• I do feel that we do have so much going on, it's so	• N8
easy to forget about something	
implemented successfully	
 I think if it's interesting, I have the enthusiasm; it 	• P9
depends on what the topic is, what the area is.	• E1
There are certain subject areas I'd be much more	• 0C1
interested in than others.	0.01
• I would need a lot of encouragement and a lot of	• P9
support it really does come down to staff,	• E1
myself and staff enthusiasm and interest in what	
the area is.	
 just having time for it, because I think time is a 	• N3
massive issue.	
 I would have the enthusiasm to do it once 	• R4
something is up and running and started	• P9
would need that little bit of push	
definitely would need a targeted time or a targeted	• E1
acjunicity would need a targeted time of a targeted	• OC1

reminder that this is our evaluation and this is what we're working towards, one slot a week or two times a week where we're going to work on this.	Anti OC8
 I do very little taking out school self-evaluation documents and reading them. 	N4OC10
 I think if there was a department directive coming out to everybody,something very specific, and told, this is the area you're going to improve on,	 R4 R1 E1 OC3 OC11 Anti OC8
 I just think it's putting the onus back on us, whereas it should be more coming from above, , this is where we saw your school had a weakness, or, pick a strength and say we want you to improve on it. And that they should be deciding it, I feel, more so than the staff. 	N8N5
 we are coming back to look at it. So that you really feel that motivation, that, okay, they're coming to look at poetry or oral Irish, or handwriting – right, I need to up my game. So that you have that kind of motivation to drive you on, because sometimes it can be hard to find motivation 	 R1 E1 OC3 Anti OC8
 You've so much more going on, and so much to teach throughout the day, 	• N8
 It's follow-through as well for them, isn't it? you're living off your nerves, Everyone is putting their best foot forward, and the whole school is making such an effort. you're trying to show off all the work you've been doing to show the school to the best of your ability. They come in, you're doing grand here, but this, this and this needs work – and then they go off and they leave you to it, come back at some stage to follow up, more often than not, they don't. So, it's just coming in, I feel, they come in criticising, 	 N7 N10 N8 R1 E1 Anti OC11

and go it's the magical 'work on that', and then they go, it's left to us to	
 theysay,you know your school best,, when they're putting work back on schools. And then, they're very quick at the same time to turn around and say, well, we know best when it comes to, this new curriculum, new documents, new everything to do. 	R1Anti OC11N8
 I certainly think we have the capacity, but I think schools are very busy places, and quite stressful placesschool management, teachers, everyone is working flat out, all the days we're here, and the days we're at home you don't leave here at three o'clock and leave it all behind you. 	 N7 N8 P3 E1 L1 OC8
 there's a tendency to kind of pass things off onto the schools it's felt that everything in the last few years all put back onto the schools. 	N8E1
 Even if they could come once a year for half a day, in an inspector could come in and sit down with the school, or even for a Croke Park hour, and look through where we're at now, and talk with the teachers and the staff and work together on it. 	 R1 R4 OC11
 I think we all have the capability to engage. we're all professionals everyone here is here with the best of intentions, works hard and wants the best, for our students pride in our school we want the school to be best that it can be	 P3 P3 E1 OC6
 We certainly have the means to assess within our own school and to know, assess our own classrooms and to know the children that we're teaching, I think we all know our children very well. 	P3P8
 the main thing schools are missing is just guidance and someone to come into sit with a staff and look through and think, right, what is the best way for you to use this time, and use this effort to get the best results for your specific school. 	 OC11 OC10 N4 R1 R4 L1
 I think sometimes when we're left to it ourselves, it's very hard to see, when you're one step back from any situation, it's easier to see it clearly. I think sometimes we're so into it, and so involved, and you're kind of protective of your school, 	 E1 OC6 OC7 OC11

 that sometimes you need someone to come in and even to look through and say, right, I see where you're going in this area, but maybe you're being just, an hours' chat, I think, would make the world of difference If they were to give very clear directives on what they want to see we could analyse it ourselves and say, well, we think we need to improve this, and what could we do, and decide ourselves. And they could still come back in and say, oh, well that's not really good enough, because we've seen a lot better in other schools	 E1 N4 OC11 OC10 R1 R2 P3 P8 P2 Anti OC7
 mightn't necessarily be what they wanted us to improve. professional enoughtostand over what we say?I think we would,it has to be a whole school thing if we're all in, we all have to be on the same page. And if everyone feels confident enough to be able to say that, it would be great. 	 R4 P3 P8 OC6
 but I think that's a confidence thing, and I think that's probably something that comes back from when you're starting your teaching career, you have the dip year, or most of us did, where they come in and you're nearly afraid to breathe too loudly, because whatever they say goes. And it's completely rewiring that, and rewiring that almost fear. I mean, we go back to like children in school. 	 R1 E1 Anti OC11
 I'd say confidence is the main thinga belief 	 R4 P3 P8 E1 OC6
 you'd need to have some evidence to show them. concrete evidenceassessment, evidence this all the timeit's the skillsets that we use every day. 	 OC7 P8 P3
 I think it's probably ingrained in us since we were training,you live in fear of the inspector coming in. And they come in then, and criticise. We definitely seek praise, and you seek positive reinforcementare you doing it to the best of your ability,are you doing it to the best of your ability,	 R1 E1 Anti OC11 R3 N10 E1

weakness or an area to develop, and you want to do it, and be told, yes, you've done an amazing job, when you're analysing it yourself, , we could pick it out, but it's more about this person that's going to come in from the outside, and maybe knock you down, even though you've done the best that you can do there's always a constant fear there that you're not going to get the appraisal and the gratitude for what you've put in	 Anti OC11 P3 P8
 important that everyone, certainly engages and 	• R4
comes together	• P9
	• OC6
 who keeps everyone together, and rounds it up, and mediates, I	• L1
• but then equally, everyone has to row in	• R4
and support each other.	• P5
	• P8
	• P9
	• OC6
• there is that tendency to focus on your own little	• R4
department and your own classroom, but I think it's	• P5
really important that teachers build each other up	• P9
as well,	• OC6
 I think we need to build each other up a little bit 	• R4
more, and to give each other a bit of praise when	• R4 • P9
we need it, and stand over each other and give	
each other a help when we need it.	• P5
each other a help when we need it.	• E1
	• OC6
I think it's important that everyone has an active	• R4
part, and everyone probably feels that they have a	• P5
voice	• P9
	• P2
	• E1
	• OC6
 the areas that you need a little bit of support in, 	• R4
that there's someone there to give it to you	• P5
	• P9
	• E1
 working in smaller teams, and then feeding back – 	• R4
probably like we did with the handwriting. I did feel	• P2
that was helpful, kind of funnelling it through,	• P9
so it's not just one big staff meeting and everyone's sitting there,	• P5

in speaking probably, and probably the people who have been doing it longer feel a little bit more confident in doing itI think for younger teachers, it's quite difficult in that setting to put themselves forward, even though they might have a lot of newer ideas, maybe, if they can bring fresh things to the table.	OC6Anti OC9
 it absolutely has to be a whole school approach. 	 R4 P5 OC6
 hear of schools where the principal decides, we're doing this, and this is happening, and then the teachers are all left scurrying around trying to do it and trying to implement it in the classroom, and it just fails because everyone isn't in it together. So, I think it is really important that everyone just is on the same page and everyone is part of the team that works on it. 	 R4 P5 P9 L1 OC6
 I do believe that leadership, whether it's principal and vice-principal, the main push comes from them, but that everybody needs to be on board. It has to be a whole everybody has to be a team. 	 R4 P5 P9 L1 OC6
• I think with the school self-evaluation that we've been doing, everybody has been involved, everybody has been asked an opinion, whether they want to or not, I think from that regard, it's been operating very successfully where everyone has been involved in the process	 R4 P5 P9 P2 L1 OC6 Anti OC9
 But I do believe that the leadership, I prefer the main drive to come from the leadership side of things, once staff are involved and know what's happening, right, this is where we're going with it, this is the direction we're going to take. We're going to do this once a week, we're going to do this on a daily, and then that the push comes from management, from a leadership, I need that driver, that motivation. 	 R4 P5 P8 L1 OC6 Anti OC8
 It can split, and go different directions, where some people are doing more, and other people are doing a lot less. Whereas, I think if everybody knows, this is what's expected, 	 R4 P5 P9

• clear guidelines. So, if it isn't to come from the	• N4
Department, and it was to come from the leaders in school,, I think that would work very well.	P8OC10
 everyone has to do their own little part. 	 R4 R1 P8 P9
 you still need someone reining everybody in, as you do in everything else,, in teams or anything else you need a leader to focus, channel the focus, I think, it stops everyone just going around in circles. 	 P5 OC10 P8 P9 L1 OC6
Focus Group Interview Two	
 positive without being enthusiastic 	• N4 • E1
 that I think it makes sense that schools can identify their own problems and you know go from there but 	 R3 P8 P5
 I don't enjoy the process I don't enjoy that kind of work in terms of policies, policies and paperwork 	 N4 E1 OC3 OC10
 is not an area of personal interest to me but I still think it's a fairly good idea for schools to do? 	 P7 E1 OC1
 It's not something I would enjoy,I have to do with work I'd have no feelings of excitement 	 N4 E1 OC3 OC10
 it'swhat schools do or should be doing anyway all the time, seeing how we can improve things but	 N4 R2 P7 E1 OC10
 An awful lot of investment for a very little a very specific little area 	 OC10 N4
 and I can see the point of it and I can see the democratic element of it 	 P5 P7
 I often think,just tell me what to do, I'll do it, 	 R1 N4 OC2 OC6 OC10
 to improve the learning process for our children to improve our practice and 	R2P7

children	• P6
achieve more or achieve their potential. any good school with a positive atmosphere, this	• R3
happens, you see we've a little weakness, it's mentioned	• P5
at a staff meeting and we do something to change it,	• P7
	• P2
	• P1
on board with it and I guess then you need proof in the	• P8
form of paperwork	• P9
	 OC6
	• N6
have always done it as in we see something, lower	• R3
results in tests or some attitude problem on the kid or	• P1
something and we kind of naturally focus in, next year	• P5
let's try to do a bit better on that.	
	• P6
	• P7
	• P8
	• P9
	• OC6
	 OC7
improvement at school level or classroom level	• R2
i'd say both,i'd say both	• P1
the topics you chose, we've picked	• P5
something very practical, handwriting,	
	• P7
it's good for the school because we've all had	• P8
a look at ourselves and said it's something we can	• P9
improve as a school and then the kids will definitely feel	 OC6
it in class because it's very practical this	• OC7
time next year we could look at the copies and say,	• OC10
'Look at that' There's the result,	
I find it hard to engage with When there was a	• N4
meeting, it was SSE, then I go out and it's kind of,	
'Oh gone', you know unless I'm reminded again about it.	
I feel like I'm always so against the clock I	• N3
believe in the decisions made, I want to do it in the	
classroom,the day just goes by, so I guess	
that's why I did like that we did something simple, like	
handwriting, I think maybe if it was perhaps	
more classroom specific	
	• D4
I'd say there is scope for a whole school use of it like we	• R4
were just talking about a lovely schoolhad a	• P5
general feeling and attitude that people in the classroom	• P7
benefited from so I guess we could look at things like,	
uniforms or punctuality andPoliteness, manners.	
issues on yard, so I'd say there is scope for	
tangible whole school improvements	
I'd say potential for it for using both, so far, it's been	• R2
classroom based and I'd it has led to improvements.	• R4
	• P7
	- 17
	• P5

	• OC7
 I'm always a little nervous about plans that are operate 	Anti OC6
over a few years,you've a changeover of	Anti OC5
teachers, you could have a new principal, maternity	• L1
leaves, you've everything and those things get lost and	• E1
so I just love when something can be implemented	• N4
between September and June and done and it's	• N1
there Anything that's a little longer term I am	• OC10
starting to lose enthusiasm for because I've started to	• 0010
realise, they just fade away.	
 I think the intention is very good I think it has to 	• R2
be a positive thing you know because we're always	• P7
trying to improve,	• P9
,	• E1
e it can be a hit yague if it's too hig	• N1
 it can be a bit vague if it's too big, I think small 	
concise areas or maybe an overall general thing in the school if you say for year 1, 2,	• N4
3, and 4, we're going to implement thisit	• OC10
	• P7
starts off great but inevitably people come and go and	• E1
puff we forget But the intention is	
good.	52
 it's good to keep looking at ourselves and looking at our 	• R2
practice	• P7
	• P2
	• P9
	• P3
	• OC6
	Anti OC7
 we do get to suggest those weaknesses 	• P5
	• R3
 There's a very supportive environment here in this 	• R4
schoolI think nobody's afraid to ask for help	• R2
if they need help people have different	• P5
strengths And no one's afraid to say, 'I'm	• P8
bad at something'.	• P9
	• E1
	• OC6
	Anti OC7
e and food into the other	
one feed into the other	• P5
 I'd say for SSE classroom based probably works better 	• R4
because it's hard to find a practical measurable way of	
add to school culture or atmosphere,	
when you talk about SSE it's more	
suitable for classroom teaching and it's probably more	
likely more effectiveBecause they're more	
precise measurableharder within the	
bounds of school self-evaluation process.	
enable you to do school self-evaluation?one	• R4
thing that's very practical is the time out that we had	• P2
from the classroom to be able to talk in focus groups	• P9

 aboutit's very hard as you know if you're trying to do that at lunchtime or after school. Like we were given time and plenty of time to – so that helped to engage and we could all agree in our group A long time ago. I just remember the cold in the library, that's what I remember about	 P5 P1 L1 OC6 OC9 N4 OC10
 very small area' That's all I remember, we were trained we were ready but you know the strike then it was called I felt like I'd emotionally invested in it and then it was gone, then it came back, This is just another thing that will come and go 	 OC11 N2 OC3
 the topics we picked had been topics I feel I can talk about any of our topics so far actually I've had opinions on I know there's an effort to have every voice heard there was so much talk and discussion and meetings to get to a few quite basic points you know so I guess that's the part of it that if an instinct to resist	 P9 P8 Anti OC7 N3 P5 P9 P8 P2 L1 OC3 Anti OC6 Anti OC7 OC8 Anti OC9
 I wouldn't be keen to get more training on it really. I assume you go off to these places and you know what's required and it filters down to us in terms of what you 	 N4 OC10 N4 L1
know what we have to do so I'm comfortable enough with that process, I'm probably not eager enough to go and study those things myself.	• OC10
 my lack of enthusiasm about the whole thing,. , I'm very happy to do whatever. 	 N4 E1 OC1 OC3 OC10
 a kind of an apathy towards it, where do you think that stemmed from, I think people want to become a teacher I think people want to become a teacher Ipicture myself as a teacher whenyou've got a book in your hand and you're talking to the class and they're answering questions and next thing the conversation moves on, you're down with the map and next thing you're talking about Brazil and that's when I feel really good in my job and I like to be in the classroom	 N4 OC10 N9 E1 R1 P5 P8 P9 OC3 OC7

I feel best when I do something	
like I read a book see something in an art	
gallery I'm doing it in school I	
actually feel – blissful moments	
everything else I try to argue to have it as minimum as	
possible and I'm being paid for being here so I accept	
that I should have to do this stuff but I don't know if I	
could ever really feel enthusiastic about that.	
only thing that would encourage enthusiasm is that if it's	
something you would personally do next year, like the	
handwriting, I do feel positive about that next year, I feel	
good that we'll all be doing the same thing or that we all	
know how to rule our copies the same way but I think	
the general apathy is just if maybe if you asked a pile of	
accountants to do this kind of work you might get some enthusiasm.	
	- N/4
things we have to write down that I feel like you write it	• N4
down and probably it will never ever be read but you're	• N9
doing it in case some legal thing comes up or someone	• OC3
comes in and there's no soul in it,	• OC10
this I wouldn't feel negative about this because it	• P1
is,actually, going to be used next year	• P9
	• P8
	• E1
something I have to write downand away it	• OC10
goes in a folder and probably eyes will never be cast on	• N4
it again, that frustrates me, anything to do with	• N9
legal, you know the whole world of insurance and	• E1
compo culture and all that and everything to do to	 OC3
prevent that is just a very negative area in my mind you	
know. If you were to think about that you wouldn't	
teach really. You'd change your job.	
Well, I love teaching. I love my job. I love the children. I	• R2
love being in there, getting you know making a	
difference in their lives and getting them enthused	
about something, that I'm enthusiastic, bringing out of	
them – bringing out what's the best in them that's real	
to me and I really love that and this	
Box ticking exercise dusty old doesn't do anything for	• OC10
me. I do it because I have to do it because that's my	• N4
terms and conditions	• R1
	• E1
	• OC2
	• OC3
I like integrated learning	• R2
I don't know where you put that in boxes, I'm	• OC10
not a box ticking person.	• N4
	• N9

 every form or folder reduces the time you have in the day to do those things she's done nothing up there because you know what I mean I haven't ticked all 	 N4 N9 R1
the boxes but yet I feel I have done a lot but you can't quantify a lot of it.	• OC10
 your best teaching moments are the things when you veer offSpontaneous. It's not really worth it to go back and change your plans you know what happened and you know and I'd say as well the parents anything they value that you do, none of that's in your folders either 	 N4 N9 OC10
 a good – a positive approach, a positive outlook really, open-mindedgood social skills to be able to engage and to listen and to take on new things and a fresh prepared to look at things in a different way really positive attitude to you know engage with the processnot a negative approach. 	• P9 • E1
 you need a leader who's able to go and do the research, fill in the forms, someone who's good at, making the tables and charts and working it all out, I think you need one person like that and the other people compliant. 	 N4 L1 OC10
 good interpersonal relationships prepare to engage with each other be honest no point in pretending. 	 P9 E1 OC6
 would lean more towards positive than negative anyway a little resistant I feel a sense of resistance,to anything that has to go into a folder you know documents, and accountability 	 N4 E1 OC3 OC10
 whole thing can be a little distasteful not very appealing to me, 	 N4 OC3 OC10
 Acceptance Acquiescence. Resignation but I like the group thing we can all you know share ideas and you know not get too het up about it and start small you know and I like the idea of it, I can see on paper the whole idea of it but I would be no good to do it on my own you know I might as well be looking into a pool of deep water	 N4 R4 P5 E1 OC3 OC6 OC10
 I could think of ways in my own class to do a thing but notYeah but you do all the time don't you?And every – after Halloween and every January and after Easter I always come back with this new thing I'm going to do and change because you know you look back and say, 'God that was so wrong', you knowIt's interesting you should say that though, we're doing it all the timeWe don't reflect on it. We don't stop and sayThis is 	 P5 P7 P9 P8 P1 P2 P3

	aluation	
a s de kn	no should be responsible for school self-evaluation in school? I think the principal, with the help of the puty head I'd say and filter it down from there. You ow to structure it, 'You guide it and eer it and then everybody else come in on board'.	 N4 L1 OC10
yo co the the or the slc slc	hink it's managementit would start with u Vinnie and then the likes of the Vice and then if you uld whittle it down to the tangible choices as in, ere's the three handwriting books, staff have a look at em, chat to your partner, have a vote, again this occess we've done here so far and there were times here I didn't think we all need to be part of it, that you you and Catherine might have got that bit done and en you know sometimes just too many cooks can just ow a whole process down and especially as anagement are more clued into these things anyway d possibly care more you know.	 L1 OC6 Anti OC9 N4 OC10

Phase 3 – Generating initial themes

This phase involved sorting the different codes (and the actual data extract quotes) into potential themes. To help analyse the codes and consider how they might combine to form overarching themes, Braun and Clarke (2006) recommend the use of a visual representation (p.89). I therefore listed all the codes with their title description into a table as follows:

Code	Title description	Code	Title description
R1	Accountability	N1	Too long term
R2	Improvement	N2	No momentum/No energy in process
R3	Evaluate (weakness and good practice)	N3	No time
R4	Planning	N4	Meaningless
		N5	Overworked staff
P1	Practical	N6	More paperwork
P2	Promotes reflection and discussion	N7	More stress
P3	Promotes professionalism	N8	Staff overworked

P4	SNA	N9	Not realistic
P5	Whole school review and work on weaknesses	N10	Lowers morale
P6	Focus on better outcomes	N11	Takes from classroom teaching and learning time
P7	Focus on improvement		
P8	Autonomy	OC1	Participate yes!
Р9	Engagement with teaching and learning	OC2	Tell me what to do
		OC3	Tiresome
E1	Emotions	OC4	Irrelevant
L1	Leadership	OC5	Teaching and learning planning more than policy please
		OC6	Involves everyone
		OC7	Once target area improves
		OC8	Pressure
		OC9	Proper sub cover
		OC10	School self- evaluation has to be done not sure why
		OC11	Partnership with inspectorate would be good

The next task involved thinking about the relationship between the codes. I used colour coding once again to link codes I felt had something in common, collating them then into their own thematic box for ease of analysis and further refinement as follows:

Code	Title description	Sub themes
R1	Accountability	
P2	Promotes reflection and discussion	
Р3	Promotes professionalism	
P8	Autonomy	
P9	Engagement with teaching and learning	
OC1	Participate yes!	
OC11	Partnership with inspectorate would be good	
R2	Improvement	
R3	Evaluate (weakness and good practice)	
R4	Planning	
P1	Practical	
P5	Whole school review and work on weaknesses	
P6	Focus on better outcomes	
P7	Focus on improvement	
OC6	Involve everyone	
OC7	Once target area improves	
R1	Accountability	
N1	Too long term	
N4	Meaningless	

OC2Tell me what to doOC4IrrelevantOC10School self- evaluation has to be done not sure whyOC11Partnership with inspectorate would be goodP4School self- evaluation is meaningless to SNAsN2No momentum/no energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers morale
OC10School self- evaluation has to be done not sure whyOC11Partnership with inspectorate would be goodP4School self- evaluation is meaningless to SNAsN2No momentum/no energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
evaluation has to be done not sure whyOC11Partnership with inspectorate would be goodP4School self- evaluation is meaningless to SNAsN2No momentum/no energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
inspectorate would be goodP4School self- evaluation is meaningless to SNAsN2No momentum/no energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
evaluation is meaningless to SNAsN2No momentum/no energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
N5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
N6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
N7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
N8 Staff overworked N10 Lowers morale N11 Takes from classroom teaching
N10 Lowers morale N11 Takes from classroom teaching
N11 Takes from classroom teaching
classroom teaching
······ ·······························
OC3 Tiresome
OC5 Teaching and learning planning more than policy please
OC8 Pressure
OC9 Proper sub cover
E1

L1	Leadership can motivate the involvement of staff in school self- evaluation	

The orange codes represented what school self-evaluation gives to a school. The yellow codes I felt were representative of the perceived reality of school self-evaluation on the ground by those implementing it. The purple codes represented what, according to the literature, school self-evaluation in theory gives to school staff. I felt the blue codes represented the attitude of staff members to school self-evaluation, whilst the green code represented the leadership of school self-evaluation.

Phase 4 – Reviewing and developing themes

This phase involved the refinement of potential themes. To do this I revisited the actual data extracts and the literature, to thus ensure, that the themes worked in relation to our data set and that we had accurate representations for the purposes of this study. The outcome of this process was as follows:

Code	Title description	Sub themes
R1	Accountability	• Internal accountability
P2	Promotes reflection and discussion	
Р3	Promotes professionalism	
P8	Autonomy	
P9	Engagement with teaching and learning	
OC1	Participate yes!	
OC11	Partnership with inspectorate would be good	

R2	Improvement	• Evaluate for improvement
R3	Evaluate (weakness and good practice)	
R4	Planning	
P1	Practical	
OC7	Once target area improves	
P6	Focus on better outcomes	
P7	Focus on improvement	
OC6	Involve everyone	Whole school approach
Р5	Whole school review and work on weaknesses	
R1	Accountability	External accountability
N1	Too long term	
N4	Meaningless	
N9	Not realistic	
OC2	Tell me what to do	
OC4	Irrelevant	
OC10	School self- evaluation has to be done not sure why	
OC11	Partnership with inspectorate would be good	
P4	School self- evaluation is meaningless to SNAs	• SNA
N2	No momentum/no energy in the process	Workload

N3	No time	
N5	Overworked staff	
N6	More paperwork	
N8	Staff overworked	
N11	Takes from classroom teaching and learning time	
OC3	Tiresome	
OC5	Teaching and learning planning more than policy please	
OC9	Proper sub cover	
N7	More stress	Emotions
N10	Lowers morale	
OC8	Pressure	
E1	Emotions	
L1	Leadership can motivate the involvement of staff in school self- evaluation	• Leadership and staff motivation

Phase 5 – Refining, defining and naming themes

In this phase Braun and Clarke (2006) say we must identify the essence of what recognised themes are about and determine what aspect of the data set each theme captures (pg.92). They contend it is important to identify the 'story' that each theme tells in relation to each other and how it fits into the overall story of the research question. Using this approach in my analysis of the data, I generated two key themes to frame a thematic analysis of the data.

The first theme 'Attitudes to school self-evaluation', is about the people tasked with implementing the school self-evaluation process to achieve the end goal of improved outcomes

in teaching and learning. Whilst the second theme, *'Leadership and staff motivation'* describes the important role school leaders play in motivating and sustaining staff to engage meaningfully with the school self-evaluation process.

Colour codes

Yellow	accountability
Dark-purple	Attitudes
Red	Reflection
Pink	improvement
Purple	whole school review v class room work
Sky-blue	whole school planning
Grass-green	practical change and best outcomes
Dusty-pink	Professionalism and autonomy
Grey	SNA
Light-orange	Time consuming
Dark-green	Paperwork
Light-blue	over-stretched staff
Dark-orange	Workload
Light-green	Stress
Salmon	Not realistic.
Dark-blue	Morale

Appendix 14

	_	
Actual data extract quotes	Code	Description title
Reason/Purpose of school self-evaluation?	- :	
Accountability	R 1	Accountability
Box ticking exercise		
Tick the box exercise		
Improvement	R 2	Improvement
• Improvement in teaching and learning		
• A need to improve a school for both		
teachers and learners		
To improve		
• Improve		
• Identify areas that need improvement		
• Improve the school to enhance		
learning		
• Improve outcomes for the better of		
each pupil and the whole school		
Improvement plans based on findings		
• To evaluate for improvement	R3	Evaluate (weakness
• Identify an area of weakness		and good practice)
• Identify areas of weakness		
Illustrate good practice		
 Identify area of weakness 		
• Critically evaluate teaching and		
learning		
• Put a plan into action	R4	Planning
Whole school planning		
Improvement plans		
Positive aspects of school self-evaluation?		
Practical changes	P1	Practical
To reflect	P2	Reflect and discuss
Reflection and discussion		
Professional	P3	Professional
As an SNA I am unaware of SSE	P4	SNA
• As an SNA my role and SSE have no		
relevance to each other		
Whole school review	P5	Whole school
• Focus on teaching in areas of		review and work on
weakness		weaknesses
• Delivering the best possible outcomes	P6	Outcomes
• Improving the teaching, learning and	P7	Improvement
management of a school		L
Improvement of learning		
Improve learning and management		
Improvement of teaching and learning		
 Improvement in learning 		

• To improve the school's teaching and		
learning		
Taking ownership	P8	Autonomy
• Autonomy		
Engagement	P9	Engagement
		(with teaching and
		learning
		improvement)
Negative aspects of school self-evaluation?		
• Very long term aims	N1	Too long term
• Staff turnover is high. Would prefer		
one item done and dusted each year		
• Roll out of SSE was paused and took	N2	No
the energy out of the process		momentum/energy
	NO	in the process
• Very time consuming	N3	No time
• Time consuming		
• <i>Time required to do it</i>		
• Time consuming		
• Time consuming		
• Time consuming		
Time consuming	NT 4	Maaninalaas
Abstract and quite meaningless	N4	Meaningless
• Over stretched staff	N5	Overstretched staff
• Over stretched staff to over perform		
beyond their means	NC	Mana maman manual
Additional paperwork	N6	More paper work
Paperwork		
Paperwork		
Increased paperwork		
Increased paperwork	NT7	More stress
• Staff stress	N7	whole stress
Increased stress		
Negative stress		
Unnecessary stress		
• stress	NTO	Over worked staff
Over worked	N8	Over worked staff
 Added workload Addition to an already full workload 		
Addition to an already full workload	NIO	Not realist.
Not realistic	N9	Not realistic
Affects morale	N10	Lowers morale
• Morale		
Morale	NT11	
• Takes away from classroom planning	N11	Takes from
time		classroom teaching
Taking away from teacher		and planning time
planning/class time		

Other comments		
• Happy to participate in SSE	OC 1	Participate yes!
• Happy to be told what to do if it saves on time/discussion/meetings	OC 2	Tell me what to do
 Would like a directive from Dept. saying what to do/how and when in each subject 		
 I find it quite tiresome in general 	OC3	Tiresome process
 As an SNA it seems irrelevant to me 	OC4	Irrelevant
• SNAs are not involved in SSE. I don't know whether or not it would be beneficial for an SNA to be included		
• It would be more beneficial to discuss planning rather than policy	OC5	Teaching and learning planning more than policy please
• Great for management to involve everyone in the process	OC6	Involve everyone
• Once the target subject area improves that's the main thing	OC7	Once target area improves
• SSE is added pressure	OC8	Pressure
• Valuable if a sub was provided to ensure work is still covered as teachers return from meetings	OC9	Proper sub cover
• It is seen as an approach of something that just 'has to be done' as a school approach, rather than knowing the specific benefits overall for teaching and learning at a smaller scale.	OC10	SSE has to be done but not sure why
• When SSE was first introduced, we invited the inspector to deliver the seminar, most beneficial in terms of the clarity it brought	OC11	Partnership with inspectorate would be good
Emotions		
 Confused Frustrated Calm Annoyed 	E1	All references to emotions
Leadership		
 it's the principal or the whole staff, decide what the school needs to improve 	L1	All references specific to leadership

Appendix 15



Information Sheet

Purpose of the Study. I am Vincent Thorpe, a doctoral student, in the Department of Education, Maynooth University.

As part of the requirements for a Doctorate in Education, I am undertaking a research study` under the supervision of Dr. Anthony Malone.

The study is concerned with organisational change and improvement. I believe our school can do selfevaluation better. By leading the school to engage with the process in more collaborative, dialogical and co-constructivist ways, the process could become more meaningful with genuine school improvement possible.

What will the study involve? The study will involve research participants carrying out an Action Research cycle to improve school self-evaluation in our school. This will involve gathering data, generating strategies for action, implementing the action, monitoring and gathering data on the action, evaluating the evidence gathered, reflecting on the outcomes and generating strategies for action or modified enquiry questions for another cycle.

Who has approved this study? This study has been reviewed and received ethical approval from Maynooth University Research Ethics committee. You may have a copy of this approval if you request it.

Why have you been asked to take part? You have been asked because I greatly value working in collaboration with you to improve the leadership of school self-evaluation in our school. Having worked with you all over the past 6 years, I am very aware of just how knowledgeable and talented you are. Your input can only be seen as a positive thing as you bring a wealth of in-depth experience, skills and insight to the project. I am therefore adopting a collaborative, dialogical and co-constructivist approach where you, my fellow work colleagues are invited to be co-researchers and share in the ownership and direction setting of this thesis.

Do you have to take part?

No, you are under no obligation whatsoever to take part in this research. However, we hope that you will agree to take part and give us some of your time to (i) complete a questionnaire regarding your attitude towards school self-evaluation (ii) take part in a taped focus group interview to explore your experience of and relationship with school self-evaluation to date (iii) carry out an Action Research cycle to improve school self-evaluation in our school and (iv) reflect on your thinking and learning over the duration of the project. It is entirely up to you to decide whether or not you would like to take part. If you decide to do so, you will be asked to sign a consent form and given a copy and the information sheet for your own records. If you decide to take part, you are still free to withdraw at
any time without giving a reason and/or to withdraw your information up until such time as the research findings are anonymised by January 31st 2021. A decision to withdraw at any time, or a decision not to take part, will not affect your relationship with the school.

What information will be collected? (i) Your attitude towards school self-evaluation (ii) Your experience of and relationship with school self-evaluation to date (iii) accounts of your reflective critical thinking and learning over the duration of the project.

Will your participation in the study be kept confidential? Yes, all data that is collected about you during the course of the research will be kept confidential. No names will be identified at any time. All hard copy information will be held in a locked cabinet at the researchers' place of work, electronic information will be encrypted and held securely on MU PC or servers and will be accessed only by Vincent Thorpe.

No information will be distributed to any other unauthorised individual or third party. If you so wish, the data that you provide can also be made available to you at your own discretion.

'It must be recognised that, in some circumstances, confidentiality of research data and records may be overridden by courts in the event of litigation or in the course of investigation by lawful authority. In such circumstances the University will take all reasonable steps within law to ensure that confidentiality is maintained to the greatest possible extent.'

What will happen to the information which you give? All the information you provide will be kept at Maynooth University in such a way that it will not be possible to identify you. On completion of the research, the data will be retained on the MU server. After ten years, all data will be destroyed (by the PI). Manual data will be shredded confidentially and electronic data will be reformatted or overwritten by the PI in Maynooth University.

What will happen to the results? The research will be written up and presented as a summary report, discussed at internal group meetings, presented at National and International conferences and may be published in scientific journals. A copy of the research findings will be made available to you upon request.

What are the possible disadvantages of taking part? I don't envisage any negative consequences for you in taking part or it is possible that talking about your experience may cause some distress.

What if there is a problem? At the end of the taped interview/action research cycle/questionnaire, I will discuss with you how you found the experience and how you are feeling. If you experience any distress following the taped interview/action research cycle/questionnaire I will provide you with contact details for a counseling service. You may contact my supervisor Dr. Anthony Malone (Anthony.Malone@mu.ie) if you feel the research has not been carried out as described above.

Any further queries? If you need any further information, you can contact me: Vincent Thorpe, (087)2663414, Vincent.Thorpe.2018@mu.ie

If you agree to take part in the study, please complete and sign the consent form overleaf.

Thank you for taking the time to read this

Consent Form

Iagree to participate in Vincent Thorpe's research study titled "A living Theo Leading School Self-Evaluation".	ory of
Please tick each statement below:	
The purpose and nature of the study has been explained to me verbally & in writing. I've been able to questions, which were answered satisfactorily.	o ask
I am participating voluntarily.	
I give permission for my interview to be audio recorded	
I understand that I can withdraw from the study, without repercussions, at any time, whether that is before it starts or while I am participating.	
I understand that I can withdraw permission to use the data right up to	
anonymization on January 31 st 2021	
It has been explained to me how my data will be managed and that I may access it on request.	
I understand the limits of confidentiality as described in the information sheet	
I understand that my data, in an anonymous format, may be used in further research projects and an subsequent publications if I give permission below:	y D
[Select as appropriate]	
I agree for my data to be used for further research projects and subsequent publications	
I do not agree for my data to be used for further research projects and subsequent publications	

Signed.	 	
0.0	 	

Date.....

Participant Name in block capitals

I the undersigned have taken the time to fully explain to the above participant the nature and purpose of this study in a manner that they could understand. I have explained the risks involved as well as the possible benefits. I have invited them to ask questions on any aspect of the study that concerned them.

Signed.....

Date.....

Researcher Name in block capitals

If during your participation in this study you feel the information and guidelines that you were given have been neglected or disregarded in any way, or if you are unhappy about the process, please contact the Secretary of the Maynooth University Ethics Committee at <u>research.ethics@mu.ie</u> or +353 (0)1 708 6019. Please be assured that your concerns will be dealt with in a sensitive manner.

For your information the Data Controller for this research project is Maynooth University, Maynooth, Co. Kildare. Maynooth University Data Protection officer is Ann McKeon in Humanity house, room 17, who can be contacted at <u>ann.mckeon@mu.ie</u>. Maynooth University Data Privacy policies can be found at <u>https://www.maynoothuniversity.ie/data-protection</u>.

Phase 1 – Becoming familiar with the data

I immersed myself in this data to become familiar with the depth and breadth of its content. This involved repeated reading of the data in an active way by making notes and searching for meanings and patterns.

Phase 2 – Data coding

This phase involved the production of initial codes from the data. This initial coding was undertaken by using gerunds and keeping the codes active and as close to the original statements as possible (Charmaz, 2006, pg.42-71). To begin, I underlined what I felt was every key word and phrase in the survey data. I then combined every underlined word and phrase into a colour coded group. For example, any words or phrases associated with staff relationships were colour coded yellow, whilst any words or phrases associated with time were colour coded blue etc., for example:

	evaluation in our school
Strengths	Weaknesses
 Staff relationships, motivation and interest Manageability of school size Frequent meetings and communication amongst staff Involving everyone Works well Specific Focus on learning and teaching Knowledgeable and professional staff 	 Interruptions with exceptional even Turnover of staff perceived lack of confidence in stamembers I'm not sure there are weaknesses, great job is being done Time consuming Extra stress and the feeling of 'juanother area of work that has to done' Lack of interest due to constamoving of goalposts regarding SSE

Opportunities	Threats (obstacles)
 Potential to organise more teams to help with each area of school planning Great opportunity to identify and agree on areas that need our focus and attention Works well Clearer layout from the Department of Education Increased supports for schools Good opportunity to harness staff energy and interest if led well 	 Willingness of staff Availability of sub cover to do it Not having all staff on board Time Support from Department of Education Lack of interest Feelings that teachers' voices won't be heard and respected

The colour coded words and phrases were then combined into one master document for ease of thematic analysis. They were then each assigned an alphanumeric code and description title. This removed the need for colour coding. Identifying the codes and matching them with the actual data extracts that demonstrate that code is an important part of phase two, as can be seen below.

The process of school self-	evaluation in our sch	ool
Strengths	Code	Description title
 Staff relationships Manageability of school size Frequent meetings and communication amongst staff 	S1	Staff involved due to school size, with good communication
Involving everyoneMotivation and interest	S2	Staff are motivated and interested
Works well	S3	Works well
• Specific focus on learning and teaching	S4	Focus on teaching and learning
Knowledgeable and professional staff	S5	Motivation and interest are there
Weaknesses	Code	Description title

• Interruptions with exceptional events	W1	Events taking away from teaching and learning
• Turnover of staff	W2	Lack of interest
• Lack of confidence in staff members		
• A great job is being done	W3	Great job being done
• Time consuming	W4	Lack of time
Extra stress	W5	Stress
• The feeling of just another area of work that has to be done	W6	Increased workload
Lack of interest	W7	Lack of interest
Opportunities	Code	Description title
Organise more teams	01	Increase staff involvement
• To identify and agree on areas that need our focus and attention	O2	To focus on agreed areas needing attention
• Works well	O3	Works well
 Clearer layout from the department of education Increased supports for schools 	O4	Increased support from the Department
Opportunity to harness staff energy and interest	O5	Potential of staff energy and interest
• If led well	06	Leadership can influence motivation of staff
Threats (obstacles)	Code	Description title
 Willingness of staff Not having all staff on board Lack of interest 	T1	Lack of interest and willingness of staff
Availability of sub coverSupport from department of education	T2	Lack of support from the Department
• Time	T3	Lack of time
• Feelings that teachers voice won't be heard and respected	T4	Staff won't be listened to

As can be seen from the example above, identifying the codes and matching them with the actual data extract quote that demonstrate that code was an important part of phase two.

With these new alphanumeric codes, I now turned my attention to coding the illustrative graph data from the survey

Phase 3 – Generating initial themes

This phase involved sorting the different codes (and the actual data extract quotes) into potential themes. To help analyse the codes and consider how they might combine to form overarching themes, Braun and Clarke (2006) recommend the use of a visual representation (p.89). I therefore listed all the codes with their title description into a table as follows:

Code	Title description	Code	Title description
S 1	Staff involved due to	01	Increase staff
	school size, with		involvement
	good communication		
S2	Staff are motivated	O2	To focus on agreed
	and interested		areas needing
			attention
S3	Works well	03	Works well
S4	Focus on teaching	O4	Increased support
	and learning		from the Department
S5	Motivation and	O5	Potential of staff
	interest are there		energy and interest
W1	Events taking away	O6	Leadership can
	from teaching and		influence motivation
	learning		of staff
W2	Lack of interest	T1	Lack of interest and
			willingness of staff
W3	Great job being done	T2	Lack of support
			from the Department
W4	Lack of time	T3	Lack of time
W5	Stress	T4	Staff won't be
			listened to
W6	Increased workload		
W7	Lack of interest		

The next task involved thinking about the relationship between the codes. I used colour coding once again to link codes I felt had something in common, collating them then into their own thematic box for ease of analysis and further refinement as follows:

Code	Title description	Sub themes
S 1	Staff involved due to	
	school size, with	
	good communication	
01	Increase staff	
	involvement	
T4	Staff won't be	
	listened to	
	•	

W6	Increased workload	
	Increased workload	
S2	Staff are motivated	
52	and interested	
W7	Lack of interest	
05	Potential of staff	
05	energy and interest	
T1	Lack of interest and	
11	willingness of staff	
	winnighess of start	
W5	Stress	
VV J	500055	
	Lack of time	
<u> </u>		
04	Increased support from the Department	
T2	Lack of support	
12	from the Department	
T3	Lack of time	
15		
S3	Works well	
	Great job being done	
03	Works well	
05	WOIKS WEII	
S5	Motivation and	
	interest are there	
W2	Lack of interest	
¥¥∠	Lack of Interest	
S4	Focus on teaching	
57	and learning	
W1	Events taking away	
	from teaching and	
	learning	
O2	To focus on agreed	
	areas needing	
	attention	
06	Leadership can	
	influence motivation	
	of staff	
		<u>L</u>

The yellow codes represent the involvement of school staff in the school self-evaluation process, whilst the dark green code represents the perceived workload involved. The light green codes refer to the motivation of school staff to engage in the school self-evaluation process, whilst the red code represents their emotional state. The orange codes represent the perceived resources a school needs (including time) to implement school self-evaluation, whilst the pink codes represent school staff who believe that the current process of school self-evaluation in

our school is more than adequate. The professionalism and knowledge of school staff is represented by the purple codes, whilst the light blue codes represent the desire of school staff to focus on teaching and learning in any school self-evaluation process. The dark blue code represents the leadership of school self-evaluation and the leader's ability to motivate school staff to engage with the process.

Phase 4 – Reviewing and developing themes

This phase involved the refinement of potential themes. To do this I revisited the actual data extracts and the literature, to thus ensure, that the themes worked in relation to our data set and that we had accurate representations for the purposes of this study. The outcome of this process was as follows:

Code	Title description	Sub themes
S 1	Staff involved due to	• Staff involvement
	school size, with	
	good communication	
O1	Increase staff	
	involvement	
T4	Staff won't be	
	listened to	
S2	Staff are motivated	Motivation
	and interested	
W7	Lack of interest	
O5	Potential of staff	
	energy and interest	
T1	Lack of interest and	
	willingness of staff	
S5	Motivation and	
	interest are there	
W2	Lack of interest	
W4	Lack of time	Resources
O4	Increased support	
	from the Department	
T2	Lack of support	
	from the Department	
T3	Lack of time	
W6	Workload	
\$3	Works well	- Emetion
	Works well	Emotion
W3	Great job being done	
03	Works well	
W5	Stress	

S4	Focus on teaching and learning	• Teaching and learning
W1	Events taking away from teaching and learning	
02	To focus on agreed areas needing attention	
O6	Leadership can influence motivation of staff	• Leadership

Phase 5 – Refining, defining and naming themes

In this phase Braun and Clarke (2006) say we must identify the essence of what recognised themes are about and determine what aspect of the data set each theme captures (pg.92). They contend it is important to identify the 'story' that each theme tells in relation to each other and how it fits into the overall story of the research question. Using this approach in my analysis of the data, I generated three key themes to frame a thematic analysis of the data.

The first, entitled '*Staff involvement in whole school decision-making*', was about the people who drive the implementation of school self-evaluation to achieve the end goal of improved outcomes in teaching and learning.

The second theme '*Resourcing school self-evaluation*', was about the daily experiences of people working in a school doing school self-evaluation and all that entails, physically, mentally and emotionally.

The potential outcomes of both of these themes are grounded in the effective enactment of the final and third theme, '*Leadership and staff motivation*'. School leadership plays an important role in motivating staff to engage meaningfully with the school self-evaluation process by facilitating the development and provision of any resources required and tapping into the energy and interest of staff members.

Vinitable Action Reflection Significance New Action Mohiff Thoughts/Thinking What Ilearned Meerned Intervention Schein Observation Modernet Modernet Intervention Coghian Observation Reaction Modernet Intervention Coghian Observation Reaction Modernet Intervention Coghian Consume Provide Schein Intervention Intervention Coghian Consume Provide Schein Intervention Intervention Coghian Consume Provide Schein Intervention Intervention Coghian Consume Provide Schein Provide Schein Intervention Coghian Consume Provide Schein Provide Schein Provide Schein Coghian Consume Provide Schein Provide Schein Provide Schein Coghian Consume Provide Schein Provide Schein Provide Schein Consume Provide Schein Provide Schein Provide S
Whitehead Action Reflection Significance New Atom back to the start/left side of the grin/left side side side side side side side side
Michael Observation Observation Indegenent Indegenent <thindegenent< th=""> <thindegenent< th=""> Indegenent</thindegenent<></thindegenent<>
Schein (taken form Coghian) Description Perection (bow diay ou coghian) Description Perection Perecti
Freues J was J really J learned Viny (10 show how your learning from one episode of action the returning the trace of the start of the
with my conticul I was later there the friend the amoved network and add real thing with myself taking and add can be very built muse and imp

1.55	No. of Concession, Name	The second second	Vinny's Refle	ctive Journal		New Action
	Whitehead	Action	Reflection		Significance	(Leads you back to the start/Left side of the grid)
-	McNiff		Thoughts/Thinking		What I learned	Intervention
	Schein (taken form Coghlan)		Observation (what did you observe? Can you describe it?)	Reaction (how did you react? What feelings were aroused in you?)	Judgement (what was your judgement about what happened? What thoughts or evaluations did the event trigger?)	(what did you do about it? How did you intervene? Remember: doing nothing or remaining silent is also an intervention)
		we met as a whole staff	I was very excited for this meeting. This was apother efformunity to address as a south teaching	9 Reefit went well. 9 enjoyed it. 9 mag tives 1	I indeed the staff to have entraced and chigge the opportunity	This box is crucial y Vinny (i) to show how your learning from one episode of action reflection feeds back into new practice. (ii) to show how y your learning influences new actions
-		conduct a Swot analysis of Mo	and lacerning issues . The Bethe scribe happily. Again this	after it	Liscussion and Lebyte Sudging by Antociptica	actions (iii) to show you taking action based on your thinking. You could regard the new action as the beginning of a new cycle. This process
		T+L of handnenting	time and space to share ladeth and observe end fisten more		and their Shot and sis responses. Jam	emphasises the cyclical nature of 'action-reflection' (A core aspect of action research)
		school.	Me debate was lively. Shif seemed		ry listening Skills as centainly fiched up	As part of the AR cycle feedba stage, it will be
			The Logen Act to toution we was simple and derefore did not infinge an	Hel	people can view and intermet data different Mato is	9 analyse the J data unbid y. whilst achar my fors
), C	momentiem. Knowledge was perly being shared purt	1	not something I would have noticed or picked up	herer 2
			Challer.		n before. As a plack, research I	tioner skerver can analys
				(a 4	interpret	an see lata

	The second second second		Vinny's Refle	ctive Journal	Sec. 20	New Action	1.2.2
ę	Whitehead	Action	Reflection	100/2	Significance	(Leads you back to the start/Left side of the grid)	
			Thoughts/Thinking		What I learned	1. to method	
	McNiff Schein (taken form Coghlan)		Observation (what did you observe? Can you describe it?)	Reaction (how did you react? What feelings were aroused in you?)	Judgement (what was your judgement about what happened? What thoughts or evaluations did the event trigger?)	Intervention (what did you do about it? How did you intervene? Remember: doing nothing or remaining silent is also an intervention)	
		Duning Jan 27th - Jan 31st J showed and snas the sst	I felt it was infostant to made each member individually to allow Phem allow Phem aust individual questions in a safer secure	I found this, action great as a secure and a leader was their first time the sec the	tatter of fitte confused ces mon willows Alis was a, poor way to explore these documents.	This box is crucial Vinny (i) to show how your learning from one episode of action reflection feeds back into new practice. (ii) to show how your learning influences new actions	
		circular, guide lives and LAOS. This was done on an individual basis. This was Action Step (D.	Environment i.e. not in front of anoyone. It also allowed ne one offering to reach and con firm consensis on the sst focus tery in	SSE documents In terms of AR this was a very real production intervention	delivergita as a knowledgable experts official the soff in a progine tole. This in essence, was	(iii) to show you taking action based on your thinking. You could regard the new action as the beginning of a new cycle. This process emphasises the cyclical nature of 'action-reflection' (A core aspect of action research) I The Deinty	Coursing for
		Each member individually with sub cover, thomald by the D.P.+ SETS	n confidential riay and uncar dny group thing fensuel ethical standards were followed all menters seemed somfortable. Most were supprised ab	r leason -	what of said to charge (in undereding the or R. Havenes & wo aware drat of hupose of the onversation, was to simply introduce the documents. Storias por interded to	me kere value of a neflectiv and how my rang my judg am more stoff b	ras on the reing a f provident tann my leades tent by carned and carned
		and i	the amend decements re SSE from The Dept.	*	training ser s took com in the fact full scriber g Diese du	ruments	l research politiquer egoslinge have als parmed the mportant
					nould occu the school in an sse through collectiv project	engaged cycle this of A.C.	critical friend com play in tel saugs be wither act

Handwriting Survey for Parents (and results)



Q2 My child has opportunities to use their handwriting for real reasons at home e.g. shopping lists, thank you letters, greeting cards etc. Multiple Choice



Q3 I know what style of handwriting my child uses at school. Multiple Choice

C ^









C ^



Q7 My child needs help with their handwriting. Multiple Choice

e ~



Q8 My child is good at handwriting. Multiple Choice

C ^







c ^

Q10 I know what I can do to help my child improve their handwriting.



342

Handwriting Survey for Pupils (and results)



Q2 I have opportunities to use my handwriting at home e.g. shopping lists, thank you letters, greeting cards et





•



Q4 I know what writing implement (e.g. pencil/pen) my teacher would like me to use. Multiple Choice

• •











Q5

Q7 I need help with my handwriting. Multiple Choice













C ^

C ^

Q10 I do my best handwriting in all my school work. Multiple Choice



Handwriting Survey for Teachers (and results)

Survey Tally Sheet

School Self-Evaluation 2020

Focus:	<u>Iandwriting in our school – What does it look like now at this moment in time in your class?</u>
Q. 1	A. Is there a handwriting book/scheme in use? 5 Yes 1 No
	B. If yesPlease specify:
	- Cursive script 1
	- Handwriting today <mark>1</mark>
	- Handwriting made easy 1
	- Don't use a book <mark>1</mark>
Q. 2	A. Are the children required to write using a specific style? 6 Yes 0 No
	B. If yesPlease specify:
	- Jolly phonics style <mark>1</mark>
	- Cursive style <mark>4</mark>
	 Jr. infants pre-writing activities and letter formation 1
	- Sr. infants development and capitals 1
	- 3 rd class cursive introduced 1
	- 4 th class cursive <mark>1</mark>
	- 5 th and 6 th class fluent personal style 1
Q. 3	A. Do you, the teacher, write on the whiteboard etc. using the same style as
	the children? <mark>5 Yes</mark> <mark>0 No</mark>
	B. If yesPlease specify:
	 Doesn't always happen, but I try to 5
	- I use print also 1
	- Jolly phonics <mark>1</mark>

Q. 4	A. Do the children write with specific writing tools/implements? 4 Yes 3 No
	A. If yesPlease specify:
	- Pencils <mark>4</mark>
	- Pencil grip <mark>2</mark>
	- Pen <mark>3</mark>
	- Chubby pencil <mark>1</mark>
Q. 5	A. Do you teach handwriting formally? 3 Yes 2 No
	B. If yesHow often?
	- Formal lessons once a week <mark>2</mark>
	- Formal lessons once a month <mark>1</mark>
Q. 6	A. Do you use any additional resources to teach/improve handwriting? (e.g.,
	pencil grippers, H.W. copies etc.) <mark>4 Yes</mark> <mark>1 No</mark>
	B. If yesPlease specify:
	- Sound copy letter formation 4
	- Pencil grips <mark>3</mark>
	- Jolly phonics <mark>1</mark>
	- Posture <mark>1</mark>
	- IWB/software 1
	- Instruction/assessment letter formation sheets 1
Q. 7	A. Are you aware of the stages of handwriting development in our school?
	(i.e., what is taught and when) <mark>1 Yes</mark> <mark>2 No</mark>
	- Unsure <mark>3</mark>

Q. 8	A. Please comment/suggest any improvements you feel could be made to
	benefit the teaching and learning of handwriting in our school:
	 Teachers should model at all times the style being taught 2
	 Uniformity of style and stages across the school 5
	 A book handwriting program for all stages 1
	 A scheme of detailed proof of progression 3
Q. 9	A. Any other comments?
	- Less focus on book <mark>1</mark>
	 School/class handwriting awards 1
	 Correct handwriting to be used by pupils in everyday work 2

Question 1:

Please indicate

- a) What would be your preferred Style(s)?
- b) Should there be a change of style and when, at what point?
- c) What would be your preferred writing implement(s)?
- d) What would be your preferred resources i.e., school wide handwriting workbook or handwriting copies?

Question 2:

Would you prefer a school wide, age-appropriate approach to the following;

• Pre-writing activities: Yes
No

What would this look like to you?

• Language used when teaching or assessing handwriting: Yes \square No \square

What would this look like to you?

• Presentation: Yes
No

What would this look like to you? Where would this be done?

• What type of copy? Yes
No

What would this look like to you?

• What type of writing implement(s) used? Yes
No

What would this look like to you?

Question 3:

Do we want to adopt a publisher's handwriting scheme? Why? Who's scheme?

Please indicate your preference:

Question 4:

What aspects of our current policy would you like to retain?

Handwriting Policy

Introductory Statement & Rationale:

This policy is formulated in line with the Primary Language Curriculum and it will be implemented with immediate effect.

In Scoil Bhríde Nurney we acknowledge that the developments of handwriting skills are important in our pupils' education. We support the current research, which outlines the benefits of teaching cursive handwriting. These include that it:

- Prevents reversals and confusion of letters
- Enhances spelling ability
- Diminishes potential for errors
- Enforces the skills for patterns in reading and writing
- Prevents erratic spaces between letters and words
- Helps left-handed children
- Is quicker and easier to use than print.

Aims of the Handwriting Policy:

- To develop and improve fine motor skills in the early years
- To ensure good letter formation from Junior Infants to Sixth Class
- To encourage that all children are using the correct pencil grip
- To ensure legibility and neat presentation of all written work
- To help develop speed, accuracy and writing fluency
- To ensure uniformity and consistency in letter formation from class to class
- To provide children with a relevant life skill
- To help in the improvement of spelling.

General Writing Guidelines:

Children will be encouraged to remember the **Four P's**- Preparation, Pencil Grip, Posture and Practice.

All classrooms should have a continuous cursive handwriting frieze displayed in their room. All classrooms should have a Literacy/Numeracy presentation sample displayed in their room.

All children should hold the pen or pencil at least 2 cm away from the tip so that they can see what they are writing. For left-handers, writing from left to right can be difficult. It is important that the class teacher ensures that the child is seated on the left side of the desk.

Triangular grip pencils are to be used in the infant classes.

Regular grip pencils are to be used in $1^{st} - 4^{th}$ class.

Pens are to be used in $5^{th} - 6^{th}$ class.

In all classes, pencil must be used to complete all Maths activities.

Teachers should model good hand writing at all times.

Handwriting Programme:

Scoil Bhríde Nurney has adopted the Educational Company Handwriting scheme "Ready to Write" for all classes.

Class	Literacy	Numeracy
6 th	Normal lined	Small square
5 th	Normal lined	Small square
4 th	Normal lined	Small square
3 rd	Normal lined	Small square
2 nd	Normal lined	1cm sq.
1 st	Thick lined (15mm J09) moving to Normal lined	1cm sq.
Senior infants	 Project copies 15A Thick lined (15mm J09) 	20mm square
Junior infants	 Blue project copies (preformed) Plain project copies (no lines) 	Maths folder

Copies for Literacy and Numeracy

Children with Special Needs

This handwriting policy aims to meet the needs of all the children in the school. This will be achieved by teachers differentiating the content, outcome and process to ensure learning for all pupils. The requirements of children with special needs will be taken into account when planning class lessons and related activities.

Pencil Grip

The pencil is grasped between the tips of the thumb and the index finger and is supported against the side of the middle finger (tripod grip). The last two fingers are curved and rested against the table surfaces. The wrist is bent back slightly and the pencil is held about 2cm" from the tip (slightly more for left-handed children). The 'counter held in the palm of the hand whilst writing' technique can be used.

Assessment and Success Criteria

- There will be constant teacher observation of pupils' handwriting in copy books at school and also in homework copies.
- There will be a focus on continual improvement.
- Teacher should display the children's writing emphasis should be placed on improvement made in handwriting. Children should be encouraged to write final pieces of work in their best handwriting.
- The end of year handwriting should be compared with the handwriting the child was producing at the beginning of the year, for assessment purposes and for possible School Report comment.

<mark>Survey</mark>

Thank You for taking the time to complete this questionnaire regarding:

School Self-Evaluation

Your help is most appreciated.

- DES = Department of Education and Skills
- SSE = School Self-Evaluation

Instructions:

In each question, please tick the box that feels most right to you and please fill in as best you can the description boxes. If you need more space, please write your answer on another piece of paper with the question number beside it.

- 1. The existing resources provided by the Department of Education and Skills (DES) are useful for School Self-Evaluation (SSE)
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 2. More resources are required from the DES on how to conduct SSE
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **3.** Rather than each school spending time and resources developing their own internal evaluation procedures, schools should be provided with a generic set of tools to assist with the implementation of SSE
 - □ Disagree Strongly
 - □ Disagree

- □ Indifferent
- □ Agree
- □ Agree Strongly
- 4. Staff at this school have the necessary skills required to carry out SSE
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 5. Principals and Deputy Principals need more training on how to conduct SSE
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 6. Teachers need more training on how to conduct SSE
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 7. Staff at this school have the capacity to analyse quantitative data
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 8. Staff at this school have the capacity to analyse qualitative data
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly

- **9.** The Principal and Deputy Principal of this school have the necessary training required to carry out peer review (teacher observation)
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **10.** Teachers of this school have the necessary training to carry out peer review (teacher observation)
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **11.** The Board of Management of this school have the necessary skills required to carry out evaluation and planning duties required of Board of Managements.
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **12.** Results from externally devised standardised tests (e.g., literacy and numeracy tests) should be used as part of the self-evaluation process of schools.
 - **Disagree Strongly**
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 13. Peer review is used as part of the SSE process in this school
 - □ Yes
 - □ No
- 14. Does your school have a set of procedures for carrying out SSE?
 - □ Yes
 - □ No

- 15. Does your school have an SSE policy?
 - □ Yes
 - □ No

16. The process of SSE is easy to understand

- □ Disagree Strongly
- □ Disagree
- □ Indifferent
- □ Agree
- □ Agree Strongly
- 17. The SSE guidelines developed by the DES are easy to understand
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 18. The Principal and Deputy Principal conduct SSE on a regular basis in this school
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly

19. Teachers conduct SSE on a regular basis in this school

- □ Disagree Strongly
- □ Disagree
- □ Indifferent
- □ Agree
- □ Agree Strongly

20. SSE involves all staff

- □ Disagree Strongly
- □ Disagree
- □ Indifferent
- □ Agree
- □ Agree Strongly
- **21.** SSE reports should be published on the internet

- □ Disagree Strongly
- □ Disagree
- □ Indifferent
- □ Agree
- □ Agree Strongly
- 22. SSE results in better management
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- 23. SSE results in better teaching and learning
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **24.** SSE places a lot of stress on staff
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **25.** SSE increases staff morale
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
- **26.** SSE takes up a lot of time
 - □ Disagree Strongly
 - □ Disagree
 - □ Indifferent
 - □ Agree
 - □ Agree Strongly
27. SSE is popular with the majority of staff in this school

- □ Disagree Strongly
- □ Disagree
- □ Indifferent
- □ Agree
- □ Agree Strongly

28. How would you describe your attitude to SSE in general?

- □ Positive
- □ Negative
- □ Indifferent
- □ Other

If 'Other' please describe:

29. Please describe briefly, in the box below, what you think is the purpose of SSE

30. Please list, in the box below, any useful policy documents that you know of regarding SSE

31. Please describe in the box below, what in your opinion are the positive aspects of SSE as a government policy

32. Please describe below, what in your opinion are the negative aspects of SSE as a government policy

33. In the box below, please describe any other opinions, thoughts, ideas, observations or comments you have in relation to your experience of SSE to date in your educational career. (*if any*)

Thank You for taking the time to complete this survey. Your help is most appreciated.

Appendix 22





INPUT























PROCESS















OUTCOMES





UNINTENDED CONSEQUENCES











Appendix 23

Focus Group Interview Questions

- 1. What is your attitude to School Self-Evaluation? Why do you think this is?
- 2. What skills do you think are needed to engage or partake in School-Self Evaluation?
- 3. What emotions come to your mind when you think of School Self-Evaluation?
- 4. Who should be responsible for School Self-Evaluation in a school?
- 5. What in your opinion is the purpose of School Self-Evaluation?
- 6. Does School Self-Evaluation, in your opinion, bring improvement at school level or classroom level? Or both? And if so...How?
- 7. In your opinion, should School Self-Evaluation be aimed at school level improvement or classroom level improvement? Or both? Why do you think this?
- 8. From your teaching career experience to date, has the School Self-Evaluation process led to an improvement at school level that you can describe to us?
- 9. From your teaching career experience to date, has the School Self-Evaluation process led to an improvement in classroom teaching and learning that you can describe to us?
- 10. Of the two outcomes above, was one of more value than the other? If so...Why? If not...Why?
- 11. How do you feel about your capacity to engage with School Self-Evaluation? What is enabling this? (e.g., training, leadership, enthusiasm, school climate) What is disabling this? (e.g., time, resources, training, skill set, your ability, apathy)

Appendix 24

Phase 1 – Becoming familiar with the data

I immersed myself in this data to become familiar with the depth and breadth of its content. This involved repeated reading of the data in an active way by making notes and searching for meanings and patterns.

Phase 2 – Data coding

This phase involved the production of initial codes from the data. This initial coding was undertaken by using gerunds and keeping the codes active and as close to the original statements as possible (Charmaz, 2006, pg.42-71). To begin, I underlined what I felt was every key word and phrase in both the survey data and the interview transcripts. I then combined every underlined word and phrase into colour coded group. For example, any words or phrases associated with making improvements were colour coded pink, whilst any words or phrases associated with paperwork were colour coded dark green etc., for example:

	Survey part 1 (multiple-choice questions)
Question 1	 INPUT: The <u>existing resources</u> provided by the DES are <u>usefu</u>l for SSE (<u>80% indifferent</u>, 10% agree)
Question 2	 INPUT: More resources are required from the DES on how to conduct SSE (70% yes, 20% indifferent)
Question 3	 INPUT: <u>Schools should be provided with generic tools for SSE</u> rather than creating their own (<u>yes 60%</u>, indifferent 20%, no 10%)
Question 4	 INPUT: <u>Staff have the necessary skills</u> to carry out SSE (<u>80%</u> <u>yes</u>, 10% indifferent, 10% no)
Question 5	 INPUT: <u>Management needs more training</u> on how to conduct SSE (<u>yes 50%, indifferent 50%</u>) <u>indifferent result may be due</u> <u>to power relations</u>
Question 6	 INPUT: <u>Staff need more training</u> on how to conduct SSE (<u>70%</u> <u>yes</u>, 20% indifferent, 10% no)

Question 7	 INPUT: <u>Staff can analyse quantitative data (90% yes</u>, 10% indifferent)
Question 8	 INPUT: <u>Staff can analyse qualitative data (90% yes</u>, 10% indifferent)
Question 9	• INPUT: School <u>management have</u> the necessary <u>training</u> <u>required to</u> carry out <u>peer review</u> (teacher observation) (<u>40% yes</u> , 30% no, 30% indifferent)
Question 10	• INPUT: <u>Staff have</u> the necessary <u>training to</u> carry out <u>peer</u> <u>review</u> (teacher observation) (<u>60% no</u> , 10% yes, 30% indifferent)
Question 11	 INPUT: <u>BOM have the skills to do SSE</u> (<u>no 50%</u>, indifferent 30%, agree 20%)
Question 12	 INPUT: <u>Standardised test should be used as part of SSE (90% yes</u>, 10% no)
Question 13	 PROCESS: <u>Peer review is</u> used as <u>part of the SSE process</u> in this school (<u>100% no</u>)
Question 14	 PROCESS: has the <u>school a set of procedures for doing SSE</u> (60% yes, 10% no, indifferent 30%)
Question 15	 PROCESS: has the school an SSE policy (90% yes, 10% no)
Question 16	 PROCESS: The process of SSE is easy to understand (no 40%, indifferent 50%, yes 10%)
Question 17	 PROCESS: The <u>SSE guidelines</u> developed by the DES re <u>easy to</u> <u>understand</u> (<u>no 80%</u>, 10% indifferent, 10% yes)
Question 18	 PROCESS: School <u>management conduct SSE on a regular basis</u> in this school (90% yes, 10% indifferent)
Question 19	 PROCESS: <u>Teachers conduct SSE on a regular basis in this</u> <u>school (90% yes</u>, 10% indifferent)
Question 20	• OUTPUT: <u>SSE involves all staff (90% yes</u> , 10% indifferent)
Question 21	 PROCESS: <u>SSE reports</u> should be <u>published on the internet (no 90%</u>, indifferent 10%)

Question 22	 OUTCOMES: <u>SSE results in better management</u> (no 30%, indifferent 50%, <u>ves 20%</u>)
Question 23	 OUTCOMES: <u>SSE results in better teaching and learning (yes 60%</u>, indifferent 20%, 20% no)
Question 24	 UNINTENDED CONSEQUENCES: SSE places a lot of <u>stress</u> on staff (<u>ves 90%</u>, 10% no)
Question 25	 UNITENDED CONSEQUENCES: SSE <u>increase staff morale</u> (<u>no</u> <u>90%</u>, 10% yes)
Question 26	 UNITENDED CONSEQUENCES: SSE <u>takes up a lot of time</u> (<u>yes 90%</u>, 10% indifferent)
Question 27	 UNINTENDED CONSEQUENCES: SSE is popular with the majority of staff in our school (no 90%, 10% indifferent)
Question 28	 UNINTENDED CONSEQUENCES: How would you describe your <u>attitude to SSE in general (positive 20%</u>, indifferent 80%)

Survey part 2 (open-ended questions)			
Question 29: What do you think is the purpose of SSE?			
S3 Respondent 1	Give schools a sense of accountability		
	Give a sense of agency to schools to work on what actually needs improvement		
	Cut down on unnecessary investigations/ revisions in schools where they aren't needed		
	Take pressure off Department		
S3 Respondent 2	To look at teaching and learning on our school: identify opportunities for improvement in practice and results		
S3 Respondent 3	Another <u>box ticking exercise</u>		
S3 Respondent 4	I think it's a <u>tick the box exercise</u> and I am not really sure to be honest		
S3 Respondent 5	A whole school plan to evaluate what areas we as a school need to improve on and start to facilitate in our daily school lives, for both teachers and learners alike		

S3 Respondent 6	For a school to identify an area of weakness
	and put a plan into action of how to improve
	the area
S3 Respondent 7	To identify areas of weakness and improve as
	needed. Can be helpful regarding whole-school
	planning
S3 Respondent 8	Illustrate good practice. Identify areas that
	need improvement. Schools to identify and
	develop their own improvement areas.
S3 Respondent 9	(no answer)
S3 Respondent 10	To <u>improve</u> all aspects of the school to lead to
	enhanced learning
S3 Respondent 11	To identify an area of weakness to improve on
	as a school. Target specific areas – improved
	outcomes for the better of each pupil and
	<mark>school</mark> as a whole
S3 Respondent 12	The purpose of SSE is to enable schools to
	critically evaluate teaching and learning within
	their organisation and to put improvement
	plans in place based on their findings
Question 20: Any useful policy do	numents that you know of regarding SSE2
Question 30: Any useful policy do	cuments that you know of regarding SSE?
	cuments that you know of regarding SSE?
S3 Respondent 1	No
S3 Respondent 1 S3 Respondent 12	
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3	No (no answer) None
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4	No (no answer) None None
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3	No (no answer) None None (no answer)
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6	No (no answer) None None (no answer) SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7	No (no answer) None None (no answer) (no answer) SSE guidelines SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 8	No (no answer) None None (no answer) SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9	No (no answer) None None (no answer) SSE guidelines SSE guidelines SSE guidelines SSE guidelines SSE guidelines (no answer) (no answer)
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10	No (no answer) None None (no answer) SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11	No (no answer) None None (no answer) SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines (SSE guidelines (SSE guidelines (SSE guidelines SSE guidelines (SSE guidelines SSE guidelines SSE guidelines SSE guidelines SSE guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines (Purple book) (no answer) SSE guidelines SSE guidelines SSE guidelines LAOS – looking at our schools
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12	No (no answer) None None (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines (Purple book) (no answer) SSE guidelines SSE guidelines SSE guidelines LAOS – looking at our schools
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positive	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines e aspects of SSE as a government policy?
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positiv S3 Respondent 1	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines e aspects of SSE as a government policy? Allows schools to make practical changes
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positive	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines (Purple book) (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines e aspects of SSE as a government policy? Allows schools to make practical changes Allows time for staff to reflect and be objective
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positiv S3 Respondent 1	No (no answer) None None (no answer) SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines e aspects of SSE as a government policy? Allows schools to make practical changes Allows time for staff to reflect and be objective and professional as we review all aspects of
S3 Respondent 1 S3 Respondent 12 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5 S3 Respondent 6 S3 Respondent 7 S3 Respondent 7 S3 Respondent 8 S3 Respondent 9 S3 Respondent 10 S3 Respondent 11 S3 Respondent 12 Question 31: What are the positiv S3 Respondent 1	No (no answer) None (no answer) SSE guidelines SSE guidelines SSE guidelines (Purple book) (no answer) SSE guidelines (Purple book) (no answer) SSE guidelines LAOS – looking at our schools School self-evaluation guidelines e aspects of SSE as a government policy? Allows schools to make practical changes Allows time for staff to reflect and be objective

S3 Respondent 4	As an SNA I don't really know the positive aspects of SSE, as my role and SSE have no relevance to each other
S3 Respondent 5	As a government policy the positive aspects are; it allows for a whole school review of our own personal teachings and going forward delivering the best possible outcomes for our
	learners
S3 Respondent 6	It is a positive way of improving a school, the teaching, learning and overall management
S3 Respondent 7	It does improve learning and management within a school. Provides opportunities for reflection and discussion.
S3 Respondent 8	Improvement of learning methodologies and strategies. Identify areas for development. Taking ownership of curriculum
S3 Respondent 9	(no answer)
S3 Respondent 10	Improvement of teaching, learning methodologies and enhancement of engagement
S3 Respondent 11	Improvement in learning overall. More focus on teaching areas of weakness. Methodologies and skills improved. Specific and easy to back by data and show improvements. SMART goals - attainable
S3 Respondent 12	All schools in the country are expected to partake. It allows schools the <u>autonomy</u> to choose areas of most concern to them and their organisation. It acknowledges the inbuilt desire of the vast majority of teachers to improve their schools teaching and learning.
Question 32: What are the negative aspects of	SSE as a government policy?
S3 Respondent 1	Policy includes <u>very long term aims</u> e.g., in 2023 we'll look at, stage 2 of our English policy happens next yearetc. <u>In that time teachers</u> and principals change, move, retire, take leave. <u>Would prefer one item done and dusted each</u> year entirely
	Also, we had just got into it when it was <mark>paused</mark> for strike which was demoralising and <u>took the</u> energy out of the whole process.
S3 Respondent 2	Very time consuming; lots of abstract, repetitive 'splitting of hairs'; quite meaningless exercise at times
S3 Respondent 3	Additional paperwork on already stretched staff
S3 Respondent 4	Paper work. Time consuming
S3 Respondent 5	The negative aspects of SSE as a government policy are; paperwork to follow paperwork, stresses on teachers to over perform beyond their means where they are already over- stretched. Scare tactics used by inspectors to

	deliver outcomes by over worked and
	underappreciated teachers and not being
	realistic.
S3 Respondent 6	Added workload and time required to do it
S3 Respondent 7	Time consuming – takes a lot of meeting times. Increased paperwork equals increased stress which affects morale.
S3 Respondent 8	Negative stress, time for analysis (timetabling), time consuming, lot of meeting time, takes away from classroom planning time, increased paperwork
S3 Respondent 9	(no answer)
S3 Respondent 10	Time consuming, morale, work overload limits time for planning, stress
S3 Respondent 11	Time consuming. Unnecessary stress/ staff morale. Taking away from teacher planning/class time
S3 Respondent 12	Continuous CPD regarding the process for teachers new to the profession has not been a feature. It is a (formal) <u>addition to an already</u> full workload.
Question 33: What emotions come to mi	ind when you think of SSE?
Question 33: What emotions come to m	ind when you think of SSE?
S3 Respondent 1	ind when you think of SSE? Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow instructions
S3 Respondent 1 S3 Respondent 2	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow
S3 Respondent 1	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow
S3 Respondent 1 S3 Respondent 2	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow
S3 Respondent 1 S3 Respondent 2 S3 Respondent 3	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow instructions Overwhelmed, stressedwhere do I find the time? on one hand but on the other hand a feeling of achievement and happy knowing that I was involved in the teamwork to try and make
S3 Respondent 1 S3 Respondent 2 S3 Respondent 3 S3 Respondent 4	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow instructions Overwhelmed, stressedwhere do I find the time? on one hand but on the other hand a feeling of achievement and happy knowing that I was involved in the teamwork to try and make the school better Inspired, motivated, determined but at the
S3 Respondent 1 S3 Respondent 2 S3 Respondent 3 S3 Respondent 4 S3 Respondent 5	Calm, content, mostly positive, somewhat inattentive and detached. Irritated (with stop- start, training followed by strike) unenthusiastic but willing to participate, have input, follow instructions Overwhelmed, stressedwhere do I find the time? on one hand but on the other hand a feeling of achievement and happy knowing that I was involved in the teamwork to try and make the school better Inspired, motivated, determined but at the

S3 Respondent 9	Frustration, dissatisfaction, disappointment, annoyance
S3 Respondent 10	· · ·
S3 Respondent 11	At Dept. level Vague, unclear, uncertainty, bewilderment, complex, time consuming, work overload, unsupported, lack of communication, under resourced, inconsistent, lack of staff morale, paperwork but with the possibility at our own level of Potential, achievement, success, positivity, focus, guidance, support, attainable, boost in staff morale when progress and figures in areas improve
S3 Respondent 12	Don't have any strong emotions positive or negative really
-	other opinions, thoughts, ideas, observations or comments
	ience of SSE to date in your educational career
S3 Respondent 1	Happy to participate in SSE but also happy to be told what to do at times if it saves on time/discussion/meetings
S3 Respondent 2	I find it quite tiresome in general and would welcome a <u>directive from the Dept. of Ed</u> <u>outlining what to do/how and when in each</u> <u>subject –</u> "there's great freedom in having no choice"
S3 Respondent 3	SSE does not seem to be a 'whole school process' and as an SNA it seems irrelevant to me
S3 Respondent 4	I don't have any thoughts other than SNAs are not involved in SSE and I don't know whether or not it would be beneficial for an SNA to be included or not
S3 Respondent 5	I do feel it would be <u>more beneficial to discuss</u> pressing matters regarding <u>planning</u> , schemes etc. going forward <u>rather than policy.</u>

S3 Respondent 6	I think it's great for management to involve
	everyone in the process (and that this will
	always draw things out and make it hard to
	please everyone) At the end of the day <u>once</u>
	the target subject area improves that's the
	main thing.
S3 Respondent 7	(no answer)
S3 Respondent 8	(no answer)
S3 Respondent 9	SSE is added pressure from the Department.
	Taking time out to discuss items with staff
	could be more valuable if a sub was provided to
	ensure work is still covered as teachers (SET
	and class) return
S3 Respondent 10	(no answer)
S3 Respondent 11	Aware of SSE in a whole school approach and
	its benefits. Unaware of resources/further
	information to enhance the benefits of SSE. I
	feel as though it is seen as an approach of
	something that just 'has to be done' as a school
	approach overall, rather than knowing the
	specific benefits overall for teaching and
	learning at a smaller scale.
S3 Respondent 12	When SSE was first introduced, the
	inspectorate made themselves available to
	their own schools to explain the process to staff
	as a whole. There was no compulsion to invite
	the inspector to deliver the seminar, but in our
	school, we did and it was most beneficial in
	terms of the clarity it brought.
	terms of the clarity it brought.

The colour coded words and phrases were then combined into one master document for ease of thematic analysis. They were then each assigned an alphanumeric code and description title. This removed the need for colour coding. Identifying the codes and matching them with the actual data extracts that demonstrate that code is an important part of phase two.

Actual data extract quotes	Code	Description title
Purpose/Reason		
Accountability	R1	Accountability
Box ticking exercise		
• Tick the box exercise		
Improvement	R 2	Improvement
• Improvement in teaching and learning		
• A need to improve a school for both		
teachers and learners		
To improve		
Improve		

• Identify areas that need improvement		
• Improve the school to enhance		
learning		
• Improve outcomes for the better of		
each pupil and the whole school		
Improvement plans based on findings		
• To evaluate for improvement	R3	Evaluate (weakness
• Identify an area of weakness		and good practice)
• Identify areas of weakness		
Illustrate good practice		
• Identify area of weakness		
• Critically evaluate teaching and		
learning		
Put a plan into action	R4	Planning
Whole school planning		
Improvement plans		
Positive aspects of SSE		
Practical changes	P1	Practical
To reflect	P2	Reflect and discuss
• Reflection and discussion		
Professional	P3	Professional
• As an SNA I am unaware of SSE	P4	SNA
• As an SNA my role and SSE have no		
relevance to each other		
Whole school review	P5	Whole school
• Focus on teaching in areas of		review and work on
weakness		weaknesses
• Delivering the best possible outcomes	P6	Outcomes
• Improving the teaching, learning and	P7	Improvement
management of a school		
• Improvement of learning		
• Improve learning and management		
• Improvement of teaching and learning		
• Improvement in learning		
• To improve the school's teaching and		
learning		
Taking ownership	P8	Autonomy
Autonomy		
Engagement	P9	Engagement

		(with teaching and learning improvement)
Negative aspects of SSE	NT1	The laws to wa
• Very long term aims	N1	Too long term
• Staff turnover is high. Would prefer		
one item done and dusted each year	NO	No
• Roll out of SSE was paused and took	N2	
the energy out of the process		momentum/energy in the process
Very time consuming	N3	No time
 Time consuming 	110	
 Time consuming Time required to do it 		
 Time required to do th Time consuming 		
 Time consuming Time consuming 		
Time consuming		
Time consuming		
Abstract and quite meaningless	N4	Meaningless
Over stretched staff	N5	Overstretched staff
Over stretched staff to over perform		
beyond their means		
Additional paperwork	N6	More paper work
Paperwork		
Paperwork		
Increased paperwork		
Increased paperwork		
Staff stress	N7	More stress
Increased stress		
• Negative stress		
Unnecessary stress		
• stress		
Over worked	N8	Over worked staff
Added workload		
• Addition to an already full workload		
Not realistic	N9	Not realistic
Affects morale	N10	Lowers morale
Morale		
Morale		
• Takes away from classroom planning	N11	Takes from
time		classroom teaching
		and planning time

Taking away from teacher		
planning/class time		
Other comments		
• Happy to participate in SSE	OC 1	Participate yes!
 Happy to be told what to do if it saves on time/discussion/meetings Would like a directive from Dept. saying what to do/how and when in each subject 	OC 2	Tell me what to do
• I find it quite tiresome in general	OC3	Tiresome process
 As an SNA it seems irrelevant to me SNAs are not involved in SSE. I don't know whether or not it would be beneficial for an SNA to be included 	OC4	Irrelevant
• It would be more beneficial to discuss planning rather than policy	OC5	Teaching and learning planning more than policy please
• Great for management to involve everyone in the process	OC6	Involve everyone
• Once the target subject area improves that's the main thing	OC7	Once target area improves
• SSE is added pressure	OC8	Pressure
• Valuable if a sub was provided to ensure work is still covered as teachers return from meetings	0C9	Proper sub cover
• It is seen as an approach of something that just 'has to be done' as a school approach, rather than knowing the specific benefits overall for teaching and learning at a smaller scale.	OC10	SSE has to be done but not sure why
• When SSE was first introduced, we invited the inspector to deliver the seminar, most beneficial in terms of the clarity it brought	OC11	Partnership with inspectorate would be good
Emotions		
 Confused Frustrated Calm Annoyed 	E1	All references to emotions
Leadership		

• <i>it's the principal or the whole staff,</i>	L1	All references
decide what the school needs to		specific to
improve		leadership

As can be seen from the example above, identifying the codes and matching them with the actual data extract quote that demonstrate that code was an important part of phase two.

With these new alphanumeric codes, I now turned my attention to coding the illustrative graph data from the questionnaire

Survey part 1 (multiple-choice questions)		
Question 1	 INPUT: The <u>existing resources</u> provided by the DES are <u>useful</u> for SSE (<u>80%</u> <u>indifferent</u>, 10% agree) 	N4Anti OC11
Question 2	 INPUT: <u>More resources are required</u> from the DES on how to conduct SSE (<u>70% yes</u>, 20% indifferent) 	N9OC9OC11
Question 3	 INPUT: <u>Schools should be provided with</u> <u>generic tools for SSE</u> rather than creating their own (<u>ves 60%</u>, indifferent 20%, no 10%) 	N9N6
Question 4	 INPUT: <u>Staff have the necessary skills</u> to carry out SSE (<u>80% yes</u>, 10% indifferent, 10% no) 	P3Anti OC11
Question 5	 INPUT: <u>Management needs more training</u> on how to conduct SSE (<u>ves 50%,</u> <u>indifferent 50%)</u> indifferent result may <u>be due to power relations</u> 	 N4 N9 P3 P8 P9 L1 OC11
Question 6	 INPUT: <u>Staff need more training</u> on how to conduct SSE (<u>70% yes</u>, 20% indifferent, 10% no) 	 N4 N9 P3 P8 P9 OC11

Question 7	 INPUT: <u>Staff can analyse quantitative</u> <u>data (90% yes</u>, 10% indifferent) 	• P3
Question 8	 INPUT: <u>Staff can analyse qualitative data</u> (90% yes, 10% indifferent) 	• P3
Question 9	 INPUT: School <u>management have</u> the necessary <u>training required to</u> carry out <u>peer review</u> (teacher observation) (<u>40%</u> <u>yes</u>, 30% no, 30% indifferent) 	 N3 N5 N6 N8 N9
Question 10	• INPUT: <u>Staff have</u> the necessary <u>training</u> <u>to</u> carry out <u>peer review</u> (teacher observation) (<u>60% no</u> , 10% yes, 30% indifferent)	 N3 N5 N6 N8 N9
Question 11	 INPUT: <u>BOM have the skills to do SSE</u> (<u>no 50%</u>, indifferent 30%, agree 20%) 	N4N9
Question 12	 INPUT: <u>Standardised test should be used</u> as part of SSE (90% yes, 10% no) 	 P5 P1 R3
Question 13	 PROCESS: <u>Peer review is</u> used as <u>part of</u> <u>the SSE process</u> in this school (<u>100% no</u>) 	 N3 N5 N6 N8 N9
Question 14	 PROCESS: has the school a set of procedures for doing SSE (60% yes, 10% no, indifferent 30%) 	• R4 • P3
Question 15	 PROCESS: has the <u>school an SSE policy</u> (90% yes, 10% no) 	 P3 R4 OC10
Question 16	 PROCESS: The process of SSE is easy to understand (no 40%, indifferent 50%, ves 10%) 	N4OC11
Question 17	 PROCESS: The <u>SSE guidelines</u> developed by the DES re <u>easy to understand</u> (<u>no</u> <u>80%</u>, 10% indifferent, 10% yes) 	N4OC11

	•	
Question 18	 PROCESS: School <u>management conduct</u> <u>SSE on a regular basis in this school</u> (90% yes, 10% indifferent) 	 P3 R3 P8 P9 L1 OC10
Question 19	 PROCESS: <u>Teachers conduct SSE on a</u> <u>regular basis in this school (90% yes</u>, 10% indifferent) 	 P3 R3 P8 P9 OC10
Question 20	• OUTPUT: <u>SSE involves all staff (90% yes</u> , 10% indifferent)	 P3 P9 R4 OC6 Anti OC9 OC10
Question 21	 PROCESS: <u>SSE reports</u> should be <u>published on the internet (no 90%</u>, indifferent 10%) 	• R1
Question 22	 OUTCOMES: <u>SSE results in better</u> <u>management</u> (no 30%, indifferent 50%, <u>ves</u> <u>20%</u>) 	N1N9L1
Question 23	 OUTCOMES: <u>SSE results in better</u> <u>teaching and learning (yes 60%</u>, indifferent 20%, 20% no) 	• P7
Question 24	 UNINTENDED CONSEQUENCES: SSE places a lot of <u>stress</u> on staff (<u>yes 90%,</u> 10% no) 	N7E1OC8
Question 25	• UNITENDED CONSEQUENCES: SSE increase staff morale (no 90%, 10% yes)	N10E1
Question 26	• UNITENDED CONSEQUENCES: SSE <u>takes up a lot of time</u> (<u>yes 90%</u> , 10% indifferent)	N3N1

Question 27	 UNINTENDED CONSEQUENCES: SSE is popular with the majority of staff in our school (<u>no 90%</u>, 10% indifferent) 	N4OC3OC10
Question 28	 UNINTENDED CONSEQUENCES: How would you describe your <u>attitude to SSE in general (positive 20%</u>, indifferent 80%) 	 N4 E1 OC10

My attention now turned to the interview transcriptions. I proceeded to underline what I felt was every key word and phrase. For ease of analysis, I also assigned each interviewee's voice a colour code of red, green, yellow and purple. These key words and phrases were then combined into one master interview document for ease of analysis. The alphanumeric codes were then applied to each keyword and phrase.

Focus Group Interview One	
attitudes are one of confusion to it	• N4
	• E1
	• OC10
 it was taken off the tablehard to get that 	• N2
momentum back, to get back into the swing of it.	• E1
	• OC3
 a lot of confusion wasn't much clear guidance 	• N4
<mark>in the first place</mark>	• E1
	• OC10
 something very broad, I know is what we were told 	• N1
at one stage, and don't go too specific well,	• N4
how broad do we go, how specific do we get.	• OC10
I think the level of engagement from school to	• N9
school varies greatly, just from speaking to my own	• N4
<mark>friends</mark>	• N1
	• P9
	• OC10
 I think, what they're supposed to be doing mostly, 	• OC10
across the board is the big part of the problem	• N4
• one other thing that was being thrown at schools	• N8
from the department, as if there wasn't really	• N6
enough already.	• N5
• Rather than, the Department coming in and doing	• N8
their WSE and deciding, this area, you need to	• R2
improve, and this is what we want you to do. They	• R1
were just saying, okay schools, you need to do this	Anti OC11
<mark>yourselves.</mark>	

 own attitude is, I suppose one of indifference, 	• N4
• own attrade is, i suppose one of margerence,	• E1
	• OC10
 there's so much going on between new language 	• N8
<i>curriculum already beingtalks of a maths</i>	• OC8
one, different programmes being introduced the	• 000
whole time, So, you're constantly running	
around, doing this, doing that, and then I just feel	
it's one extra thing being added into the equation.	
	• N4
 on board with doing whatever I have to do, but I'm not over interested in it. 	• R1
not over interested in it.	• K1 • OC10
	• OC10
	• P3
	• P9
	• E1
• to identify areas of weakness	• R3
	• R5 • P5
	• 0C7
• it's the principal or the whole staff, decide what the	• R2
school needs to improve	• R2 • P7
school needs to improve	• L1
	• OC6
 what the school needs to improve 	• R2
	• P7
	• OC7
 for a particular two years 	• N1
	• R4
 pick something and improve it. 	• R2
	• P7
	• OC7
 schools are doing anyway, because in our job, that's 	• R2
what we do every day.	• R3
	• R4
	• P3
	• P9
	• P8
	• P2
 getting everyone on the same – singing off the 	• R4
same hymn sheet, school-wide,	• P6
	• OC6
 You do that day to day in your own classroom and 	• P8
in your setting that we're watching, and seeing	• P9
what children need to work on.	• P3
	• R2
	• R3
	• R4

• I do see the benefit of it having one specific thing that the school overall has a focus on,	• P5
	- D4
	• R4
	• R2
	• P5
	• P7
 school self-evaluation then is more aimed at 	• R2
	• R2 • P5
school level, or is it aimed at classroom level	• 13
aimed at both, but	D 0
• it's happening anyway, on a classroom level on	• P8
a band level, within classes, teachers talk about	• P9
things we know the areas of strengths and	• P3
weaknessesand look at our own test results,	• R2
or their own assessment, and see where a class	• R3
needs to go	• R4
	• P2
 it pulls all that together and gives focus to the 	• R4
school of where to go.	• P5
	• OC6
• starts at class level, individual classes,, strength	• R4
can become a whole school thing. For example, with	• P5
the handwriting; if people start to implement	• OC6
changes within their rooms, then that's shown	
throughout the school, whether that's in displays or	
handwriting competitionsbecomes a whole	
school result	
 anything we've certainly looked at would have been 	• R4
areas that even we would all have said, oh	• P9
yeah, that's definitely an area. And I think that's	• P8
what got is started on the handwriting, was that	• P5
	• P2
• we had all identified that ourselves. So, I suppose	• R2
you're first identifying things from your own	• R3
experience, before you look at it as a whole school	• R4
thing	• P5
	• P8
	• P9
• all have our own little zone, then when	• R4
we come together then in staff meetings or in	• OC6
discussions, then it becomes a whole school issue	
 certainly, anything that we've looked at, 	• R2
	• R3
in my own experience, would have flagged in my	• R4
	• P3
own classes as areas that would have needed	
own classes as areas that would have needed things.	• P9

 it forces you, we need to improve N4 this, what can we do, you mightn't necessarily get R2 around to doing itit does force you to do R1 Something about an issue that exists P7 especially the handwriting I think I would always P9 have been something that was always on the back OC7 burner, I must try and improve OC10 that. It's like everything else, you've only so many hours N3 	
around to doing it.it does force you to do• R1something about an issue that exists.• P7especially the handwriting I think I would always• P9have been something that was always on the back• OC7burner,I must try and improve• OC10that.• it's like everything else, you've only so many hours• N3	
 something about an issue that exists. especially the handwriting I think I would always have been something that was always on the back burner, I must try and improve that. it's like everything else, you've only so many hours N3 	
 especially the handwriting I think I would always P9 have been something that was always on the back burner, I must try and improve OC10 that. it's like everything else, you've only so many hours N3 	
 have been something that was always on the back burner, I must try and improve • OC7 • OC10 • it's like everything else, you've only so many hours • N3 	
burner, I must try and improve • OC10 that. • it's like everything else, you've only so many hours • N3	
that. • it's like everything else, you've only so many hours • N3	
it's like everything else, you've only so many hours N3	
in the week, and trying to get everything done, that OC8	
l suppose	
this probably brings it to the forefront and makes • R4	
you actually focus on it a bit more than you would • R2	
have done otherwise. • P7	
• P9	
• OC7	
 what are we actually doing well first, with a R2 	
positive stand first. Would you ever see it like • P7	
that? I always think of, what do we need	
to improve	
we're so paperwork focused and paperwork driven N6	
at the minute,	
you're constantly thinking, of inspectors N7 N10	
and WSEs that the focus • N10	
suppose the focus externally would be on areas to • R1	
improve. It's very rarely they come in and say, well, • R2	
you did great on such a thing, you know. Then tend • R3	
to come in and they will focus on the weaknesses, • P5	
and they will pull you up on things • E1	
Anti OC11	
• a positive start , but if you started with it • Anti OC11	
said, you're doing these things really, really • N10	
they'd still be saying, but you need to • R2	
improve. • R1	
• P5	
 If we were to choose an area that's really strong, R2 	
but step it up one more notch <mark>,</mark> would • R4	
probably change people's attitudes • P7	
• P9	
• E1	
 with that mind set. You're kind of automatically, R2 	
looking to improve something, you're • R3	
thinking of it as an area that you're falling down on. • P5	
Maybe the attitudes would be different R2	
board if you came in looking from a point of view • P7	

of, we're doing really well in this, so let's try to	• P9
maintain that, or keep developing it.	• E1
• a few years ago, in specific areas of maths, and part	• R3
of it was to raise the overall score, say in maths and	• P5
standardised tests, and we put focus on the areas	• P7
that would have been traditionally weaker. And we	• P9
definitely did see a huge improvement At	• P1
school level, we did see a big	• OC7
improvement as a result when you're	
focused on something, you get more results.	
 we'd said aside time every week to do a little bit on 	• R2
that Kind of stuck it to the more	• P1
practical areas of maths <mark>,every week</mark>	• P7
we'd take one day and touch on thatit	• P9
helped to focus everyone <mark>,</mark>	• OC7
you touched on it throughout the term every week,	
you were actually teaching it properly and coming	
back and revisiting things	
improvement?Absolutely, definitely.	
 from a school level, 	• R1
pick it up in the area, or I need to focus on this.	• R4
nobody wants to be seen to be	• P5
letting the side down, and you want to pull your	• P8
weight, if it's on a whole school level. So, it does	• P9
impact your classroom teaching,	
• I think there's an accountability as well, when it's a	• R1
whole school thing.	• P8
	• P9
	• OC6
• I do feel that we do have so much going on, it's so	• N8
easy to forget about somethingto be	
implemented successfully	
 I think if it's interesting, I have the enthusiasm; it 	• P9
depends on what the topic is, what the area is.	• E1
There are certain subject areas I'd be much more	• 0C1
interested in than others.	
 I would need a lot of encouragement and a lot of 	• P9
support it really does come down to staff,	• E1
myself and staff enthusiasm and interest in what	- L1
the area is.	
 just having time for it, because I think time is a 	• N3
massive issue.	
 I would have the enthusiasm to do it once 	• R4
something is up and running and started	• R4 • P9
would need that little bit of push	
definitely would need a targeted time or a targeted	• E1
acjunicity would need a targeted time of a targeted	• OC1

reminder that this is our evaluation and this is what we're working towards, one slot a week or two times a week where we're going to work on this.	Anti OC8
 I do very little taking out school self-evaluation documents and reading them. 	N4OC10
 I think if there was a department directive coming out to everybody,something very specific, and told, this is the area you're going to improve on,	 R4 R1 E1 OC3 OC11 Anti OC8
 I just think it's putting the onus back on us, whereas it should be more coming from above, , this is where we saw your school had a weakness, or, pick a strength and say we want you to improve on it. And that they should be deciding it, I feel, more so than the staff. 	N8N5
 we are coming back to look at it. So that you really feel that motivation, that, okay, they're coming to look at poetry or oral Irish, or handwriting – right, I need to up my game. So that you have that kind of motivation to drive you on, because sometimes it 	 R1 E1 OC3 Anti OC8
 can be hard to find motivation You've so much more going on, and so much to teach throughout the day, 	• N8
 It's follow-through as well for them, isn't it? you're living off your nerves, Everyone is putting their best foot forward, and the whole school is making such an effort. you're trying to show off all the work you've been doing to show the school to the best of your ability. They come in, you're doing grand here, but this, this and this needs work – and then they go off and they leave you to it, 	 N7 N10 N8 R1 E1 Anti OC11

and go it's the magical 'work on that', and then they go, it's left to us to	
 theysay,you know your school best,, when they're putting work back on schools. And then, they're very quick at the same time to turn around and say, well, we know best when it comes to, this new curriculum, new documents, new everything to do. 	R1Anti OC11N8
 I certainly think we have the capacity, but I think schools are very busy places, and quite stressful placesschool management, teachers, everyone is working flat out, all the days we're here, and the days we're at home you don't leave here at three o'clock and leave it all behind you. 	 N7 N8 P3 E1 L1 OC8
 there's a tendency to kind of pass things off onto the schools it's felt that everything in the last few years all put back onto the schools. 	N8E1
 Even if they could come once a year for half a day, in an inspector could come in and sit down with the school, or even for a Croke Park hour, and look through where we're at now, and talk with the teachers and the staff and work together on it. 	 R1 R4 OC11
 I think we all have the capability to engage. we're all professionals everyone here is here with the best of intentions, works hard and wants the best, for our students pride in our school we want the school to be best that it can be staff are willing and are enthusiastic and care enough about the school as a whole 	 P3 P3 E1 OC6
 We certainly have the means to assess within our own school and to know, assess our own classrooms and to know the children that we're teaching, I think we all know our children very well. 	P3P8
 the main thing schools are missing is just guidance and someone to come into sit with a staff and look through and think, right, what is the best way for you to use this time, and use this effort to get the best results for your specific school. 	 OC11 OC10 N4 R1 R4 L1
 I think sometimes when we're left to it ourselves, it's very hard to see, when you're one step back from any situation, it's easier to see it clearly. I think sometimes we're so into it, and so involved, and you're kind of protective of your school, 	 E1 OC6 OC7 OC11

 that sometimes you need someone to come in and even to look through and say, right, I see where you're going in this area, but maybe you're being just, an hours' chat, I think, would make the world of difference	 E1 N4 OC11 OC10 R1 R2 P3 P8 P2 Anti OC7
 professional enoughtostand over what we say?I think we would,it has to be a whole school thing if we're all in, we all have to be on the same page. And if everyone feels confident enough to be able to say that, it would be great. 	 R4 P3 P8 OC6
 but I think that's a confidence thing, and I think that's probably something that comes back from when you're starting your teaching career, you have the dip year, or most of us did, where they come in and you're nearly afraid to breathe too loudly, because whatever they say goes. And it's completely rewiring that, and rewiring that almost fear. I mean, we go back to like children in school. 	 R1 E1 Anti OC11
 I'd say confidence is the main thinga belief 	 R4 P3 P8 E1 OC6
 you'd need to have some evidence to show them. concrete evidenceassessment, evidence this all the timeit's the skillsets that we use every day. 	 OC7 P8 P3
 I think it's probably ingrained in us since we were training,you live in fear of the inspector coming in. And they come in then, and criticise. We definitely seek praise, and you seek positive reinforcementare you doing it to the best of your ability,are you doing it to the best of your ability. 	 R1 E1 Anti OC11 R3 N10 E1

weakness or an area to develop, and you want to do it, and be told, yes, you've done an amazing job, when you're analysing it yourself, , we could pick it out, but it's more about this person that's going to come in from the outside, and maybe knock you down, even though you've done the best that you can do there's always a constant fear there that you're not going to get the appraisal and the gratitude for what you've put in	 Anti OC11 P3 P8
 important that everyone, certainly engages and 	• R4
comes together	• P9
	• OC6
 who keeps everyone together, and rounds it up, and mediates, I? I suppose the leadership has to come from the top, 	• L1
• but then equally, everyone has to row in	• R4
and support each other.	• P5
	• P8
	• P9
	• OC6
• there is that tendency to focus on your own little	• R4
department and your own classroom, but I think it's	• P5
really important that teachers build each other up	• P9
as well,	• 0C6
 I think we need to build each other up a little bit 	• R4
more, and to give each other a bit of praise when	• P9
we need it, and stand over each other and give	• P5
each other a help when we need it.	• E1
	 OC6
 I think it's important that everyone has an active 	• R4
part, and everyone probably feels that they have a	• P5
voice	• P9
	• P2
	• E1
	• OC6
 the areas that you need a little bit of support in, 	• R4
that there's someone there to give it to you	• P5
	• P9
	• E1
 working in smaller teams, and then feeding back – 	• R4
probably like we did with the handwriting. I did feel	• P2
that was helpful, kind of funnelling it through,	• P9
so it's not just one big staff meeting and everyone's	• P5
sitting there, the voices who are confident	
in speaking probably, and probably the people who have been doing it longer feel a little bit more confident in doing itI think for younger teachers, it's quite difficult in that setting to put themselves forward, even though they might have a lot of newer ideas, maybe, if they can bring fresh things to the table.	OC6Anti OC9
--	---
 it absolutely has to be a whole school approach. 	 R4 P5 OC6
 hear of schools where the principal decides, we're doing this, and this is happening, and then the teachers are all left scurrying around trying to do it and trying to implement it in the classroom, and it just fails because everyone isn't in it together. So, I think it is really important that everyone just is on the same page and everyone is part of the team that works on it. 	 R4 P5 P9 L1 OC6
 I do believe that leadership, whether it's principal and vice-principal, the main push comes from them, but that everybody needs to be on board. It has to be a whole everybody has to be a team. 	 R4 P5 P9 L1 OC6
 I think with the school self-evaluation that we've been doing, everybody has been involved, everybody has been asked an opinion, whether they want to or not, I think from that regard, it's been operating very successfully where everyone has been involved in the process 	 R4 P5 P9 P2 L1 OC6 Anti OC9
 But I do believe that the leadership, I prefer the main drive to come from the leadership side of things, once staff are involved and know what's happening, right, this is where we're going with it, this is the direction we're going to take. We're going to do this once a week, we're going to do this on a daily, and then that the push comes from management, from a leadership, I need that driver, that motivation. 	 Anti OC9 R4 P5 P8 L1 OC6 Anti OC8
 It can split, and go different directions, where some people are doing more, and other people are doing a lot less. Whereas, I think if everybody knows, this is what's expected, 	 R4 P5 P9

• clear guidelines. So, if it isn't to come from the	• N4
Department, and it was to come from the leaders in school,, I think that would work very well.	P8OC10
 everyone has to do their own little part. 	 R4 R1 P8 P9
 you still need someone reining everybody in, as you do in everything else,, in teams or anything else you need a leader to focus, channel the focus, I think, it stops everyone just going around in circles. 	 P5 OC10 P8 P9 L1 OC6
Focus Group Interview Two	
 positive without being enthusiastic 	• N4 • E1
 that I think it makes sense that schools can identify their own problems and you know go from there but 	 R3 P8 P5
 I don't enjoy the process I don't enjoy that kind of work in terms of policies, policies and paperwork 	 N4 E1 OC3 OC10
 is not an area of personal interest to me but I still think it's a fairly good idea for schools to do? 	 P7 E1 OC1
 It's not something I would enjoy,I have to do with work I'd have no feelings of excitement 	 N4 E1 OC3 OC10
 it'swhat schools do or should be doing anyway all the time, seeing how we can improve things but– I wouldn't get excited about it now 	 N4 R2 P7 E1 OC10
 An awful lot of investment for a very little a very specific little area 	 OC10 N4
 and I can see the point of it and I can see the democratic element of it 	 P5 P7
 I often think,just tell me what to do, I'll do it, 	 R1 N4 OC2 OC6 OC10
 to improve the learning process for our children to improve our practice and 	R2P7

children	• P6
achieve more or achieve their potential. any good school with a positive atmosphere, this	• R3
happens, you see we've a little weakness, it's mentioned	• P5
at a staff meeting and we do something to change it,	• P7
	• P2
	• P1
on board with it and I guess then you need proof in the	• P8
form of paperwork	• P9
	 OC6
	• N6
have always done it as in we see something, lower	• R3
results in tests or some attitude problem on the kid or	• P1
something and we kind of naturally focus in, next year	• P5
let's try to do a bit better on that.	
	• P6
	• P7
	• P8
	• P9
	• OC6
	 OC7
improvement at school level or classroom level	• R2
i'd say both,i'd say both	• P1
the topics you chose, we've picked	• P5
something very practical, handwriting,	
	• P7
it's good for the school because we've all had	• P8
a look at ourselves and said it's something we can	• P9
improve as a school and then the kids will definitely feel	 OC6
it in class because it's very practical this	• OC7
time next year we could look at the copies and say,	• OC10
'Look at that' There's the result,	
I find it hard to engage with When there was a	• N4
meeting, it was SSE, then I go out and it's kind of,	
'Oh gone', you know unless I'm reminded again about it.	
I feel like I'm always so against the clock I	• N3
believe in the decisions made, I want to do it in the	
classroom,the day just goes by, so I guess	
that's why I did like that we did something simple, like	
handwriting, I think maybe if it was perhaps	
more classroom specific	
	• D4
I'd say there is scope for a whole school use of it like we	• R4
were just talking about a lovely schoolhad a	• P5
general feeling and attitude that people in the classroom	• P7
benefited from so I guess we could look at things like,	
uniforms or punctuality andPoliteness, manners.	
issues on yard, so I'd say there is scope for	
tangible whole school improvements	
I'd say potential for it for using both, so far, it's been	• R2
classroom based and I'd it has led to improvements.	• R4
	• P7
	- 17
	• P5

	• OC7
• I'm always a little nervous about plans that are operate	Anti OC6
over a few years,you've a changeover of	Anti OC5
teachers, you could have a new principal, maternity	• L1
leaves, you've everything and those things get lost and	• E1
so I just love when something can be implemented	• N4
between September and June and done and it's	• N1
there Anything that's a little longer term I am	• OC10
starting to lose enthusiasm for because I've started to	
realise, they just fade away.	
 I think the intention is very goodI think it has to 	• R2
be a positive thing you know because we're always	• P7
trying to improve,	• P9
	• E1
• it can be a bit vague if it's too big, I think small	• N1
concise areas or maybe an overall general	• N4
thing in the school if you say for year 1, 2,	• OC10
3, and 4, we're going to implement thisit	• P7
starts off great but inevitably people come and go and	• E1
puff we forget But the intention is	
good.	
• it's good to keep looking at ourselves and looking at our	• R2
practice	• P7
· · · · · · · · · · · · · · · · · · ·	• P2
	• P9
	• P3
	• OC6
	Anti OC7
 we do get to suggest those weaknesses 	• P5
• We do get to suggest those weaknesses	• R3
There's a very supportive environment here in this	• R4
school I think nobody's afraid to ask for help	• R4 • R2
if they need help people have different	
strengths And no one's afraid to say, 'I'm	• P5
bad at something'.	• P8
	• P9
	• E1
	• OC6
	Anti OC7
one feed into the other	• P5
I'd say for SSE classroom based probably works better	• R4
because it's hard to find a practical measurable way of	
add to school culture or atmosphere,	
when you talk about SSE, it's more	
suitable for classroom teaching and it's probably more	
likely more effectiveBecause they're more	
precise measurableharder within the	
bounds of school self-evaluation process.	
 enable you to do school self-evaluation?one 	• R4
thing that's very practical is the time out that we had	• P2
from the classroom to be able to talk in focus groups	• P9

about it's very hard as you know if you're trying to do that at lunchtime	• P5
or after school. Like we were given time and plenty of time to – so that helped to engage and we could all	 P1 L1
agree in our group	OC6OC9
 A long time ago. I just remember the cold in the library, that's what I remember about 'Pick a very small area' That's all I remember, 	 N4 OC10 OC11
 we were trained we were ready but you know the strike then it was called I felt like I'd emotionally invested in it and then it was gone, then it came back, This is just another thing that will come and go 	• N2 • OC3
 the topics we picked had been topics I feel I can talk aboutany of our topics so far actually I've had opinions on 	 P9 P8 Anti OC7
 I know there's an effort to have every voice heard there was so much talk and discussion and meetings to get to a few quite basic points you know so I guess that's the part of it that if an instinct to resist. it's just the time and the all the democracyOn the flip side, if we weren't given all that time, we'd all be moaning wouldn't we. 	 N3 P5 P9 P8 P2 L1 OC3 Anti OC6 Anti OC7 OC8 Anti OC9
 I wouldn't be keen to get more training on it really. 	N4OC10
 I assume you go off to these places and you know what's required and it filters down to us in terms of what you know what we have to do so I'm comfortable enough with that process, I'm probably not eager enough to go and study those things myself. 	 N4 L1 OC10
 my lack of enthusiasm about the whole thing,., I'm very happy to do whatever. 	 N4 E1 OC1 OC3 OC10
 a kind of an apathy towards it, where do you think that stemmed from, I think people want to become a teacher I think people want to whenyou've got a book in your hand and you're talking to the class and they're answering questions and next thing the conversation moves on, you're down with the map and next thing you're talking about Brazil and that's when I feel really good in my job and I like to be in the classroom anything that goes into a folder for me is just the stuff I have to do to get paid, people became teachers	 N4 OC10 N9 E1 R1 P5 P8 P9 OC3 OC7

I feel best when I do something	
like I read a book see something in an art	
gallery I'm doing it in school I	
actually feel – blissful moments	
everything else I try to argue to have it as minimum as	
possible and I'm being paid for being here so I accept	
that I should have to do this stuff but I don't know if I	
could ever really feel enthusiastic about that.	
only thing that would encourage enthusiasm is that if it's	
something you would personally do next year, like the	
handwriting, I do feel positive about that next year, I feel	
good that we'll all be doing the same thing or that we all	
know how to rule our copies the same way but I think	
the general apathy is just if maybe if you asked a pile of	
accountants to do this kind of work you might get some enthusiasm.	
	- N/4
things we have to write down that I feel like you write it	• N4
down and probably it will never ever be read but you're	• N9
doing it in case some legal thing comes up or someone	• OC3
comes in and there's no soul in it,	• OC10
this I wouldn't feel negative about this because it	• P1
is,actually, going to be used next year	• P9
	• P8
	• E1
something I have to write downand away it	• OC10
goes in a folder and probably eyes will never be cast on	• N4
it again, that frustrates me,anything to do with	• N9
legal, you know the whole world of insurance and	• E1
compo culture and all that and everything to do to	 OC3
prevent that is just a very negative area in my mind you	
know. If you were to think about that you wouldn't	
teach really. You'd change your job.	
Well, I love teaching. I love my job. I love the children. I	• R2
love being in there, getting you know making a	
difference in their lives and getting them enthused	
about something, that I'm enthusiastic, bringing out of	
them – bringing out what's the best in them that's real	
to me and I really love that and this	
Box ticking exercise dusty old doesn't do anything for	• OC10
me. I do it because I have to do it because that's my	• N4
terms and conditions	• R1
	• E1
	• OC2
	• OC3
I like integrated learning	• R2
I don't know where you put that in boxes, I'm	• OC10
not a box ticking person.	• N4
	• N9

 every form or folder reduces the time you have in the day to do those things she's done nothing up there because you know what I mean I haven't ticked all 	 N4 N9 R1
the boxes but yet I feel I have done a lot but you can't quantify a lot of it.	• OC10
 your best teaching moments are the things when you veer offSpontaneous. It's not really worth it to go back and change your plans you know what happened and you know and I'd say as well the parents anything they value that you do, none of that's in your folders either 	 N4 N9 OC10
 a good – a positive approach, a positive outlook really, open-mindedgood social skills to be able to engage and to listen and to take on new things and a fresh prepared to look at things in a different way really positive attitude to you know engage with the processnot a negative approach. 	• P9 • E1
 you need a leader who's able to go and do the research, fill in the forms, someone who's good at, making the tables and charts and working it all out, I think you need one person like that and the other people compliant. 	 N4 L1 OC10
 good interpersonal relationships prepare to engage with each other be honest no point in pretending. 	 P9 E1 OC6
 would lean more towards positive than negative anyway a little resistant I feel a sense of resistance,to anything that has to go into a folder you know documents, and accountability 	 N4 E1 OC3 OC10
 whole thing can be a little distasteful not very appealing to me, 	 N4 OC3 OC10
 Acceptance Acquiescence. Resignation but I like the group thing we can all you know share ideas and you know not get too het up about it and start small you know and I like the idea of it, I can see on paper the whole idea of it but I would be no good to do it on my own you know I might as well be looking into a pool of deep water	 N4 R4 P5 E1 OC3 OC6 OC10
 I could think of ways in my own class to do a thing but notYeah but you do all the time don't you?And every – after Halloween and every January and after Easter I always come back with this new thing I'm going to do and change because you know you look back and say, 'God that was so wrong', you knowIt's interesting you should say that though, we're doing it all the timeWe don't reflect on it. We don't stop and sayThis is 	 P5 P7 P9 P8 P1 P2 P3

 evaluationThat's withough isn't it? You're improving a who should be responsible for sch a school? I think the principal, deputy head I'd say and filter it dow know to structure it, steer it and then everybody else comparison 	It the time. • N4 ool self-evaluation in • N4 with the help of the • L1 wn from there. You • OC10 'You guide it and • OC10
 I think it's management you Vinnie and then the likes of the could whittle it down to the tangib here's the three handwriting books them, chat to your partner, have a process we've done here so far and where I didn't think we all need to or you and Catherine might have g then you know sometimes just too slow a whole process down and es management are more clued into t and possibly care more you know. 	 L1 Vice and then if you OC6 Anti OC9 s, staff have a look at vote, again this N4 OC10 d there were times be part of it, that you ot that bit done and many cooks can just pecially as L1 OC6 Anti OC9 OC10

Phase 3 – Generating initial themes

This phase involved sorting the different codes (and the actual data extract quotes) into potential themes. To help analyse the codes and consider how they might combine to form overarching themes, Braun and Clarke (2006) recommend the use of a visual representation (p.89). I therefore listed all the codes with their title description into a table as follows:

Code	Title description	Code	Title description
R1	Accountability	N1	Too long term
R2	Improvement	N2	No momentum/No energy in process
R3	Evaluate (weakness and good practice)	N3	No time
R4	Planning	N4	Meaningless
		N5	Overworked staff
P1	Practical	N6	More paperwork
P2	Promotes reflection and discussion	N7	More stress
P3	Promotes professionalism	N8	Staff overworked

P4	SNA	N9	Not realistic
P5	Whole school review and work on weaknesses	N10	Lowers morale
P6	Focus on better outcomes	N11	Takes from classroom teaching and learning time
P7	Focus on improvement		
P8	Autonomy	OC1	Participate yes!
Р9	Engagement with teaching and learning	OC2	Tell me what to do
		OC3	Tiresome
E1	Emotions	OC4	Irrelevant
L1	Leadership	OC5	Teaching and learning planning more than policy please
		OC6	Involves everyone
		OC7	Once target area improves
		OC8	Pressure
		OC9	Proper sub cover
		OC10	School self- evaluation has to be done not sure why
		OC11	Partnership with inspectorate would be good

The next task involved thinking about the relationship between the codes. I used colour coding once again to link codes I felt had something in common, collating them then into their own thematic box for ease of analysis and further refinement as follows:

Code	Title description	Sub themes
R1	Accountability	
P2	Promotes reflection and discussion	
Р3	Promotes professionalism	
P8	Autonomy	
P9	Engagement with teaching and learning	
OC1	Participate yes!	
OC11	Partnership with inspectorate would be good	
R2	Improvement	
R3	Evaluate (weakness and good practice)	
R4	Planning	
P1	Practical	
P5	Whole school review and work on weaknesses	
P6	Focus on better outcomes	
P7	Focus on improvement	
OC6	Involve everyone	
OC7	Once target area improves	
R1	Accountability	
N1	Too long term	
N4	Meaningless	

OC2Tell me what to doOC4IrrelevantOC10School self- evaluation has to be done not sure whyOC11Partnership with inspectorate would be goodP4School self- evaluation is meaningless to SNAsN2No momentum/no energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers morale
OC10School self- evaluation has to be done not sure whyOC11Partnership with inspectorate would be goodP4School self- evaluation is meaningless to SNAsN2No momentum/no energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
evaluation has to be done not sure whyOC11Partnership with inspectorate would be goodP4School self- evaluation is meaningless to SNAsN2No momentum/no energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
inspectorate would be goodP4School self- evaluation is meaningless to SNAsN2No momentum/no energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
evaluation is meaningless to SNAsN2No momentum/no energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
energy in the processN3No timeN5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
N5Overworked staffN6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
N6More paperworkN7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
N7More stressN8Staff overworkedN10Lowers moraleN11Takes from classroom teaching
N8 Staff overworked N10 Lowers morale N11 Takes from classroom teaching
N10 Lowers morale N11 Takes from classroom teaching
N11 Takes from classroom teaching
classroom teaching
······ ·······························
OC3 Tiresome
OC5 Teaching and learning planning more than policy please
OC8 Pressure
OC9 Proper sub cover
E1

L1	Leadership can motivate the involvement of staff in school self- evaluation	

The orange codes represented what school self-evaluation gives to a school. The yellow codes I felt were representative of the perceived reality of school self-evaluation on the ground by those implementing it. The purple codes represented what, according to the literature, school self-evaluation in theory gives to school staff. I felt the blue codes represented the attitude of staff members to school self-evaluation, whilst the green code represented the leadership of school self-evaluation.

Phase 4 – Reviewing and developing themes

This phase involved the refinement of potential themes. To do this I revisited the actual data extracts and the literature, to thus ensure, that the themes worked in relation to our data set and that we had accurate representations for the purposes of this study. The outcome of this process was as follows:

Code	Title description	Sub themes
R1	Accountability	• Internal accountability
P2	Promotes reflection and discussion	
Р3	Promotes professionalism	
P8	Autonomy	
P9	Engagement with teaching and learning	
OC1	Participate yes!	
OC11	Partnership with inspectorate would be good	

R2	Improvement	• Evaluate for improvement
R3	Evaluate (weakness and good practice)	
R4	Planning	
P1	Practical	
OC7	Once target area improves	
P6	Focus on better outcomes	
P7	Focus on improvement	
OC6	Involve everyone	Whole school approach
Р5	Whole school review and work on weaknesses	
R1	Accountability	External accountability
N1	Too long term	
N4	Meaningless	
N9	Not realistic	
OC2	Tell me what to do	
OC4	Irrelevant	
OC10	School self- evaluation has to be done not sure why	
OC11	Partnership with inspectorate would be good	
P4	School self- evaluation is meaningless to SNAs	• SNA
N2	No momentum/no energy in the process	Workload

N3	No time	
N5	Overworked staff	
N6	More paperwork	
N8	Staff overworked	
N11	Takes from classroom teaching and learning time	
OC3	Tiresome	
OC5	Teaching and learning planning more than policy please	
OC9	Proper sub cover	
N7	More stress	Emotions
N10	Lowers morale	
OC8	Pressure	
E1	Emotions	
L1	Leadership can motivate the involvement of staff in school self- evaluation	• Leadership and staff motivation

Phase 5 – Refining, defining and naming themes

In this phase Braun and Clarke (2006) say we must identify the essence of what recognised themes are about and determine what aspect of the data set each theme captures (pg.92). They contend it is important to identify the 'story' that each theme tells in relation to each other and how it fits into the overall story of the research question. Using this approach in my analysis of the data, I generated two key themes to frame a thematic analysis of the data.

The first theme 'Attitudes to school self-evaluation', is about the people tasked with implementing the school self-evaluation process to achieve the end goal of improved outcomes

in teaching and learning. Whilst the second theme, *'Leadership and staff motivation'* describes the important role school leaders play in motivating and sustaining staff to engage meaningfully with the school self-evaluation process.

Definitions of School Self-Evaluation found in a review of the literature

"a systematic process, largely initiated by the school itself, where participants describe and evaluate the functioning of the school for the purposes of making decisions or undertaking initiatives in the context of (aspects of) overall school (policy) development"

(Van Petegem, 2005, pg.104; Vanhoof and Van Petegem, 2010, pg.20)

"SSE can be described as a process, in large part initiated by the school, whereby highly eligible participants systematically describe and judge the functioning of the school in order to make decisions or adopt initiatives within the framework of school development"

(Vanhoof and Petegem, 2010, cited in Faddar et al., 2017, pg.398)

"SSE is considered as an attempt involving various changes made by the schools in order to ensure a better teaching and learning environment, improve students' academic achievement and also strengthen the schools' ability to implement those changes"

(Hopkins, 2005, cited in Hamzah and Tahir, 2013, pg.50)

"SSE is considered as an internal and formative evaluation which is based on a collection of evidence. Or in other words the schools have to base their judgments on all the evidence gathered to identify the effectiveness of the implementation of the schools' programs...... SSE stresses that it is a reflection process by the schools on their own practice. SSE needs to be operated in a systematic and transparent manner in order to achieve its aims to improve the students' achievement and enhance the schools' professional and organizational learning"

(Hamzah and Tahir, 2013, pg.51)

"School self-evaluation (SSE) is an internal process which aims to ensure quality, improve the teaching–learning process and increase school performance"

(Hofman et al., 2009, cited in O'Brien et al., 2019, pg.2)

"School Self-Evaluation is, by definition, something that schools do to themselves, by themselves and for themselves'. School Self-Evaluation (SSE) involves examining teaching and learning strategies, the performance and development culture and other aspects of school operations so they can be strengthened and supported to improve student outcomes. It also provides an opportunity for the whole school community, including learners, parents and all staff, to reflect on the learner outcomes in light of their goals, targets and key improvement strategies from the previous planning cycle"

(Swaffield and MacBeath, 2005, pg.239; Smith, 2012, cited in Setlalentoa, 2014, pg.525)

"To arrive at an appraisal of their current functioning (strengths and weaknesses) as a point of departure for a plan or vision for the future. Self-evaluation is a procedure which is initiated and carried out by the school in order to describe and evaluate its own functioning"

(Vanhoof et al., 2009, pg.21) (Blok et al., 2005, pg.3)

"The primary purpose of the school is student learning. Evaluation, a tool that promotes and supports both the individual and the system, can be very helpful......to have a sound basis for decision-making about potential developments to increase student learning. It can provide the school with clear guidelines and parameters to follow in order to make positive changes that facilitate this learning.... Moreover, the school can develop and improve through a reflective practice which can help stimulate the decision-making processes"

(Dahler-Larsen, 2006; Sivesend et al., 2006, cited in Emstad, 2011, pg.271-272)

"a situated analysis of educational quality"

(Simons, 2013, pg.9)

"School self-evaluation is a process of conceiving, collecting, analysing and communicating information to: (i) inform decision-making within a school (ii) ascribe value or worth (iii) establish public confidence in the school (iv) demonstrate professional self-accountability (v) to meet the purposes of accountability, development, knowledge and encourage creativity in schools"

(Simons, 2013, pg.16-17)

"Evaluation for accountability purposes involves reporting on goals and standards (including checking on compliance matters) while an improvement focus involves assisting schools to improve through their self-review"

(Mutch, 2013, pg.83)

"Self-review involves investigating evidence about student outcomes and current ways of doing things to find out where improvement is needed"

(Ministry of Education, 2003, cited in Mutch, 2013, pg.84)

"a procedure involving systematic information gathering that is initiated by the school itself and intends to assess the functioning of the school and the attainment of its educational goals for purposes of supporting decision making and learning and for fostering school improvement as a whole"

(Schildkamp, 2007, pg.4)

"a process mainly initiated by the school to collect systematic information about the school functioning, to analyse and judge this information regarding the quality of the school's education and to make decisions that provide recommendations"

(Devos, 1998, pg.51-52, cited in Demetriou et al., 2012, pg.150)

"Teachers and schools are encouraged to formulate their own development plan, to discuss their own perceived strengths and weaknesses, thus fostering collegiality and cohesiveness in terms of the school's own mission and aspirations for improvement"

(Brady, 2016, pg.524)

"School self-evaluation is a process by which members of staff in a school reflect on their practice and identify areas for action to stimulate improvement in the areas of pupil and professional learning"

(Chapman and Sammons, 2013, pg.2)

"Self-evaluation is a process of reflection on practice, made systematic and transparent, with the aim of improving pupil, professional and organisational learning"

(MacBeath, 2005, pg.4)

"School self-evaluation is a collaborative, inclusive, and reflective process of internal school review. An evidence-based approach, it involves gathering information from a range of sources, and then making judgements. All of this with a view to bring about improvements in students' learning."

(DES, 2016, pg.10)

"Its primary goal is to help schools to maintain and improve through critical self-reflection. It is concerned to equip teachers with the know-how to evaluate the quality of learning in their classrooms so that they do not have to rely on an external view, yet welcome such a perspective because it can enhance and strengthen good practice"

Definitions of School Self-Evaluation

A definition of School Self-Evaluation for our school

Please click the three definitions of School Self-Evaluation below which you most agree with:

- □ "a systematic process, largely initiated by the school itself, where participants describe and evaluate the functioning of the school for the purposes of making decisions or undertaking initiatives in the context of (aspects of) overall school (policy) development"
- □ "SSE can be described as a process, in large part initiated by the school, whereby highly eligible participants systematically describe and judge the functioning of the school in order to make decisions or adopt initiatives within the framework of school development"
- "SSE is considered as an attempt involving various changes made by the schools in order to ensure a better teaching and learning environment, improve students' academic achievement and also strengthen the schools' ability to implement those changes"
- "SSE is considered as an internal and formative evaluation which is based on a collection of evidence. Or in other words the schools have to base their judgments on all the evidence gathered to identify the effectiveness of the implementation of the schools' programs...... SSE stresses that it is a reflection process by the schools on their own practice. SSE needs to be operated in a systematic and transparent manner in order to achieve its aims to improve the students' achievement and enhance the schools' professional and organizational learning"
- □ "School self-evaluation (SSE) is an internal process which aims to ensure quality, improve the teaching–learning process and increase school performance"
- □ "School Self-Evaluation is, by definition, something that schools do to themselves, by themselves and for themselves'. School Self Evaluation (SSE) involves examining teaching and learning strategies, the performance and development culture and other aspects of school operations so they can be strengthened and supported to improve student outcomes. It also provides an opportunity for the whole school community, including learners, parents and all staff, to reflect on the learner outcomes in light of their goals, targets and key improvement strategies from the previous planning cycle"

- □ "To arrive at an appraisal of their current functioning (strengths and weaknesses) as a point of departure for a plan or vision for the future. Self-evaluation is a procedure which is initiated and carried out by the school in order to describe and evaluate its own functioning"
- □ "The primary purpose of the school is student learning. Evaluation, a tool that promotes and supports both the individual and the system, can be very helpful......to have a sound basis for decision-making about potential developments to increase student learning. It can provide the school with clear guidelines and parameters to follow in order to make positive changes that facilitate this learning.... Moreover, the school can develop and improve through a reflective practice which can help stimulate the decision-making processes"
- □ "a situated analysis of educational quality"
- "school self-evaluation is a process of conceiving, collecting, analysing and communicating information to: (i) inform decision-making within a school (ii) ascribe value or worth (iii) establish public confidence in the school (iv) demonstrate professional self-accountability (v) to meet the purposes of accountability, development, knowledge and encourage creativity in schools"
- □ "Evaluation for accountability purposes involves reporting on goals and standards (including checking on compliance matters) while an improvement focus involves assisting schools to improve through their self-review"
- □ "Self-review involves investigating evidence about student outcomes and current ways of doing things to find out where improvement is needed "
- □ "a procedure involving systematic information gathering that is initiated by the school itself and intends to assess the functioning of the school and the attainment of its educational goals for purposes of supporting decision making and learning and for fostering school improvement as a whole"
- □ "a process mainly initiated by the school to collect systematic information about the school functioning, to analyse and judge this information regarding the quality of the school's education and to make decisions that provide recommendations"

- □ "Teachers and schools are encouraged to formulate their own development plan, to discuss their own perceived strengths and weaknesses, thus fostering collegiality and cohesiveness in terms of the school's own mission and aspirations for improvement"
- □ "School self-evaluation is a process by which members of staff in a school reflect on their practice and identify areas for action to stimulate improvement in the areas of pupil and professional learning"
- □ "Self-evaluation is a process of reflection on practice, made systematic and transparent, with the aim of improving pupil, professional and organisational learning"
- □ "School self-evaluation is a collaborative, inclusive, and reflective process of internal school review. An evidence-based approach, it involves gathering information from a range of sources, and then making judgements. All of this with a view to bring about improvements in students' learning."
- □ "Its primary goal is to help schools to maintain and improve through critical selfreflection. It is concerned to equip teachers with the know-how to evaluate the quality of learning in their classrooms so that they do not have to rely on an external view, yet welcome such a perspective because it can enhance and strengthen good practice"

Please click the three definitions of School Self-Evaluation below which you most agree with.			
Reference	Definition	Votes in agreement	Overall ranking
participants de the purposes of context of (asp	process, largely initiated by the school itself, where escribe and evaluate the functioning of the school for f making decisions or undertaking initiatives in the pects of) overall school (policy) development" , 2005, pg.104) (Vanhoof and Van Petegem, 2010,	3	
school, wherel and judge the or adopt initia	described as a process, in large part initiated by the by highly eligible participants systematically describe functioning of the school in order to make decisions tives within the framework of school development" Petegem, 2010, cited in Faddar et al., 2017, pg.398)	0	
by the schools environment, strengthen the	dered as an attempt involving various changes made s in order to ensure a better teaching and learning improve students' academic achievement and also schools' ability to implement those changes" 5, cited in Hamzah and Tahir, 2013, pg.50)	3	
is based on a c have to base th the effective programs schools on th systematic and improve the professional a	dered as an internal and formative evaluation which collection of evidence. Or in other words the schools heir judgments on all the evidence gathered to identify hess of the implementation of the schools' SSE stresses that it is a reflection process by the eir own practice. SSE needs to be operated in a d transparent manner in order to achieve its aims to students' achievement and enhance the schools' nd organizational learning" Tahir, 2013, pg.51)	6	1
ensure quality, school perform	valuation (SSE) is an internal process which aims to improve the teaching-learning process and increase nance" , 2009, cited in O'Brien et al., 2019, pg.2)	4	3
to themselves, Evaluation (S strategies, the aspects of sci supported to opportunity fo	Evaluation is, by definition, something that schools do by themselves and for themselves'. School Self SSE) involves examining teaching and learning performance and development culture and other hool operations so they can be strengthened and improve student outcomes. It also provides an or the whole school community, including learners, Il staff, to reflect on the learner outcomes in light of	5	2

their goals, targets and key improvement strategies from the previous planning cycle" (Swaffield and MacBeath, 2005, pg.239; Smith, 2012, cited in Setlalentoa, 2014, pg.525)		
"To arrive at an appraisal of their current functioning (strengths and weaknesses) as a point of departure for a plan or vision for the future. Self-evaluation is a procedure which is initiated and carried out by the school in order to describe and evaluate its own functioning" (Vanhoof et al., 2009, pg.21) (Blok et al., 2005, pg.3)	2	
"The primary purpose of the school is student learning. Evaluation, a tool that promotes and supports both the individual and the system, can be very helpfulto have a sound basis for decision- making about potential developments to increase student learning. It can provide the school with clear guidelines and parameters to follow in order to make positive changes that facilitate this learning Moreover, the school can develop and improve through a reflective practice which can help stimulate the decision-making processes" (Dahler-Larsen, 2006; Sivesend et al., 2006, cited in Emstad, 2011, pg.271-272)	3	
<i>"a situated analysis of educational quality"</i> (Simons, 2013, pg.9)	0	
"school self-evaluation is a process of conceiving, collecting, analysing and communicating information to: (i) inform decision-making within a school (ii) ascribe value or worth (iii) establish public confidence in the school (iv) demonstrate professional self-accountability (v) to meet the purposes of accountability, development, knowledge and encourage creativity in schools" (Simons, 2013, pg.16-17)	1	
"Evaluation for accountability purposes involves reporting on goals and standards (including checking on compliance matters) while an improvement focus involves assisting schools to improve through their self-review" (Mutch, 2013, pg.83)	0	
"Self-review involves investigating evidence about student outcomes and current ways of doing things to find out where improvement is needed" (Ministry of Education, 2003, cited in Mutch, 2013, pg.84)	1	
"a procedure involving systematic information gathering that is initiated by the school itself and intends to assess the functioning of the school and the attainment of its educational goals for purposes of supporting decision making and learning and for fostering school improvement as a whole" (Schildkamp, 2007, pg.4)	5	2

"a process mainly initiated by the school to collect systematic information about the school functioning, to analyse and judge this information regarding the quality of the school's education and to make decisions that provide recommendations" (Devos, 1998, pg.51-52, cited in Demetriou et al., 2012, pg.150)	0	
"Teachers and schools are encouraged to formulate their own development plan, to discuss their own perceived strengths and weaknesses, thus fostering collegiality and cohesiveness in terms of the school's own mission and aspirations for improvement" (Brady, 2016, pg.524)	2	
"School self-evaluation is a process by which members of staff in a school reflect on their practice and identify areas for action to stimulate improvement in the areas of pupil and professional learning" (Chapman and Sammons, 2013, pg.2)	1	
"Self-evaluation is a process of reflection on practice, made systematic and transparent, with the aim of improving pupil, professional and organisational learning" (MacBeath, 2005, pg.4)	4	3
"School self-evaluation is a collaborative, inclusive, and reflective process of internal school review. An evidence-based approach, it involves gathering information from a range of sources, and then making judgements. All of this with a view to bring about improvements in students' learning." (DES, 2016, pg.10)	1	
"Its primary goal is to help schools to maintain and improve through critical self-reflection. It is concerned to equip teachers with the know-how to evaluate the quality of learning in their classrooms so that they do not have to rely on an external view, yet welcome such a perspective because it can enhance and strengthen good practice"	1	

Please click the one definition of School Self-Evaluation below which you most agree with.			
Reference	Definition	Votes in agreement	Overall ranking
(Hamzah and Tahir, 2013, pg.51)	"SSE is considered as an internal and formative evaluation which is based on a collection of evidence. Or in other words the schools have to base their judgments on all the evidence gathered to identify the effectiveness of the implementation of the schools' programs SSE stresses that it is a reflection process by the schools on their own practice. SSE needs to be operated in a systematic and transparent manner in order to achieve its aims to improve the students' achievement and enhance the schools' professional and organizational learning"	2 (previous votes:6)	3 (previous ranking:1)
(Hofman et al., 2009, cited in O'Brien et al., 2019, pg.2)	"School self-evaluation (SSE) is an internal process which aims to ensure quality, improve the teaching-learning process and increase school performance"	0 (previous votes:4)	4 (previous ranking:3)
(Swaffield and MacBeath, 2005, pg.239; Smith, 2012, cited in Setlalentoa, 2014, pg.525)	"School Self-Evaluation is, by definition, something that schools do to themselves, by themselves and for themselves'. School Self- Evaluation (SSE) involves examining teaching and learning strategies, the performance and development culture and other aspects of school operations so they can be strengthened and supported to improve student outcomes. It also provides an opportunity for the whole school community, including learners, parents and all staff, to reflect on the learner outcomes in light of their goals, targets and key improvement strategies from the previous planning cycle"	7 (previous votes:5)	1 (previous ranking:2)
(Schildkamp, 2007, pg.4)	"a procedure involving systematic information gathering that is initiated by the school itself and intends to assess the functioning of the school and the attainment of its educational goals for purposes of supporting decision making and learning and for fostering school improvement as a whole"	3 (previous votes:5)	2 (previous ranking:2)
(MacBeath, 2005, pg.4)	"Self-evaluation is a process of reflection on practice, made systematic and transparent, with the aim of improving pupil, professional and organisational learning"	0 (previous votes:4)	4 (previous ranking:3)

Definitions of School Self-Evaluation

A definition of School Self-Evaluation for our school

The following definition of school self-evaluation has come from a process of reducing nineteen definitions down to five definitions using your votes. The key words and phrases from the final five definitions have been interwoven into the one definition below. Please read the definition of school self-evaluation offered below and then answer the questions which follow. Thank you.

An agreed statement of what "School Self-Evaluation" means to us for our school

"School Self-Evaluation is a cyclical procedure we do to ourselves, by ourselves and for ourselves. It provides an opportunity for the whole school community, including learners, parents and all staff to reflect on our practice and systematically gather information to examine, assess and make decisions on our teaching and learning strategies, the performance and development culture and other aspects of our school's operations, so that they can be strengthened and supported to foster quality, improved teaching and learning and enhanced professional and organisational performance"

8. Do you agree with this statement?

- □ Yes
- □ No
- □ Not sure

9. Have you any comments, thoughts or ideas on this statement?

10. Have you any comments, thoughts or ideas on the way we reached this statement? (*i.e.*, *involving all teaching staff, voting from a list of definitions, combining the common words etc.*)

Definitions of School Self-Evaluation

A definition of School Self-Evaluation for our school

The following definition of school self-evaluation has come from a process of reducing nineteen definitions down to five definitions using your votes. The key words and phrases from the final five definitions were interwoven into the one definition and then adjusted based on your feedback. Please read the definition of school self-evaluation offered below and then answer the questions which follow. Thank you.

An agreed statement of what "School Self-Evaluation" means to us for our school

"In [our school name] school self-evaluation is a cyclical procedure, which provides an opportunity for the whole school community, including learners, parents and all staff to reflect on our practice and systematically gather information to examine, assess and make decisions on our teaching and learning strategies, the performance and development culture and other aspects of our school's operations, so that they can be strengthened and supported to foster quality, improved teaching and learning and enhanced professional and organisational performance"

11. Do you agree with this statement?

- □ Yes
- □ No
- □ Not sure

12. Have you any further comments, thoughts or ideas on the agreed statement? (*if*

you cannot think of anything please just write 'pass')

Question 1 in that are driven or condu	What are all the activities that you can think of that our school engages in that are driven or conducted externally for accountability purposes? (if you cannot think of anything please just write 'pass')		
Respondent	Answer		
S8 respondent 1	Pass		
S8 respondent 2	Pass		
S8 respondent 3	Pass		
S8 respondent 4	Pass		
S8 respondent 5	 Inspectorate advisory teams / individual enhancing progress of whole school development PDST supports guidance Comparing national figures of child development / statistics to current school levels etc. 		
S8 respondent 6	• DES inspection parent questionnaire (e.g., amount of homework)		
S8 respondent 7	 Teaching and learning School self-evaluation Special Education Needs and inclusion Mentoring and Droichead process Monitoring attendance Curricular changes Active and green school flags Vetting procedures Employment Data protection – GDPR Health and safety Child protection and welfare NEPS Anti-bullying procedures Legislation FSSU 		
S8 respondent 8	Credit union quiz		
S8 respondent 9	Pass		

Question 2	What are all the activities you can think of that our school engages in that we select and conduct for self-accountability purposes? (if you cannot think of anything please just write 'pass')		
	Respondent	Answer	
	S8 respondent 1	• Planning and assessment for learning	
	S8 respondent 2	Parent / Teacher meetings	
	S8 respondent 3	 Standardised testing at all levels. Analysis of results. Targeting area for improvement based on results Whole school approach to improving Gaeilge ó bhéal and handwriting 	
	S8 respondent 4	Pass	
	S8 respondent 5	 Sigma-T/Micra-T results Teacher in class assessment Parent Teacher conferences Teacher / Teacher conferences Teacher / Pupil statistics from previous year results Progress / goal update for each given School Self-Evaluation focus Whole School Evaluation plan of School Self-Evaluation 	
	S8 respondent 6	 Individual teacher self-reflection Continuous Professional Development Whole School Self-Evaluation Staff meetings Great communication between colleagues Planning 	
	S8 respondent 7	• Assessment of and for learning, teaching and learning activities	
	S8 respondent 8	Credit union quiz	
	S8 respondent 9	• Ensuring that planning is carried out fully in accordance with the curriculum and that all policies are regularly updated and followed	

Question 3 in that we selec	What are all the activities that you can think of that our school engages in that we select and conduct for school self-improvement purposes? (<i>if you cannot think of anything please just write 'pass'</i>)	
Respondent	Answer	
S8 respondent 1	Pass	
S8 respondent 2	Pass	
S8 respondent 3	 Handwriting – uniformity of approach and practice Píosa as Gaeilge performed by each class at assembly (pre COVID) 	
S8 respondent 4		
S8 respondent 5		
S8 respondent 6		
S8 respondent 7	ideas, Continuous Professional Development	
S8 respondent 8	Pass	
S8 respondent 9	• Regular meetings to discuss the areas in which we are performing well and are happy with and also identifying the areas upon which we can improve to ensure we are improving as a school	

Question 4	What are all the activities you can think of that our school engages in that are conducted or supported externally for improvement purposes? (if you cannot think of anything please just write 'pass')		
	Respondent	Answer	
	S8 respondent 1	Pass	
	S8 respondent 2	 Courses Training programmes Upskilling 	
	S8 respondent 3	Pass	
	S8 respondent 4	• Asking suppliers of new schemes subject based to come in and we examine and discuss materials involved	
	S8 respondent 5	Pass	
	S8 respondent 6	Continuous Professional DevelopmentDES inspections	
	S8 respondent 7	 Continuous Professional Development Revision of curricula Child protection procedures 	
	S8 respondent 8	Pass	
	S8 respondent 9	• Courses carried out in the Education Centre in order to upskill ourselves	

A graphic diagram to communicate and explain all the evaluation activities undertaken in our school

The following diagram has come from a process of listing all the evaluation activities undertaken in our school based on its purpose (i.e., improvement or accountability) and who conducts it (i.e., internally or externally). Please read the diagram and then answer the questions which follow. Thank you.

A graphic diagram to communicate and explain all the evaluation activities
undertaken in our school
13. Do you agree with this diagram?
\Box Yes
\Box No
\Box Not sure
14. Have you any further comments, thoughts or ideas on the agreed diagram?
(if you cannot think of anything please just write 'pass')